Neurofeedback of theta and beta frequencies: Effects on selective attention and response inhibition

Miguel Pimenta

Submitted in total fulfilment of the requirements of the degree of

Doctor of Philosophy

Goldsmiths, University of London

Declaration

I declare that the work presented in this thesis is my own.

Acknowledgments

I would like to start by expressing my gratitude to my supervisors John Gruzelier and Jan de Fockert for the opportunity to start this enriching adventure and for the support through the various stages of planning, implementation and writing of this thesis. I am also especially indebted to Joydeep Bhattacharya for his availability to discuss the aspects of the experimental work related to electrophysiological methods and analysis. I would also like to extend my gratitude to James McDonald, Caroline Luft, Luke Mason, Job Lindsen and Nicolas Farrugia that introduced me to the realms of psychophysiology technique and data analysis. During the lengthy hours of data collection, I counted with the help and dedication of Anneke Kikkert and with the invaluable contribution and refined sense of humor of Chris van Run. I must as well express my sincere appreciation to all the Goldsmiths students that volunteered to take part in the experimental studies. I would also like to thank my old and new friends Carlos 'Glorioso' Pina, Catarina Gamito, Paulo Martinho, Page McLean, Luigi Gioia, Diana Ramalho, Maria Eugenia Valdivia, Pawel Szynkarczuk, José Manuel Reis, Francisco Faria and Dinam Sbardelott for their support and encouragement during these years. My family, especially my parents Cassilda Garcia and José Pimenta, my sister Cristina, my brother André, my nephew Gonçalo and my niece Carolina were a continuous source of strength and encouragement. Finally, I am grateful to my lovely Elisavet for the support and for making me look forward to new experiences in life.

Abstract

employment of electroencephalographic Despite the widespread (EEG) neurofeedback (NF) in clinical and cognitive enhancement contexts, its impact on selective attention and response inhibition remains poorly understood. The present research investigated the influence of theta frequency suppression and low-beta frequencies enhancement on these cognitive functions. A first study investigated the differential impact of the sensorimotor rhythm (SMR) and beta1 amplitude enhancement NF on event-related potentials (ERP) and behavioral measures indexing selective attention and response inhibition in the three-stimuli oddball and in the cued-Go/Nogo tasks. The learning curves evinced training-specific amplitude increments in the beta1 but not in the SMR frequency. However, SMR NF was associated with increased Go-P3 amplitude, decreased reaction time mean (RT) and standard-deviation (RT-SD), while control and beta1 NF were associated with increased false alarm rates in the cued-Go/Nogo. A second study attempted to understand whether performance increments in selective attention and response inhibition could be explained by theta suppression NF when compared to beta1 NF in the same task conditions. Within-session theta amplitude was decreased in theta relative to beta1 NF in passive resting state but not during feedback trials. However, for both theta and beta1 NF there was no evidence of training-specific amplitude changes relative to controls. Regarding selective attention, the mean RT was increased following beta1 NF and decreased after theta NF but not in the same task conditions. This study also failed to provide evidence of increased or decreased performance in response inhibition. In conclusion, the present research was not conclusive regarding the NF conditions that might have contributed to improvements in target processing efficiency and cancelation of a previously prepared response in previous studies. Specific proposals to address several methodological limitations that might have hindered the possibility of detecting frequency-specific amplitude changes and cognitive improvements were advanced.

Contents

Acknowledgments			
Abstract 4			
List of Abbreviations			
List of Tables			
List of Figures	17		
Chapter 1			
1. Introduction	20		
1.1. The functional significance of EEG oscillations	21		
1.1.1. Electrophysiological and behavioral studies of selective attention	25		
1.1.1.1. Electrophysiological indices of selective attention in the oddball task	25		
1.1.1.2. Theta and beta oscillatory activity in selective attention	28		
1.1.1.3. Behavioral correlates of selective attention in the oddball task	31		
1.1.2. Electrophysiological and behavioral studies of response inhibition	32		
1.1.2.1. Electrophysiological indices of response inhibition in the Go/Nogo task	33		
1.1.2.2. Theta and beta oscillatory activity in response inhibition	35		
1.1.2.3. Behavioral correlates of response inhibition in the Go/Nogo task	38		
1.2. Neurofeedback of theta and beta frequencies	39		
1.2.1. Definition and conceptual aspects of Neurofeedback	39		
1.2.2. Models of low-beta/theta NF effects on attention and response inhibition	42		
1.2.3. Learning specificity	44		
1.2.3.1. Within- and across-session changes in active feedback periods	45		
1.2.3.2. Across-session changes in passive periods	51		
1.2.3.3. Frequency-specific effects	52		
1.2.4. Low-beta/theta NF effects on attention and inhibition	54		
1.3. Research questions and goals	66		
Chapter 2	70		
Study 1: The effects of SMR NF and beta1 NF on selective attention and response inhibition	70		

2.1. Introduction	70	
2.1.1. SMR NF effects on selective attention and response inhibition		
SMR learning	70	
Cognitive effects of SMR NF	71	
2.1.2. Beta1 NF effects on selective attention and response inhibition	71	
Beta1 learning	71	
Cognitive effects of beta1 NF	71	
2.2. Outline of Study 1	72	
Chapter 3	74	
Experiment 1: SMR and beta1 learning	74	
3.1. Hypotheses	74	
3.2. Methods and materials	76	
3.2.1. Participants	76	
3.2.2. General procedure	76	
3.2.3. Neurofeedback Procedure	77	
3.2.4. Neurofeedback training	78	
3.2.5. EEG-Neurofeedback Data Processing	80	
3.2.6. Statistical Analysis	81	
3.3. Results	81	
3.3.1. Training-specific learning	81	
3.3.1. Frequency-specific learning	88	
3.4. Discussion	90	
3.4.1. Training- and frequency-specificity in active periods	90	
3.4.1.1. Training-specific within-session amplitude changes	90	
3.4.1.2. Frequency-specific within-session active amplitude changes	92	
3.4.1.3. Across-session active amplitude changes	93	
3.4.2. Training- and frequency-specificity in passive periods	94	
3.4.3. Summary	95	
Chapter 4	96	
Experiment 2: The effects of SMR and beta1 NF on selective attention	96	
4.1. Hypotheses	96	

4.2. Methods	97
4.2.1. Participants	97
4.2.2. Stimulus and procedure	97
4.2.3. EEG recording and analysis	98
EEG recording	98
EEG data analysis	98
Percent of correct ERP trials.	99
4.2.4. Behavioral data analysis	99
Reaction time data preparation	99
d' data preparation	100
4.2.5. Statistical analysis	100
4.3. Results	101
4.3.1. Hypothesis 1: Increased P3a and P3b amplitude as a function of SMR and beta1 NF	101
4.3.2. Hypothesis 2: Decreased RT and RT-SD as a function of SMR and beta1 NF	106
4.3.3. Hypothesis 3: Increased performance rates and perceptual sensitivity as a function of SMR and beta1 NF	109
4.4. Discussion	112
4.4.1. Electrophysiological and behavioral effects of beta1 NF	112
4.4.2. Electrophysiological and behavioral effects of SMR NF	115
4.4.3. Summary	117
Chapter 5	119
Experiment 3: The effects of SMR and beta1 NF on response inhibition	119
5.1. Hypotheses	119
5.2. Methods	120
5.2.1. Participants	120
5.2.2. Stimulus and procedure	120
5.2.3. EEG recording and analysis	121
EEG recording	121
EEG data analysis	121
Percent of correct ERP trials.	122
5.2.4. Behavioral data recording and analysis	122
Reaction time data preparation	122

d-prime data preparation	123
5.2.5. Statistical analysis	123
5.3. Results	123
5.3.1. Hypothesis 1: Increased Go- and Nogo-P3 amplitude as a function	
of SMR and beta1 NF	123
5.3.2. Hypothesis 2: Decreased RT and RT-SD as a function of SMR and beta1 NF	130
5.3.3. Hypothesis 3: Increased performance rates and perceptual sensitivity as a function of SMR and beta1 NF.	131
5.4. Discussion	134
5.4.1 Differential effects of low-beta NE on response inhibition	134
5.4.2 Differential effects of low-beta NE on selective atention	136
5.4.3. Summary	137
Chapter 6	139
General discussion and conclusions of study 1	139
6.1. Overview of the main findings	139
SMR NF was associated with increased attention to targets but not with inhibition	139
The cognitive effects of SMR NE do not necessarily reflect increased SMR activity	141
Beta1 and RE NE were associated with increased inhibitory difficulties	141
6.2 Limitations and future directions	142
6.2.1 Eailure to demonstrate independence of SMR and beta1 amplitude increments	142
6.2.2. Failure to replicate selective attention and response inhibition findings	1/12
Chapter 7	145
Study 2: The effects of theta NE and heta1 NE on selective attention and response inhibition	140
7.1. Introduction	140
7.1.1 Rete1 NE offects on coloctive attention and records in hibition	140
7.1.1. Betai NF effects on selective attention and response inhibition	140
Beta I learning	146
	147
7.1.2. Theta NF effects on selective attention and response inhibition	148
Theta learning	148
Cognitive effects of theta NF	149
7.2. Study 2 outline	149
Chapter 8	151

Experiment 4: Theta and beta1 learning15		
8.1. Hypotheses		
8.2. Methods and materials	3	
8.2.1. Participants	3	
8.2.2. General procedure	3	
8.2.3. Neurofeedback Procedure	4	
8.2.4. Neurofeedback training 154	4	
8.2.5. EEG-Neurofeedback Data Processing156	6	
8.2.6. Statistical Analyses	7	
8.3. Results	7	
8.3.1. Training-specific learning	7	
8.3.1. Frequency-specific learning	4	
8.4. Discussion	6	
8.4.1. Within-session training- and frequency-specificity	6	
8.4.1.1. Training-specific within-session amplitude changes	6	
8.4.1.2. Frequency-specific within-session amplitude changes	9	
8.4.1.3. Across-session training-specific effects	0	
8.4.4. Summary	1	
Chapter 9	2	
Experiment 5: The effects of theta and beta1 on selective attention	2	
9.1. Hypotheses	2	
9.2. Methods	3	
9.2.1. Participants	3	
9.2.2. Stimulus and procedure	3	
9.2.3. EEG recording and analysis	4	
EEG recording	4	
EEG data analysis	4	
Percent of correct ERP trials174	4	
9.2.4. Behavioral data analysis		
Reaction time data preparation 17		
Performance rates and d-prime data preparation	6	

	9.2.5. Statistical analysis	176	
9.3. Results			
	9.3.1. Hypothesis 1: Increased P3a and P3b amplitude as a function of theta and beta1 NF	176	
	9.3.2. Hypothesis 2: Decreased RT and RT-SD as a function of theta and beta1 NF	182	
	9.3.3. Hypothesis 3: Increased performance rates and perceptual sensitivity		
	as a function of theta and beta1 NF	183	
9	4. Discussion	186	
	9.4.1. Electrophysiological and behavioral effects of beta1 NF	187	
	9.4.2. Electrophysiological and behavioral effects of theta NF	190	
	9.4.3. Summary	191	
С	hapter 10	192	
E	xperiment 6: The effects of theta and beta1 on response inhibition	192	
1	0.1. Hypotheses	192	
1	0.2. Methods	193	
	10.2.1. Participants	193	
	10.2.2. Stimulus and procedure	193	
	10.2.3. EEG recording and analysis	193	
	EEG recording	193	
	EEG data analysis.	193	
	Percent of correct ERP trials.	193	
	10.2.4. Behavioral data recording and analysis	194	
	Reaction time data preparation	194	
	d-prime data preparation	194	
	10.2.5. Statistical analysis	194	
1	0.3. Results	194	
	10.3.1. Hypothesis 1: Increased Go and Nogo-P3 amplitude as a function of theta and beta1 NF	194	
	10.3.2 Hypothesis 2: Decreased RT and RT-SD as a function of theta and beta1 NE	199	
	10.3.3 Hypothesis 3: Increased performance rates and percentual sensitivity		
	as a function of theta and beta1 NF	202	
1	0.4. Discussion	204	
	10.4.1. Differential effects of theta and beta1 NF on response inhibition	204	
	•		

10.4.2. Differential effects of theta an beta1 NF on Go stimulus processing	206
10.4.3. Summary	207
Chapter 11	209
General discussion and conclusions of study 2	209
11.1. Overview of the main findings	209
Beta1 NF was associated with task specific increased RT	209
Control NF was associated with task specific increased hit rate and d'	210
Delta and theta up- and down-regulation may be associated with distinct effects on selective attention and response inhibition	211
11.2. Limitations and future directions	212
Chapter 12	213
Conclusions	213
12.1. General discussion of studies 1 and 2	213
12.1.1. Learned modulation of theta and beta frequencies	213
12.1.2 Central midline theta and sensorimotor beta effects on selective attention and response inhibition	214
Selective attention	214
Response inhibition	216
12.2. Theoretical and practical implications	218
12.3. Methodological limitations and future directions	221
12.3.1. Factors affecting learning-specificity and cognitive performance effects	221
12.3.2. Difficulty in determining the effects of NF on voluntary and involuntary aspects of selective attention	225
12.3.3. Failure to independently manipulate Go response probability and response priming	226
12.3.4. Statistical comparison of different ERP components	227
12.4. Final conclusions	227
13. References	229
14.1. Appendix A: Screening questionnaire	250
14.2. Appendix B: Neurofeedback instructions	251
14.3. Appendix C: Examples of feedback displays	252

List of Abbreviations

- ADHD Attention Deficit and Hyperactivity Disorder
- ANOVA Analysis of Variance
- BOLD Blood-oxygen-level dependent
- CPT Continuous Performance Task
- d' d-prime
- DAT Divided attention task
- EEG Electroencephalography
- EMG Electromyography
- ERP-Event-Related Potential
- ERS Event-Related Synchronization
- fm Frontal midline
- fMRI Functional Magnetic Resonance Imaging
- IAF Individual Alpha Frequency
- Hz-Hertz
- ms miliseconds
- NF Neurofeedback
- nVB Ventrobasal nuclei
- PCA Principal Component Analysis

- RF Random-Frequency
- RT Reaction Time
- RT-SD Reaction Time Standard Deviation
- SCP Slow Cortical Potentials
- SMR Sensorimotor Rhythm
- SST Stop Signal Task
- TACS Transcranial Alternate Current Stimulation
- TMS Transcranial Magnetic Stimulation
- TOVA Test of Variables of Attention
- $\mu V microvolt$

List of Tables

- 1.1. Summary of the absolute and relative amplitude changes involving theta, 46 SMR, beta1 frequencies in previous studies. The changes are denoted by the overall trend towards increase (Inc), decrease (Dec) or no change (NC) for each EEG parameter (EEG) during active periods (within- and across-session) and passive periods in pre-training resting state baseline (Baseline) and from pre- to post-training qEEG (qEEG) following different NF protocols including Random Frequency (RF) NF, sham NF.
- 1.2. Outcome of studies investigating NF effects on selective attention and response inhibition. The pre- to post-training change was classified as Increased (Inc), Decreased (Dec) or No Change (NC) for the variables Mean Reaction Time (RT), Reaction Time Standard Deviation (RT-SD), Accuracy, Hit rate, False Alarm (FA) rate, Go-P3 amplitude and Nogo-P3 amplitude in different cognitive tasks.
- 3.1. Mean (and SD) of SMR and beta1 amplitude change (μV) for each period
 83 (normalized to period 1 of each session) for control NF, SMR NF and beta1
 NF; independent t-tests (Bonferroni corrected p-values) between control NF, SMR NF and beta1 NF.
- 3.2. Mean (and SD) of normalized SMR and beta1 amplitude change (μV) in active and passive periods for each session and for control NF, SMR NF and beta1 NF; independent t-tests (Bonferroni corrected p-values) between control NF, SMR NF and beta1 NF.
- 3.3. Means (and SD) of the mean within-session amplitude change (μV) in active
 and passive periods for each frequency; statistical values of the between-subjects one-way ANOVAs and multiple comparisons by independent t-tests (Bonferroni corrected p-values) between control NF, SMR NF and beta1
 NF.
- 4.1. Descriptive statistics of age (mean and standard deviation) and gender 97 (frequency) by NF protocol after excluding four participants with insufficient number of correct trials per stimulus condition.

- 4.2. Means (standard deviations) and statistical results of the pre-training oneway ANOVAs and the pre- to post-training two-way ANOVAs on Mean P3 amplitude to Target-Standard (T-S) and Nontarget-Standard (N-S), Hit Rate (H), False Alarm Rate (FA) to Standards (S) and Nontargets (N), d-prime (d') to Target-Standard (T-S) and Nontarget-Standard (N-S), Reaction Time (RT) Mean and Standard Deviation (RT-SD), percentage of Correct Trials (CT) to Targets (T), Standards (S), and Nontargets (N) included in ERP analyses after artifact rejection.
- 5.1. Means (standard deviations) and statistical results of the pre-training one128 way ANOVAs and the pre- to post-training two-way ANOVAs on Mean
 Nogo- and Go-P3 amplitude, Hit Rate (H), False Alarm Rate (FA), d-prime (d'), Reaction Time (RT) Mean and Standard Deviation (RT-SD), percentage of Correct Trials (CT) to Go and Nogo stimuli included in ERP analyses after artifact rejection.
- 8.1. Mean (and SD) of theta and beta1 amplitude change (μV) for each period 159 (normalized to period 1 of each session) for control NF, theta NF and beta1 NF; independent t-tests (Bonferroni corrected p-values) between control NF, theta NF and beta1 NF.
- 8.2. Mean (and SD) of normalized theta and beta1 amplitude change (μV) in 161 active and passive periods for each session and for control NF, theta NF and beta1 NF; independent t-tests (Bonferroni corrected p-values) between control NF, theta NF and beta1 NF.
- 8.3. Means (and SD) of the mean within-session amplitude change (μV) in active 163 and passive periods for each frequency; statistical values of the between-subjects one-way ANOVAs and multiple comparisons by independent t-tests (Bonferroni corrected p-values) between control NF, theta NF and beta1 NF.
- 9.1. Descriptive statistics of age (mean and SD), gender (frequency) and IAF 173 (mean, SD and range) by NF protocol after excluding one participant with insufficient number of correct trials per stimulus condition.
- 9.2. Means (standard deviations) and statistical results of the pre-training oneway ANOVAs and the pre- to post-training two-way ANOVAs on TargetStandard (T-S) and Nontarget-Standard (N-S) P3 amplitude, Hit Rate (H),
 False Alarm Rate (FA) to Standards (S) and Nontargets (N), d-prime (d') to
 Target-Standard (T-S) and Target-Nontarget (T-N), Reaction Time (RT)

Mean and Reaction Time Standard Deviation (RT-SD), percentage of Correct Trials (CT) to Targets (T), Standards (S) and Nontargets (N) included in ERP analyses after artifact rejection for each of the three NF protocols.

10.1. Means (standard deviations) and statisitical results of the pre-training one200 way ANOVAs and the pre- to post-training two-way ANOVAs on Mean
Nogo- and Go-P3 amplitude, Hit Rate (H), False Alarm Rate (FA), d-prime
(*d'*), Reaction Time (RT) Mean and Reaction Time Standard Deviantion
(RT-SD), percentage of Correct Trials (CT) to Go and Nogo stimuli included
in ERP analyses after artifact rejection for each of the three NF protocols.

List of Figures

3.1.	General NF Procedure.	77
3.2.	Within-session SMR and beta1 amplitude change in active periods for control NF, SMR NF and beta1 NF.	83
3.3.	Across-session SMR and beta1amplitude change for control NF, SMR NF and beta1 NF in active periods.	84
3.4.	Across-session SMR and beta1amplitude change for control NF, SMR NF and beta1 NF in passive periods.	87
3.5.	Within-session amplitude change in different frequency bands for control NF, SMR NE and betal NE protocols	89
4.1.	Grand average of the stimulus-locked ERP of target-standard and nontarget- standard difference waves in midline frontal, central and parietal regions at pre-training irrespective of the NF protocol.	102
4.2.	Two-way interaction of Stimulus and Region in mean P3 amplitude at pre- training.	103
4.3.	Grand-average of the target-standard and nontarget-standard difference ERP waveforms in the parietal region for Control, SMR and beta1 NF at pre- and	105
4.4.	post-training. Mean P3 amplitude of the target-standard and nontarget-standard difference waveforms for Control, SMR and beta1 NF at pre- and post-training.	106
4.5.	Two-way interaction of 2 Time x 3 Protocol for reaction time.	107
4.6.	Two-way interaction of 2 Time x 3 Protocol for d-prime.	110
4.7.	Two-way interaction of 2 Time x 3 Protocol for performance rates.	111
5.1.	Grand average of the stimulus-locked ERP original waveforms elicited by the Go and Nogo stimuli in midline frontal, central and parietal regions at pre-training irrespective of the NF protocol.	124

5.2.	Two-way interaction of Stimulus and Region in mean P3 amplitude at pre- training.	126
5.3.	Grand-average of the parietal Go and central Nogo ERP waveforms for Control, SMR and beta1 NF at pre- and post-training.	129
5.4.	Mean parietal Go- and central Nogo-P3 amplitudes for control, SMR and beta1 NF at pre- and post-training.	130
5.5.	Two-way interaction of 2 Time x 3 Protocol for reaction time.	131
5.6.	Two-way interaction of 2 Time x 3 Protocol for d' and performance rates.	133
8.1.	Within-session theta and beta1 amplitude change in active periods for control NF, theta NF and beta1 NF.	158
8.2.	Across-session theta and beta1 amplitude change for control NF, theta NF and beta1 NF in active periods.	160
8.3.	Across-session theta and beta1 amplitude change for control NF, theta NF and beta1 NF in passive periods.	163
8.4.	Within-session amplitude change in different frequency bands for control NF, theta NF and beta1 NF protocols.	165
9.1.	Grand average of the stimulus-locked ERP of target-standard and nontarget- standard difference waves in midline frontal, central and parietal regions at pre-training irrespective of the NF protocol.	177
9.2.	Two-way interaction of Stimulus and Region in mean P3 amplitude at pre- training.	178
9.3.	Grand-average of the parietal target-standard difference and average nontarget-standard difference ERP waveforms for control, theta and beta1 NF at pre- and post-training.	181
9.4.	Mean P3 amplitude of the target-standard and nontarget-standard difference waveforms for control, theta and beta1 NF at pre- and post-training.	181

9.5.	Two-way interaction of 2 Time x 3 Protocol for reaction time.	182
9.6.	Two-way interactions of 2 Time x 3 Protocol for d-prime.	184
9.7.	Two-way interaction of 2 Time x 3 Protocol for performance rates.	185
10.1.	Grand average of the stimulus-locked ERP original waveforms elicited by the Go and Nogo stimuli in midline frontal, central and parietal regions at pre-training irrespective of the NF protocol.	196
10.2.	Two-way interactions of Stimulus and Region in mean P3 amplitude at pre- training.	197
10.3.	Grand-average of the parietal Go and central Nogo ERP waveforms for control, theta and beta1 NF at pre- and post-training.	198
10.4.	Mean parietal Go- and central Nogo-P3 amplitudes for control, theta and beta1 NF at pre- and post-training.	199
10.5.	Two-way interaction of 2 Time x 3 Protocol for reaction time.	201
10.6.	Two-way interaction of 2 Time x 3 Protocol for d' and performance rates.	203

Chapter 1

1. Introduction

The capacities to self-regulate the allocation of attention resources and dominant response tendencies are among the most distinctive features of the human mind. The growing evidence that plasticity across the life span is a fundamental property of the brain (Pascual-Leone, Amedi, Fregni, & Merabet, 2005) has encouraged the search for neurostimulation techniques that might enhance cognitive performance in both healthy and clinical populations. Differently from other forms of neurostimulation, such as the transcranial magnetic stimulation (TMS) and the transcranial electrical stimulation (TES), neurofeedback (NF) presupposes the active engagement of the participant in the modulation of the ongoing brain activity as measured by the electroencephalographic (EEG) and blood-oxigen-leveldependent (BOLD) signals. In this context, EEG NF has received considerable interest for being based on a direct measure of cortical oscillatory activity (i.e., the EEG) that can be related to specific cognitive functions and for its potential in improving cognitive performance through self-regulation. In particular, the NF modulation of EEG amplitude to simultaneously suppress theta and amplify beta frequencies may hold promise for enhancing selective attention and response inhibition in attention deficit hyperactivity disorder (ADHD; for a recent review see Arns, Heinrich, & Strehl, 2014) and in healthy functioning adults (for a recent review see Gruzelier, 2014a, 2014b) as suggested by changes in the amplitude of endogenous event-related potentials (ERPs) and behavioural responses.

However, theta and beta oscillations reflect distinct aspects of neural computation suggesting that they might contribute differently to these cognitive improvements. Moreover, the neural mechanisms by which the NF-related changes in spontaneous oscillatory activity may affect the amplitude and latency of ERPs as well as the speed and accuracy of the behavioural responses remain unclear. Thus, the present thesis aims to explore the independent effects of EEG NF (henceforth NF) in theta and beta frequencies on ERP and behavioural correlates of selective attention and response inhibition.

1.1. The functional significance of EEG oscillations

The brain oscillatory activity can be defined in terms of its frequency, amplitude and instantaneous phase (Klimesch et al., 2007). The different oscillatory frequencies, traditionally organized into five bands - delta (0-4 Hz), theta (4-8 Hz), alpha (8-13 Hz), beta (13-30 Hz) and gamma (>30 Hz) – possibly reflect the size of specific brain networks and the extent of the functional integration of local and distributed neural processing systems (Von Stein & Sarnthein, 2000). Low frequency oscillatory activity has been proposed to reflect the modulation of brain activity over long-range connections and long temporal windows, while high frequencies might influence small spatial regions in short temporal windows (Canolty & Knight, 2010). In support of this view, gamma oscillations have been associated with local visual processing, beta synchronization with modulation of the activity in temporal and parietal neighboring cortical areas during semantic processing, and slow oscillations (e.g., alpha and theta) with the implementation of long range communication between distant fronto-parietal areas during working memory and mental imagery (Von Stein & Sarnthein, 2000).

Even though a direct association between a given cognitive function and a specific frequency band may be difficult and unrealistic to establish, it is widely accepted that different oscillations reflect distinct perceptual sensorimotor and cognitive processes (Buzsáki, 2006; Engel & Fries, 2010; Nunez, 2000; Steriade, 2006). In the resting state, the synchronized activity in high-frequencies is generally taken to be indicative of aroused states, while the synchronization of low-frequencies typically signals a state of cortical deactivation (Steriade, Amzica, & Nunez, 1993).

The amplitude component of oscillations in a certain frequency range reflects the extent to which the underlying neural populations are involved in specific cognitive processes. Variations in amplitude (or power) of the ongoing EEG activity have been shown to reflect cortical activation or deactivation in the sensory and motor system (Doppelmayr, Klimesch, Pachinger, & Ripper, 1998; Grabska-Barwińska & Zygierewicz, 2006; Müller et

al., 2003; Neuper & Pfurtscheller, 2001; Pfurtscheller & Lopes da Silva, 1999; Pfurtscheller, Stancák, & Neuper, 1996; Stancák & Pfurtscheller, 1996).

The phase synchronization between pre- and post-synaptic neural populations has been proposed as a mechanism of neural communication by which the activity of distributed neural areas is integrated and gives rise to meaningful cognitive representations (Fries, 2005, 2015; Varela, Lachaux, Rodriguez, & Martinerie, 2001). Inputs that arrive at phases of high postsynaptic excitability benefit from enhanced connectivity relative to those arriving at random phases of excitability. Thus, selective communication is implemented by the reciprocal interaction between neural assemblies in which post-synaptic neural groups communicate preferentially with the coherent pre-synaptic neural groups. Phase coherence provides an estimation of the difference in instantaneous phase angle between two electrodes within the same frequency and an indication of the functional association between neuronal populations.

Local phase coupling in different frequencies (also known as cross-frequency coupling) has been proposed to subserve the synchronization between neuronal networks with different resonance frequency (Canolty et al., 2006; Canolty & Knight, 2010; Holz, Glennon, Prendergast, & Sauseng, 2010; Sauseng, Klimesch, Gruber, & Birbaumer, 2008). In particular, it has been proposed that the phase-amplitude cross-frequency coupling by which a relationship is established between the low-frequency phase and the high-frequency amplitude might provide a mechanism for transient coupling of distinct neuronal populations and functional systems. For instance, theta phase resetting enables the amplitude modulation of theta and gamma oscillations which probably promotes the exchange of top-down (e.g., internal expectancies) and bottom-up (e.g., visual input) information between different networks (Sauseng et al., 2008).

The role of brain oscillations on event-related potentials

Despite evidence that beta/theta NF might contribute to enhance the amplitude of late P3 ERP components subserving the allocation of attention resources and response inhibition (Egner & Gruzelier, 2001; Egner & Gruzelier, 2004; Kropotov et al., 2005; Studer et al., 2014), it is still unclear how the self-regulated changes in the amplitude of theta and beta

frequencies (i.e., in the absence of stimulus modulation) might impact the stimulus related neural responses.

A likely reason for this state of affairs is that the stimulus-evoked neural responses have been traditionally considered to be of an additive nature and inherently independent from the ongoing oscillatory activity. According to this view, the ERP components reflect the neural activity in localized brain regions responsible for processing particular aspects of stimuli and responses. The flow of current resulting from the activation of a sufficient number of dendrites with the same orientation in a certain brain region creates a dipole (i.e., a pair of positive and negative charges), the position and orientation of which defines the distribution of positive and negative voltages recorded at the scalp (Luck, 2014). The temporal sequence of the averaged evoked responses, in which peaks of positive and negative polarity appear in a timely ordered manner, suggests that these brain regions become sequentially activated within certain latencies relative to stimulus onset.

Following this proposal, NF might contribute to increase/decrease the neural activity in the particular brain regions that contribute to the generation of the dipole resulting in amplitude changes of ERP components. In support of this mechanism it has been shown that the NF suppression of alpha activity, generally taken as indicating cortical deactivation, was associated with increase cortical excitability in sensorimotor areas as indicated by increased motor-evoked potentials (MEP) induced by TMS (Ros, Munneke, Ruge, Gruzelier, & Rothwell, 2010). NF might also impact the latency of the evoked neural responses through the modulation of attentional states. For instance, the enhanced top-down control may be associated with reduced transmission times leading to shorter latencies of the ERP components.

An alternative view suggests that ERP components reflect the phase reorganization of ongoing EEG oscillations (Makeig, Debener, Onton, & Delorme, 2004; Sauseng et al., 2007; Sauseng & Klimesch, 2008). Differently from the "evoked model", in which oscillations are proposed to be noise and ERP components emerge from specific locations, the "phase reorganization model" proposes that oscillations are intimately related to the generation of the ERP components and coordinate the neural processing in distributed neural assemblies (Fries, 2015; Varela et al., 2001).

The fundamental idea of this model is that task-relevant oscillations pre-existing the stimulus presentation (i.e., ongoing) and/or elicited by the stimulus (i.e., evoked) become functionally coordinated in time by transient phase alignment. Phase resetting provides a mechanism by which task-relevant ongoing oscillations reorganize their phase in order to be at the maximal excitability phase when the stimulus is processed. An ERP is produced by averaging a number of trials in which the task-relevant oscillations are phase locked to the stimulus onset (Makeig et al., 2002; Sayers, Beagley, & Henshall, 1974).

This model makes several predictions concerning the contribution of the oscillatory activity to the amplitude and latency of brain evoked response (Klimesch, Sauseng, Hanslmayr, Gruber, & Freunberger, 2007; Sauseng et al., 2007; Sauseng & Klimesch, 2008). Firstly, because the ERP components reflect the timing and frequency of the functional relevant oscillations, an increase or decrease in the average ERP latency may be determined by the inter-trial variability of phase-locking and by changes in oscillatory frequency. A reduction in inter-trial variability would contribute to increase the phase concentration around a certain time point leading either to increased or decreased latency. Moreover, because the interpeak latency is smaller for faster frequencies and larger for slower frequencies, an increase or decrease in oscillatory frequency would determine a shorter or longer ERP latency respectively.

On the other hand, the ongoing oscillatory activity at pre-stimulus might determine the magnitude of the event-related response. Importantly, phase-resetting requires a significant level of ongoing activity in the pre-stimulus period (Shah et al., 2004). As an example, the contribution of the evoked alpha activity to early ERPs amplitude was larger in participants with higher pre-stimulus alpha power (Min et al., 2007). Although there is evidence to suggest that NF might contribute to an increase in post-stimulus evoked theta activity (Enriquez-Geppert, Huster, Figge, & Herrmann, 2014), it is currently unclear whether this increase might be explained by an higher pre-stimulus theta activity induced by NF. Thus, NF might have an effect on the amplitude and latency of the ERP components by increasing or decreasing the ongoing activity of certain oscillatory frequencies relevant for stimulus processing and by decreasing the phase variability between trials.

1.1.1. Electrophysiological and behavioral studies of selective attention

From a neurobiological perspective selective attention comprises voluntary (topdown) and involuntary (bottom-up) attention processes (Corbetta & Shulman, 2002; Escera, Alho, Schröger, & Winkler, 2000; Näätänen, Paavilainen, Rinne, & Alho, 2007; Posner, 1980; Sarter, Givens, & Bruno, 2001; Womelsdorf & Fries, 2007). Voluntary attention concerns the goal-directed processing of stimulus information according to knowledge and expectations as well as the decision process leading to response selection. In contrast, involuntary attention is driven by the properties of the stimulus such as its salience, novelty and potential survival value. Importantly, these attention processes have been related to distinct brain networks and predict distinct patterns of behavior (Corbetta & Shulman, 2002).

1.1.1.1. Electrophysiological indices of selective attention in the oddball task

In both auditory and visual modalities of the three-stimuli oddball task two distinct components could be distinguished in relation to infrequent stimuli: a frontal P3a component with a short peak latency (250-400ms) associated with the distracter task-irrelevant stimulus; and a parietal P3b component, with a longer peak latency (300-600ms), elicited by the target task-relevant stimulus (Courchesne, Hillyard, & Galambos, 1975; Squires, Squires, & Hillyard, 1975). Subsequent studies have suggested that the neural response to infrequent task-relevant targets (P3b) and task-irrelevant nontargets (P3a) may reflect functionally distinct voluntary and involuntary attention processes (Debener, Kranczioch, Herrmann, & Engel, 2002; Debener, Makeig, Delorme, & Engel, 2005; Escera et al., 2000; Escera, 1990; Sawaki & Katayama, 2009).

P3b

The ERP P3b component was first described in response to an unexpected auditory tone in a background of frequent standard stimuli with maximal amplitude in central-parietal sites (Sutton, Braren, Zubin, & John, 1965). The amplitude and latency of the P3b have been found to reflect different processes (for reviews see Polich, 2007, 2012). The P3b amplitude has been found to be sensitive to manipulations task-relevance (Squires, Donchin, Herning, & McCarthy, 1977) and of global and local sequence probability with amplitude increasing as a function of both increased standard/target ratio and target-to-target interval (Polich,

2007, Polich, 2012). In the classical oddball task, the physiological processes underlying the P3b latency appear to be sensitive to the difficulty of target discrimination (Kutas,, McCarthy, & Donchin, 1977) but to be independent from those related to response selection and execution as demonstrated by unchanged latencies with varying degrees of target-response incompatibility (Magliero, Bashore, Coles, & Donchin, 1984).

In spite of being one of the most well researched ERP components, there is still some debate regarding the functional significance of the P3b. The context-updating theory suggests that the P3b amplitude reflects the update of a schema upon detection of a conflict between new information and expectations (Kamp, Brumback, & Donchin, 2013). In other words, in the context of high global and local sequence standard stimulus probability, a deviant stimulus will require an endogenous attention-driven comparison process to update the current representation. If no change is detected in the stimulus category the stimulus context is maintained and only sensory processes are involved (Donchin & Coles, 1998).

The P3b amplitude is also believed to provide information on the neural activity underlying allocation of attentional resources and processing capacity (Kok, 2001). Evidence that the P3 is modulated by the amount of attentional resources allocated comes from dual task performance studies showing that the P3 amplitude decreases and its latency gets longer as the difficulty of the primary task increases (Isreal, Chesney, Wickens, & Donchin, 1980; Kramer, Wickens, & Donchin, 1985). In the same way, increased memory load has also been associated with reductions of the P3 amplitude (García-Larrea & Cézanne-Bert, 1998; Kok, 2001). Thus, it has been assumed that the P3b reflects the degree of involvement of voluntary aspects of attention during the task.

Whether this representation update will serve any function in terms of response selection is still a matter of debate. An important proposal of the context-updating theory is that P3b reflects a planning for future responses (i.e., a strategic process) rather than the immediate decision to respond or the generation of a response to a target (i.e., a tactical process). However, other accounts suggest that it also reflects decision processes associated with response selection (Kelly & O'Connell, 2014; Nieuwenhuis, Aston-Jones, & Cohen, 2005; Verleger, Hamann, Asanowicz, & Śmigasiewicz, 2015; Verleger, Metzner, Ouyang, Śmigasiewicz, & Zhou, 2014).

Taken together these findings appear to support the idea that the P3b latency is sensitive to difficulty of stimulus categorization but remains unaffected by factors influencing response selection, whereas the P3b amplitude is sensitive to both factors that influence target discriminability and the strength of association between target and response.

РЗа

The P3a designates a family of positive waveforms observed in response to infrequent stimuli without an associated task in the three-stimuli oddball paradigm. Given the absence of an overt response, the P3a necessarily reflects a stimulus-driven process (Polich, 2012). This component presents a variable centro-parietal and fronto-central scalp distribution depending on the type of distracter stimulus and task demands (Polich, 2007, 2012). A frontal-central "novelty P3" has been first described by Courchesne and collaborators (1975) associated with infrequent unexpected novel stimuli (i.e., non-repeated complex visual stimulus) presented in a series of frequent standards and infrequent targets. This component habituates rapidly as demonstrated by decreased amplitude with repeated presentation. Therefore, the novelty P3 has been associated with the orienting response (Courchesne et al., 1975; Friedman, Cycowicz, & Gaeta, 2001; Rushby, Barry, & Doherty, 2005).

On the other hand, a "Nogo-P3" component has been identified in response to deviant typical stimuli (e.g., auditory tones of varying degrees of salience to standard and target stimuli) with variable topography depending on the attentional task demands. In contexts of relatively low attentional task demands (i.e., easy perceptual discrimination between target and standards), the P3 amplitude to deviant nontargets (P3a) elicits a similar centro-parietal topography and has been proposed to have the same functional significance as the P3b (Bennington & Polich, 1999; Jeon & Polich, 2001; Oades, Zerbin, & Dittmann-Balcar, 1995; Wronka, Kaiser, & Coenen, 2008). However, in this context the P3a component has a smaller amplitude and shorter latency relative to the P3b, reflecting the difference in the amount of change required to update the memory representation (Donchin & Coles, 1998; Hagen, Gatherwright, Lopez, & Polich, 2006).

In contexts of difficult target-standard discrimination, the P3a component exhibits a fronto-central distribution as well as an higher amplitude and shorter latency when compared to the P3b (Comerchero & Polich, 1999; Katayama & Polich, 1999; Sawaki & Katayama,

2006, 2008, 2009). The amplitude of the "Nogo-P3a" has been considered to reflect automatic capture and orientation of attention towards a deviant stimulus (Sawaki & Katayama, 2008) involuntary switching of attention (Escera et al., 2000; Escera, Alho, Winkler, & Naatanen, 1998; Näätänen, 1992) and a useful physiological index of distractibility (Berti, Roeber, & Schröger, 2004). In line with the argument that the P3a reflects an automatic process (i.e., independent of the availability of attentional resources), it was found that the P3a amplitude to an auditory stimulus was not significantly modulated by changes in difficulty of a concurrent visual task (Muller-Gass, Macdonald, Schröger, Sculthorpe, & Campbell, 2007). It is currently assumed that the "novelty P3" and the "Nogo-P3" in a difficult target-standard discrimination context are variants of the same ERP component (Polich, 2007), as suggested by the similar frontal-central topography (Polich & Comerchero, 2003) and the failure to distinguish between the two in Principal Component Analysis (PCA) (Simons, Graham, Miles, & Chen, 2001).

An alternative interpretation proposes that the P3a reflects the automatic response inhibition upon detection of deviance. This account assumes that the P3a reflects in part the detection of deviance and a response inhibition process upon proper identification of the stimulus as a nontarget (Azizian, Freitas, Parvaz, & Squires, 2006; Goldstein, Spencer, & Donchin, 2002).

1.1.1.2. Theta and beta oscillatory activity in selective attention

Theta

In healthy adults, theta rhythms are difficult to detect in the fully awake and functional resting state (Klimesch, 1999). Elevations of theta power in the resting state EEG have been attributed to the slowing of alpha activity in pathological cerebral oxygenation and blood flow (Ingvar et al., 1976), poor cognitive performance both in children and cognitively impaired adults (Klimesch, 1999) hypnagogic states (Gruzelier, 2009), transition to sleep (De Gennaro, Ferrara, & Bertini, 2001) and mind wandering states (Braboszcz & Delorme, 2011). Moreover, pre-stimulus theta power increases with decreased alertness, fatigue and drowsiness and has often been associated with increased reaction time (RT) and increased error rates (Gevins & Smith, 1999; Huang, Jung, Delorme, & Makeig, 2008; Makeig & Jung, 1996; Paus et al., 1997; Takahashi et al., 1997; Wascher et al., 2013). Increased occipital

theta power has also been related to vigilance decrements (assessed by omission errors and increased detection latency) during auditory and visual tasks (O'Hanlon & Beatty, 1977; Williams, Granda, Jones, Lubin, & Armington, 1962) which inspired the first investigations in theta suppression NF (Beatty, Greenberg, Deibler, & O'Hanlon, 1974; Williams, Beatty, & O'Hanlon, 1975). The time course of delta and theta power was also found to be increased during prolonged resting state EEG recordings (Brismar, 2007).

At the functional level, frontal midline (fm) theta oscillations have been proposed as a candidate mechanism by which the prefrontal cortex exerts top-down control (Basar-Eroglu & Demilrap, 2001; Besserve et al., 2008; Cavanagh & Frank, 2014; Mitchell, McNaughton, Flanagan, & Kirk, 2008). For instance, increased fronto-parietal synchrony in the lower frequencies was associated with attentional top-down control during visual search tasks (Buschman & Miller, 2007).

Pre- and post-stimulus theta oscillations have been proposed to play an important role in deviant stimulus detection. The pre-stimulus theta and alpha power have been related to stimulus detection. For instance, a low absolute tonic theta power in the pre-stimulus baseline correlated positively with the extent of event-related theta synchronization during target detection in a simple oddball task (Doppelmayr et al., 1998). Consistently, Makeig & Jung (1996) found evidence that pre-stimulus increases in central-parietal theta were predictive of decreased vigilance in a monitoring task, while decreased pre-stimulus theta power was associated with increased detection of targets. Pre-stimulus spectral power in the alpha band was positively correlated with P3 amplitude and negatively correlated with P3 latency (Intriligator & Polich, 1994; Jasiukaitis & Hakerem, 1988).

The oscillatory response to the deviant targets in the oddball paradigm has been characterized by increased theta evoked amplitude and phase-locking relative to the prestimulus period. Both evoked theta (Başar-Eroglu, Başar, Demiralp, & Schürmann, 1992; Basar-Eroglu & Demilrap, 2001; Cacace & McFarland, 2003; Choi et al., 2013; Fuentemilla, Marco-Pallarés, Münte, & Grau, 2008; Mazaheri & Picton, 2005; Yordanova & Kolev, 1998; Yordanova, Rosso, & Kolev, 2003) and delta power (Barry et al., 2007; Cacace & McFarland, 2003; Güntekin & Başar, 2010) have been implicated in the generation of the P3 amplitude during auditory information processing. For instance, a decrease in wavelet entropy reflecting the transient dominance in the theta frequency ERP component over other frequency components was demonstrated for both targets and standards (Yordanova et al., 2002, 2003). The prominence of theta synchronization over other EEG frequencies in the 250-300 ms post-stimulus period was hypothesized to reflect a basic processing stage during which working memory operations are facilitated and interference from other processes is minimized. A more specific involvement of the theta frequency in deviance detection was evinced by the demonstration that the P3a component and frontal-central theta exhibit similar frontal topographies and that theta activity was significantly higher for novel nontargets compared to typical nontargets in the easy and difficult three-stimuli oddball task (Demiralp, Ademoglu, Comerchero, & Polich, 2001). In line with these findings, theta power and phase alignment were found to contribute to the generation of the automatic deviance detection Mismatch Negativity frontal and temporal components (Fuentemilla et al., 2008) and strong phase synchrony in the theta range between frontal and temporal areas was evinced during the automatic detection of deviants (Choi et al., 2013).

Beta

The pioneering work of (Moruzzi & Magoun, 1949) demonstrated that stimulation of reticular formation of the brainstem was associated with the blockage of synchronized slow wave activity and with the appearance of fast low voltage oscillations characteristic of vigilance states. More recently, beta oscillations have been proposed to play a role in top-down-processing.

Computational models provide supportive evidence to the involvement of beta cortical rhythms in attentional signal processing gain by exerting endogenous/top-down control over sensory areas through the activation of ascending synaptic projections (Lee, Whittington, & Kopell, 2013). Similarly, beta oscillations have been proposed to play a role in the maintenance of the current perceptual or cognitive set. The entrainment of neuronal circuits to enhance beta power might result in decreased behavioral flexibility by raising the threshold and slowing down the response to novel and unexpected stimuli while promoting cognitive control in predictable tasks (Engel & Fries, 2010).

More specifically, increased beta power has also been related to increased vigilance. The studies of Wróbel and colleagues in cats and in humans highlighted the role of beta oscillations in controlling the top-down and bottom-up processes in the visual attention system (Bekisz & Wróbel, 1999; Gola, Kamiński, Brzezicka, & Wróbel, 2012; Wróbel, 2000). Their studies proposed that beta activity may induce a background excitation in cortico-thalamic circuits of the visual system (primary visual cortex, higher visual areas and lateral posterior and pulvinar complex). For instance, phase coupling (i.e., the synchronization across different frequencies) between bursts of beta and gamma activity (Bekisz & Wróbel, 1999) suggests that beta activity may function as a carrier of visual attention by allowing for high frequency (i.e., gamma) synchronization during feature binding (Fell, Fernández, Klaver, Elger, & Fries, 2003). Moreover, the increased prestimulus fronto-parietal synchrony in both beta and gamma bands was significantly correlated with correct target detection which was interpreted as reflecting a state of increased visual attention with an impact on visual perception (Hanslmayr et al., 2007). In line with the previous results, reduced beta phase synchronization in fronto-parietal areas has been associated with attentional lapses (Gross et al., 2004). More recently, the increased beta activity in the foreperiod of cued-Go/Nogo tasks has been proposed to be involved in the facilitation of cue processing (Kilavik, Zaepffel, Brovelli, MacKay, & Riehle, 2013).

1.1.1.3. Behavioral correlates of selective attention in the oddball task

Low-beta/theta NF has also been associated with improvements in selective attention as indicated by decreased RT and reaction time standard deviation (RT-SD), increased hit rate and overall increased perceptual sensitivity (d') in continuous performance task (CPT), Go/Nogo and divided attention tasks (DAT) (Doppelmayr & Weber, 2011; Egner & Gruzelier, 2001, 2004).

In the context of the oddball task, the same conditions that contribute to the facilitation of the voluntary attention mechanisms indexed by the P3b (e.g, discrimination difficulty, confidence in the detection of the target) have been found to affect the behavioral response. Moreover, the amplitude of the P3b was found to correlate positively with decreased RT (Nieuwenhuis et al., 2005; Pfefferbaum, Ford, Weller, & Kopell, 1985; Verleger, Jaskowski, & Wascher, 2005), decreased RT-SD (Comerchero & Polich, 1999;

Comerchero & Polich, 1998; Ramchurn, de Fockert, Mason, Darling, & Bunce, 2014) increased hit rate and *d*' and negatively with target omissions and false alarms (Hillyard, Squires, Bauer, & Lindsay, 1971; Wilkinson & Seales, 1978).

Importantly, despite being positively correlated the mean RT and response accuracy reflect different processes from the P3b.The mean RT depends on connectivity between premotor and primary motor areas involved in the selection and preparation of movement (Koch et al., 2006) which are not involved in the generation of the P3b. On the other hand, RT-SD has been recognized in recent years as an index of preserved frontal functioning and top-down control of attention in healthy, neurodevelopmental and brain damaged populations (Karalunas, Geurts, Konrad, Bender, & Nigg, 2014; Stuss, Murphy, Binns, & Alexander, 2003; Vaurio, Simmonds, & Mostofsky, 2009). There is currently evidence that the theta inter-trial phase coherence is related to the variability of RT in ADHD children and typically developing individuals across the life-span (McLoughlin, Palmer, Rijsdijk, & Makeig, 2014; Papenberg, Hämmerer, Müller, Lindenberger, & Li, 2013). Although it is unclear whether low-beta/theta NF might have an impact of theta phase, an increase in inter-trial theta phase-coherence might explain the decreases in RT-SD observed in previous studies (see table 1.2.).

In sum, the existing evidence suggests any changes in theta and beta oscillatory activity may have been responsible for the electrophysiological and behavioral improvements observed in previous low-beta/theta NF in the context of the oddball task.

1.1.2. Electrophysiological and behavioral studies of response inhibition

Inhibition is a crucial aspect of adaptive behavior in overcoming automatic responses to changing circumstances in the environment. Response inhibition implies the voluntary control or cancelation of a previously prepared motor response. Since successful cancelation of a response is not associated with an overt behavioral response, the ERP technique is especially suited to investigate the chain of cognitive events leading to response inhibition. A standard task designed to investigate response inhibition in cognitive neuroscience is the Go/Nogo paradigm in which participants are required to respond with a button-press to a Go stimulus, while refraining from responding to a lower (or equal) probability Nogo stimulus (Pfefferbaum et al., 1985). The response inhibition demands imposed by the Go/Nogo task can be defined in terms of action restraint (i.e., the ability to withhold a prepotent response tendency). Other aspects of response inhibition such as action cancelation (i.e., the ability to cancel a response after being initiated) are better described by measures of latency of inhibition (i.e. the stop signal RT) in the Stop Signal Task (SST) (Bari & Robbins, 2013; Schachar et al., 2007).

1.1.2.1. Electrophysiological indices of response inhibition in the Go/Nogo task

The investigation of the neural correlates of response inhibition has been dominated by two major research questions. The first question concerns the functional significance of the ERP components associated with the Nogo stimulus. In comparison to the Go stimulus, the Nogo stimulus elicits a larger fronto-central negative deflection in the time window around 200 and 300 ms after stimulus onset (Nogo-N2), followed by a larger central positive P3 component (Nogo-P3) component with a peak latency around 400-600 ms (Eimer, 1993; Falkenstein, Hoormann, & Hohnsbein, 1999; Pfefferbaum et al., 1985; Sasaki & Gemba, 1986; Simson, Vaughan, & Ratter, 1977).

Recent accounts currently favor the interpretation that the Nogo-N2 and Nogo-P3 components may reflect dissociable processes of response conflict and response inhibition (Bruin, Wijers, & Van Staveren, 2001; Dimoska, Johnstone, & Barry, 2006; Groom & Cragg, 2015; Randall & Smith, 2011). In Go/Nogo tasks in which the probability of the Go stimuli is higher than that of the Nogo stimuli the amplitude of the ERP components may equally reflect response inhibition or increased conflict between response representations (Botvinick, Cohen, & Carter, 2004). Recent findings suggest that the processes of response conflict and response inhibition can be dissociated in cued-Go/Nogo paradigms in which conflict can be elicited in the absence of inhibition by increasing the response activation (Donkers & van Boxtel, 2004; Randall & Smith, 2011). The N2 component was found to be increased to rare unexpected stimuli in situations in which an increase in response activation (GO) rather than response inhibition was required (Donkers & van Boxtel, 2004). The N2 amplitude was more negative when the required response was different from the planned response based on the cue (i.e., Nogo target after Go cue or to Go target after Nogo cue). The Nogo-P3 amplitude was increased when the previously planned response to the Go stimulus (due to a cue, a prepotent Go response or fast responding) had to be inhibited (Bruin et al., 2001; Dimoska et al., 2006; Randall & Smith, 2011). Furthermore, it has been consistently found that the Nogo-P3 amplitude is increased for fast responders compared to slow responders in cued Go/Nogo (Smith, Johnstone, & Barry, 2006) and in the SST paradigms (Dimoska et al., 2006), possibly reflecting the greater need for inhibitory activation with increase response preparation. Thus, the N2 to Nogo stimulus may reflect the need for a different response – response conflict rather than response inhibition *per se*. The Nogo-P3 is enhanced when a increment in response preparation has to be subsequently inhibited.

Further evidence supporting the differential role of the N2 as a marker response conflict and of the P3 as an index of response inhibition comes from an hybrid Go/Nogo flanker task (Groom & Cragg, 2015) in which the effects of response activation-inhibition and of high-low conflict could be dissociated. The N2 amplitude was enhanced on high response conflict trials (i.e., incongruent) relative to those involving low response conflict (i.e., congruent) but not in those requiring response inhibition. Moreover, P3 amplitude was enhanced in frontal sites on trials requiring response inhibition relative to response activation but not in those involving the resolution of the conflict effect. Thus, the N2 might reflect the need to select between competing responses (which in a situation of prepotent Go response is also present in the Go/Nogo task), while the frontal P3 to Nogo trials might reflect the active suppression of a motor response.

The second question that has been debated in the literature concerns whether the increased evoked responses to the Nogo stimulus reflects cognitive or movement aspects of inhibition. More specifically, some authors (Salisbury, Rutherford, Shenton, & McCarley, 2001; Salisbury, Griggs, Shenton, & McCarley, 2004) have suggested that the increased P3 amplitude in Nogo trials relative to Go trials (Nogo effect), may be related to the presence of negative movement related deflections in Go trials. Smith, Johnstone and Barry (2008) directly tested this prediction by calculating Nogo and Go difference waves in the count and press conditions. A difference between Go press and Go count waveforms, would suggest that the Nogo effect would be associated with movement-related potentials on the Go trials. Contrary to the prediction, there was no difference between Go count and Go press in the P3 amplitude window. Moreover, the Nogo-P3 in the "press" conditions was significantly higher than the in the "count" condition suggesting the Nogo effect was sensitive to response inhibition demands. Moreover, the simultaneous recording of ERP and functional Magnetic

Resonance Imaging (fMRI) confirmed that the increased Nogo-P3 during "press" conditions relative to "count" conditions was associated with a significant deactivation of motor areas related to inhibition such as the inferior frontal gyrus, the precentral gyrus and the supplementary motor area (Smith, Jamadar, Provost, & Michie, 2013).

Taken together these results suggest that the Nogo-P3 amplitude may be considered an electrophysiological index of response inhibition of a prepotent response in the context of the Go/Nogo task.

1.1.2.2. Theta and beta oscillatory activity in response inhibition

Oscillatory activity in the theta and beta range has been associated with response inhibition and executive control.

Theta

Increased fm-theta activity has been associated with both the realization for the need to control and the instantiation of control in several executive tasks (Cavanagh & Frank, 2014; Nigbur, Ivanova, & Stürmer, 2011). Evidence for the differential contribution of prestimulus delta and theta oscillations for the amplitude and latency of several ERP components associated with the Go and Nogo responses was obtained in a recent study (De Blasio & Barry, 2013a). High pre-stimulus delta amplitude was found to contibute to enhancements of both early and late components irrespective of the stimulus type. In contrast, the pre-stimulus theta amplitude was significantly modulated by stimulus type. An inverse modulation was observed for the Go stimulus (i.e., low pre-stimulus theta was associated with high Go-P3 and vice versa). However, low pre-stimulus theta activity was associated with increased Nogo-N2 and reduced Nogo-P3. Taking into consideration that the Nogo-P3 appears to be a more direct measure of response inhibition than the Nogo-N2, these results suggest that pre-stimulus theta may be involved in attention (Go-P3) but not response inhibition functions (Nogo-P3).

Other studies have proposed distinct roles for delta and theta oscillations as markers of cognitive control. In one study, the PCA decomposition revealed that post-stimulus delta and theta activity accounted for a significant proportion of the variance of the Nogo minus Go differences scores of N2 and P3 components. However, the moderate correlation between

delta and theta components in Nogo minus Go differences scores suggested that the two components may reflect dissociable process (Harper, Malone, & Bernat, 2014). The specific role of post-stimulus delta and theta synchronization in sustained attention and response inhibition was revealed in another study comparing the equiprobable Go/Nogo and the cued-CPT tasks. In both tasks, the time-frequency decomposition using wavelet transform, revealed that the Nogo condition evoked higher fm-theta coefficients than the Go condition in the early post-stimulus period (corresponding to the N1 and P3 components time windows) probably reflecting response inhibition demands. In the CPT, the theta oscillatory activity was observed to be superimposed in the delta activity suggesting that the delta component in the latency range of the N2 and P3 components may reflect sustained attention demands (Kirmizi-Alsan et al., 2006).

Beta

From a functional point of view, sensorimotor beta oscillations have been related to motor activity. Sensorimotor beta oscillations have been found to desynchronize during selfpaced or stimulus-related movement and steady-state muscle contractions (Pfurtscheller & Lopes da Silva, 1999; Stancák & Pfurtscheller, 1996) and to be transiently increased after movement cessation, a phenomenon known as beta rebound (Pfurtscheller, Neuper, Brunner, & Lopes da Silva, 2005; Pfurtscheller & Solis-Escalante, 2009). The beta rebound effect may reflect an active inhibition of the motor system as suggested by a decrease in corticospinal excitability (Chen, Corwell, & Hallett, 1999) and a positive linear correlation between concentration of the inhibitory neurotransmitter γ -Aminobutyric acid (GABA) and beta rebound power (Gaetz, Edgar, Wang, & Roberts, 2011). Alternatively, it may represent a period of recalibration of the motor system after one movement and the preparation for the next (Baker, 2007; Gaetz & Cheyne, 2006). These observations contributed to the conceptualization of beta oscillations as an "idling rhythm" of the sensorimotor areas (Pfurtscheller et al., 1996). However, rather than just reflecting movement execution beta desynchronization may be associated with perceptual and cognitive aspects of motor control (Kilavik et al., 2013; Müller et al., 2003).

More recently, it has been proposed that beta oscillations promote the maintenance of the current sensorimotor state at the expense of new movements (Engel & Fries, 2010;
Jenkinson & Brown, 2011). This view is supported by evidence that increased beta synchrony is associated with the reinforcement of existing movements and posture, and with impaired speed in the execution of new movements (Gilbertson et al., 2005). In consonance with this hypothesis, the enhancement of beta and gamma activities by transcranial alternating current stimulation (TACS) resulted in dissociable behavioral effects. The increase in beta activity was associated with antikinetic and slower movements, while the increase in gamma activity resulted in prokinetic behavior as demonstrated by measures of force in responding to a Go/Nogo task (Joundi, Jenkinson, Brittain, Aziz, & Brown, 2012).

Beta activity has also been associated with enhanced response inhibition in both humans and primates. In Go/Nogo tasks, beta activity is increased when a previously prepared response is followed by the presentation of a Nogo stimulus (Alegre et al., 2004; Kühn et al., 2004; Swann et al., 2009; Zhang, Chen, Bressler, & Ding, 2008). In the monkey, beta desynchronization was initially observed in relation to response preparation to Go and Nogo stimuli, while beta synchronization (beta rebound) was observed exclusively in relation to response inhibition in the Nogo condition and exhibited a similar coherent pattern to that observed during the pre-stimulus condition (Zhang et al., 2008). Moreover, an increase in fronto-central beta synchronization was observed after the decision to withhold a response (Alegre et al., 2004). Recently, a study using transcranial magnetic stimulation and simultaneous EEG recording (Picazio et al., 2014) found enhanced cortical connectivity in the beta frequency between the prefrontal and motor cortex (where the motor plans are inhibited or executed) and increased beta activity in the fronto-central areas following Nogo trials providing direct evidence of the inhibitory influences of the prefrontal cortex on the motor areas.

De Blasio & Barry (2013b) also suggested a differential contribution of pre-stimulus alpha and beta oscillations for the amplitude and latency of several early and late ERP components associated with the Go and Nogo responses. The pre-stimulus alpha amplitude was associated with corresponding modulations of the positive ERP components irrespective of the stimulus type (i.e., high pre-stimulus alpha was associated with increased P1, P2 and P3 amplitude and vice versa). In agreement with the role of beta oscillations in early attentional gain (Iversen, Repp, & Patel, 2009; Lee et al., 2013), the increased pre-stimulus beta amplitude was associated with increased with increased pre-stimulus beta amplitude was associated with increased amplitude of the exogenous P1 and N1

component (and decreased latency of the N1 component). However, the pre-stimulus beta amplitude was not predictive of the modulation of the late ERP components associated with the differential responses to the Go and Nogo stimulus.

In sum, the current evidence suggests the involvement of different oscillations in response inhibition. Although in the post-stimulus period both theta and beta synchronization have been associated with response inhibition (Harper et al., 2014; Kirmizi-Alsan et al., 2006; Picazio et al., 2014; Zhang et al., 2008), in the pre-stimulus period other oscillations (delta, alpha) may contribute to the ERP components associated with Go and Nogo stimuli (De Blasio & Barry, 2013a, 2013b).

1.1.2.3. Behavioral correlates of response inhibition in the Go/Nogo task

As shown in table 1.2., low-beta/theta NF has been associated with decreased false alarm rate in Go/Nogo tasks (Bakhshayesh, Hänsch, Wyschkon, Rezai, & Esser, 2011; Egner & Gruzelier, 2001; Gruzelier, Foks, Steffert, Chen, & Ros, 2014; Kropotov et al., 2005), which can be related to the ability to withhold a prepotent response tendency, a crucial aspect of response inhibition.

In the Go/Nogo task the difficulty of withholding a prepotent response tendency is determined by the degree of previous response preparation (e.g., whether a warning cue was present or absent) and the Go stimulus probability (Bruin & Wijers, 2002; Eimer, 1993). Inhibition is assumed to be successful when a correct non-response is associated with the Nogo stimulus. Everything being equal and assuming a valid cue, higher demands on response inhibition are expected when a Nogo stimulus is preceded by response priming compared to when it is unexpected. On the other hand, the higher probability of the Go stimulus probably enhances the response preparation to the upcoming trials, therefore increasing the demands on response inhibition to the less probable Nogo stimulus. (Smith, Johnstone, & Barry, 2004). Additionally, response inhibition demands may be affected by individual differences and strategies in response speed. For instance, explicit instructions to increase speed at the expense of accuracy generated more commission errors than speed and accuracy instructions (Band, Ridderinkhof, & van der Molen, 2003).

Behaviorally, the false alarm or commission error rate (i.e., the percentage of incorrect responses to the Nogo stimulus) has been taken as a measure of unsuccessful

response inhibition (Falkenstein et al., 1999). Although variations in Go/Nogo stimulus probability equally affect Nogo-P3 amplitude and the false alarm rate (i.e., both increase as a function of increasing Go/Nogo probability ratio), a functional relation between the two is difficult to establish because these measures are based on different data. In fact, while false alarm rate takes into consideration both correct and incorrect non-responses, the Nogo-P3 represents an average of the correct non-responses after excluding false alarms. Moreover, other ERP components such as the ERN (Falkenstein, Hohnsbein, Hoormann, & Blanke, 1991; Gehring, Goss, Coles, Meyer, & Donchin, 1993) may constitute a more approximated measure of the failure of inhibitory processes.

Additionally, individual differences in RT-SD (but not in mean RT) to Go responses in the context of Go/Nogo tasks have been found to be a strong predictor of successful response inhibition as demonstrated by increased activation of frontal regions in fMRI (Bellgrove, Hester, & Garavan, 2004).

1.2. Neurofeedback of theta and beta frequencies

Despite consistent evidence that improvements in cognitive performance may result from the learned modulation of certain brain signals (Birbaumer, Ruiz, & Sitaram, 2013; Fetz, 1969; Gruzelier, 2014b; Salari, Buchel, & Rose, 2014), the demonstration that the modulation of theta and low-beta amplitudes results from specific experimental manipulations, rather than from unspecific factors affecting the EEG amplitude, has faced considerable conceptual and methodological challenges.

1.2.1. Definition and conceptual aspects of Neurofeedback

A pervasive question in cognitive neurosciences is that the association between changes in brain activity and cognitive performance does not necessarily imply a causal relationship and that the direction of such correlations is often difficult to determine (Seitz, 2013). However, as discussed in sections 1.1.1.2. and 1.1.2.2., there is a substantive body of research demonstrating the functional significance of brain oscillations for attention and inhibition, thus, supporting the assumption that voluntary changes in theta and beta oscillation may affect specific cognitive functions. Although it is commonly accepted that the voluntary induction of certain mental states is associated with the modulation of EEG oscillations (Buzsáki, 2006; Nunez, 2000), it is still debated whether the direct manipulation of brain oscillations may affect the pattern of synchronization of localized and distributed neural networks that subserve specific cognitive operations.

In NF, additional difficulties in establishing a causal link between voluntary changes in brain activity and behavior result from (1) the complexity of disentangling intended changes in brain activity from epiphenomenal changes related to unspecific cognitive processes and (2) the absence of evidence that the intended changes in brain oscillatory activity are aligned in time with changes in cognitive performance.

Although it still debated the extent to which NF implicates some form of top-down self-regulation (Bagdasaryan & Quyen, 2013; deCharms, 2008) or is essentially a procedural learning skill dependent on the basal ganglia plasticity (Birbaumer et al., 2013; Ruiz, Buyukturkoglu, Rana, Birbaumer, & Sitaram, 2014; Sherlin et al., 2011), it is currently assumed that, at least in humans, unspecific cognitive processes (such as feedback learning and self-monitoring) probably mediate the learned modulation of any electrophysiological signal (Gruzelier, 2014b).

One question raised here is whether it is possible to disentangle the contribution of top-down from that of bottom-up processes to the resulting changes in brain activity. In fact, the implementation of self-regulatory processes probably recruits low frequencies implicated in cognitive control. This in turn, is likely to constrain the activity of superimposed local high-frequencies governed by cross-frequency phase-amplitude coupling mechanisms. On the other hand, the neurostimulation of the lower levels of the cognitive organization conceivably gives rise to the emergence of higher level conscious experience subserved by low frequencies (Bagdasaryan & Quyen, 2013). Two ubiquitous findings in NF research illustrate this complexity. Firstly, NF appears to have an impact on the activity of the frontal lobes irrespective of the training frequencies and training sites (Angelakis et al., 2007; Egner, Zech, & Gruzelier, 2004; Keizer, Verment, & Hommel, 2010) possibly reflecting the intervention of self-regulatory processes. On the other hand, NF probably implicates complex relations across frequencies as suggested by the amplitude modulation of frequencies that were not targeted by the intervention (Enriquez-Geppert, et al., 2014a; Keizer, Verschoor, Verment, & Hommel, 2010; Nan et al., 2012; Ros et al., 2013).

Thus, considerable difficulties in establishing cause-and-effect relationships stems from the possibility that cognitive changes may constitute a by-product of non-specific selfregulatory processes rather than a primary effect of the modulation of brain activity (Caria, 2016). Here, the comparison of NF with other forms of neurostimulation (such as TACS) that do not implicate self-regulation processes but provide considerable frequency-specificity may help solve this conundrum. Although the two techniques may be difficult to compare in terms of their local specificity and intensity of the stimulation, their differential cognitive impact can be attributed to self-regulatory or other epiphenomenal effects associated with NF.

For the establishment of cause-and-effect relationships between NF modulation and changes in cognitive performance it is also important to demonstrate a temporal association between the two. As will be seen in the next sections, previous reports of NF efficacy have relied on correlational evidence between measures of successful modulation of EEG activity during training and pre- to post-training changes in cognitive tasks which are typically obtained at different times (sometimes with intervals of days). Given the non-stationarity of the EEG signal and the influence of pre-stimulus oscillatory activity on ERPs amplitude and latency (see sections 1.1.1.2. and 1.1.2.2.), these distal NF learning measures might not be a reliable predictor of event-related EEG changes. This might explain the relatively small evidence of positive correlations between NF learning and cognitive outcome obtained in NF applications in optimal cognitive performance (Gruzelier, 2014a).

The establishment of this temporal link would be possible by using EEG-dependent interactive ERP analysis. In this method, the presentation of a stimulus eliciting ERP component is dependent on the spontaneous (Price, 1997; Rahn & Basar, 1993) or TMS induced (Price, 2004) state of the background EEG immediately preceding it. The background activity (e.g., amplitude, instantaneous phase) is then treated as an independent variable and a causal effect on the ERP dependent variable can be argued. In line with this proposal, a recent real-time fMRI neurofeedback study suggests that participants trained to control the level of spontaneous activation of the visual cortex showed a selective increase in stimulus-evoked activity when able to invoke a previously learned up-regulated state at the trained location (Scharnowski, Hutton, Josephs, Weiskopf, & Rees, 2012). Similarly, a causal relationship between NF-induced changes and ERP amplitude and latency could be achieve through the voluntary activation of the desired pattern of EEG background activity trained

during transfer trials (i.e., without feedback) and then used in interactive ERP recordings as an independent variable.

1.2.2. Models of low-beta/theta NF effects on attention and response inhibition

Over the years, several theoretical models have been proposed to explain the alleged influence of NF on attention and response inhibition. These models have been heavily influenced by demonstrations of successful operant conditioning of neuronal spike activity and brain oscillations in animals (Fetz, 1969). Therefore, these models emphasize the implicit nature of brain-regulation and provide only limited understanding of how these processes may be mediated by volitional self-regulation.

The operant conditioning of low-beta frequencies was first demonstrated in the context of the learned suppression of a previously rewarded behavioral response in cats (Roth, Sterman, & Clemente, 1967; Sterman, Wyrwicka, & Roth, 1969; Wyrwicka & Sterman, 1968). The reinforcement of a rhythmic 12-14 Hz EEG activity in the sensorimotor cortex, thereafter labeled Sensorimotor Rhythm (SMR), was associated with a behavioral pattern of quiet alertness, while the reinforcement of higher frequencies and the extinction of SMR activity led to restless and agitated behavior. Sterman and colleagues (Harper & Sterman, 1972; Howe & Sterman, 1972) hypothesized that the SMR originates in the ventrobasal nuclei (nVB) of the thalamus which is associated with conducting the afferent sensorimotor information to the cortex. The behavioral suppression of somatomotor activity and decrease in muscular tone causes the nVB firing pattern to shift from a fast and nonrhythmic to systematic burst discharges that characterize the SMR in intracranial and scalp EEG recordings. Moreover, it was proposed that the SMR was modulated by the brainstem neuromodulatory systems which, during the waking activity, maintain the depolarization of the thalamic structures and prevent the occurrence of the rhythmic activity. Thus, SMR synchronization may reflect an active state of behavioral inhibition characterized by the absence of movement or of intention to move. The SMR activity was also proposed to influence the attentional processes by reducing the interference of sensory and somatomotor information processing in other brain areas (Mann, Sterman, & Kaiser, 1996).

With the advent of NF applications to the management of resistant epileptic seizures in humans (for a review and meta-analysis see Sterman & Egner, 2006 and Tan et al., 2009), the reinforcement of SMR was complemented by the suppression of theta and high-beta activities justified by the need to signal epileptiform slow and spike activity respectively (Lubar & Bahler, 1976). The goal of enhancing beta relative to theta amplitude was further grounded on the hypothesis that a low-beta/theta ratio may be a physiological marker of cortical arousal responsible for the inattention and impulsivity/hyperactivity symptomatology of ADHD (Mann, Lubar, Zimmerman, Miller, & Muenchen, 1992). Lubar (1997) expanded this conceptualization by proposing that the increase in higher frequencies relative to lower frequencies would facilitate the transition from a hypercoupled to a hypocoupled state (i.e., from a resonate mode characterized by long range communication to more localized and specialized information processing). In support of this theory Lubar, Swartwood, Swartwood, & O'Donnell (1995) reported that ADHD children evinced behavioural improvements in composite measures of attention and response inhibition in the Test of Variables of Attention (TOVA, Greenberg, Kindschi, & Corman, 1996; for a description of the task see section 1.2.4.1.) following an intensive beta1/theta NF regime. Interestingly, these improvements were significantly higher in those participants that showed a higher decrement in theta activity. However, given the composite nature of the cognitive measures the exact implications of theta suppression for the processes of attention and response inhibition remained elusive.

Othmer, Othmer, & Kaiser (1999) later proposed that SMR/theta NF in the right hemisphere and beta1/theta NF in the left hemisphere would have distinct implications for response inhibition and attention processes. This distinction was justified by "a crucial fulcrum point" that "appears to exist in the vicinity of 15 Hz, above which the subjects will be moved toward sympathetic arousal and below which parasympathetic dominance will be promoted" (pp. 282). Based on a model of hemispheric specialization (Tucker & Williamson, 1984), beta1/theta NF in the left hemisphere (C3) was proposed to compensate the underactivation responsible for inattentiveness, while SMR/theta NF in the right hemisphere (C4) would regulate the excessive right hemispheric activity responsible for impulsivity. Although the distinct behavioral and cognitive outcomes of SMR/theta NF and beta1/theta NF have been confirmed in subsequent studies (Doppelmayr & Weber, 2011; Egner & Gruzelier, 2001, 2004), the assumptions regarding hemispheric specialization and frequency specificity remain to be confirmed.

1.2.3. Learning specificity

The claims that SMR and beta1 NF might be related to specific cognitive outcomes are insufficiently supported by evidence of training- and frequency-specific effects. Following recent proposals (Gruzelier, 2014b; Zoefel, Huster, & Herrmann, 2011) the demonstration of specific learning effects should encompasses two aspects: *trainability* or *training-specificity* (i.e., the demonstration that the EEG changes resulted from the training and not from unspecific factors) and *independence* or *frequency specificity* (i.e., the demonstration that the training selectively impacted the EEG parameters targeted by the intervention when other frequencies in the EEG spectrum are taken into consideration). Thus, the disentanglement of specific from unspecific learning effects should be critically dependent on the demonstration of changes in the experimental NF manipulation relative to a comparison or control condition. For example, *training-specificity* was defined as a significant directional change in the frequency of interest (e.g., increments in SMR amplitude) in response to a specific experimental manipulation (e.g., SMR NF) when compared with unspecific comparison or control conditions (e.g., beta1 or control NF).

Table 1.1. presents a summary of the absolute and relative amplitude changes involving theta, SMR, beta1 frequencies in previous studies. As shown in the table, the success of the NF experimental manipulations has been assessed through both changes during active feedback periods and in the passive resting state. In active periods, learning has been defined by both within- and across-sessions changes, while in the passive resting state it has been determined by the across session change in pre-training baseline or by the difference in QEEG power between the beginning and the end of the training.

The within- and across-session distinction evinces the dynamic EEG changes in different time scales. During neurofeedback learning trials, the reinforcement of EEG activity is achieved by rewarding directional changes in the EEG parameter (i.e, amplitude increases or decreases) in relation to a reference period (e.g., pre-training resting state baseline). The repetition of this associative learning process across several training trials in the same training session defines the *within-session learning* process. To the extent that the re-

establishment of the previously learned association is facilitated through repetition over a variable number of sessions a carry-over effect from one session to the next may result, evincing across-session learning. A second aspect that must be taken into consideration when determining the specificity of learning concerns the state of the organism during the EEG recording. EEG exhibits distinct Given that patterns of synchronization/desynchronization during the resting state and task performance it is reasonable to assume that the learning process may affect differently the EEG amplitude during *passive resting state* and *active feedback* periods.

1.2.3.1. Within- and across-session changes in active feedback periods

1.2.3.1.1. SMR

Within-session

Despite abundant demonstrations of successful SMR conditioning in single-case and non-randomized controlled studies (Lubar & Bahler, 1976; Lubar & Shouse, 1976; Lubar et al., 1981; Sterman & Friar, 1972; Sterman, Howe, & Macdonald, 1970; Sterman et al., 1974), only recently within-session SMR amplitude increments were demonstrated in controlled studies. In two of these, within-session SMR amplitude increments were evinced by significantly positive slopes from baseline to the last training period in the SMR/theta NF, while the same correlational analysis was not significant in sham NF conditions (Kober et al., 2015; Witte, Kober, Ninaus, Neuper, & Wood, 2013). In two other studies, the cognitive effects of SMR/theta NF were compared with other NF and control conditions (see table 1.2.) but within-session amplitude changes in SMR, theta and SMR/theta ratio were only investigated for the SMR/theta NF (Ros et al., 2009; Vernon et al., 2003). Interestingly, both studies showed SMR/theta ratio amplitude increments but Ros and collaborators. (2009) could not find evidence of SMR amplitude increments. Other studies suggest that despite SMR/theta ratio amplitude increments the SMR amplitude was decreased relative to the initial baseline period in the SMR/theta NF (de Zambotti, Bianchin, Magazzini, Gnesato, & Angrilli, 2012; Gruzelier, Foks, Steffert, Chen, & Ros, 2014).

Table 1.1. Summary of the absolute and relative amplitude changes involving theta, SMR, beta1 frequencies in previous studies. The changes are denoted by the overall trend towards increase (Inc), decrease (Dec) or no change (NC) for each EEG parameter (EEG) during active periods (within- and across-session) and passive periods in pre-training resting state baseline (Baseline) and from pre- to post-training qEEG (qEEG) following different NF protocols including Random Frequency (RF) NF, sham NF. The number of NF sessions (No.) per condition is also indicated.

Cturdu.		No	EEG —	Ac	tive	Passive		
Study	NF FIOLOCOI	NO.		Within	Across	Baseline	qEEG	
Vernon et al. (2003)	SMR/(T+B2)	8	SMR/T	Inc	n/a	n/a	n/a	
			SMR	Inc	n/a	n/a	n/a	
			Т	Dec	n/a	n/a	n/a	
			SMR/B2	NC	n/a	n/a	n/a	
Egner et al. (2004)	SMR/(T+HB)	10	SMR	n/a	n/a	n/a	NC	
			Other	n/a	n/a	n/a	NC	
	B1/(T+HB)	10	B1	n/a	n/a	n/a	NC	
			Other	n/a	n/a	n/a	NC	
Hoedlmoser et al.	SMR	10	Abs.SMR	n/a	NC	NC	NC	
(2008)			Rel.SMR	n/a	Inc	NC	NC	
	RF	10	Abs.SMR	n/a	NC	NC	NC	
			Rel.SMR	n/a	NC	NC	NC	
Ros et al. (2009)	SMR/(T+HB)	8	SMR/T	Inc	Inc *	n/a	n/a	
			SMR	NC	n/a	n/a	n/a	
			Т	Dec	n/a	n/a	n/a	
Doppelmayr &Weber (2011)	SMR/(LT+HB)	30	SMR	n/a	Inc	n/a	n/a	
			т	n/a	NC	n/a	n/a	
			B1	n/a	NC	n/a	n/a	
			B1/T	n/a	NC	n/a	n/a	
	B1/T Ratio	30 &	SMR	n/a	NC	n/a	n/a	
	& RF	30	т	n/a	NC	n/a	n/a	
			B1	n/a	NC	n/a	n/a	
			B1/T	n/a	NC	n/a	n/a	
Weber et al. (2011)	SMR/(LT+HB)	25	SMR	n/a	Inc	n/a	n/a	
	SMR/(LT+HB)	25	SMR	n/a	NC**	n/a	n/a	
de Zambotti et al.	SMR/T	16	SMR/T	Inc	Inc	Dec	n/a	
(2012)			SMR	Dec	Inc	n/a	n/a	
			Т	Dec	NC	n/a	n/a	
Becerra et al. (2012)	Tsupression	30	Abs.T	n/a	n/a	n/a	Dec	
			Abs.A	n/a	n/a	n/a	Inc	
	Sham	30	Abs.T	n/a	n/a	n/a	NC	
			Abs.A	n/a	n/a	n/a	NC	
	Tsupression	30	Rel. T	n/a	n/a	n/a	Dec	
			Rel. A	n/a	n/a	n/a	Inc	
	Sham	30	Rel. T	n/a	n/a	n/a	Dec	
			Rel. A	n/a	n/a	n/a	Inc	

Study	NE Protocol	No	FEG	Ac	tive	Passive		
Study	NEFICIOCO	NO.		Within	Across	Baseline	qEEG	
Kober et al. (2013)	SMR	10	SMR	n/a	Inc	n/a	n/a	
			G	n/a	NC	n/a	n/a	
	Gamma	10	SMR	n/a	NC	n/a	n/a	
			G	n/a	NC	n/a	n/a	
Witte et al. (2013)	SMR/(T+HB)	10	SMR	Inc	NC	NC	n/a	
	Sham		SMR	NC	NC	NC	n/a	
Gruzelier et al.	SMR/(T+HB)	10	SMR/T	Inc	NC	n/a	n/a	
(2014a)			SMR	Dec	n/a	n/a	n/a	
			т	Dec	n/a	n/a	n/a	
			SMR/HB	NC	NC	n/a	n/a	
Gruzelier et al. (2014b)	SMR/(T+HB)	10	SMR/T	NC	NC	n/a	n/a	
			SMR/HB	Inc	Inc	n/a	n/a	
Schabus et al. (2014)	SMR	10	SMR	n/a	Inc	n/a	n/a	
			В	n/a	Inc	n/a	n/a	
	RF	5	SMR	n/a	NC	n/a	n/a	
			В	n/a	NC	n/a	n/a	
Kober et al. (2015)	SMR/(T+HB)	10	SMR	Inc	NC	n/a	n/a	
			т	NC	NC	n/a	n/a	
			HB	Inc	n/a	n/a	n/a	
	Sham	10	SMR	NC	NC	n/a	n/a	
			т	NC	NC	n/a	n/a	
			НВ	NC	n/a	n/a	n/a	

Table 1.1. continued

Low -theta (LT):3-5Hz; Theta (T):4-7Hz; Alpha (A):8-12Hz; SMR:12-15Hz; Beta (B):16-25Hz; B1:15-18HZ; B2:18-22Hz; High-beta (HB):22-30Hz; Gamma (G):40-43Hz; Abs.=absolute amplitude; Rel.= Relative amplitude; * = first half of training; ** = non-learners

Across-session

There is mixed evidence regarding across-session SMR learning following both SMR/theta and SMR (i.e., without theta and high-beta suppression) NF. Doppelmayr and Weber (2011) investigated normalized amplitude changes in SMR, beta1 and theta as function of three NF conditions: SMR/theta, beta1/theta ratio and random-frequency (RF NF). For each of the frequency bands, the mean amplitude was calculated for six blocks of five sessions (each block corresponding to a week of training) and normalized to the mean amplitude in the first block. The SMR amplitude change was significantly higher in the final relative to the initial training block in the SMR/theta NF but not in the beta/theta ratio NF and RF NF suggesting training specific SMR learning.

Another study investigating relative and absolute SMR amplitude changes across ten sessions of SMR NF and RF NF suggests that the demonstration of SMR learning might depend on the method of calculating across-session amplitude changes (Hoedlmoser et al., 2008). In this study, the relative SMR amplitude (calculated by dividing the mean amplitude during active trials for each session by the mean amplitude of the pre-training resting state) was found to increase from early to late training sessions in SMR but not in RF NF. However, the SMR absolute amplitude (calculated by simply averaging the within-session amplitude in active trails) in SMR NF was not reliably increased across-sessions when compared to the RF NF.

In line with these results, Kober and collaborators (2013) revealed linear increases in SMR amplitude across-sessions (as evinced by significant regression slopes in SMR NF), while no increases were observed in a gamma-enhancement NF. Other studies have provide evidence of SMR amplitude increments following SMR/theta NF but the failure to report SMR amplitude changes in comparison conditions (de Zambotti et al., 2012; Ros et al., 2009) or the unequal number of sessions in the control condition (Schabus et al., 2014) prevents further conclusions regarding training-specificity.

In contrast with this body of evidence, recent controlled studies (Kober et al., 2015; Witte et al., 2013) have failed to demonstrate across-session SMR amplitude increments, in spite of evidence of within-session learning. Here, the linear increments in SMR amplitude across ten SMR/theta NF sessions were not significantly different from those obtained by Sham NF conditions. However, given that only within-subjects analysis were reported, these results should be interpreted with caution when considering its implications for the training specificity of across-session SMR increments.

There are several possible explanations for these inconsistent findings. Besides the different methods of defining across-session learning (e.g., relative vs. absolute) mentioned above, the low number of NF sessions might have affected the development and consolidation of an appropriate self-regulatory strategy in some studies (e.g., Kober et al., 2015; Ros et al., 2009). However, other studies with equivalent number of sessions have provided evidence of training-specific SMR increments (e.g., Hoedlmoser et al., 2008; Kober, Witte, & Ninaus, 2013). Additionally, the failure to observe across-session SMR learning may be related to individual differences in learning. One study indicates that a

percentage as high as 50% of the participants may fail to show evidence of across-sessions SMR amplitude increments and that SMR learners can only be reliably discriminated from non-learners after more than ten sessions (Weber et al., 2011). Finally recent studies suggest that individual differences in strategy adoption and consistency of training may play a significant role in across-session learning (Ros et al., 2009; Witte et al., 2013).

1.2.3.1.2. Beta1

Within-session

As shown in table 1.1., no controlled studies to date investigated the within-session amplitude changes in the sensorimotor beta1 frequency.

Across-session

Doppelmayr and Weber (2011) failed to provide supportive evidence of acrosssession beta1 learning after thirty beta1/theta NF sessions relative to SMR/theta NF and RF NF. However, as noted above, it is unclear whether the absence of significant across-session beta1 amplitude changes may be associated with the method of calculating across-session amplitude changes, individual differences in learning and/or any other factors affecting the development of a consistent self-regulatory strategy. Speculatively, the failure to inhibit highbeta frequencies in the beta1/theta NF could have contributed to a lesser control over states of excessive cortical excitability incompatible with successful sensorimotor beta conditioning (Wyrwicka & Sterman, 1968).

1.2.3.1.3. Theta

Within-session

Jackson Beatty and colleagues (Beatty, Greenberg, Deibler, & O'Hanlon, 1974; O'Hanlon & Beatty, 1977; Williams, Beatty, & O'Hanlon, 1975) first demonstrated the possibility of voluntarily modulating relative occipital theta amplitude (i.e., 3-7 Hz/3-30 Hz Hz) during prolonged radar monitoring tasks. In these studies, healthy adult volunteers were able to suppress and enhance theta ratio amplitude relative to a condition were feedback was not provided. These pioneering findings are, however, difficult to integrate with the subsequent literature focusing on functional distinct central and fm-theta rhythms (EnriquezGeppert et al., 2014a; Wang & Hsieh, 2013). In fact, posterior theta rhythms often associated with drowsiness and mental effort (Schacter, 1977; Vogel, Broverman, & Klaiber, 1968; Williams, Granda, Jones, Lubin, & Armington, 1962), may be functional distinct from the central-midline theta activity that has been related to executive control. Moreover, because of the lengthier NF sessions those early studies are difficult to compare with subsequent ones in terms of the time course of theta amplitude, i.e., the tendency for theta amplitude to increase over time especially during monotonous and long duration task (Huang, Jung, Delorme, & Makeig, 2008; Maltez, Hyllienmark, Nikulin, & Brismar, 2004; Wascher et al., 2013).

More recently, the investigation of theta-suppression in central electrodes following low-beta/theta NF independent from task performance has provided conflicting evidence. As shown in table 1.1., in several SMR/theta NF studies in healthy adults the within-session increases in SMR/theta ratio amplitude were underpinned by theta decrements (de Zambotti et al., 2012; Gruzelier, Hirst, Holmes, & Leach, 2014; Ros et al., 2009; Vernon et al., 2003). In all of these studies, theta amplitude decreased linearly from the beginning to the end of the training session. However, none of these studies reported within-session theta amplitude changes in comparison or control conditions. Thus, from these studies it remains unclear whether theta amplitude decrements may reflect NF unspecific effects such as time on task, cognitive effort or placebo. Importantly, other SMR/theta NF studies suggest that withinsession theta amplitude decrements are not necessarily a by-product of unspecific NF effects. In fact, other studies failed to provide evidence of within-session theta amplitude decrements or changes in the SMR/theta ratio in active or sham NF (Kober et al., 2015; Gruzelier, Foks, Steffert, Chen, & Ros, 2014).

Across-session

As shown in table 1.1., few studies to date examined across-session theta amplitude changes in the context of low-beta/theta NF. However, the available evidence suggests that theta amplitude changes remain constant across sessions (de Zambotti et al., 2012; Doppelmayr & Weber, 2011; Kober et al., 2015), i.e., the extent of theta desynchronization from the beginning to the end of the sessions was not higher as NF training progresses.

1.2.3.2. Across-session changes in passive periods

As previously noted, the impact of theta suppression and low-beta enhancement NF on the resting state theta and beta activity has been assessed exclusively through measures of across session change in pre-training baseline or by the difference in QEEG power between the beginning and the end of the training.

1.2.3.2.1. SMR

Contrary to active periods, Hoedlmoser and collaborators (2008) found no evidence of across-session relative SMR amplitude changes in resting state pre-training baseline. Similarly, subsequent studies either failed to evince SMR amplitude increments or even showed SMR amplitude decrements across pre-training baselines, despite observable increments in active periods (de Zambotti et al., 2012; Witte et al., 2013). Moreover, in line with one previous study (Egner, Zech, & Gruzelier, 2004), Hoedlmoser and collaborators (2008) failed to provide supportive evidence of the impact of SMR NF on the resting state SMR activity as measured by QEEG at the beginning and at the end of training irrespective of whether relative or absolute SMR power was considered. Taken together, these findings support the interpretation that successful SMR amplitude increments might depend on an active self-regulatory mechanism (probably involving the control of voluntary movement) which has little impact on the resting state SMR amplitude.

1.2.3.2.2. Beta1

Similarly to SMR, one previous study failed to provide supportive evidence of beta1 amplitude changes as measures by pre- and post-training resting state QEEG following beta1/theta NF (Egner et al., 2004). Moreover, a recent review of the controlled studies examining the effects of beta1/theta NF on ADHD did not find evidence to support the claims of enduring changes in the theta/beta ratio in QEEG measures (Arns et al., 2014).

1.2.3.2.3. Theta

Regarding theta amplitude changes in passive resting state, only one study provided supportive evidence of successful theta supression in normal elderly subjects relative to a agematched control group receiving non-contingent feedback (Becerra et al., 2012). For the theta suppression NF group the abnormally elevated absolute theta power was identified across the scalp in relation to a normative database and feedback was provided based on the EEG location where the z-score was most deviated from the normal. After thirty sessions of NF, the absolute theta power in frontal and midline electrodes was significantly reduced in the experimental group relative to controls. However, the relative theta power was significantly reduced in the suppression (over frontal electrodes) and sham NF (across the scalp). Given the varying locations of these effects and the lack of homogeneity of the training sites care must be taken when extrapolating these findings to central midline theta rhythms.

1.2.3.3. Frequency-specific effects

The self-regulation of EEG activity is a complex cognitive task involving specific mental processes associated with the functional properties of target frequency (e.g., active motor inhibition in SMR conditioning) and unspecific mental processes also present in other forms of feedback learning such as attention and performance monitoring, cognitive control and motivational aspects (Gruzelier, 2014b). Some of these unspecific aspects are known to have a direct effect on the EEG such as post-reinforcement fast wave synchronization during EEG conditioning (Clemente, Sterman, & Wyrwicka, 1964) and increases in theta power following negative feedback and in beta power following positive feedback in feedback learning (Cohen & Ranganath, 2007; Luft, Nolte, & Bhattacharya, 2013). Moreover, since any complex mental activity is probably the result of a concertation of multiple oscillatory frequencies (Canolty & Knight, 2010) it comes with no surprise that changes in frequencies other than the ones targeted by the NF interventions has been a prevalent observation in research (Gruzelier, 2014c).

1.2.3.3.1 SMR frequency-specific effects

Besides the already mentioned theta amplitude decrements, SMR/theta NF has been associated with amplitude changes in beta and high-beta frequencies. As shown in table 1.1., previous studies have found conflicting evidence regarding the effects of SMR/theta NF on beta and high beta frequencies. Even despite beta and/or high-beta amplitude suppression instructions, some studies found increments in these frequencies (Kober et al., 2015; Schabus et al., 2014) or could not find evidence that the amplitude in these frequencies became lower

than in SMR as evinced by unchanged SMR/beta and SMR/high-beta ratio amplitudes (Gruzelier, et al., 2014a; Vernon et al., 2003). Furthermore, two of these studies suggest that SMR/theta NF specifically increased the amplitude of higher beta frequencies relative to control conditions (Kober et al., 2015; Schabus et al., 2014). The latter results are surprising in light of the proposed antithetic effects of SMR and higher frequencies conditioning demonstrated in previous non-randomized controlled studies (Lubar & Shouse, 1976; Wyrwicka & Sterman, 1968). The reduced number of NF sessions in some studies may explain this lack of frequency-specificity. In fact, previous non-randomized controlled studies suggest that SMR amplitude increments become independent from increments in adjacent alpha and high beta frequency bands (Lubar & Shouse, 1976; Sterman & Friar, 1972).

In contrast, other studies found no evidence of concomitant changes in beta1 amplitude (Doppelmayr & Weber, 2011) or in gamma frequency (Kober et al., 2013) and found an increase in SMR/high-beta ratio amplitude (Gruzelier et al., 2014b) suggestive of a differentiation between SMR and high beta amplitude after SMR/theta NF.

1.2.3.3.2. Beta1 frequency-specific effects

As shown in table 1.1., fewer studies investigated the specificity of beta1 amplitude increments relative to other frequencies. Doppelmayr and Weber (2011) found no evidence of significant across-session amplitude changes in theta, SMR and beta1 frequencies in beta1/theta NF relative to other NF protocols. Similarly Egner and collaborators (2004) could not find evidence of pre- to post-training QEEG changes in beta1 power or in other frequencies ranging from delta to high-beta. Consequently, these studies were inconclusive regarding the frequency-specificity of beta1 amplitude increments.

1.2.3.3.3. Theta frequency-specific effects

A shown in table 1.1., one previous study found that theta suppression NF was associated with significant absolute and relative alpha amplitude increments (Becerra et al., 2012). However, while the absolute amplitude measures suggest that theta suppression and alpha increments were observed in the theta suppression but not in sham NF, the relative amplitude measures showed that sham NF was also associated with significant theta decrements and alpha increments. Thus, theta suppression NF may be associated with specific decrements in theta power but also with power changes in other frequencies, namely with an increase in alpha power. However, it is unclear whether unspecific effects of training may explain these changes.

1.2.4. Low-beta/theta NF effects on attention and inhibition

Table 1.2. summarizes the specific effects of SMR/theta and beta1/theta NF on electrophysiological and behavioral measures of selective attention and response inhibition in healthy adults and in ADHD children. To the exception of two studies (Egner & Gruzelier, 2001; Kropotov et al., 2005) only those that could provide evidence of the distinct cognitive effects of the two NF protocols will be discussed. Other studies that provided evidence of cognitive improvements in these domains following mixed protocols combining SMR/theta with beta1/theta NF (Fuchs, Birbaumer, Lutzenberger, Gruzelier, & Kaiser, 2003; Lévesque, Beauregard, & Mensour, 2006; Rossiter, & Lavaque, 1995), and beta1/theta with slow cortical potentials (SCP) (Gevensleben, Holl, Albrecht, Schlamp, et al., 2009; Gevensleben, Holl, Albrecht, Vogel, et al., 2009; Liechti et al., 2012; Wangler et al., 2011) will not be considered because of difficulties in interpreting the protocol specificity of the results. It is worth noting that, although other NF interventions have been associated with cognitive improvements¹, low-beta/theta NF has received by far the most attention from the literature as an intervention capable of specifically influencing the thalamocortical mechanisms responsible for variations in alertness and behavioral inhibition in both healthy and clinical populations.

1.2.4.1. Mixed low-beta frequencies effects on attention and response inhibition

The impact of low-beta/theta NF on the amplitude of ERP components indexing selective attention (P3b) and response inhibition (Nogo-P3) was demonstrated in both healthy adults and ADHD children.

¹ Cognitive improvements in the young and elderly healthy adults have been associated with frontal-midline theta (Enriquez-Geppert, Huster, Figge, et al., 2014; Wang & Hsieh, 2013) alpha (Angelakis et al., 2007; Escolano, Aguilar, and Minguez, 2011; Hanslmayr et al., 2005; Nan et al., 2012; Zoefel et al., 2011) and gamma enhancement (Keizer, Verment, et al., 2010; Keizer, Verschoor, et al., 2010; Salari and Rose, 2013; Staufenbiel et al., 2014).

 Table 1.2.
 Outcome of studies investigating NF effects on selective attention and response inhibition. The pre- to post-training change was classified as Increased (Inc), Decreased (Dec) or No Change (NC) for the variables Mean Reaction Time (RT), Reaction Time Standard Deviation (RT SD), Accuracy, Hit rate, False Alarm (FA) rate, Go-P3 amplitude and Nogo-P3 amplitude in different cognitive tasks. Information on the type of NF protocol, electrode location and number of NF sessions (No.) is also shown.

Study	NF (electrode)	No.	Task	Pre- to post-training change							
				RT	RTV	Accuracy	Hit rate	FA rate	Go-P3	Nogo-P3	
Egner & Gruzelier (2001)*	SMR **	10	TOVA	NC	NC	Inc	NC	Dec	n/a	n/a	
			auditory oddball	n/a	n/a	n/a	n/a	n/a	Inc	n/a	
	B1 **	10	TOVA	NC	NC	Dec	NC	Inc	n/a	n/a	
			auditory oddball	n/a	n/a	n/a	n/a	n/a	Inc	n/a	
Vernon et al. (2003)	SMR/B2 (Cz)	8	Digit sequence recognition	n/a	n/a	n/a	Inc	n/a	n/a	n/a	
	T/A (Cz)	8	,	n/a	n/a	n/a	NC	n/a	n/a	n/a	
	Wait list	0		n/a	n/a	n/a	NC	n/a	n/a	n/a	
Egner & Gruzelier (2004)	SMR/T (Cz)	10	TOVA	NC	Dec (marginal)	Inc	NC	NC	n/a	n/a	
			divided attention	Dec	Dec	Inc	Inc	NC	n/a	n/a	
			auditory oddball	n/a	n/a	n/a	NC	NC	NC	n/a	
	B1/T (Cz)	10	TOVA	Dec	Dec (marginal)	NC	NC	NC	n/a	n/a	
			divided attention	Dec	NC	NC	NC	NC	n/a	n/a	
			auditory oddball	n/a	n/a	n/a	NC	NC	Inc	n/a	
	Alexander	10	TOVA	NC	NC	NC	NC	NC	n/a	n/a	
	reennique		divided attention	Dec	NC	NC	NC	NC	n/a	n/a	
Kropotov et al. (2005)*	B1(C3-Fz) + SMR (C4-Pz)	15-22	Go/Nogo	Dec	Dec	n/a	Dec	Dec	Inc	Inc	
()	0		Go/Nogo***	NC	NC	n/a	Dec	Dec	NC	NC	
Ros et al. (2009)	SMR/(T+HB)(Cz)	8	Time on task	Dec	n/a	n/a	n/a	n/a	n/a	n/a	
	A/T (Pz)		Time on task	Dec (marginal)	n/a	n/a	n/a	n/a	n/a	n/a	
Doppelmayr & Weber (2011)	SMR/(LT+HB) (C3-C4)	30	simple reaction time	Dec	n/a	n/a	n/a	n/a	n/a	n/a	
	()		choice reaction time	Dec	n/a	n/a	n/a	n/a	n/a	n/a	
			spatial rotation	n/a	n/a	n/a	Inc	n/a	n/a	n/a	
			d2	n/a	n/a	n/a	NC	NC	n/a	n/a	
	B1/T Ratio (C3-C4)	30	simple reaction time	NC	n/a	n/a	n/a	n/a	n/a	n/a	
			choice reaction time	NC	n/a	n/a	n/a	n/a	n/a	n/a	
			spatial rotation	n/a	n/a	n/a	NC	n/a	n/a	n/a	
			d2	n/a	n/a	n/a	NC	NC	n/a	n/a	
	RF (C3-C4)	30	simple reaction time	Inc	n/a	n/a	n/a	n/a	n/a	n/a	
	(00 04)		choice reaction time	Inc	n/a	n/a	n/a	n/a	n/a	n/a	
			spatial rotation	n/a	n/a	n/a	NC	n/a	n/a	n/a	
			d2	n/a	n/a	n/a	NC	NC	n/a	n/a	

Table 1.2. continued

Study	NF (electrode)	No.	Task -	Pre- to post-training change						
,	(RT	RTV	Accuracy	Hit rate	FA rate	Go-P3	Nogo-P3
Bakhshayesh et al. (2011)	Beta1/Theta (CPz-FCz)	30	CPT	Dec	NC	n/a	NC	NC	n/a	n/a
			cancellation	Inc	n/a	Inc	Inc	Dec	n/a	n/a
	EMG- Biofeedback	30	CPT	NC	NC	n/a	NC	NC	n/a	n/a
			cancellation	NC	n/a	NC	NC	NC	n/a	n/a
Gruzelier et al. (2014b)	SMR/(T+HB)(Cz)	10	TOVA	NC	NC	NC	NC	Dec (marginal)	n/a	n/a
	A/T (Pz)	10		NC	NC	NC	NC	Dec	n/a	n/a
	Wait list	0		NC	NC	NC	NC	n/a	n/a	n/a
Kober et al. (2015)	SMR/(T+HB)	8	auditory oddball	NC	n/a	Inc	n/a	n/a	n/a	n/a
	(02)		Sternberg	n/a	n/a	Inc	n/a	n/a	Inc	n/a
	Sham-NFT	8	auditory oddball	Dec	n/a	NC	n/a	n/a	n/a	n/a
	(02)		Sternberg	n/a	n/a	Inc	n/a	n/a	NC	n/a

Low-theta (LT):3-5Hz; Theta (T):4-7Hz; Alpha (A):8-12Hz; SMR:12-15Hz; B1:15-18HZ; B2:18-22Hz;High-beta (HB):22-30Hz; *= SMR+beta1/(T+HB) NF; **= amplitude; ***= non-learners

As shown in table 1.2., to date, just one study investigated the effects of lowbeta/theta NF on the Nogo-P3 amplitude (Kropotov et al., 2005). In each session, the ADHD children started by attempting to enhance the right-hemispheric SMR amplitude relative to lower (1-11 Hz) and higher frequencies (19-30 Hz) for 7-10 minutes. In the second half of the sessions the enhancement of beta1 amplitude relative to the same lower and higher frequencies was trained for 20 minutes. For statistical analysis, the participants were divided in two groups (learners and non-learners) according their ability to increase the neurofeedback parameter by 25% in the training periods relative to the resting state baseline in more than 60% of the sessions. At pre- and post-training the participants performed an auditory equiprobable cued-Go/Nogo task. The analyses revealed significant pre- to posttraining increases in Go-and Nogo-P3 amplitude only in the learners group. At the behavioural level, separate statistical analysis for learners and non-learners revealed significant decreases in omission and commission errors in both groups. However, just the learners group was associated with decreased RT and RT-SD. Taken together, these results suggest that the successful enhancement of relative beta amplitude (i.e., the learners group) may be associated with improvements in successful response inhibition (Nogo-P3) but not necessarily with decreasing unsuccessful response inhibition (commission errors). Moreover, increased processing efficiency of task-relevant information (i.e., increased Go-P3, decreased RT and RT-SD) suggest a modification in the responding strategy of learners. A strategic increase in response speed, may have imposed higher demands on response inhibition that could have been reflected in increased Nogo-P3 amplitude (Dimoska et al., 2006; Smith et al., 2006).

Several methodological limitations recommend a careful interpretation of these findings. Firstly, the failure to provide the relevant statistical interactions of time (pre vs. post-training), stimulus (Go vs. Nogo) and group (learners vs. non-learners) precludes the interpretation that the response inhibition effects were specific to the learners group. Another limitation of this study was the variable number of sessions (as shown in table 1.2.). Although the authors provide the criteria for the different moments to terminate the treatment (according to age, ADHD type, learning curves and parent reports), the number of sessions was not taken into account in the statistical analysis comparing learners and non-learners.

In healthy adults, Egner and Gruzelier (2001) first provided evidence that SMR and beta1 amplitude increments had significant positive correlations with improvements in response inhibition and selective attention respectively. Here, each thirty-minute session consisted of two equal periods of SMR/theta NF at C4 and beta1/theta NF at C3 in counterbalanced order over the ten training sessions. The P3b amplitude in a divided attention auditory oddball task was taken as a measure of improved selective attention. The oddball task consisted of auditory tones of different pitch and frequency of occurrence presented in two blocks. In the first block participants were instructed to attend to the task-relevant channel (left ear) and to ignore the simultaneous stimulus presentation in the task irrelevant channel (right ear). In the second block the task relevance of the channels was reversed. Stimulus discrimination context was made difficult by the frequency proximity of the pure sinusoidal auditory tones (Standard = 1000 Hz; Target = 1100 Hz). In the absence of a control condition, the pre- to post-training comparisons revealed significant increments in the peak P3 amplitude (averaged across nine electrodes distributed across frontal, central and parietal regions) in the post-stimulus 250-400ms time window.

Moreover, the behavioural performance was investigated in the TOVA. In this choice RT task the target and non-target visual stimuli were presented in two fixed order conditions (infrequent target and frequent target) with a Go/Nogo ratio of 3.5/1 (reversed in infrequent target condition). The results revealed that the mixed low-beta/theta NF was associated with significant reductions in commission errors and increases in perceptual discrimination (*d'*), while no changes were observed in omission errors, mean RT and RT-SD from pre- to post-training.

As shown in table 1.2., further *post-hoc* exploratory analysis attempted to understand whether enhancements in SMR and beta1 amplitude could differentially predict variations in electrophysiological and behavioural performance. The ratio between successive withinsession SMR and beta1 amplitude increments and decrements was taken as an index of SMR and beta1 learning. Partial correlations analysis revealed that both SMR and beta1 enhancements were positively correlated with increased peak P3b amplitude. However, the two learning indices predicted different behavioural outcomes. The SMR learning was positively correlated with reduced commission errors and increased d' when controlling for beta1 learning, whereas beta 1 learning was significantly correlated with increased commission errors and reduced d' when controlling for corresponding enhancements in SMR amplitude. However, the practical significance of these partial correlations may be questioned by the fact that SMR and beta1 enhancement were also positively correlated. These results provided support for the interpretation that SMR and beta1 might be functionally distinct frequencies and motivated a subsequent investigation of this hypothesis in a between subjects design.

1.2.4.2. SMR and beta1 NF differential effects on attention and response inhibition

To date, the specific effects of SMR/theta NF and beta1/theta NF on attention and inhibition were compared in two studies (section 1.2.4.2.1). Other randomized controlled studies investigated these effects separately for SMR/theta NF (section 1.2.4.2.2) and for beta1/theta NF (section 1.2.4.2.3).

1.2.4.2.1. Comparison between SMR/theta NF and beta1/theta NF cognitive outcomes

Based on evidence that SMR and beta1 amplitude increments could predict different patterns of behavioral outcomes (Egner & Gruzelier, 2001), Egner and Gruzelier (2004) compared the effects of SMR/theta NF and beta1/theta NF in a between-subjects design using the same electrophysiological and behavioral measures (see section 1.2.4.1.). Differently from that previous study, the electrode placement was changed to Cz (with left earlobe reference) for both NF protocols and an active control condition was included.

In the auditory oddball task, the P3b amplitude was increased in the beta1/theta NF across the scalp (average of frontal, central and parietal derivations), while no significant changes were observed in the SMR/theta NF. However, because the amplitude of the P3b component was not calculated for other stimulus categories (i.e., standard stimulus in the attended channel, target and standards in the non-attended channel) it is unclear whether the P3 amplitude was selectively increased to the target stimulus relative to other stimulus categories. Moreover, it was not possible to determine whether the P3 amplitude to task-irrelevant (i.e., presented in the unattended channel) frequent or infrequent stimulus (P3a) was changed as a function of any of the NF protocols. Thus, from these results it was not possible to determine whether voluntary or involuntary selective attention processes were influenced.

The changes in behavioral performance were investigated in two attention tasks: the TOVA and the DAT. In the TOVA, the mean RT was significantly modulated by the type of NF, with the *post-hoc* comparisons revealing a significant decrease in beta1 NF. However, because only global mean RT scores were reported (i.e., average of frequent and infrequent target conditions) it was unclear whether this effect could have been modulated by different target frequency contexts. The RT-SD was significantly decreased from pre- to post-training across all groups. In the DAT the participants were instructed to respond to left ear channel high pitch tones and right ear channel low pitch tones with left or right button presses respectively while ignoring the other stimulus category in the relevant channel. Importantly, targets and nontargets were presented with equal probability in each channel and were difficult to discriminate (given the physical proximity of the tones 1000 Hz and 1100 Hz). Differently from the TOVA, in the DAT the analysis failed to provide evidence of a significant modulation of mean RT by the type of NF. Also differently from the TOVA, the

RT-SD was significantly decreased following SMR/theta NF relative to the other training conditions. Taking into consideration the increased P3b amplitude and the decreased mean RT in the beta1/theta NF, these results were interpreted as indicating an increased cortical excitability/arousal state due to enhanced beta activity and improved modulation of the locus coeruleus-norepinephrine system (Nieuwenhuis et al., 2005).

The changes in performance accuracy were assessed by *d'*, omission and commission errors in the TOVA and in the DAT. Both tasks revealed significant *d'* increases following the SMR/theta NF. Moreover, this group was associated with a significant decrease in omission errors in the DAT suggesting that the physiological regulation of sensorimotor activity might contribute to reduce processing interference and subsequently increase the resources available for higher order attentional processing. However, these effects were not protected by significant omnibus effects. Moreover, this study did not corroborate the hypotheses advanced by the previous study (Egner & Gruzelier, 2001) that the SMR/theta NF would be associated with improvements in response inhibition due to the reduced cortical excitability in sensorimotor areas (Sterman, 1996) and that the beta1/theta NF would contribute to an increase in commission errors. However, it is unclear whether the failure to observe differential effects on response inhibition might have been related to the low response inhibition demands of the tasks for the healthy adults.

Overall, these results suggested that the SMR/theta NF and the beta1/theta NF were associated with a facilitation of different parameters of target information processing efficiency under different tasks conditions. However, it was unclear whether the differential behavioral performance could be explained by the different attention mechanisms, the stimulus modality, the difficulty of target-nontarget discrimination or the target probability imposed by the tasks. Importantly, the failure to provide evidence of training- and frequency-specific effects of SMR/theta NF and beta1/theta NF in the QEEG assessments reported in Egner and collaborators (2004), recommends a cautious interpretation of these findings.

In another study, Doppelmayr and Weber (2011) provided further evidence of differential effects of SMR/theta NF and beta1/theta NF in the attentional domain. In comparison with a RF NF condition, only the SMR/theta NF was associated with decreased RT in a simple target visual detection task (button-press to targets presented with a variable inter-stimulus interval), decreased RT in a choice RT visual oddball task (button-press to

targets and no response to non-targets; target/non-target = 1/5.34) and increased accuracy in a spatial rotation task. However, none of the NF protocols was associated with significant differences in omission and commission errors in a standardized sustained visual attention task (d2,Brickenkamp & Zillmer, 1988). Thus, this study suggests that SMR/theta NF was specifically associated with increased processing speed in sustained and selective attention tasks without affecting performance accuracy. These results are in disagreement with those obtained by Egner and Gruzelier (2004). Since that study suggested that the SMR/theta NF may be implicated in increased performance accuracy but not necessarily with increased processing speed. Regarding beta1/theta NF, Doppelmayr and Weber (2011) could not replicate the protocol-specific increments in processing speed observed in the TOVA (Egner & Gruzelier, 2004).

The different conclusions of the two studies might be explained by methodological differences and limitations. As discussed in section 1.2.3.1.2., a major limitation of this study was the failure to demonstrate successful beta1/theta learning, therefore, critically affecting the possibility of comparing the differential effects of the two low-beta NF protocols. Differently from Egner and Gruzelier (2004), in which separate absolute amplitude signals were delivered for each frequency band (i.e., beta1, theta and high-beta), in Doppelmayr and Weber's (2011) study a single feedback signal was based on the beta1/theta ratio amplitude. Also differently from that previous study, there were more SMR/theta NF sessions (thirty vs. ten) and the theta frequency was suppressed between 3 and 5 Hz rather than between 4 and 8 Hz. However, it is unclear which of these factors might have contributed to the differences in processing speed between the two studies. The two studies also differed in the type of control condition. While in Egner and Gruzelier (2004) the control condition did not involve NF training, in Doppelmayr and Weber's (2011) study the RF involved the suppression of low theta (3-5 Hz) and high-beta (variable frequency band) as well as the counterbalanced enhancement and suppression of target 1 Hz frequency (randomly selected from the 6-35 Hz frequency range) in each half of the training session. Thus, besides the different control for NF placebo effects, the two control conditions might have involved different levels of cognitive effort. Moreover, the cognitive effects associated with the suppression of low-theta and high-beta activity in RF NF could have masked cognitive improvements in the beta1/theta ratio NF.

1.2.4.2.2. Outcome specificity of SMR/theta NF

Several studies have provided supportive evidence of facilitated processing of taskrelevant information following SMR/theta NF relative to control or comparison conditions (Gruzelier et al., 2014b.; Kober et al., 2015; Ros et al., 2009; Vernon et al., 2003). Vernon and collaborators (2003) investigated the effects of SMR/theta NF complemented by beta2 inhibition (18-22 Hz), theta enhancement with concomitant suppression of delta and alpha amplitude and wait-list controls in a two-digit sequence detection task. After eight NF sessions, only the SMR/theta NF demonstrated a significant increase in the percentage of correct detections. However, in the absence of a significant omnibus time x group effect, these results must be cautiously interpreted.

In line with these results, Kober and collaborators (2015) recently provided supportive evidence of the contribution of SMR/theta NF to increased accuracy. SMR/theta NF was associated with decreased error rates but not with decreased RT in a two-stimuli auditory oddball task. Notably, sham NF was associated with decreased RT but not with increased accuracy, suggesting that SMR/theta NF might promote accuracy at the cost of speed. In the same study, there was evidence that the N1 and P3 amplitudes elicited by the encoding phase of the Sternberg recognition task were significantly increased in SMR/theta NF but not in sham NF. In this task, the participants memorized a set of four or six digits for 1000ms (encoding phase) and were subsequently probed to indicate by button-press whether a single digit was part of the memory set (recognition phase). Despite the absence of improvements in performance accuracy relative to sham NF, the increased amplitude of the N1 and P3 components in the encoding phase was interpreted as indicating increased strength and availability of processing resources to allocate attention during encoding. Moreover, the P3 amplitude increments at encoding were positively correlated with training-specific withinsession increments in SMR amplitude.

Additionally, the authors attempted to test the hypothesis that SMR/theta NF would positively impact cognitive performance by increasing the availability of attentional resources and reducing processing interference (Sterman, 1996). Unlike controls, SMR/theta NF was associated with reduced SMR resting state functional connectivity between the motor and visual areas, which was taken as an indication of reduced sensorimotor interference. However, the relation between the reduced connectivity between motor and visual areas and improved cognitive performance was not entirely clear because the SMR/theta NF seemed to benefit performance accuracy in auditory tasks (e.g., oddball task) but not in visual tasks (e.g., Sternberg task).

SMR/theta NF was associated with enhanced visuo-motor processing speed during surgical procedures as judged by expert independent raters relative to alpha/theta NF (Ros et al., 2009). Importantly, the successful within-session SMR/theta learning was driven by within-session theta amplitude decrements rather than by SMR increments (see section 1.2.3.1.1.). However, school children treated with SMR/theta NF did not outperform those that undergone alpha/theta NF or no-treatment controls in attention and response inhibition scores measured by the TOVA (Gruzelier et al., 2014b). On the contrary, commission errors were significantly decreased in the alpha/theta NF and marginally decreased in SMR/theta NF. Here, the failure to increase the SMR/theta amplitude ratio may partly explain the absence of significant cognitive improvements.

In conclusion, there is conflicting evidence of SMR/theta NF contribution to increased efficiency of target detection. In fact, there is no current evidence that SMR/theta NF may be associated with improved processing of targets as indexed by the P3 amplitude in different cognitive paradigms (Egner & Gruzelier, 2004; Kober et al., 2015). At the behavioural level, some studies provided evidence of increased processing speed (Doppelmayr & Weber, 2011; Ros et al., 2009) and decreased RT-SD (Egner & Gruzelier, 2004) to target stimulus, while other studies failed to corroborate this evidence (Egner & Gruzelier, 2001, 2004; Kober et al., 2015). These inconsistencies were observed under similar task conditions. For instance, in the oddball tasks, decreased RT was observed in one study (Doppelmayr & Weber, 2011) but not in another (Kober et al., 2015). Given that both studies provided evidence of SMR learning and equivalent training contingencies, the higher number of sessions in the former might explain the different outcomes. On the other hand, there is some consensus regarding the SMR/theta NF contribution to increased performance accuracy (Doppelmayr & Weber, 2011; Egner & Gruzelier, 2001, 2004; Kober et al., 2015; Vernon et al., 2003). However, the absence of significant omnibus effects in some studies suggest that increased target detection may not correspond to specific SMR/theta NF outcome (Egner & Gruzelier, 2004; Vernon et al., 2003). The available evidence does not

support a specific SMR/theta NF contribution to decreased false alarm rate (Doppelmayr & Weber, 2011; Egner & Gruzelier, 2004; Gruzelier et al., 2014b).

1.2.4.2.3. Outcome specificity of beta1/theta NF

There is increasing evidence to support the specificity of beta1/theta NF in managing inattentiveness and impulsivity/hyperactivity symptoms in paediatric ADHD in comparison with pharmacological interventions according to parent reports and that the treatment effects may be maintained at a 6-month follow-up according to parents and teachers ratings (Duric, Assmus, Gundersen, & Elgen, 2012; Meisel, Servera, Garcia-Banda, Cardo, & Moreno, 2013; Monastra, Monastra, & George, 2002; for a meta-analysis Arns, de Ridder, Strehl, Breteler, & Coenen, 2009).

However, the impact of the beta1/theta NF on objective measures of attention ad response inhibition is less clear. Monastra and collaborators (2002) revealed, that compared to pharmacological intervention, the combination of NF and medication had superior effects in maintaining the reductions in the theta/beta ratio and the improvements in the TOVA indices of attention and response inhibition after a one-week medication washout. However, because the group assignment was non-randomized and the children in the NF intervention group had more contact with a therapist these results could equally be explained by a placebo effect.

Bakhshayesh and collaborators (2011) showed that ADHD children increased the response speed to infrequent targets in a visual cue-CPT task (Knye et al., 2004) following thirty sessions of beta1/theta NF in central scalp locations (FCz/CPz), but not following electromyographic (EMG)-biofeedback targeted at reducing EMG artifacts. In disagreement with the interpretation of a facilitated processing of targets, there was no evidence of increased hit rate or decreased RT-SD in the cued-CPT and the number of hits per time was reduced in a paper-and-pencil cancelation task suggesting an increase in RT. There was also conflicting evidence regarding the effects of beta1/theta NF on response inhibition. In comparison with the control group, there was evidence of decreased commission errors in the paper-and-pencil cancelation task, but not in the cued-CPT. However, the weak correlation of the cued-CPT with response inhibition (Maoz, Aviram, Nitzan, Segev, & Bloch, 2015) warrants caution in the interpretation of these findings.

In sum, beta1/theta NF resulted in unchanged or decreased commission errors effects depending on whether healthy adults (Egner & Gruzelier, 2004) or ADHD children (Monastra et al., 2002) were assessed in the TOVA. This conflicting evidence may be related to differences in sensitivity of the measures to detect improvements in response inhibition in the two populations. In these relatively non-demanding task, healthy adults have typically produced a negligible percentage of false alarm errors resulting in "floor effects" (e.g., Egner & Gruzelier, 2004), while ADHD children are expected to commit significantly more errors than controls (Greenberg, Kindschi, & Corman, 1996). Notwithstanding, there was no evidence of decreased commission errors when healthy adults were tested in more demanding tasks like the DAT (Egner & Gruzelier, 2004).

Moreover, several studies suggest that beta1/theta NF might be associated with decreased RT across different task conditions in ADHD (Bakhshayesh et al., 2011; Monastra et al., 2002) and healthy adults (Egner & Gruzelier, 2004). However, other studies failed to corroborate these findings possibly due to unsuccessful beta1/theta NF learning (Doppelmayr & Weber, 2011). The beta1/theta NF was associated with increased P3b amplitude suggesting increased efficiency in processing task-relevant information in a low Go-stimulus probability context (Egner & Gruzelier, 2004). Moreover, there is mixed evidence of the effects of bea1/theta NF on target detection and on response inhibition (Bakhshayesh et al., 2011; Egner & Gruzelier, 2004).

1.2.4.3. Theta suppression effects on attention and response inhibition

Previous studies suggest that decreases in theta and theta/SMR amplitude may play an important role in improving selective attention (Vernon et al., 2003), motor control and response inhibition (Gruzelier et al., 2014b; Ros et al., 2009). Additionally, higher pre-training theta activity and larger decreases in theta activity from pre to post-training predicted the positive impact of mixed beta/theta and SCP NF on inattention and hyperactivity (Gevensleben et al., 2009a). However, from these studies it remains unclear whether theta amplitude decrements were training-specific because no comparison with controls was provided. In one recent study, Becerra and collaborators (2012) found that theta supression was associted with improved general cognitive functioning in healthy elderly (as measured by the WAIS-III and the NEUROPSI). However, these improvements were indistinguishable

from those obtained by sham NF control group. Moreover, because theta amplitude was suppressed in individualized scalp locations (corresponding to the maximal theta amplitude in QEEG topographical maps) these findings provide only limited evidence of the possible role of central midline theta suppression to cognitive improvement.

Other studies went further in establishing a causal relationship between the voluntary modulation of relative theta amplitude and improvements in cognitive performance. In the classical investigations of Beatty and colleagues the suppression of occipital theta (3-7 Hz) amplitude relative to other EEG frequencies (3-30 Hz) was associated with increased performance in radar monitoring tasks (Beatty et al., 1974; O'Halon, Royal, & Beatty, 1977; Williams et al., 1975). On the other hand, in line with demonstrations of increased occipital theta and vigilance decrement (O'Hanlon & Beatty, 1977), the reversing of the above NF conditions resulted in performance deterioration as measured by increased error rates. However, it remains unclear whether the improved performance may reflect the facilitation of specific mechanisms involved in the modulation of visual attention (Yamagishi et al., 2003) or increased cognitive control more generally (Cavanagh & Frank, 2014). In line with the latter, recent studies suggest that frontal-midline theta enhancement NF may have a significant impact on executive functioning in healthy adults (Enriquez-Geppert et al., 2014a.,2014b; Wang & Hsieh, 2013). However, only tasks involving proactive cognitive control (such as Task-Switching) but not those involving reactive inhibitory control (such as the SST) were associated with increased post-stimulus theta power and behavioral performance (Enriquez-Geppert et al., 2014b) possibly reflecting the active instructions to develop self-regulatory and metacognitive strategies. Taken together these studies suggest that both theta enhancement (in frontal midline) and suppression (in posterior electrodes) has been implicated in improved cognitive control. However, the specific functional role of central midline theta suppression remains unclear.

1.3. Research questions and goals

Having reviewed the theoretical background and empirical investigations that suggest improvements in selective attention and response inhibition following neurofeedback of lowbeta amplitude enhancement and theta amplitude suppression two main research questions emerge. The first question concerns the specific contribution of these frequencies for the pattern of cognitive improvements observed following NF. The second question is related to the specific nature of the processes of selective attention and response inhibition that may be influence by low-beta and theta frequencies.

What are the specific contributions of low-beta (SMR and beta1) enhancement and theta suppression for the improvements in selective attention and response inhibition?

As was apparent in the literature review of section 1.2., there are a number of caveats with respect to the distinct cognitive effects of SMR/theta and beta1/theta NF protocols in healthy adults. In fact, the attempts to disentangle the cognitive effects of the two protocols have produced contradictory evidence (Doppelmayr & Weber, 2011; Egner & Gruzelier, 2004) and did not take into consideration possible hemispheric specificities as proposed by Othmer and colleagues (1999). In fact, increases in attention have been associated with both SMR/theta NF (Doppelmayr & Weber, 2011) or beta1/theta NF (Egner & Gruzelier, 2004). Regarding response inhibition, the improvements associated with SMR amplitude increments and deterioration associated beta1 amplitude increments (Egner & Gruzelier, 2001) were not replicated in subsequent studies. Moreover, the failure provide evidence of learning-specificity for both NF protocols in one study (Egner & Gruzelier, 2004) and for beta1/theta ratio NF in the other (Doppelmayr & Weber, 2011) limits the conclusions that can be drawn from these studies.

Additionally, the examination of the several indices of learning suggested that the cognitive enhancements associated with the low-beta/theta NF protocol could be explained by significant theta suppression as much as by low-beta enhancements (de Zambotti et al., 2012; Gevensleben et al., 2009; Gruzelier, Foks, Steffert, Chen, & Ros, 2014; Lubar et al., 1995; Ros et al., 2009). Although the independent supression of theta activity may be associated with increased vigilance (Beatty et al., 1974; Becerra et al., 2012; Williams et al., 1975) the existing evidence could not be related to the central-midline theta activity that is usually suppressed in the low-beta/theta NF protocol. Taking into consideration that both low pre-stimulus (or tonic) and high post-stimulus (or phasic) fronto-central theta amplitude were associated with improved selective attention and response inhibition (see sections 1.1.1.2. and 1.1.2.2.), the present thesis will attempt to clarify whether a NF intervention

aimed at suppressing theta amplitude might explain the performance enhancement in these cognitve domains.

What are the specific mechanisms of selective attention and response inhibition associated with the self-regulated modulation of theta and low-beta frequencies?

As seen in previous sections, the extent to which the selective attention and response inhibition improvements may be revealed in Go/Nogo paradigm critically depends on the cognitive demands imposed by the Go stimulus probability of the task. The selective attention demands are increased when the Go stimulus probability in low and an occasional distracter stimulus captures attention (e.g., in the oddball task). In contrast, response inhibition is required when an habitual response to a frequent Go stimulus has to be withhold (Go/Nogo task).

However, previous studies reported commission and omission errors in the TOVA irrespective of the Go stimulus probability (Egner & Gruzelier, 2001, 2004; Gruzelier, Foks, Steffert, Chen, & Ros, 2014; Fuchs et al., 2003; Rossiter and Lavaque, 1995) or in equiprobable Go/Nogo tasks (Kropotov et al., 2005) in which the specific selective attention and response inhibition demands are difficult to ascertain. Moreover, the relatively low response inhibition demands imposed by the TOVA in healthy adults might have determined "ceiling and floor effects" in the behavioral measures of selective attention and response inhibition (Egner & Gruzelier, 2001, 2004). Thus, it is currently unclear whether the contradictory evidence within and between studies regarding the cognitive implications of SMR and beta1 NF might have been related to methodological limitations concerning the specificity and sensitivity of the tasks.

Additionally, the examination of ERP components reflecting selective attention and response inhibition was not conclusive regarding the impact of the low-beta/theta NF in these processes. The study of single electrophysiological measures of selective attention in the context of the classical oddball (i.e., P3b) did not allow for a distinction between voluntary or involuntary mechanisms of selective attention indexed by the P3b and P3a respectively. Moreover, the specificity of the P3b increments relative to other task conditions (i.e., standard tones) could not be established (Egner & Gruzelier, 2001, 2004). Also, the P3 amplitude increments to both Go and Nogo stimuli in ADHD (Kropotov et al., 2005) could be interpreted as reflecting practice effects rather than improved response inhibition.

Moreover, given that a mixed low-beta/theta NF protocol was used these effects could not be related to theta suppression and/or SMR- and beta1 enhancement. Finally, the differential effects of theta suppression and low-beta enhancement on successful response inhibition remain to be explored in a healthy adult population.

Aims of the thesis

The present thesis attempted to further investigate the differential selective attention and response inhibition effects of theta- suppression and SMR- and beta1-enahncement NF inhibition in healthy adults. A primary concern of the thesis was to support the assertions regarding the specific cognitive effects on evidence of training- and frequency-specific effects. In order to achieve this goal, the feasibility of inducing specific EEG amplitude change in the different frequencies was investigated in within- and across-session learning indices and in active and passive restive state. This also implied that the relative contribution of theta suppression was disentangled from that of SMR and beta1 enhancement in different experimental conditions. This was implemented in two studies. In the first study (see Chapter 2), the effects of SMR and beta1 NF in the cognitive processes of interest were compared with a RF NF control condition. In the second study (see Chapter 7), the effects of thetainhibition NF protocol (theta NF) and beta1 NF in the selective attention and response inhibition were compared with a RF NF control condition.

The other main goal of the thesis was to better understand the differential impact of these NF protocols on specific mechanisms of selective attention and response inhibition. The selective attention and response inhibition demands were experimentally manipulated by varying the Go response probability of two Go/Nogo tasks. The differential effects of the NF protocols on voluntary and involuntary selective attention mechanisms were investigated in the three-stimuli oddball task, while the same effects on the cancelation of a previously prepared and prepotent Go response were examined in the cued Go/Nogo task.

Chapter 2

Study 1: The effects of SMR NF and beta1 NF on selective attention and response inhibition

2.1. Introduction

The proposal that SMR enhancement would improve the regulation of inhibitory functions, whereas beta1 enhancement would increase alertness and attention to the external environment was previously investigated in two studies (Doppelmayr & Weber, 2011; Egner & Gruzelier, 2004). These studies confirmed distinct cognitive outcomes in the two low-beta frequencies, but provided incompatible results in terms of the facilitation of selective attention and were inconclusive regarding the improvements in response inhibition. Moreover, the ability to differentiate between treatment effects was confounded by the failure to provide evidence of training-specific effects in beta1/theta NF (Doppelmayr & Weber, 2011) or in both NF protocols (Egner et al., 2004; Egner & Gruzelier, 2004). Following Othmer and collaborators (1999) the current study attempted to clarify whether the SMR in the right hemisphere and beta1 NF in the left hemisphere would be associated with (1) training- and frequency-specific SMR and beta1 amplitude increments (2) and with specific improvements in selective attention and response inhibition.

2.1.1. SMR NF effects on selective attention and response inhibition

SMR learning

The present study attempted to replicate previous within-session (Vernon et al., 2003; Ros et al., 2009; Kober et al., 2015) and across-session (e.g., Doppelmayr & Weber, 2011; Hoedlmoser et al., 2008; Kober et al., 2013) SMR amplitude increments during active feedback trials. Although previous studies failed to provide evidence of across-session training-specific SMR amplitude increments in passive periods (Arns et al., 2012; Egner et al., 2004; Hoeldmoser et al., 2008), the present study investigated within- and across-session SMR amplitude changes in passive periods.

Finally, differently from previous studies (e.g., Kober et al., 2015; Schabus et al., 2014) the frequency-specificity of SMR amplitude changes was examined in a large spectrum of frequencies and not only in the frequencies targeted by the NF intervention.

Cognitive effects of SMR NF

The present study attempted to replicate previous evidence of the association of SMR/theta NF with decreased mean RT (Doppelmayr & Weber, 2011; Egner & Gruzelier, 2004), decreased RT-SD (Egner & Gruzelier, 2004), increased perceptual sensitivity and hit rate (Egner & Gruzelier, 2004; Kober et al., 2015; Ros et al., 2009; Vernon et al., 2003), and decreased false alarm rate (Egner & Gruzelier, 2001; Gruzelier et al., 2014a). Moreover, these effects were investigated in two Go/Nogo tasks imposing different demands on selective attention and response inhibition.

2.1.2. Beta1 NF effects on selective attention and response inhibition

Beta1 learning

The present study attempted to provide novel evidence of betal learning. In fact, previous studies either did not investigate within-session betal amplitude changes or failed to provide evidence of across-session learning during active feedback (Doppelmayr & Weber, 2011) or passive periods (Egner et al., 2004) following betal/theta NF.

Cognitive effects of beta1 NF

The present study attempted to replicate previous evidence that beta1/theta NF associated with increased P3b amplitude in the oddball task (Egner & Gruzelier, 2004), decreased mean RT (Egner & Gruzelier, 2004), decreased perceptual sensitivity (Egner & Gruzelier, 2001) and increased hit rate (Bakhshayesh, Hänsch, Wyschkon, Rezai, & Esser, 2011). Moreover, the current study attempted to clarify whether beta1/theta NF might contribute to increased (Egner & Gruzelier, 2001) or decreased false alarm rate (Bakhshayesh et al., 2011). In sum, it was investigated whether beta1 NF may promote a fast but not

necessarily accurate response mode (Egner & Gruzelier, 2004) in Go/Nogo tasks imposing different selective attention and response inhibition demands.

2.2. Outline of Study 1

Experiment 1 sought to determine whether the amplitude changes in the SMR and beta1 frequencies were training-specific (i.e., could be attributed to the experimental manipulation and rather than to unspecific factors). Given that the adoption of single learning measure may result in the failure to apprehend non-overlapping learning phenomena occurring in the different time scales and in different EEG parameters, this implied an exploratory investigation of different methods (within- and across-session) and types of periods (active feedback periods and passive resting state periods). The combination of these resulted in four learning-indices: (1) within-session amplitude changes in active periods, (2) across-session amplitude changes in active periods, (3) within-session amplitude changes in passive periods, and (4) across-session amplitude changes in passive-periods. Moreover, the frequency-specificity of amplitude changes (i.e., the extent to which the experimental manipulation selectively affects the frequency of interest) was investigated by exploring the effects of SMR NF and beta1 NF in other EEG frequencies. Frequency-specificity was investigated for active and passive periods based on the average within-session amplitude change.

Experiment 2 examined the relative contribution of SMR and beta1 NF to selective attention in three-stimuli oddball task. Increments in P3b amplitude were taken as suggestive of improvement in endogenous /voluntary attention (Donchin & Coles, 1998), while increments in P3a amplitude were assumed to reflect an increase in involuntary attention (e.g., Courchesne, Hillyard, & Galambos, 1975; Sawaki & Katayama, 2009) and response-inhibition (e.g., Goldstein, Spencer, & Donchin, 2002).

Experiment 3 examined the relative contribution of SMR and beta1 NF to response inhibition in a cued-Go/Nogo task. An increment in the Nogo-P3 amplitude was assumed to reflect a more efficient decision to withhold the previously prepared response. Additionally, the false alarm rate was taken as a measure of unsuccessful inhibition of a previously prepared response. Because unsuccessful inhibition is not reflected in the Nogo-P3 amplitude
(which consists of averaged correct non-responses) and changes in efficiency of successful inhibition cannot be measured by a null behavioral response, the electrophysiological and behavioral measures are believed to provide complementary information regarding the successfulness of the inhibitory process.

Increases in performance accuracy were described in terms of increased hit rate, decreased false alarm rate and increased perceptual sensitivity or d' (e.g., Katayama, & Polich, 1996; Hillyard et al., 1971). Increases in response speed have been proposed to reflect the difficulty of target discrimination and response selection imposed by the task. The mean RT was expected to decrease as a function of a faster categorization of the stimulus and closure of the decision process associated with the response (Comerchero & Polich, 1998; Comerchero & Polich, 1999). Finally, decreases in RT-SD to Go responses were regarded as reflecting a higher moment-to-moment consistency in behavioral responses subserved by improvements in top-down and inhibitory control (Bellgrove, Hester, & Garavan, 2004).

Chapter 3

Experiment 1: SMR and beta1 learning

3.1. Hypotheses

Hypothesis 1: Training-specificity of within-session SMR and beta1 amplitude changes in active periods.

It was hypothesized that:

- (1) SMR amplitude would be increased within-sessions in active periods in SMR NF relative to beta1 NF and control NF.
- (2) Beta1 amplitude would be increased within-sessions in active periods in beta1 NF relative to SMR NF and control NF.

Hypothesis 2: Training-specificity of across-session SMR and beta1 amplitude changes in active periods.

It was hypothesized that:

- SMR amplitude would be increased across-sessions in active periods in SMR NF relative to beta1 NF and control NF.
- (2) Beta1 amplitude would be increased across-sessions in active periods in beta1 NF relative to SMR NF and control NF.

Hypothesis 3: Training-specificity of within-session SMR and beta1 amplitude changes in passive periods.

It was hypothesized that:

- SMR amplitude would be increased within-sessions in passive periods in SMR NF relative to beta1 NF and control NF.
- (2) Beta1 amplitude would be increased within-sessions in passive periods in beta1 NF relative to SMR NF and control NF.

Hypothesis 4: Training-specificity of across-session SMR and beta1 amplitude changes in passive periods.

It was hypothesized that:

- (1) SMR amplitude would be increased across-sessions in passive periods in SMR NF relative to beta1 NF and control NF.
- (2) Beta1 amplitude would be increased across-sessions in passive periods in beta1 NF relative to SMR NF and control NF.

Hypothesis 5: Frequency-specificity of SMR and beta1 amplitude changes in active periods.

It was hypothesized that:

- (1) SMR NF would be associated with increased mean SMR amplitude but not with significant amplitude changes in other frequencies relative to beta1 NF and control NF on the average of active periods within-sessions.
- (2) Beta1 NF would be associated with increased mean beta1 amplitude but not with significant amplitude changes in other frequencies relative to SMR NF and control NF on the average of active periods within-sessions.

Hypothesis 6: Frequency-specificity of SMR and beta1 amplitude changes in passive periods.

It was hypothesized that:

- (1) SMR NF would be associated with increased mean SMR amplitude but not with significant amplitude changes in other frequencies relative to beta1 NF and control NF on the average of passive periods within-sessions.
- (2) Beta1 NF would be associated with increased mean beta1 amplitude but not with significant amplitude changes in other frequencies relative to SMR NF and control NF on the average of passive periods within-sessions.

3.2. Methods and materials

3.2.1. Participants

Thirty-one adult students from Goldsmiths, University of London volunteered to take part in the experiment. Twenty-five participants were first-year psychology undergraduate students recruited through a Research Participation Scheme in return of course credit. Six participants were postgraduate students recruited through advertisement in the Campus and received monetary compensation of £10 at completion. All participants were right-handed and had normal or corrected-to-normal vision. None of the participants had previous experience with NF. Prior to the pre-assessment general information regarding the goals of the study and about EEG and NF procedures was given. Participants remained blind to the details of the experimental design and to the frequency-bands of interest. Participants gave written informed consent for participation in the study. The experimental procedures were approved by the Department of Psychology Ethics Commission. Participants were screened for neuropsychiatric disorders and psychoactive substances use (see Appendix A). Participants were randomly allocated to one of three experimental conditions resulting in a control group consisting of 10 participants (6 female, mean age=21.30 years; standarddeviation (SD)=1.64 years), an SMR group consisting of 10 participants (5 female, mean age=23.0 years; SD=5.57 years) and a beta1 group consisting of 11 participants (5 female, mean age=21.91 years; SD=4.16 years). A one-way ANOVA confirmed that the mean age did not differ between the NF protocols (F(2,28)=0.41, p=0.66, $\eta_p^2=0.03$).

3.2.2. General procedure

As shown in figure 3.1., for each participant the experiment consisted of 8 sessions (1 pre-assessment session, 6 NF training sessions and 1 post-assessment session) taken on different days. The sessions were arranged according to the participants' availability and distributed over a period no longer than four weeks. The final assessment session had to be arranged within one to three days after the last training session. Before the pre-assessment session, the participants were contacted via email with general information about the experiment and returned the completed neuropsychiatric screening questionnaire. Pre- and post-assessment sessions consisted of two resting state (eyes closed and eyes open) 3-minutes

EEG recordings and two cognitive tasks with simultaneous EEG recording (3-stimuli auditory oddball task and cued-Go/Nogo task) performed in counterbalanced order between participants. For each participant, the order of the tasks was kept unchanged on the second assessment.



Figure 3.1. General NF procedure. **(A)** For each participant the experiment consisted of three blocks: one pre-assessment session, followed by 6 NF sessions, and finally one post-assessment session. **(B)** NF sessions consisted of ten equivalent periods of 3 minutes: a passive resting state period, followed by eight active feedback periods, and finally by a second passive resting state period.

3.2.3. Neurofeedback Procedure

Participants received six sessions of either SMR NF, beta1 NF or control NF distributed over two to three weeks. Participants, but not the experimenter, were kept blind to the specific frequencies targeted by the NF. To minimize the influence of circadian rhythms variation in the EEG, effort was taken to ensure the sessions were taken at the same time of the day (with an acceptable variation of 1 to 2-hours). As shown in Fig. 3.1, each of the NF training sessions consisted of an initial pre-training passive resting state period, followed by eight active feedback training periods and a post-training passive resting state period. To ensure comparability between resting state and feedback conditions all EEG recordings had an equal length of three minutes (160 seconds) and were performed with eyes open. Each

period consisted of 150 seconds of EEG recording and a resting and blink pause of 10 seconds. Participants were allowed to rest for longer periods if needed.

3.2.4. Neurofeedback training

EEG recording. NF was conducted using a Biograph ProComp Infiniti (Thought technology, Ltd.; Montreal, Canada) differential amplifier. Signal was acquired at a 256 Hz sampling rate, A/D converted, band-pass filtered (IIR Butterworth Filter) and peak-to-peak amplitudes were calculated to continuously extract theta (4-7Hz), low alpha (8-10Hz), high alpha (10-12Hz), SMR (12-15Hz), beta1 (15-18Hz), beta2 (19-22Hz), beta3 (23-26Hz), beta4 (27-30Hz) and electromyographic activity (EMG) (52-58Hz) components. Amplitude measures in the filter-bands were transformed online into visual representations and fed back to the participants via a Hannspree 18.5in LCD (16:9 - 5 ms) monitor positioned at a distance of ~1.5 m. The scalp electrodes were placed at C3 for beta1 NF, at C4 for SMR NF (with reference electrode placed on the contralateral and ground electrode place in the ipsilateral earlobe) and at Cz for control NF (with reference electrode placed on the left and ground electrode place in the right earlobe) according to the 10-20 electrode placement system. Impedance was kept below 5 K Ω .

Neurofeedback task. After the electrodes were attached the participants were instructed in the neurofeedback task. Participants were asked to sit comfortably and instructed to relax, minimize movement and blinking to prevent artifacts. The three groups received similar instructions regarding training goals and procedures (see Appendix B). Feedback consisted of visual representations of peak-to-peak amplitude variations (see *Feedback screens* above). Participants were asked to try to learn from the visual representations by letting themselves be guided by the feedback screen while trying to maximize the positive feedback. A brief explanation of the meaning of the visual signal was provided at the beginning of the first session (see Appendix B).

Feedback screens. In order to keep the participants motivated during NF session two feedback screens were used. During the first four training periods of each session visual feedback consisted of a video game that would respond to the variations in peak-to-peak amplitude by either moving or stopping according to the reward contingencies (see Reward

contingencies section below), while in the last four training periods visual feedback consisted of a circle that enlarged or shrunk according to the absolute amplitude of the target frequency (examples of feedback screens are presented in the Appendix C). The displays used for the passive resting state recordings were similar to the ones used in the first half of the training.

Feedback instructions and groups. In the first screen the participants were instructed to keep the video game moving for as long as possible while trying to keep EMG activity within range. In the circle feedback display participants were asked try to keep the circle as large as possible and above a visible threshold, while also trying to keep EMG activity within range. The SMR and beta1 NF groups aimed to increase the amplitude of the target frequency. In control NF group a bidirectional training strategy was used within the same training session. In each session the participants started by attempting to decrease (initial four active periods) and then to increase (last four active periods) the amplitude of a different EEG frequency. Since during the initial active inhibition period no explicit indication was provided regarding amplitude variations (video game screen), the participants were not aware that the training conditions were reversed in the second part of the session (circle screen). Six different EEG frequencies other than SMR and beta1 frequencies (theta, alpha1, alpha2, beta2, beta3, beta4) were selected from the range of 4 to 30Hz. The order of the training frequencies in the control group was kept constant between participants. In order to reduce the chances of transferring practice effects in a particular frequency range from one session to the next the training sequence of low and high frequencies was counterbalanced as follows: theta, beta2, alpha1, beta3, alpha2, beta4. The purpose of varying target frequency bands in each session was to minimize the chances of inducing systematic neurofeedback-related changes yet providing participants with real feedback.

Reward contingencies. Operant contingencies were such that reward was gained whenever the participants' EEG activity changed in the requested direction relative to the 150-seconds pre-feedback baseline measure. In the SMR and beta1 NF rewards were contingent on increasing the amplitude in the respective frequency, while in control NF they were either contingent on increasing or decreasing the amplitude of the frequency selected for any given session. After extracting the average amplitude of the target frequency during the first passive period (baseline) of each session, the reward threshold for the training periods was calculated. This was set was set at 0.7 times the baseline average amplitude for enhancement frequencies and at 1.3 times the baseline average amplitude for inhibition frequencies. EMG threshold was kept below 3.5 μ v during resting state periods and at 1.2 times the baseline amplitude during the training periods. In accordance with previous studies (e.g., Ros et al., 2009), threshold settings remained adjustable by the experimenter in steps of 0.05 throughout the training sessions, in order to ensure that the reward obtained would neither frustrate nor satiate the participants. From the first training period onwards the threshold adjustment decisions were made based on the absolute change of target-frequency compared to the previous training period.

3.2.5. EEG-Neurofeedback Data Processing

Data preparation. Analysis of the single channel EEG data from NF-training sessions was performed using custom software, programmed using MATLAB (v8.0, The MathWorks Inc., Natick, MA, 2012b). The batch-processing MATLAB script mimicked the NF-training software, using the same band-pass filter and similar peak-to-peak amplitude computation. Single channel peak-to-peak amplitude data of 150 seconds length was segmented in 1 second epochs (i.e., 150 epochs per recording).

Artifact detection and rejection. Three types of artefacts per epoch were identified. Firstly, electro-ocular activity (EOG) was computed as the range of the Z-transformed signal in an epoch after a Butterworth filter (1-15 Hz, 5th order) was applied. After visually inspecting four randomly selected sessions for each participant an individual threshold (ranging from 5 to 8) was set. Epochs were rejected if the Z-score surpassed the individually set threshold. Secondly, electromuscular activity (EMG) was computed by Z-transforming the filtered signal used for the EOG but with band limits set to 52-58 Hz. An epoch was rejected on the basis of EMG if it contained a Z-score larger than 4. Thirdly, epochs with amplitude perturbations larger than 60 μ V were rejected if they were still present after the application of a finite impulse response filter (600th order, 1-45 Hz) to the signal. The resulting epoch-rejection rate per participant ranged from 1.5% to 11.5% (Mean=6.5%, SD=2.5%). The percentage of rejected trials did not differ between groups as demonstrated by one-way ANOVA (F(2,28)=0.27, p=0.76, $\eta_p^2=0.02$).

Peak-to-peak amplitude calculation. Peak-to-peak amplitude per frequency band was calculated per epoch after applying a 4th order Butterworth filter. For each recording an average frequency-band amplitude was calculated by averaging all non-rejected epochs.

3.2.6. Statistical Analysis

Statistical analyses were performed with SPSS (Version 22.0. Armonk, NY: IBM Corp., 2013). Statistical significance was assumed at the 0.05 alpha level (two-tailed). Effect sizes were estimated by partial eta square (η_p^2) for the mixed-design Analysis of Variance (ANOVA). All reported ANOVAs used Greenhouse-Geisser degrees of freedom correction for violations of the sphericity assumption (as indicated by significant Mauchly's test of sphericity) and Bonferroni corrected probability values (*p*-values) for multiple comparisons.

Data reduction and analysis. Normalized measures of within-session amplitude change in any given frequency band were calculated by dividing the EEG amplitude of the active and passive periods by the EEG amplitude of the first active or passive period of each session. The average within-session amplitude change of each session was used as a measure of across-session changes.

3.3. Results

3.3.1. Training-specific learning

The amplitude changes in SMR and beta1 frequencies of the three NF groups were examined in separate analyses for within-session and across-session amplitude changes and for active and passive trials. The normalized within-session amplitude change in active periods was calculated for 8 periods. For passive periods there was only a single measure of normalized within-session change. Thus, four different analyses were performed (1) two-way 8 PERIOD x 3 PROTOCOL for within-session active periods, (2) one-way 3 PROTOCOL for within-session passive periods, (3) two-way 6 SESSION x 3 PROTOCOL for across-

session active periods, and (4) two-way 6 SESSION x 3 PROTOCOL for across-session passive periods.

Hypothesis 1: Training-specificity of within-session SMR and beta1 amplitude changes in active periods.

Active changes within-session in each of the NF protocols were investigated for both SMR and beta1 frequencies in separate two-way 8 PERIOD x 3 PROTOCOL ANOVAs with the amplitude change of each period relative to the first training period collapsed across sessions as the dependent measure.

Figures 3.2A and 3.2B show the within-session amplitude in the SMR and beta1 frequencies for the three NF protocols. These were investigated in separate two-way 8 PERIOD x 3 PROTOCOL ANOVAs for SMR and beta1 frequencies. The analyses indicated a main effect of PROTOCOL for beta1 (F(2,28)=4.79, p=0.02, $\eta_p^2=0.26$) but not for SMR (F(2,28)=1.08, p=0.36, $\eta_p^2=0.07$). As hypothesized (Hypothesis 1), pairwise comparisons by independent *t*-tests revealed training-specific enhancements of mean beta1 amplitude in beta1 NF when compared to control NF (t(19)=3.11, p=0.01). However, contrary to hypothesis the mean increments in beta1 amplitude were not significantly higher in beta1 NF when compared to SMR NF (t(20)=1.50, p=0.45).

As shown in figures 3.2A and 3.2B, the analyses revealed significant PERIOD x PROTOCOL interaction effects in the beta1 (F(5.54,77.50)=2.67, p=0.02, $\eta_p^2=0.16$) but not in the SMR frequency (F(4.95,69.27)=1.55, p=0.17, $\eta_p^2=0.10$). Descriptive statistics of SMR and beta1 amplitude change and statistical results of between-subjects multiple comparisons for each period are presented in table 3.1. As shown in the table, the comparisons between beta1 and control NF groups indicated that beta1 amplitude change was significantly increase in periods 4 and 6 through 8 and a marginally increased in the remaining periods. Moreover, in period 4 beta1 amplitude change was marginally increased in the beta1 relative to SMR NF. In beta1 NF within-subject pairwise comparison failed to provide significant beta1 amplitude changes between periods (all ps>0.16). These were also non-significant in the control (ps>0.45) and SMR NF (ps>0.28). Finally , the main effects of PERIOD were not significant in any of the frequencies (SMR: F(2.47,69.27)=0.40, p=0.71, $\eta_p^2=0.01$; beta1: F(2.77,77.50)=0.58, p=0.62, $\eta_p^2=0.02$).



Figure 3.2. Within-session SMR and beta1 amplitude change in active periods for control NF, SMR NF and beta1 NF. The graphs represent the mean amplitude change in eight active periods (from P1 to P8) collapsed across six training sessions relative to the first active period of each session for SMR(A) and beta1 frequencies (B). Error bars depict standard error of the mean. Statistical significance level is represented by asterisks ($^{+}=p<0.10$, $^{*}=p<0.05$, $^{*}=p<0.01$).

Table 3.1. Mean (and SD) of SMR and beta1 amplitude change (µV) for each period (normalized to period 1 of each session) for control NF
SMR NF and beta1 NF; independent t-tests (Bonferroni corrected p-values) betw een control NF, SMR NF and beta1 NF.

				Prot	Inc	Independent <i>t</i> -test				
		Control		SMR		Beta1		Control	Control	SMR
Frequency	Period	М	SD	М	SD	М	SD	vs. SMR	vs. Beta1	vs. Beta1
SMR	1	1.00	0.00	1.00	0.00	1.00	0.00	_	_	_
	2	0.99	0.05	1.01	0.04	1.01	0.05	0.87	0.94	0.06
	3	1.00	0.05	1.00	0.04	0.99	0.06	0.15	0.31	0.16
	4	1.00	0.05	0.99	0.05	0.99	0.06	0.17	0.21	0.04
	5	0.96	0.08	1.04	0.11	0.99	0.07	2.07	0.76	1.35
	6	0.98	0.09	1.03	0.10	1.01	0.10	1.37	0.90	0.50
	7	0.96	0.08	1.03	0.11	1.02	0.07	1.76	1.46	0.34
	8	0.98	0.06	1.04	0.09	1.02	0.07	1.64	1.09	0.59
Beta1	1	1.00	0.00	1.00	0.00	1.00	0.00	_	_	_
	2	0.99	0.03	1.01	0.03	1.02	0.03	1.62	2.47 †	0.82
	3	0.98	0.03	1.01	0.06	1.03	0.04	1.20	2.27 †	1.05
	4	0.99	0.03	0.99	0.04	1.04	0.06	0.29	2.73 *	2.42 †
	5	0.96	0.06	1.01	0.07	1.04	0.08	1.67	2.46 †	0.75
	6	0.97	0.06	1.02	0.06	1.05	0.06	1.89	2.84 *	0.93
	7	0.95	0.06	1.01	0.05	1.04	0.06	2.27	3.46 **	1.12
	8	0.97	0.08	0.99	0.06	1.05	0.07	0.79	2.72 *	1.91

†= p < 0.10, *= p < 0.05, **= p < 0.01

Hypothesis 2: Training-specificity of across-session SMR and beta1 amplitude changes in active periods.

Figures 3.3A and 3.3B show the SMR and beta1 amplitude changes in active periods across-sessions in the three NF protocols. Across-session SMR and beta1 amplitude changes as a function the type of NF protocol were investigated in separate 6 SESSION x 3 PROTOCOL mixed ANOVAs. Contrary to the hypothesis, the SESSION x PROTOCOL interactions were non-significant for both frequencies (SMR: F(10,140)=1.07, p=0.39, $\eta_p^2=0.07$; beta1: F(10,140)=0.59, p=0.82, $\eta_p^2=0.04$).



Figure 3.3. Across-session SMR and beta1amplitude change for control NF, SMR NF and beta1 NF in active periods. The graphs represent the amplitude change in six training sessions (from S 1 to S 6) for SMR (A) and beta1 frequencies (B). Amplitude change was normalized relative to the first active period of the first session. Error bars depict standard error of the mean. Statistical significance level is represented by a cross and an asterisk ($^+=p<0.10$; $^{*}=p<0.05$, $^{**}=p<0.01$).

Means (and SD) of SMR and beta1 amplitude change in active periods for each session are presented in table 3.2. As can be observed, between-subject pairwise comparisons (Bonferroni corrected p-values) revealed that in the first session the beta1 amplitude was significantly increased in beta1 relative to control NF (t(19)=3.56, p=0.004) and in SMR relative to control NF (t(18)=3.00, p=0.02). In the subsequent sessions beta1 amplitude changes were non-significantly different between NF groups (all ps>0.05). In the SMR frequency none of the between-subjects pairwise comparisons were significant (all ps>0.05).

		Protocol					Independent <i>t</i> -test			
	-	Cor	ntrol	SM	/IR	Be	ta1	Control	Control	SMR
Frequency	Session	М	SD	м	SD	м	SD	vs. SMR	vs. Beta1	vs. Beta1
SMR	1	0.99	0.05	1.06	0.09	1.07	0.11	1.63	1.91	0.24
(active)	2	1.05	0.14	1.05	0.14	1.06	0.13	0.01	0.18	0.17
	3	1.06	0.21	0.99	0.13	1.10	0.10	1.08	0.67	1.78
	4	1.11	0.17	1.02	0.15	1.13	0.13	1.39	0.21	1.62
	5	1.10	0.21	1.03	0.11	1.12	0.11	1.02	0.35	1.39
	6	1.13	0.21	1.08	0.16	1.15	0.11	0.69	0.31	1.02
Beta1	1	0.96	0.03	1.01	0.05	1.02	0.04	3.00 *	3.56 **	0.50
(active)	2	1.01	0.14	1.05	0.10	1.05	0.13	0.72	0.82	0.09
	3	0.99	0.18	1.02	0.11	1.03	0.13	0.41	0.55	0.13
	4	1.05	0.15	1.03	0.14	1.07	0.08	0.36	0.32	0.69
	5	1.07	0.20	1.05	0.09	1.07	0.12	0.39	0.06	0.34
	6	1.09	0.21	1.10	0.10	1.06	0.17	0.13	0.43	0.57
SMR	1	1.02	0.06	1.01	0.09	1.01	0.05	0.36	0.50	0.14
(passive)	2	1.05	0.10	1.03	0.17	1.08	0.19	0.17	0.48	0.66
	3	1.10	0.23	0.96	0.17	1.13	0.16	1.67	0.33	2.04
	4	1.10	0.08	1.00	0.21	1.13	0.15	1.51	0.39	1.93
	5	1.11	0.13	0.98	0.15	1.07	0.13	2.10	0.67	1.47
	6	1.15	0.16	1.04	0.16	1.14	0.15	1.54	0.08	1.50
Beta1	1	0.98	0.05	1.01	0.04	1.02	0.06	1.38	1.84	0.44
(passive)	2	1.02	0.13	1.03	0.12	1.03	0.12	0.14	0.23	0.08
	3	1.03	0.22	0.99	0.11	1.00	0.10	0.61	0.45	0.18
	4	1.05	0.11	1.03	0.15	1.05	0.09	0.45	0.05	0.41
	5	1.08	0.16	1.01	0.11	0.99	0.18	0.89	1.21	0.30
	6	1.07	0.19	1.08	0.15	1.01	0.16	0.17	0.84	1.01

Table 3.2. Mean (and SD) of norma	lized SMR and beta1 ampl	itude change (μV) in activ	e and passive periods for	each session and for
control NF, SMR NF and beta1 NF; ir	dependent t-tests (Bonfe	rroni corrected p-values) betw een control NF, SM	R NF and beta1 NF.

†= p < 0.10, *= p < 0.05, **= p < 0.01

The apparent linear increase in SMR and beta1 amplitude change across sessions in control NF (see figures 3.3A and 3.3B) was not supported by linear SESSION x PROTOCOL interactions for both frequencies (SMR: F(2,28)=1.62, p=0.22, $\eta_p^2=0.10$; beta1: F(2,28)=1.14, p=0.33, $\eta_p^2=0.08$). The analysis also revealed a significant main effect of SESSION for both frequencies (SMR: F(5,140)=3.13, p=0.01, $\eta_p^2=0.10$; beta1: F(5,140)=3.77, p<0.01, $\eta_p^2=0.12$). Additionally, polynomial contrasts revealed a linear effect of SESSION for both SMR and beta1 frequencies (SMR: F(1,28)=9.36, p<0.01, $\eta_p^2=0.25$; beta1: F(1,28)=9.55, p<0.01, $\eta_p^2=0.25$) indicating a linear increase in amplitude change irrespective of the NF group. Bonferroni corrected post-hoc comparisons revealed marginally

significant increases in SMR amplitude change from session 2 to 6 (t(30)=3.13, p=0.05) and in beta1 amplitude change from session 1 to 6 (t(30)=2.97, p=0.09) and from session 3 to 6 (t(30)=3.00, p=0.09). Both main effects of PROTOCOL were non-significant (SMR: F(2,28)=1.02, p=0.37, $\eta_p^2=0.07$; beta1: F(2,28)=0.12, p=0.88, $\eta_p^2<0.01$).

Hypothesis 3: Training-specificity within-session SMR and beta1 amplitude changes in passive periods.

Within-session SMR and beta1 amplitude changes in passive periods were investigated in one-way between-subjects ANOVA. Data and statistical results are presented in table 3.3. Contrary to hypothesis, the main effect of PROTOCOL was not significant for both SMR and beta1 frequencies (SMR: F(2,28)=0.12, p=0.89, $\eta_p^2<0.01$; beta1: F(2,28)=2.34, p=0.12, $\eta_p^2=0.14$). As shown in figure 3.5B, the mean within-session beta1 amplitude change in passive periods in beta1 NF relative to control NF. However, the posthoc pairwise comparisons (Bonferroni corrected p-values) did not show significant differences between the groups (ps>0.12).

Hypothesis 4: Training-specificity of across-session SMR and beta1 amplitude changes in passive periods.

Figures 3.4A and 3.4B show the SMR and beta1 amplitude changes in passive periods across-sessions in the three protocols. Across-session SMR and beta1 amplitude changes in passive periods was investigated in separate two-way 6 SESSION x 3 PROTOCOL mixed-ANOVAs. Table 3.2. presents means (and SD) of average within-session SMR and beta1 amplitude change in passive periods for each session.

In the SMR frequency, the analysis failed to reveal significant changes across sessions as a function of the type of NF (SESSION x PROTOCOL:F(10,140)=1.30, p=0.24, $\eta_p^2=0.09$). However, there was a significant main effect of SESSION (F(5,140)=2.75, p=0.02, $\eta_p^2=0.09$). Polynomial contrasts evinced a significant linear increment in SMR amplitude change across sessions irrespective of protocol (F(1,28)=10.43, p<0.01, $\eta_p^2=0.27$). Bonferroni corrected pairwise comparisons further revealed a reliable SMR amplitude increment between sessions 1 and 6 (t(30)=3.37, p=0.04). The remaining pairwise comparisons were non-significant (all ps>0.14).

		Protocol					ANOVA	Inc	dependent t-t	est	
		Con	trol	SMR Beta1			Control	Control	SMR		
Source	Frequency	м	SD	М	SD	М	SD	- F	vs.SMR	vs. Beta1	vs. Beta1
Active	Theta	0.975	0.07	0.950	0.06	0.993	0.04	1.52	1.00	0.72	1.74
	Alpha1	0.954	0.07	0.982	0.04	1.000	0.03	2.28	1.28	2.13	0.82
	Alpha2	0.989	0.06	1.026	0.07	1.027	0.05	1.34	1.39	1.45	0.03
	SMR	0.983	0.05	1.018	0.06	1.004	0.05	1.08	1.46	0.90	0.59
	Beta1	0.977	0.04	1.007	0.04	1.034	0.05	4.79 *	1.58	3.07 *	1.48
	Beta2	0.986	0.04	1.003	0.04	1.017	0.04	1.63	0.92	1.80	0.87
	Beta3	0.984	0.05	0.995	0.05	1.008	0.04	0.65	0.49	1.14	0.63
	Beta4	0.979	0.06	0.982	0.05	1.001	0.02	0.74	0.14	1.11	0.97
Passive	Theta	0.971	0.11	0.966	0.08	1.008	0.15	0.40	0.08	0.73	0.81
	Alpha1	0.987	0.11	0.967	0.04	0.999	0.07	0.41	0.55	0.34	0.90
	Alpha2	1.019	0.11	0.971	0.06	1.003	0.10	0.69	1.16	0.40	0.78
	SMR	0.990	0.08	0.973	0.08	0.984	0.09	0.12	0.48	0.19	0.30
	Beta1	0.967	0.08	0.992	0.04	1.032	0.08	2.34	0.81	2.14	1.31
	Beta2	0.975	0.07	0.975	0.04	0.990	0.05	0.26	0.01	0.62	0.61
	Beta3	0.958	0.09	0.955	0.05	0.959	0.06	0.01	0.09	0.03	0.13
	Beta4	0.947	0.10	0.944	0.05	0.976	0.06	0.63	0.08	0.92	1.01

Table 3.3. Means (and SD) of the mean within-session amplitude change (μ V) in active and passive periods for each frequency; statistical values of the between-subjects one-way ANOVAs and multiple comparisons by independent *t*-tests (Bonferroni corrected p-values) between control NF, SMR NF and beta1 NF.

† = p < 0.10; * = p < 0.05; ** = p < 0.01

In the beta1 frequency the main effect of SESSION (F(5,140)=1.09, p=0.36, $\eta_p^2=0.04$) and the interaction effect of SESSION x PROTOCOL (F(10,140)=0.87, p=0.57, $\eta_p^2=0.06$) were not reliable.



Figure 3.4. Across-session SMR and beta1amplitude change for control NF, SMR NF and beta1 NF in passive periods. The graphs represent the amplitude change in six training sessions (from S 1 to S 6) for SMR (A) and beta1 frequencies (B). Amplitude change was normalized relative to the first passive period of the first session. Error bars depict standard error of the mean.

3.3.1. Frequency-specific learning

In order to investigate frequency-specific effects, the mean within-session amplitude change for eight frequencies ranging from theta to beta4 was calculated for each NF protocol. Means (and SD) of within-session amplitude changes in the eight frequencies in active feedback and passive resting state periods are presented in table 3.3. Frequency-specific effects were analyzed with two-way 8 FREQUENCY x 3 PROTOCOL mixed-ANOVAs for active and passive periods.

Hypothesis 5: Frequency-specificity of SMR and beta1 amplitude changes in active periods.

For active feedback periods the analysis revealed a significant main effect of FREQUENCY (F(7,196)=5.01, p<0.001, $\eta_p^2=0.15$). The pairwise comparisons revealed overall significant within-session amplitude decreases in both theta (M=0.973; SD=0.06) and alpha1 (M=0.979 ; SD=0.05) relative to alpha2 (M=1.014; SD=0.06) (theta vs. alpha2: t(30)=3.72, p=0.03; alpha1 vs. alpha2: t(30)=3.89, p<0.01). The interaction effect of FREQUENCY x PROTOCOL was non-significant (F(14,196)=1.24, p=0.25, $\eta_p^2=0.08$). However, the effects of the three protocols on each of the eight frequencies were explored by one-way ANOVAs and *t*-tests. As shown in figure 3.5A, beta1 NF appears to be associated with increased within-session amplitudes in alpha1 and beta1 relative to control NF and in theta relative to SMR NF. However, as can be seen in table 3.3., apart from the previously discussed main effects of PROTOCOL in the beta1 frequency (section 3.1.1.), no significant differences between NF groups were observed in the other frequency bands (all ps>0.13). The analysis also indicated that the main effect of PROTOCOL was not reliable (F(2,28)=1.96, p=0.16, $\eta_p^2=0.12$).



Figure 3.5. Within-session amplitude change in different frequency bands for control NF, SMR NF and beta1 NF protocols. The graphs indicate the mean within-session amplitude change from theta to beta4 frequencies for active periods (normalized to the first active period of each sessions) (A) and passive periods (normalized to the first passive period of each session) (B). Error bars depict standard error of the mean. Statistical significance level is represented by asterisks (* p<0.05 and **p<0.01).

Hypothesis 6: Frequency-specificity of SMR and beta1 amplitude changes in passive periods.

The mean amplitude changes in passive periods for the frequencies ranging from theta to beta4 for each of NF protocols are shown in figure 3.5B. As shown in the figure, beta1 NF appears to be associated with frequency-specific increments in beta1 amplitude relative to control NF. However, the 8 FREQUENCY x 3 PROTOCOL mixed ANOVA indicated that the interaction effect of FREQUENCY x PROTOCOL was not reliable $(F(14,196)=0.65, p=0.68, \eta_p^2=0.04)$. The analysis further revealed a significant main effect of FREQUENCY ($F(7,196)=2.19, p=0.04, \eta_p^2=0.07$). Post-hoc analysis revealed significant within-session amplitude decreases in beta3 (M=0.957; SD=0.07) and beta4 (M=0.956 ; SD=0.07) relative to alpha2 (M=0.998 ; SD=0.09; alpha2 vs. beta3: t(30)=3.63, p=0.02; alpha2 vs. beta4: t(30)=3.50, p=0.04) and relative to beta1 (M=0.998; SD=0.07; beta1 vs. beta3: t(30)=4.00, p=0.02; beta1 vs. beta4: t(30)=3.81, p=0.02). The analysis further indicated a non-significant main effect of PROTOCOL ($F(2,28)=0.50, p=0.61, \eta_p^2=0.04$).

3.4. Discussion

3.4.1. Training- and frequency-specificity in active periods

3.4.1.1. Training-specific within-session amplitude changes

Beta1

The present study provided supportive evidence of the feasibility of using the EEG NF technique to increase the beta1 amplitude, but failed to replicate its effect in enhancing the SMR amplitude (Hypothesis 1). The results present novel evidence that, during active feedback periods, beta1 amplitude was significantly increased in the course of 20-minute-long beta1 NF sessions relative to a RF NF control condition. This finding suggests that the increase in sensorimotor beta1 activity was related to the training rather than to unspecific effects such as time on task, placebo, sensorimotor stimulation and feedback learning.

However, as suggested by figures 3.2B and 3.5A and additionally confirmed by posthoc comparisons, these beta1 amplitude increases were not significantly larger than those obtained in SMR NF. These latter findings question the training-specificity of the beta1 amplitude increments. In fact, it is conceivable that the differences in the beta1 amplitude might be explained by some common aspect that was present in the SMR and beta1 NF but not in control NF condition. For instance, the experimental manipulation of RF NF might have made it more difficult to develop and maintain a consistent self-regulatory strategy both within and across the training sessions because a new frequency was trained in every session and the training contingencies were reversed halfway through the session. The more stable conditions of the SMR and beta1 NF would be more likely to contribute to the maintenance of the current postural and cognitive set (Engel & Fries, 2010) and the development of topdown mechanisms of attentional control (Wróbel, 2000; Buschman & Miller, 2007; Lee et al., 2013) that correlate with increased beta activity. Further, this explanation appears to be consistent with the absence of beta amplitude increments in sham NF conditions (Witte et al., 2013; Kober et al., 2015) in which the unpredictability of feedback may prevent the development of self-regulatory strategies. Although similar RF NF conditions have been used in previous studies (Hoedlmoser et al., 2008; Doppelmayr & Weber, 2011; Schabus et al., 2014) little is known regarding their comparability with experimental and other control

conditions (e.g., sham NF) in terms of demands imposed on self-regulation, cognitive effort and other unspecific variables with impact on beta and other EEG frequencies. Further work is needed to better understand whether experimental and control conditions impose different self-regulatory and cognitive demands and to prevent such differences by ensuring their consistency within- and across-sessions.

SMR

The present results failed to replicate within-session SMR amplitude increments following SMR/theta (Witte et al., 2013; Kober et al., 2015). Although this is line with other studies (e.g., Ros et al., 2009), several methodological limitations may explain the lack of significant within-session SMR amplitude increments. Firstly, the different visual and kinesthetic properties of the screens used to display feedback information could have interfered specifically with the ability to increase the SMR amplitude. Because the SMR frequency is suppressed with actual or imagined voluntary motor execution (Stancák & Pfurtscheller, 1996; Gilbertson et al., 2005; Pfurtscheller et al., 2005; Kilavik et al., 2013), the movement sensation induced by the first screen (periods 1-4) might have caused difficulties to the synchronization of SMR activity when compared to the more static properties of the second screen (periods 5-8). However, this apparent difficulty (see figure 3.2A) was not confirmed by within-subject comparisons, which failed to show significant differences in SMR amplitude change between periods in SMR NF. Secondly, the shorter duration of the active feedback periods (150 seconds) relative to those previous studies (180 seconds) might also explain the differences in success of within-session SMR learning. While this limits the comparisons between studies, it must be noticed that the duration of the feedback periods was not an impediment to observe training-specific within-session betal amplitude increases in the present study. This suggests that the extent to which the duration of the feedback period affects the learning ability might significantly differ from one frequency to the other. The question of whether the duration of feedback trials differentially affects the reactivity, and therefore, the learning success in different EEG frequencies is certainly an important aspect of NF learning to be explored in future studies.

An alternative explanation for the discordant results is that SMR NF did not include explicit instructions to suppress theta and high-beta amplitude. This interpretation is in line with the proposal that theta and high-beta suppression might play a coadjuvant role in successful SMR amplitude increments by ensuring the necessary conditions of attentional focus and behavioral stability (Sterman and Shouse, 1980; Othmer et al., 1999). In light of the present evidence one cannot rule out the hypothesis that the modulation of slow or fast wave activity may play a mechanistic role in within-session SMR amplitude increments as observed in previous studies. A possible candidate mechanisms is the ability of the lower frequencies to modulate the amplitude of higher frequencies by phase-amplitude or amplitude-amplitude cross-frequency coupling (Canolty & Knight, 2010). However, in the absence of empirical studies that compare SMR NF with and without suppression of lower and higher frequencies the relative contribution of these frequencies to successful within-session SMR amplitude increments remains unclear.

3.4.1.2. Frequency-specific within-session active amplitude changes

Another goal of the present study was to investigate whether the NF protocols could selectively increase the activity in the target frequencies. Importantly, the non-significant FREQUENCY x PROTOCOL interaction did not allow the interpretation that beta1 NF was related to within-session amplitude increments in beta1 frequency but not in other frequencies (Hypothesis 5). As shown in fig. 3.5A, beta1 NF was also associated with (non-reliable) amplitude increments in other frequencies when compared to the other NF groups. However, the non-reliable main effect of PROTOCOL did not support the hypothesis of an overall within-session amplitude increment irrespective of frequency following beta1 NF either. Instead, there was a significant main effect of FREQUENCY indicating reliable within-session amplitude increases in alpha2 relative to theta and low-alpha irrespective of the NF protocol which may correspond to unspecific effects of NF.

Importantly, the present findings failed to corroborate previous evidence of frequency-specific SMR and beta1 amplitude increments relative to theta and high-beta (Vernon et al., 2003; Ros et al., 2009; Gruzelier et al., 2014a, 2014b). However, as shown in table 1.1, (1) there was conflicting evidence between studies of frequency-specificity relative to theta (e.g., Ros et al., 2009; Kober et al., 2015) and high-beta (Gruzelier et al., 2014a, 2014b), (2) frequency-specificity relative to both frequencies was never observed in the same study (e.g., Vernon et al., 2003; Kober et al., 2015) and, (3) contrary to the present study,

when more than one statistical test was used the *p*-values were not adjusted to the number of multiple comparisons.

3.4.1.3. Across-session active amplitude changes

Also contrary to hypothesis, the present study failed to provide supportive evidence of training-specific across-session amplitude increments in active feedback periods (Hypothesis 2). In fact, both SMR and beta1 amplitudes were linearly increased irrespective of the NF protocol. These results are in line with previous studies suggesting that, despite linear increments in SMR amplitude at the session level, the absolute SMR amplitude was not superiorly increased across sessions in SMR/theta relative to sham NF (Witte et al., 2013; Kober et al., 2015). They are also consistent with previous failures to demonstrate across-session beta1 learning following beta1/theta when compared to other NF protocols (Doppelmayr & Weber, 2011). However, these results are in disagreement with previous findings of increased SMR amplitude from the beginning to the end of the training following SMR and SMR/theta relative to RF NF control conditions (Doppelmayr & Weber, 2011; Hoedlmoser et al., 2008; Schabus et al., 2014).

In the current experiment, the across-session SMR and beta1 amplitude increments observed in the control NF could have masked the training-specific effects of SMR and beta1 NF. Because similar increments in the RF NF were not observed in previous studies (e.g., Doppelmayr & Weber, 2011; Hoedlmoser et al., 2008), a possible explanation for this divergent findings may rely on the greater difficulty of the present control condition. In fact, the change from down- to up-regulation midway through the session might have imposed higher demands on maintenance of a cognitive and postural set than a single up-regulation strategy (as in the SMR and beta1 NF). These higher demands might have resulted in initial lower levels of beta synchronization relative to later sessions, that despite the absence of a significant SESSION x PROTOCOL interations, were especially evident in RF NF (as shown in figures 3.3A and 3.3B).

An alternative explanation for the failure to observed across-session SMR and beta1 amplitude increments may rely on the relatively reduced number of sessions in the present study. However, whether across-session learning might be related to the number of sessions *per se* remains elusive. In fact, while some studies suggest across-session SMR amplitude

increments after ten or more sessions (Doppelmayr & Weber, 2011; Hoedlmoser et al., 2008; Kober et al., 2013; Schabus et al., 2014) with SMR/theta ratio reaching a plateau after the initial four sessions (Ros et al., 2009), others failed to yield any evidence of across-session learning after ten NF sessions (Witte et al., 2013; Kober et al., 2015).

3.4.2. Training- and frequency-specificity in passive periods

The results failed to provide supportive evidence of training-specific within-session SMR and beta1 amplitude changes between pre- and post-training passive resting state periods (Hypothesis 3). Thus, it can be concluded that, despite the within-session beta1 amplitude increments in the beta1 relative to control NF observed in the active periods, the same increments in the passive periods did not reach statistical significance (as shown in fig. 3.5B). These results are in agreement with the previous suggestion that the increased beta synchronization during the active feedback state may not necessarily culminate in increased resting state beta activity (Egner et al., 2004).

As also shown in the figure 3.5B, SMR amplitude was globally decreased from pre-to post-training. However, from the present findings it remains unclear whether the failure to observe significant within-session SMR amplitude changes in passive periods may be related to the unsuccessful SMR self-regulation during the active feedback periods. Also contrary to prediction, the results failed to provide supportive evidence of across-session amplitude SMR amplitude increments in passive periods as a function of the specific training conditions (Hypothesis 4). These results are in accordance with the failure to observe training-specific resting state SMR amplitude change across sessions, even when training-specific increments were observed during active feedback conditions (Hoedlmoser et al., 2008; Witte et al., 2013). However, there was a linear across-session increments in post-training SMR amplitude irrespective of the NF. Similarly to the active periods, it is unclear whether control NF could have masked training-specific effects or whether this increments correspond to unspecific effects of NF such as an increase in motor inhibition associated with an increasing compliance with general NF instructions of remaining still to prevent EMG artifacts.

Similarly to active periods, the analysis failed to provide support to the hypothesis of frequency-specific SMR and beta1 amplitude increments in passive periods (Hypothesis 6). The analysis revealed a main effect of FREQUENCY suggesting within-session amplitude

increments in the alpha2 and beta1 frequencies relative to the beta3 and beta4 frequencies regardless of the NF protocol. This amplitude increment in lower relative to higher frequencies may reflect an increased cortical inhibition effect of the NF sessions specific to passive periods. Speculatively, the observation of the opposite pattern of amplitude changes in the active periods (i.e., decrements in lower relative to higher frequencies) provides support to the notion that NF may have a distinct impact on the brain dynamics during task involvement (i.e., increased activation) and resting state (i.e., increased inhibition).

Additionally, the amplitude increments in beta1 relative to beta3 and beta4 suggest that unspecific factors associated with NF may have a differential impact on functionally distinct sensorimotor beta rhythms (Hari & Salmelin, 1997; Neuper & Pfurtscheller, 2001; Cheyne, 2013). However, the nature of the unspecific factors that may have contributed to these patterns of reactivity in the beta band is not entirely clear. An investigation of the possible association of this pattern with dissociable somatosensory processing and movement execution aspects of task performance (Hari & Salmelin, 1997) would contribute to a better understanding of the NF impact on low- and high-beta frequencies.

3.4.3. Summary

The current study provides empirical evidence that the validation of SMR and betal NF may benefit from taking into consideration different learning indices and supporting assertions of learning specificity in comparisons with control conditions in a broad range of frequencies. Our results provide novel evidence of the feasibility of enhancing the betal amplitude by NF. However, whether such increments may be related to specific or unspecific effects remains unclear. Beta1 amplitude increments may be related to the possibility of maintaining a consistent self-regulatory strategy throughout the session. Also in line with the interpretation that the beta1 amplitude increments in beta1 NF might have been epiphenomenal, we failed to provide evidence that those increments were larger than in other frequency bands. Our findings are consistent with previous studies that failed to demonstrate a net increase in within-session SMR amplitude. However, it is still unclear whether the differences in training-specificity of SMR and beta1 NF might be associated with methodological limitations (e.g., changes in the visual and kinesthetic properties of the screens) or reflect different patterns of reactivity to self-regulatory strategies.

Chapter 4

Experiment 2: The effects of SMR and beta1 NF on selective attention

4.1. Hypotheses

Hypothesis 1: Increased P3a and P3b amplitude as a function of SMR and beta1 NF

It was hypothesized that:

- (1) P3b amplitude would be increased in the beta1 NF relative to SMR NF and control NF conditions and in the SMR NF relative to control NF condition (beta1> SMR>control).
- (2) P3a amplitude would be increased in the SMR NF relative to beta1 and control NF and in the beta1NF relative to control NF condition (SMR>beta1>control).

Hypothesis 2: Decreased RT and RT-SD as a function of SMR and beta1 NF

It was hypothesized that:

- (1) RT would be decreased in the beta1 NF relative to SMR NF and control NF and in the control NF relative to the SMR NF (beta1<control<SMR).
- (2) RT-SD would be decreased in the beta1 NF relative to SMR NF and control NF and in the SMR NF relative to the control NF (beta1<SMR <control).

Hypothesis 3: Increased perceptual sensitivity and performance rates as a function of SMR and beta1 NF

It was hypothesized that:

- Target-standard d' and target-nontarget d' would be increased in SMR NF relative to beta1 NF and control NF and increased in beta1 NF relative to control NF (SMR>beta1>control).
- (2) Hit rate would be increased in the beta1 NF relative to SMR NF and control NF and increased in SMR NF relative to control NF (beta1>SMR>control).

(3) Standard false alarm rate and nontarget false alarm rate would be decreased in the SMR NF relative to control NF and beta1 NF and in the control NF relative to the beta1 NF (SMR<control<beta1).</p>

4.2. Methods

4.2.1. Participants

The same thirty-one participants of Experiment 1 took part in the present study. After excluding trials with artifacts and averaging the ERP waveforms four participants (two from the control NF and two from the beta1 NF) were excluded from the analysis for having less than thirty correct trials in at least one of the stimulus conditions at either pre- or post-training. Table 4.1. presents demographic data the remaining twenty-seven participants (12 males, 15 females). The mean age did not differ between NF protocols as confirmed by a one-way ANOVA (F(2,24)=0.31, p=0.73, $\eta_p^2=0.03$).

Table 4.1.

and choice and participants with insufficient named of context and per stimulas contained.									
	A	ge		Gender					
NF	Μ	SD	Male	Female	Total				
Control	21.38	1.77	4	4	8				
SMR	23.00	5.58	5	5	10				
Beta	21.89	4.371	3	6	9				
Total	22.15	4.38	12	15	27				

Descriptive statistics of age (mean and standard deviation) and gender (frequency) by NF protocol after excluding four participants with insufficient number of correct trials per stimulus condition.

4.2.2. Stimulus and procedure

The three-stimuli oddball task was presented to the subjects before and after the NF sessions (Experiment 1) in counterbalanced order with the cued Go/Nogo task (Experiment 3). The order of presentation had no effect in either the P3 amplitude for target, nontarget and standard stimuli or the behavioral measures of RT and accuracy.

Stimuli were presented electronically using the E-Prime 1.1 software (Psychology Software Tools, Pittsburgh, PA). Auditory tones were pure sinusoidal tones presented binaurally in a random series at 75 dB SPL through speakers every 1.75 seconds with a 200ms plateau and 10ms rise/fall times. EEG activity was recorded during 5 blocks, each of which consisted of 100 stimulus presentations. Each block lasted approximately 3 minutes. The stimuli differed in pitch and probability of occurrence and were defined as target (low pitch=500 Hz; probability of occurrence=0.10), standard (medium pitch=1000 Hz; probability of occurrence=0.10). Participants were instructed to respond as quickly and accurately as possible to the target stimulus by pressing a button with their right thumb. Response time and accuracy were recorded within the response interval (0 to 1750ms after stimulus presentation). Each participant was given a practice trial block consisting of 20 stimuli with the same probability rates as described above.

4.2.3. EEG recording and analysis

EEG recording.

For EEG acquisition at pre- and post-training a Biosemi Active Two A/D box ADC-9 (Biosemi, Netherlands) digital amplifier was used. EEG data was obtained from 64 electrodes mounted in an elastic cap in accordance with the international 10-20 system. Two additional electrodes, Common Mode Sense (CMS) and Driven Right Leg (DRL), were used as a "ground". (http://www.biosemi.com/faq/cmsanddrl.htm). The EEG was Direct Current-recorded with a low-pass filter of 100 Hz, a high-pass filter of 0.16 and a sampling rate of 512 Hz. Horizontal and vertical electro-oculograms were measured bipolarly from a pair of electrodes placed at the outer canthi of the eyes and a pair of electrodes placed above and below the right eye.

EEG data analysis.

EEG analysis was performed in Brain Vision Analyzer EEG analysis 1.05 software (Brain Products, Munich, Germany). After data acquisition, the EEG was digitally rereferenced to the average of the left and right earlobes and band-pass filtered (high-pass 0.5 Hz, low-pass 40 Hz). EEG continuous recordings were visually inspected to remove extreme EEG artifacts and segmented in reference to the stimuli in intervals of 1000ms (-200ms prestimulus to 800ms post-stimulus). Only correct trials (correct response to target and correct non-response to standard and non-target on the 0 to 1000ms interval post-stimulus onset) were analyzed. The data was then submitted to baseline correction (-200ms to 0ms relative to stimulus onset) and artefact rejection was performed for ocular movements and EEG fluctuations exceeding \pm 80 μ V. A minimum of 30 trials per condition were used. Subsets of four electrodes were pooled to define three regions of interest: frontal (F1, Fz, F2, FCz), central (C1, Cz, C2, CPz), parietal (P1, Pz, P2, POz). The data was then averaged per region of interest.

In order to isolate the P3a and P3b components the mean amplitudes of *target-standard difference waveform* and *nontarget-standard difference waveform* were calculated by linear subtraction. The deviant-standard difference waveform was obtained by averaging target-standard and nontarget-standard difference waveforms and then subtracting the standard amplitude from it.

Percent of correct ERP trials.

Descriptive statistics of percent of correct trials in the oddball task are presented in table 4.2 Separate two-way ANOVAs on the percentage of correct trials as a function of TIME (Pre- vs. Post-Training) and PROTOCOL (control NF vs. SMR NF vs. beta1 NF) were performed for each type of STIMULUS (target, standard and nontarget). As can be seen in table the three 2 TIME x 3 PROTOCOL ANOVAs yielded no significant main effects of or interactions (all ps>0.12) suggesting that the artifact rejection on the correct trials did not introduce systematic differences in the analysis in terms of the number of trials per condition.

4.2.4. Behavioral data analysis

Reaction time data preparation

Behavioral responses were collected using E-Prime 1.0 software (Psychology Software Tools, Pittsburgh, PA). Data preparation for RT and RT-SD analysis was as follows. In order to remove outliers from both ends of the RT distribution correct trials (i.e., responses to the target stimulus) which response latencies fell below 100 ms and above 1000ms were excluded from the analysis (rejection rate =0.07% of the trials). Only participants with a minimum of 30 correct trials were included in the analysis.

d' data preparation

In order to guarantee comparability between RT and ERP data sets, only correct responses (button press) to targets within the 100-1000ms interval were considered "hits" while responses to standard or non-target stimulus within the same interval were considered "false alarms". The hit and false alarm rates were subjected to a z-score transformation so that the distributions had similar variances and therefore d' was calculated by [d'=z(hit rate)z(false alarm rate)] where the function z(p) was the inverse of the cumulative Gaussian distribution. This was calculated in MATLAB using the norminv function [d]-norminv(hit rate)-norminv(false alarm rate)]. To prevent infinity as a consequence of the Ztransformation of perfect scores (i.e., hit rates of 1 and false alarm rates of 0) the rates were subjected to a standard correction according to the procedure proposed by Macmillan and Creelman (2005) as follows. When misses were higher than zero the hit rate (H) was calculated by [H=number of hits/(number of hits + number of misses)]. When misses were zero a standard correction was performed so the hit rate was then given by [H=(number of hits -0.5/(number of hits + number of misses)]. When false alarms were higher than zero false alarm rate (FA) was calculated by [FA=Number of false alarms/ (Number of false alarms + Number of correct rejections)]. When false alarms were zero, an analog standard correction was performed and FA was calculated by [FA=(Number of false alarms+0.5)/(Number of false alarms+Number of correct rejections)].

The signal detection accuracy in the 3-stimuli oddball may be described, as in signal detection classification experiments, by the increase in perceptual sensitivity to task-relevant stimuli (targets) in relation to both frequent (standard) and infrequent task-irrelevant stimuli (nontargets). Thus, in the present experiment two d' measures were used: the target-standard d' and the target-nontarget d'. Given that the sensitivity measures are based on standardized scores, stimuli within the same pitch tone dimension can be conveniently compared regardless of different probabilities of occurrence (Macmillan & Creelman, 2005).

4.2.5. Statistical analysis

Specifications of statistical packages and criteria were provided in section 3.2.6.

4.3. Results

4.3.1. Hypothesis 1: Increased P3a and P3b amplitude as a function of SMR and beta1 NF

The differences between the three NF protocols in the P3 amplitude of target-standard and nontarget-standard difference waveforms were investigated at pre-training (4.3.1.1.) and from pre- to post-training (section 4.3.1.2).

4.3.1.1. Pre-training differences between NF protocols

Figure 4.1. shows the grand average of the stimulus-locked ERP elicited by the targetstandard and nontarget-standard difference waves in frontal, central and parietal regions at pre-training irrespective of NF protocol. In parietal electrodes a large positive deflection starting around 250ms after stimulus onset with maximal peak amplitude around 300ms – the P3 component - could be identified for both nontarget (P3a) and target stimulus (P3b).

The three-way 2 STIMULUS (target-standard vs. nontarget-standard) x 3 REGION (frontal vs. central vs. parietal) x 3 PROTOCOL (control NF, SMR NF and beta1 NF) ANOVA for the mean P3 amplitude at pre-training yielded significant main effects of STIMULUS ($F(1,24)=23.09, p<0.01, \eta_p^2=0.49$) and REGION ($F(1.37,33.00)=29.35, p<0.01, \eta_p^2=0.55$) as well as a significant interaction of STIMULUS x REGION ($F(1.24,29.86)=10.44, p<0.01, \eta_p^2=0.30$). The main effect of PROTOCOL ($F(2,24)=0.57, p=0.57, \eta_p^2=0.05$) and the interactions involving the between-subject variable were non-significant (STIMULUS x PROTOCOL: $F(2,24)=1.19, p=0.32, \eta_p^2=0.09$; REGION x PROTOCOL: $F(2.75,33.00)=0.65, p=0.57, \eta_p^2=0.05$; STIMULUS x REGION x PROTOCOL: $F(2.49,29.86)=0.57, p=0.69, \eta_p^2=0.05$).



Fig. 4.1. *Left.* Grand average of the stimulus-locked ERP of target-standard and nontarget-standard difference waves in midline frontal, central and parietal regions at pre-training irrespective of the NF protocol. The shaded area represents the 250-400ms measurement window. Positive amplitude is displayed upwards. *Right*. Distribution of the target target-standard and nontarget-standard difference waves.

As shown in fig. 4.2., the post-hoc comparisons (Bonferroni corrected *p*-values) for the target-standard difference waves revealed a significantly higher P3 amplitude in the parietal region (M=6.97, SD=3.72) relative to central (M=4.43, SD=3.72; t(26)=5.12, p<0.001) and frontal regions (M=2.32, SD=3.14; t(26)=5.88, p<0.001) and in the central region relative to the frontal region (t(26)=4.16, p<0.01). Similarly, the nontarget-standard difference waves P3 amplitude was higher in the parietal region (M=3.24, SD=2.08) relative to central (M=2.40, SD=2.66; t(26)=2.61, p<0.05) and frontal regions (M=0.97, SD=3.21; t(26)=4.06, p<0.01) and in the central region relative to the frontal region (t(26)=4.32, p<0.01). Moreover, the post-hoc comparisons revealed that the P3 amplitude was significantly higher for the target-standard difference waves relative to the nontarget-standard difference waves in three regions (Frontal: t(26)=2.10, p<0.05; Central: t(26)=3.41, p<0.01; Parietal: t(26)=7.44, p<0.001).



Figure 4.2. Two-way interaction of Stimulus and Region in mean P3 amplitude at pre-training. The graphs shows the mean P3 amplitude (μ V) for Target-Standard and Nontarget-Standard stimulus in three Regions (Frontal, Central and Parietal) irrespective of NF protocol at pre-training. For both types of stimuli the P3 amplitude was larger at Parietal electrodes. Error bars depict standard error of the mean.

These results replicated the maximal parietal distribution of the P3a and P3b components and smaller amplitude of the P3a relative to the P3b in the easy target-standard discrimination context (Katayama, & Polich, 1999). Additionally, the analysis did not indicate significant initial differences between the NF protocols in the P3a and P3b amplitude.

4.3.1.2. Pre- to post-training differences between NF protocols

Taking into consideration the above, the effects of the three NF protocols on the P3a and P3b amplitudes were analyzed in the parietal region where these components reached the maximal amplitude. The means (and standard deviations) of the P3 amplitude of target-standard and nontarget-standard differences waves for the three NF protocols at pre- and post-training are shown in table 4.2. These effects were investigated in a two-way mixed-ANOVA with 2 TIME (pre- vs. post-training) x 3 PROTOCOL (control NF vs. SMR NF vs. beta1 NF) for target-standard and nontarget-standard difference waves.

Figures 4.3. and 4.4. present the target-standard and nontarget-standard difference waveforms and the mean P3 amplitude for each of the three NF protocols at pre- and post-training. As suggested by the figures, the statistical analysis did not support the hypothesis of a differential impact of the three NF protocols on the neural processing of targets (TIME x PROTOCOL: F(2,24)=0.64, p=0.17, $\eta_p^2=0.01$). As shown in the table 4.2., the main effects of TIME and PROTOCOL were also not reliable (ps>0.66). The post-hoc comparisons failed to provide evidence of pre- to post-training target-standard P3 amplitude changes in any of the three NF protocols (all ps>0.50).

Similarly, the type of NF did not significantly modulate the nontarget-standard P3 amplitude from pre- to post-training (TIME x PROTOCOL: F(2,24)=1.82, p=0.36, $\eta_p^2=0.08$). Again the main effects of TIME and PROTOCOL were not reliable for the nontarget-standard difference waves (*ps*>0.36). The post-hoc comparisons failed to provide evidence of pre- to post-training nontarget-standard P3 amplitude changes in any of the three NF protocols (all *ps*>0.26).



Fig. 4.3. Grand-average of the target-standard and nontarget-standard difference ERP waveforms in the parietal region for Control, SMR and beta1 NF at pre- and post-training. The shaded area represents the 250-400ms measurement window. Positive amplitude is displayed upwards.



Figure 4.4. Mean P3 amplitude of the target-standard and nontarget-standard difference waveforms for Control, SMR and beta1 NF at pre- and post-training. Both target-standard (A) and nontarget-standard P3 amplitudes (B) were not significantly modulated by the type NF protocol. Error bars depict standard error of the mean.

4.3.2. Hypothesis 2: Decreased RT and RT-SD as a function of SMR and beta1 NF

Table 4.2. summarizes the data and statistical results for RT and RT-SD at pre- and post-training for the three NF protocols. As can be observed in table, two one-way ANOVAs revealed that there were no initial differences between NF protocols in mean RT and RT-SD (all ps>0.69). The differential impact of the three NF protocols in mean RT and RT-SD were analyzed in two-way 2 TIME x 3 PROTOCOL mixed-ANOVAs.

4.3.2.1. Mean reaction time

As shown in fig. 4.5A, the three NF protocols were associated with decreased mean RT. The statistical analysis revealed a significant main effect of TIME (F(1,24)=10.91, p<0.01, $\eta_p^2=0.25$) indicating that the RT was significantly lower at post-training (M=385.19; SD=113.18) relative to pre-training (M=425.89; SD=137.56). This effect was driven by significant decreases in mean RT in control (t(7)=2.43, p=0.02) but not in the SMR and beta1 NF (all ps>0.22). However, these decrements were not reliably different as a function of the NF protocol as evinced by a non-significant interaction effect of TIME x PROTOCOL (F(2,24)=0.62, p=0.54, $\eta_p^2=0.05$). Finally, the main effect of PROTOCOL (F(2,24)=0.27, p=0.76, $\eta_p^2=0.02$) was not significant.

4.3.2.2. Reaction time standard deviation

Figure 4.5B shows the effects of the three NF protocols on RT-SD. As show in the figure, the RT-SD decreased from pre- to post-training for both control and SMR NF and increased for the beta1 NF. However, the statistical analysis did not reveal significant main effects (TIME: F(1,24)=0.08, p=0.79, $\eta_p^2 < 0.01$; PROTOCOL: F(2,24)=0.89, p=0.42, $\eta_p^2=0.07$) or interactions (TIME x PROTOCOL: F(2,24)=1.32, p=0.28, $\eta_p^2=0.10$). The post-hoc comparisons failed to provide evidence of pre- to post-training RT-SD changes in any of the three NF protocols (all *ps*>0.25).



Figure 4.5. Two-way interaction of 2 Time x 3 Protocol for reaction time. The graphs show the effects at preand post-training of the three NF protocols on mean RT and RT-SD. **(A)** A significant main effect of Time was observed for RT. **(B)** Both the Control NF and the SMR NF protocols showed a decrease in RT-SD, while the beta1 NF protocol showed an increase in RT-SD. However, the Time x Protocol interaction was non-significant. Error bars depict standard error of the mean.

Table 4.2.

One-way ANOVA Two-way ANOVA **Pre-training** Post-training Protocol Time Time x Protocol Protocol SMR Р η_p^2 F η_p^2 F Р η_p^2 F Source Control Beta1 Control SMR Beta1 F Р Р η_p^2 Mean P3 T-S 7.76 (3.98) 7.12 (3.40) 6.04 (3.79) 8.21 (4.72) 7.04 (4.26) 6.67 (3.20) 0.39 0.54 0.02 0.17 0.84 0.01 0.42 0.66 0.03 -N-S 3.31 (2.54) 2.56 (2.16) 3.86 (1.54) 3.45 (2.86) 3.22 (2.40) 3.29 (1.80) 0.05 0.83 0.00 1.06 0.36 0.08 0.29 0.75 0.02 H(%) 97.70 (3.43) 98.22 (0.97) 97.67 (3.16) 98.50 (0.97) 96.44 (7.67) 97.93 (4.43) 0.30 0.74 0.02 $0.09 \quad 0.76 \quad 0.00$ 1.17 0.33 0.09 $0.17 \quad 0.84 \quad 0.01$ FA(%) S 0.23 (0.31) 0.13 (0.00) 0.13 (0.00) 0.13 (0.00) 0.13 (0.00) 0.22 (0.29) 1.21 0.32 0.09 0.01 0.93 0.00 1.61 0.22 0.12 0.60 0.56 0.05 Ν 2.75 (2.71) 2.10 (2.81) 2.33 (2.35) 2.50 (1.69) 1.50 (0.97) 1.67 (1.00) 0.14 0.87 0.01 0.88 0.36 0.04 0.05 0.95 0.01 0.70 0.51 0.06 d'T-S 5.00 (0.66) 5.32 (0.10) 0.42 0.66 0.03 0.13 0.88 0.01 5.18 (0.36) 5.16 (0.20) 5.24 (0.20) 5.09 (0.77) 0.80 0.38 0.03 0.97 0.39 0.08 T-N 4.11 (0.53) 4.32 (0.45) 4.24 (0.45) 4.31 (0.27) 4.43 (0.23) 4.32 (0.62) 0.43 0.65 0.04 2.33 0.14 0.09 0.19 0.83 0.02 0.40 0.68 0.03 RT (ms) 415.74 (131.32) 437.08 (160.94) 424.84 (69.38) 351.58 (123.48) 407.11 (119.72) 396.87 (61.09) 0.06 0.94 0.01 7.94 0.01 0.25 0.62 0.54 0.05 0.27 0.76 0.02 Mean RT-SD 92.72 (22.46) 102.35 (29.10) 96.18 (17.65) 84.20 (28.26) 93.97 (26.39) 108.17 (32.31) 0.38 0.69 0.03 0.08 0.79 0.00 1.32 0.28 0.10 0.89 0.42 0.07 CT (%) Т 92.25 (6.18) 88.60 (7.95) 84.22 (14.68) 87.75 (10.61) 91.00 (8.39) 85.33 (16.97) 1.29 0.29 0.10 0.01 0.91 0.00 $0.50 \quad 0.62 \quad 0.04$ 1.08 0.36 0.08 S 86.13 (7.48) 79.03 (9.63) 75.89 (15.88) 78.41 (13.30) 82.65 (8.71) 79.44 (13.96) 1.71 0.20 0.13 0.006 0.94 0.00 2.32 0.12 0.16 0.47 0.63 0.04 81.00 (10.42) 81.50 (9.90) 2.09 0.15 0.15 0.57 0.46 0.02 1.47 0.25 0.11 0.83 0.45 0.06 Ν 91.25 (7.70) 79.56 (17.80) 83.60 (14.32) 79.78 (16.35)

Means (standard deviations) and statistical results of the pre-training one-way ANOVAs and the pre- to post-training two-way ANOVAs on Mean P3 amplitude to Target-Standard (T-S) and Nontarget-Standard (N-S), Hit Rate (H), False Alarm Rate (FA) to Standards (S) and Nontargets (N), d-prime (d') to Target-Standard (T-S) and Nontarget-Standard (N-S), Reaction Time (RT) Mean and Standard Deviation (RT-SD), percentage of Correct Trials (CT) to Targets (T), Standards (S), and Nontargets (N) included in ERP analyses after artifact rejection. Significant differences (p < 0.05) are highlighed in bold.
4.3.3. Hypothesis 3: Increased performance rates and perceptual sensitivity as a function of SMR and beta1 NF

Table 4.2. summarizes the data and statistical results for target-standard d', targetnontarget d', target hit rate, standard false alarm and nontarget false alarm at pre- and posttraining for the three NF protocols. As can be observed in the table, the one-way ANOVAs failed to reveal pre-training differences between NF protocols in any of these variables (all *ps*>0.32). The differential impact of the three NF protocols in these variables were analyzed in two-way 2 TIME x 3 PROTOCOL mixed-ANOVAs.

4.3.3.1. Target-standard d'

As can be observed in figure 4.6A the target-standard d' increased from pre- to post-training in the control NF and in the SMR NF and showed the opposite tendency in beta1 NF. As shown in table 4.5, the statistical analysis did not yield any significant main effects or interactions (TIME: F(1,24)=0.80, p=0.38, $\eta_p^2=0.03$; PROTOCOL: F(2,24)=0.13, p=0.88, $\eta_p^2=0.01$; TIME x PROTOCOL: F(2,24)=0.97, p=0.39, $\eta_p^2=0.08$). The post-hoc comparisons failed to provide evidence of pre- to post-training target-standard d' changes in any of the three NF protocols (all ps>0.14).

4.3.3.2. Target-nontarget d'

As shown in figure 4.6B the target-nontarget d' was slightly increased from pre- to post-training in the three NF protocols. However, the two-way ANOVA on the target-nontarget d' did not reveal significant main effects or interactions (TIME: F(1,24)=2.33, p=0.14, $\eta_p^2=0.09$; PROTOCOL: F(2,24)=0.40, p=0.68, $\eta_p^2=0.03$; TIME x PROTOCOL: F(2,24)=0.19, p=0.83, $\eta_p^2=0.02$). The post-hoc comparisons failed to provide evidence of pre- to post-training target-nontarget d' changes in any of the three NF protocols (all ps>0.38).



Figure 4.6. Two-way interaction of 2 Time x 3 Protocol for d-prime. The graphs show the effects at pre- and post-training of the three NF protocols on d' target-standard and d' target-nontarget. (A) d' target-standard d-prime was increased from pre- to post-training in the Control NF and the SMR NF, but decreased in the beta1 NF. However, the Time x Protocol interaction was non-significant. (B) d' target-standard d-prime was increased from pre- to post-training in the three NF protocols. However, the main effect of Time was non-significant. Error bars depict standard error of the mean.

4.3.3.3. Target hit rate

As shown in figure 4.7A, the mean hit rate increased as function of both control NF and SMR NF and decreased in the beta1 NF. However, the 2 TIME x 3 PROTOCOL mixed-ANOVA revealed no reliable main effects or interactions (F(1,24)=0.09, p=0.76, $\eta_p^2 < 0.01$; PROTOCOL: F(2,24)=0.17, p=0.84, $\eta_p^2=0.01$; TIME x PROTOCOL: F(2,24)=1.17, p=0.33, $\eta_p^2=0.09$). The post-hoc comparisons failed to provide evidence of pre- to post-training hit rate changes in any of the three NF protocols (all *ps*>0.30).

4.3.3.4. False alarm rate for standards

As shown in figure 4.7B from pre-to post-training the false alarm rate to standard stimulus was decreased in the control NF, changed in the SMR NF and increased in the beta1 NF. However, the two-way TIME x PROTOCOL mixed ANOVA revealed no significant main effects or interactions (TIME: F(1,24)=0.01, p=0.93, $\eta_p^2 < 0.01$; PROTOCOL: F(2,24)=0.60, p=0.56, $\eta_p^2=0.05$; TIME x PROTOCOL: F(2,24)=1.61, p=0.22, $\eta_p^2=0.12$). The post-hoc comparisons failed to provide evidence of pre- to post-training false alarm rate to standard stimulus changes in any of the three NF protocols (all ps>0.20).



Figure 4.7. Two-way interaction of 2 Time x 3 Protocol for performance rates. The graphs show the effects at pre- and post-training of the three NF protocols on hit rate (**A**), standard false alarm rate (**B**), and nontarget false alarm rate. (**A**) Hit rate was increased in the Control NF and SMR NF but decreased in the beta1NF. However, the Time x Protocol interaction was non-significant. (**B**) Standard false alarm rate was decreased in the Control NF, unchanged in the SMR NF and increased in beta1 NF. Again, the Time x Protocol interaction was non-significant. (**C**) Nontarget false alarm rate was decreased in the three NF protocols but the main effect of Time was non-significant.

4.3.3.5. False alarm rate for nontargets

As shown in figure 4.7C a decrease in the false alarm rate for non-targets was observed irrespective of the NF protocol. However, the two-way TIME x PROTOCOL mixed-ANOVA did not reveal a significant main effect of TIME (F(1,24)=0.88, p=0.36, $\eta_p^2=0.04$). As suggested by the figure, the analysis did not reveal other main effects or interactions (TIME x PROTOCOL: F(2,24)=0.05, p=0.95, $\eta_p^2=0.01$; PROTOCOL: F(2,24)=0.70, p=0.51, $\eta_p^2=0.06$). The post-hoc comparisons failed to provide evidence of pre- to post-training false alarm rate to nontarget stimulus changes in any of the three NF protocols (all ps>0.50).

4.4. Discussion

The ERP analysis did not support the hypotheses that the SMR NF and beta1 NF would be associated with differential neural response to infrequent nontarget (P3a) and target (P3b) stimuli (Hypothesis 1.1 and 1.2) which would reflect the enhancement of involuntary and voluntary aspects of selective attention respectively (Escera et al., 2000; Sawaki & Katayama, 2009).

Also contrary to hypothesized, the behavioral analyses failed to provide supportive evidence of the differential contribution of SMR and beta1 NF to improve selective attention and response inhibition. In fact, the two NF protocols did not have a distinct effect of behavioral indices of selective attention such as RT (Hypothesis 2.1), RT-SD (Hypothesis 2.2), target-standard d' and target-nontarget d' (Hypothesis 3.1) and hit rate (Hypothesis 3.2). Similarly, there were no significant differences in indices of response inhibition such as standard and nontarget false alarm rate (Hypothesis 3.3).

The next sections will examine several methodological differences relative to previous studies that might have been related to the failure replicate differential electrophysiological and behavioral effects of SMR NF and beta1 NF on selective attention and response inhibition in the context of a three-stimuli auditory oddball (Doppelmayr & Weber, 2011; Egner & Gruzelier, 2001, 2004).

4.4.1. Electrophysiological and behavioral effects of beta1 NF

ERP components

Several methodological differences may account for the failure to replicate the P3b amplitude increments associated with beta1/theta NF in one previous study (Egner & Gruzelier, 2004). Firstly, the two studies differed in the type of auditory attention mechanisms involved in the oddball task. Contrary to the current study, that previous one used a divided attention auditory oddball task in which the participants were instructed to focus the attention on the attended channel, while ignoring the stimuli presented at the unattended channel. This might have resulted in performance advantages associated with top-down biasing mechanisms (Desimone & Duncan, 1995) such as an increase in P3b amplitude. Moreover, in the current experiment the participants had to discriminate

between three types of auditory stimuli, while in the oddball task of the previous study just two stimuli were presented. However, it is unlikely that the introduction of a third stimulus might have affected the P3b amplitude effect. In fact, previous studies suggest that the amplitude of the P3a and P3b components is affected by the probability of the nontarget and targets respectively but not by the probability of the other class of stimulus (Katayama, & Polich, 1996).

Additionally, the physical properties of the stimuli determined that the discrimination between deviant (target and nontarget) and standard stimuli was easier in the current experiment (as demonstrated by the absence of a frontal P3a component) than in that previous one. The higher discrimination difficulty may explain a superior engagement of top-down mechanisms suggested by P3b amplitude increments in frontal and central regions reported by Egner & Gruzelier (2004). Thus, the current experimental situation might have prevented the observation of potential effects of the increased beta1 activity on top-down regulation. On the contrary, the higher demands on top-down regulation imposed by the oddball paradigm used by Egner and Gruzelier (2004) might have facilitated the observation of a P3 amplitude increment effect following beta1/theta NF.

Additionally, the failure to replicate previous findings of increased P3b amplitude might be explained by the different methods of ERP analysis. As seen in section 1.2.4.2.1., it was unclear whether the P3 amplitude increments observed by Egner and Gruzelier (2004) could be interpreted as a selective facilitation of task-relevant stimulus processing because the analysis focused exclusively on the neural response to the target stimuli. On the contrary, the present study isolated the neural response to targets from that associated with standard stimuli in difference waveforms. This allowed the cancellation of the possible contribution of earlier ERP components to the modulation of the P3 and a more clear interpretation of the NF effects on the differential neural response to targets in the P3 time window.

Another aspect that might have contributed to the conflicting results between the two studies was related to the different methods of statistical analysis. In fact, while the current analysis was focused on the regions where the P3 amplitude effect was larger, Egner and Gruzelier (2004) investigated the P3 amplitude in multiple single electrode sites.

The latter method increased the risk of finding a spurious significant effect. Moreover, the failure to adjust the statistical significance level to the number of multiple comparisons might have been responsible for the detection of otherwise insignificant P3 amplitude increments. Finally, because Egner and Gruzelier (2004) did not report electrophysiological results in the control condition it was unclear whether the P3b amplitude increments reflected unspecific effects of NF (e.g., placebo) or practice effects with the oddball task.

Behavioral responses

Also contrary to hypothesis, the beta1 NF was not associated with decreased RT and RT-SD (Hypothesis 2.1 and 2.2.) or increased performance accuracy (Hypotheses 3.1 and 3.2.) relative to SMR NF and control conditions. These findings are in disagreement with one study that showed decreased RT and increased performance accuracy in the cue-CPT following successful beta1/theta NF in ADHD children (Bakhshayesh et al., 2011). A possible explanation to these divergent findings may be that the beta1 NF may contribute to increased target detection efficiency in contexts in which the target is preceded by a cue but not in contexts in which the target is unpredictable. This explanation would be consistent with the role of beta oscillations in response preparation in the foreperiod between cued and target presentation (Kilavik et al., 2013). On the other hand, in situations in which the target stimulus is unpredictable the enhanced attentional control promoted by the beta activity may not represent an advantage. In fact, situations in which the low probability and absence of warning cues determine a higher degree of unpredictability of the target stimulus the facilitation of top-down mechanism may paradoxically induce an increase in RT (Boulinguez, Ballanger, Granjon, & Benraiss, 2009). However, another study suggest that the beta1/theta NF was associated with decreased RT in Go/Nogo tasks in which the stimulus presentation was not preceded by a warning cue (Egner & Gruzelier, 2004).

The failure to demonstrate increased response speed in the context of the oddball task following the beta1 NF is consistent with the hypothesis that the enhancement of beta activity might contribute to the maintenance of the current postural set at the expenses of new movements (Engel & Fries, 2010). In the context of low target probability, the enhancement of beta activity might promote the maintenance of the motor inhibition which would be consistent with the failure to observe a decrease in RT. This explanation is in line

with one previous study in which the beta1/theta NF was not associated with decreased RT in simple and choiceRT tasks with a low target probability (Doppelmayr & Weber, 2011).

The failure to demonstrate an increase in performance accuracy following the betal NF is in line with one previous study in which SMR/theta but not beta1/theta NF was associated with increased d' and increased it rate (Egner & Gruzelier, 2004). However, in the context of the present experiment, the failure to observe increased response accuracy may be related to "ceiling effects" in the percentage of hit rate. As shown in table 4.4., in the present experiment there was a negligible percentage of omission errors (<5%) in all groups at both pre- and post-training possibly reflecting the good cognitive performance of the sample.

The present results also failed to support the hypothesis that the beta1 NF might have a detrimental effect on response inhibition which was proposed to reflect excessive arousal (Egner & Gruzelier, 2004). In fact, despite a slight and negligible (<0.1%) increase in false alarm rate to standards no significant differences between NF were observed in the context of the three-stimuli oddball task. Confirming this idea, the false alarm rate to nontargets showed a slight decrease in beta1 NF comparable to the other experimental conditions.

Taken together, the electrophysiological and behavioral evidence suggests that the the enhancement of beta1 does not necessarily contribute to the facilitation of context updating indexed by the P3 amplitude or to increased response efficiency, consistency and accuracy in situations of an unpredictable target stimulus as would be predicted by an increased involvement of top-down mechanisms of selective attention.

4.4.2. Electrophysiological and behavioral effects of SMR NF

ERP components

In disagreement with the proposal that SMR amplitude enhancement would be associated with increased response inhibition, the present experiment failed to provide evidence of increased P3a amplitude following the SMR NF. However, as seen in section 1.1.1.1., because the extent to which the anterior P3a amplitude reflects response inhibition is still debated (Debener, Makeig, Delorme, & Engel, 2005; Dien, Spencer, & Donchin,

2004; Goldstein, Spencer, & Donchin, 2002) these results must be cautiously interpreted. Moreover, in the easy target-standard discrimination context of the present experiment, the parietal topography of the P3a may reflect the updating of the stimulus representation context rather than the processes of involuntary attention capture and/or response inhibition. Specific recommendations to address this limitation in future studies will be discussed in section 12.3.2.

Also contrary to hypothesis (Hypothesis 1.1.), there was no evidence of increased P3b amplitude in the SMR NF relative to controls. This is in contrast with previous evidence of increased P3b amplitude in a two-stimuli oddball task following SMR/theta NF (Arns et al., 2012). However, differently from the current experiment that study did not control for unspecific effects of NF and used a non-randomized sample of ADHD patients. The present results are, however, in agreement with previous studies that failed to show evidence of significant increments in oddball P3b amplitude following SMR/theta NF in healthy adults (Egner & Gruzelier, 2004).

Behavioral responses

The failure to show an the association of the SMR NF with decreased RT is disagreement with previous findings that indicated a more efficient target detection in simple and choice RT relative to RF NF condition (Doppelmayr & Weber, 2011). In fact, in the present experiment there was a decrease in mean RT from pre- to post-training irrespective of the NF protocol possibly reflecting the increased familiarity with the task at post-training. A reason for these conflicting results might be the difference between studies in the number of NF sessions. Noticeably, the decreased mean RT was observed following thirty daily SMR/theta NF sessions (Doppelmayr & Weber, 2011), while the current and previous studies (e.g., Kober et al., 2015) that failed to replicate that result did not exceed ten NF session. Furthermore, the different behavior of RF NF may also explain the discrepant results. While in Doppelmayr and Weber's (2011) study RF NF was associated with an increased mean RT, in the current one it was associated with a (non-significant) decrease.

The proposed functional association of the SMR with behavioral inhibition additionally suggested a positive impact in reducing the percentage of commission errors.

However, in line with previous studies, the present experiment failed to provide evidence of a specific contribution of the SMR NF to decrease commission errors relative to other NF conditions (Egner & Gruzelier, 2004; Gruzelier et al., 2014). Arguably, a "floor effect" in the false alarm rates imposed by the low response inhibition demands of the task in conjunction with the expected good inhibitory functioning of the sample might have prevented an adequate investigation of this hypothesis. As shown in table 4.4., the false detections of standard and nontarget stimuli taken together did not exceed 3% of the trials.

The present study failed to replicate previous observations of decreased RT-SD following the SMR/theta NF in the specific conditions of an equiprobable target-nontarget DAT (Egner & Gruzelier, 2004). Given that the RT-SD has been found to be modulated by task demands (Hultsch, MacDonald, & Dixon, 2002), a possible explanation for these divergent results may be that the particular cognitive demands imposed by the oddball task were not sensitive enough to the changes in performance consistency induced by the NF manipulation. These differences might be more evident in task contexts that impose demands on response inhibition (Vaurio, Simmonds, & Mostofsky, 2009).

In contrast with previous studies the SMR NF was not associated with increased performance accuracy (Egner & Gruzelier, 2001, 2004; Kober et al., 2015; Vernon et al., 2003). Curioulsy, Egner and Gruzelier (2004) found that the SMR/theta NF was associated with increased d' and hit rate in the context of higher Go stimulus probability and higher perceptual discrimination demands of the DAT (see section 1.2.5.2.1.), suggesting that tasks that impose higher discrimination and response inhibition demands may be more sensitivity to changes in performance accuracy promoted by the SMR NF.

All in all, the present findings did not support the proposal that the SMR NF would be specifically associated with a facilitation of response inhibition. A possible explanation for not observing this effect may be the relatively low response inhibition demands of the three-stimuli oddball task.

4.4.3. Summary

The current results failed to support a functional differentiation between the SMR NF and beta1 NF protocols. In fact, contrary to previous SMR/theta and beta1/theta NF

studies, the electrophysiological and behavioral measures failed to provide evidence of specific impact of any of the NF under investigation on selective attention and response inhibition mechanisms. The current results suggest that SMR and beta1 NF had no effect on the P3 amplitude when controlling for differential neural effects of targets and nontargets relative to standard stimulus and unspecific effects of NF.

Chapter 5

Experiment 3: The effects of SMR and beta1 NF on response inhibition

5.1. Hypotheses

Hypothesis 1: Increased Go and Nogo-P3 amplitude as a function of SMR and beta1 NF

It was hypothesized that:

- (1) Nogo-P3 amplitude would be increased in SMR NF relative to beta1 and control NF and in beta1 NF relative to control NF (SMR > beta1 > control).
- (2) Go-P3 amplitude would be increased in beta1 NF relative to SMR and control NF conditions and in SMR NF relative to control NF condition (beta1 > SMR > control).

Hypothesis 2: Decreased RT and RT-SD as a function of SMR and beta1 NF

It was hypothesized that:

- RT would be decreased in beta1 NF relative to the SMR and control NF and in SMR NF relative to control NF (beta1 < SMR < control).
- (2) RT-SD would be decreased in SMR NF relative to control and beta1 NF and in beta1 NF relative to control NF (SMR < beta1< control).</p>

Hypothesis 3: Increased performance rates and perceptual sensitivity as a function of SMR and beta1 NF

It was hypothesized that:

- d' would be increased in SMR NF relative to control and beta1 NF and in control NF relative to beta1 NF (SMR> control > beta1).
- (2) Hit rate would be increased in beta1 NF relative to the SMR and control NF and in SMR NF relative to control NF (beta1> SMR > control).

(3) False alarm rate would be decreased in SMR NF relative to control and beta1 NF and in control NF relative to beta1 NF (SMR < control < beta1).</p>

5.2. Methods

5.2.1. Participants

The same thirty-one participants of Experiment 1 took part in the present study. As for Experiment 2, after excluding trials with artifacts and averaging the ERP waveforms four other participants (two from control NF and two from beta1 NF) were excluded from the analysis for having less than twenty correct trials in at least one of the four stimulus conditions (Left-cue Go, Left-cue Nogo, Right-cue Go, Right-cue Nogo) at either pre- or post-training. The same participants were included in Experiments 2 and 3 (see table 4.1.).

5.2.2. Stimulus and procedure

The cued Go/Nogo task was presented to the subjects before and after the NF sessions in counterbalanced order with the three-stimuli oddball task (Experiment 2). The order of presentation had no effect on the P3 amplitude for Go and Nogo stimuli or on the behavioral measures of RT and accuracy.

Stimuli were presented electronically using the E-Prime 1.1 software (Psychology Software Tools, Pittsburgh, PA). Stimuli were cues consisting of arrows pointing either left or right and auditory "Go" and "Stop" instructions (Go and Nogo stimuli) presented through speakers. The arrows were constructed as < (Left) and > (Right) signals with an edge length of 35mm presented at the center of a PC-CRT screen in white color against a black background. Go trials (probability=0.67) occurred twice as often as Nogo trials (probability=0.33). The left and right cues were presented for 100 ms with equal probability prior to the Go or Stop auditory instruction. For each participant a total of 160 Go and 80 Nogo trials (summed over left-hand and right-hand responses) were presented which were equally preceded by a left or right cue (80 Right-cue Go; 80 Left-cue Go; 40 Right-cue Nogo; 40 Left-cue Nogo). Participants were instructed to prepare to respond to the direction of the arrows by pressing a button with their right or left thumbs after hearing the Go

instruction and to withhold the prepared response after the Nogo instruction. Auditory "Go" or "Stop" instructions were presented at a fixed interval of 1100 ms after cue onset. The participants were instructed to fixate a white cross in the centre of a black screen in the 1000ms interval after the visual cue and prior to the auditory instruction and in the 1500ms interval after hearing the auditory Go or Nogo instruction and the next visual cue. Response time and accuracy were recorded within the response interval (0 to 1500ms after auditory stimulus presentation). Each trial had the duration of 2600ms. The stimuli were presented in 4 blocks of 60 trials lasting approximately 3 minutes each with short blink-pauses between blocks. Each participant was given a practice trial block consisting of a 20 stimuli presentation with the same probability rate as described above.

5.2.3. EEG recording and analysis

EEG recording.

EEG recording specifications were identical to Experiment 2 (see section 4.2.3).

EEG data analysis.

EEG analysis was performed in Brain Vision Analyzer EEG analysis 1.05 software (Brain Products, Munich, Germany). After data acquisition, the EEG was digitally rereferenced to the average of the left and right earlobes and band-pass filtered (high-pass 0.5 Hz, low-pass 40 Hz). EEG continuous recordings were visually inspected to remove extreme EEG artifacts and segmented in reference to the stimuli in intervals of 1000ms (-200 ms pre-stimulus to 800ms post-stimulus). Only correct trials (correct response to Go stimulus and correct non-response to Nogo stimulus in the 0 to 1200ms interval post-stimulus onset) were analyzed. The data was then submitted to baseline correction (-200 ms relative to stimulus onset) and artefact rejection was performed for ocular movements and EEG fluctuations exceeding \pm 80 μ V. A minimum of 20 trials per condition were used. Subsets of four electrodes were pooled to define three regions of interest, similarly do described in section 4.2.3.

In order to visualize the brain's differential response to Go and Nogo stimuli at preand post-training *Nogo-Go difference waveforms* were calculated by linear subtraction. Percent of correct ERP trials.

Descriptive statistics of percent of rejected trials in the Go/Nogo task are presented in table 5.1. A preliminary analysis revealed that the percentage of correct Go and Nogo responses did not vary as a function of the responding hand (Go: F(1,26)=0.31, p=0.58, $\eta_p^2=0.01$; Nogo: F(1,26)=0.07, p=0.79, $\eta_p^2<0.01$) at pre-training.

As shown in the table 5.1., the one-way ANOVAs failed to provide evidence of initial differences between the NF protocols in the percentage of correct Go and Nogo trials (all *ps*>0.19). However, as shown in the table the two TIME (Pre- and Post-Training) x 3 PROTOCOL (control NF, SMR NF and beta1 NF) mixed ANOVAs revealed significant TIME x PROTOCOL interactions for both the percentage of Go (F(1,24)=4.00, p=0.03, $\eta_p^2=0.25$) and Nogo trials (F(1,24)=8.63, p<0.01, , $\eta_p^2=0.42$). For the Go stimuli, the post-hoc comparisons revealed an decrease in the percentage of correct trials from pre- to post-training in control NF (t(7)=2.44, p=0.02), while the same difference was not significant in the SMR (t(9)=1.49, p=0.15) and beta1 NF (t(8)=0.68, p=0.50). Similarly, for the Nogo stimuli, there was a significant decrease in the percentage of correct trials in control NF (t(7)=4.17, p<0.001), but not in the SMR (t(9)=1.55, p=0.13) and beta1 NF (t(8)=0.82, p=0.42).

5.2.4. Behavioral data recording and analysis

Reaction time data preparation

Behavioral responses were collected using E-Prime 1.0 software (Psychology Software Tools, Pittsburgh, PA). Data preparation for mean RT and RT-SD analysis was as follows. In order to remove outliers from both ends of the RT distribution, correct trials (i.e., responses to the target stimulus) with response latencies below 100 ms or above 1200ms were excluded from the analysis (rejection rate=0.03% of the trials).

d-prime data preparation

In order to guarantee the comparability between RT and ERP data sets only correct responses (button press) to the Go within the 100-1200ms interval were considered "hits" while responses Nogo stimuli within the same interval were considered "false alarms". The d calculation was identical to Experiment 2 described in section 4.2.4.

5.2.5. Statistical analysis

Specifications of statistical packages and criteria were provided in section 3.2.6.

5.3. Results

5.3.1. Hypothesis 1: Increased Go- and Nogo-P3 amplitude as a function of SMR and beta1 NF

After investigating initial differences between the NF protocols at pre-training (section 5.3.1.1), the effects of the three NF protocols from pre- to post-training in the Go-P3 and Nogo-P3 amplitude were examined (section 5.3.1.2).

5.3.1.1. Pre-training differences between NF protocols

Figure 5.1. shows the grand average of the stimulus-locked ERP components elicited by the frequent Go and infrequent Nogo stimuli in midline frontal, central and parietal regions at pre-training irrespective of NF protocol. The ERP waveforms (*Left panel*) and the topographical representations (*Right panel*) show that the Nogo-P3 component was especially enlarged in the central location, while the Go-P3 amplitude reached its maximal amplitude in the parietal region.

In order to investigate initial differences in P3 amplitude between Go and Nogo stimulus as a function of the scalp region in the three NF protocols, a three-way 2 STIMULUS (Go vs. Nogo) x 3 REGION (frontal vs. central vs. parietal) x 3 PROTOCOL (control NF vs. SMR NF vs. beta1 NF) ANOVA was performed.



Figure 5.1. *Left.* Grand average of the stimulus-locked ERP original waveforms elicited by the Go and Nogo stimuli in midline frontal, central and parietal regions at pre-training irrespective of the NF protocol. *Right.* Distribution of the P3 amplitude for the Go and Nogo stimuli. Positive amplitude is displayed upwards.

This analysis yielded significant main effects of STIMULUS (F(1,24)=79.31, p<0.01, $\eta_p^2=0.77$) and REGION (F(1.50,36.17)=72.93, p<0.01, $\eta_p^2=0.75$) as well as a significant interaction of STIMULUS x REGION (F(1.43,34.26)=18.19, p<0.01, $\eta_p^2=0.43$). The main effect of PROTOCOL (F(2,24)=0.76, p=0.48, $\eta_p^2=0.06$) and the interactions involving the between-subject variable were non-significant (STIMULUS x PROTOCOL: F(2,24)=0.20, p=0.81, $\eta_p^2=0.02$; REGION x PROTOCOL: F(3.04,36.17)=0.86, p=0.44, $\eta_p^2=0.07$; STIMULUS x REGION x PROTOCOL: F(2.86,34.26)=0.95, p=0.43, $\eta_p^2=0.07$).

As shown in figure 5.2., the post-hoc comparisons (Bonferroni corrected *p*-values) for the Go stimulus revealed significantly higher P3 amplitude in the parietal region (M=3.94, SD=2.14) relative to central (M=2.70, SD=2.80; t(26)=4.61, p<0.01) and frontal regions (M=-0.50, SD=2.55; t(26)=10.18, p<0.001) and in the central region relative to the frontal region (t(26)=9.15, p<0.001). However, for the Nogo-P3 amplitude was higher in the central region (M=7.64, SD=3.80) relative to parietal (M=6.78, SD=3.17; t(26)=3.37, p<0.01) and frontal regions (M=4.46, SD=3.51; t(26)=8.35, p<0.001) and in the parietal region relative to the frontal region (t(26)=4.60, p<0.001). Moreover, the post-hoc comparisons revealed that the Nogo stimulus elicited a significantly higher P3 amplitude relative to the Go stimulus in the three regions (Frontal: t(26)=8.41, p<0.001; Central: t(26)=8.72, p<0.001; Parietal: t(26)=6.40, p<0.001).

These results replicated previous findings reporting the central topography of the Nogo-P3 component and the parietal topography of the Go-P3 component in the context of the cued Go/Nogo task (Eimer, 1993; Kopp, Mattler, Goertz, and Tist, 1996; Pfefferbaum et al., 1985; Simson, Vaughan, and Ratter, 1977).



Figure 5.2. Two-way interaction of Stimulus and Region in mean P3 amplitude at pre-training. The graph shows that the mean P3 amplitude (μ V) for Nogo stimulus was significantly higher in the Central when compared to Frontal and Parietal locations irrespective of NF protocol at pre-training. For the Go stimulus P3 amplitude was significantly lower in Frontal when compared to Central and Parietal Regions. Error bars depict standard error of the mean.

5.3.1.2. Pre- to post-training differences between NF protocols

Based on the previous analysis of the scalp distribution of the Nogo- and Go-P3 amplitude, the impact of the three NF protocols was analyzed in the regions where these ERP components were maximally observed. Figure 5.3. presents the original waveforms for Nogo stimulus in the central region and for Go stimulus in the parietal region before and after control, SMR and beta1 NF.

Differences in mean parietal Go and central Nogo-P3 amplitude were investigated in a two-way mixed-design ANOVA with the factors 2 TIME (Pre- vs. Post-training) x 3 PROTOCOL (control NF vs. SMR NF vs. beta1 NF). Figure 5.4. shows the mean P3 amplitude in the three NF protocols at pre- and post-training. Data and statistical values are presented in table 5.1. As shown in the table, for the mean Nogo-P3 the two-way ANOVA did not support the hypothesis of a differential impact of the NF (TIME x PROTOCOL: F(2,24)=0.99, p=0.39, $\eta_p^2=0.08$). The main effects of TIME and PROTOCOL were also non-significant (TIME: F(1,24)=1.88, p=0.18, $\eta_p^2=0.07$; PROTOCOL: F(2,24)=0.11, p=0.90, $\eta_p^2=0.01$). The post-hoc comparisons failed to provide evidence of pre- to posttraining Nogo-P3 amplitude changes in any of the three NF protocols (all *ps*>0.10).

A similar two-way ANOVA performed for the Go stimulus indicated a significant effect of the type of NF on the mean P3 amplitude in the parietal region (TIME x

PROTOCOL: F(2,24)=4.27, p=0.03, $\eta_p^2=0.26$). As shown in figure 5.3., the post-hoc comparisons revealed a significant increase in Go-P3 amplitude from pre- to post-training in SMR NF (t(9)=2.31, p=0.03) while the decrease observed in beta1 NF was not reliable (t(8)=1.46, p=0.16). Also, there was a marginal increase in Go-P3 amplitude in control NF (t(7)=1.90, p=0.07). For the Go-P3 amplitude the main effects of TIME and PROTOCOL were also non-significant (ps>0.13).

Table 5.1.

Source							One-	NOVA				Two-	way A	NOVA				
	Pre-training			Post-training			Protocol			Time			Time x Protocol			Protocol		
	Control	SMR	Beta1	Control	SMR	Beta1	F	Р	η_p^2	F	Р	η_p^2	F	Р	η_p^2	F	Р	η_p^2
Mean P3																		
Nogo	7.88 (1.89)	6.75 (4.05)	8.30 (4.63)	8.93 (1.87)	8.53 (3.96)	8.02 (5.55)	_	_	_	1.87	0.18	0.07	0.99	0.39	0.08	0.11	0.90	0.01
Go	3.94 (1.82)	3.26 (1.83)	4.63 (2.65)	4.92 (1.51)	4.33 (1.89)	3.92 (2.59)	-	-	-	2.51	0.13	0.09	4.26	0.03	0.26	0.26	0.77	0.02
H (%)	98.81 (1.56)	99.50 (0.85)	99.39 (0.74)	99.26 (1.08)	99.38 (0.52)	99.60 (0.94)	1.00	0.38	0.08	0.47	0.50	0.02	1.16	0.33	0.09	0.96	0.40	0.07
FA (%)	1.81 (2.05)	2.65 (1.78)	2.83 (3.77)	4.44 (3.23)	2.70 (1.97)	4.17 (3.77)	0.35	0.71	0.03	9.88	0.00	0.29	3.06	0.07	0.20	0.23	0.79	0.02
d'	4.82 (0.99)	4.80 (0.72)	4.90 (0.68)	4.42 (0.69)	4.95 (0.63)	4.46 (1.03)	0.04	0.96	0.00	1.49	0.23	0.06	1.12	0.34	0.09	0.41	0.67	0.03
RT (ms)																		
Mean	362.91 (58.90)	381.87 (91.45)	356.77 (67.80)	298.69 (40.33)	322.70 (64.91)	351.11 (87.06)	0.29	0.75	0.02	11.43	0.00	0.32	2.16	0.14	0.15	0.35	0.71	0.03
RT-SD	112.61 (16.81)	110.99 (28.68)	91.00 (28.52)	97.36 (30.81)	84.55 (21.33)	100.73 (46.00)	1.95	0.16	0.140	1.98	0.17	0.08	2.09	0.15	0.15	0.37	0.69	0.03
CT (%)																		
Go	94.61 (4.51)	91.69 (4.93)	92.50 (6.12)	85.94 (13.39)	96.44 (3.16)	90.21 (12.70)	0.71	0.50	0.06	1.14	0.30	0.05	4.00	0.03	0.25	0.81	0.46	0.06
Nogo	93.59 (4.65)	83.38 (13.27)	80.56 (21.24)	74.06 (18.48)	89.88 (5.42)	76.94 (24.22)	1.77	0.19	0.13	4.70	0.04	0.16	8.63	0.00	0.42	0.68	0.52	0.05

Means (standard deviations) and statistical results of the pre-training one-way ANOVAs and the pre- to post-training two-way ANOVAs on Mean Nogo- and Go-P3 amplitude, Hit Rate (H), False Alarm Rate (FA), d-prime (d'), Reaction Time (RT) Mean and Standard Deviation (RT-SD), percentage of Correct Trials (CT) to Go and Nogo stimuli included in ERP analyses after artifact rejection. Significant differences (p<0.05) are highlighed in bold.



Figure 5.3. Grand-average of the parietal Go and central Nogo ERP waveforms for Control, SMR and beta1 NF at pre- and post-training. The shaded area represents the 250-450ms measurement window. Positive amplitude is displayed upwards.



Figure 5.4. Mean parietal Go- and central Nogo-P3 amplitudes for control, SMR and beta1 NF at pre- and posttraining. **(A)** Go-P3 amplitude was significantly increased in SMR NF, marginally increased in control NF and non-significantly decreased n beta1 NF. **(B)** Nogo-P3 amplitude was not significantly modulated by the NF protocol. Error bars depict standard error of the mean.

5.3.2. Hypothesis 2: Decreased RT and RT-SD as a function of SMR and beta1 NF

As can be observed in table 5.1., two one-way ANOVAs revealed that there were no initial differences between NF protocols in mean RT and RT-SD (all ps>0.16). The differential impact of the three NF protocols in mean RT and RT-SD were analyzed in two-way 2 TIME x 3 PROTOCOL mixed-ANOVAs.

5.3.2.1. Mean reaction time

As shown in fig. 5.5A, the mean RT decreased from pre-to post-training in the three NF protocols but predominantly in the control and SMR NF. As shown in table 5.1., the ANOVA revealed a significant main effect of TIME (F(1,24)=11.43, p<0.001, $\eta_p^2=0.32$) indicating that the RT was significantly lower at post-training (M=324.16; SD=64.10) relative to pre-training (M=367.18; SD=72.72) irrespective of the NF protocol. This effect was driven by significant decreases in mean RT in control (t(7)=2.79, p=0.01) and SMR NF (t(9)=2.84, p<0.01) but not in beta1 NF (t(8)=0.26, p=0.80). However, the TIME x PROTOCOL interaction was non-significant (F(2,24)=2.16, p=0.14, $\eta_p^2=0.15$), thus failing to provide evidence of a superior reduction of mean RT in the SMR and/or control NF

protocols relative to beta1 NF. Finally, the main effect of PROTOCOL (F(2,24)=0.35, p=0.71, $\eta_p^2=0.03$) was not significant.



Figure 5.5. Two-way interaction of 2 Time x 3 Protocol for reaction time. The graphs show the effects at preand post-training of the three NF protocols on mean reaction time (RT) and reaction time standard deviation (RT-SD). **(A)** The mean RT was significantly decreased from pre-to post-training irrespective of the NF protocol. **(B)** There were no significant differences in RT-SD from pre- to post-training as a function of the NF. Error bars depict standard error of the mean.

5.3.2.2. Reaction time standard deviation

As shown in table 5.1. the two-way TIME x PROTOCOL mixed-ANOVA on RT-SD did not reveal significant main effects or interactions (all *ps*>0.15). However, as show in the figure 5.5B, post-hoc comparisons revealed a significantly decrease RT-SD in SMR NF (t(9)=2.13, p=0.04) but not in the other NF protocols (all *ps*>0.28).

5.3.3. Hypothesis 3: Increased performance rates and perceptual sensitivity as a function of SMR and beta1 NF

Table 5.1. summarizes the data and statistical results for mean d', hit rate and false alarm rate at pre- and post-training for the three NF protocols. As can be observed in table, there were no differences between NF protocols at pre-training for any of the dependent variables (all *ps*>0.38). The differential impact of the three NF protocols in mean d', hit rate and false alarm rate were analyzed in two-way 2 TIME x 3 PROTOCOL mixed-ANOVAs.

5.3.3.1. d'

Figure 5.6A, shows the pre-to post-training changes in d' for the three NF protocols. As shown in table 5.1, the two-way ANOVA on mean d' did not provide evidence of significant main effects or interactions (all *ps*>0.23). The post-hoc comparisons between preand post-training d' scores for each NF protocol were non-reliable (all *ps*>0.19).

5.3.3.2. Hit rate

The behavioral data for mean hit rate at pre- and post-training for each of the three NF protocols is presented in table 5.1. The mean percentage of hit rate showed a ceiling effect at pre- (M=99.23%; SD=1.05%) and post-training (M=99.41%; SD=0.84%). Figure 5.6B, shows the pre-to post-training changes in hit rate for the three NF protocols. Despite the apparent decreases in beta1 NF relative to the other two NF protocols, the TIME x PROTOCOL mixed-ANOVA on the mean hit rate revealed no reliable main effects or interactions (all *ps*>0.33), as shown in the table 5.1. The post-hoc comparisons between pre-and post-training hit rate scores for each NF protocol were non-reliable (all *ps*>0.11)

5.3.3.3. False alarm rate

The mean percentage of false alarms showed a floor effect at pre- (M=2.43%; SD=2.53%) and post-training (M=3.77%; SD=2.99%). As shown in figure 5.6C, the false alarm rate was increased after control NF and beta1 NF protocols and unchanged from pre-to post-training in SMR NF. As shown in table 5.1., the two-way TIME x PROTOCOL mixed ANOVA revealed a significant main effect of TIME (F(1,24)=9.88, p<0.01, $\eta_p^2=0.29$) indicating an increase in false alarm rate from pre- to post-training irrespective of the NF protocol. Additionally, the TIME x PROTOCOL interaction was marginally significant (F(2,24)=3.06, p=0.07, $\eta_p^2=0.20$). Post-hoc comparisons revealed that the percentage of false alarms increased significantly in control NF (t(7)=3.34, p<0.01) marginally in beta1 NF (t(8)=1.82, p=0.08) and non-significantly in SMR NF (t(9)=0.07, $p_p^2=0.02$).



Figure 5.6. Two-way interaction of 2 Time x 3 Protocol for d' and performance rates. The graphs show the effects of the three NF protocols on d' (**A**), hit rate (**B**) and false alarm rate (**C**) at pre- and post-training. (**A**) d' was non-significantly changed from pre- to post-training as a function of the NF protocol. (**B**) The hit rate in beta1 NF was non-significantly decreased relative to the other NF protocols. (**C**) The false alarm rate was significantly increased in control NF, marginally increased in beta1 NF and non-significatly increased in SMR NF. Error bars depict standard error of the mean.

5.4. Discussion

The present experiment failed to provide supportive evidence of the contribution of SMR NF (or of the other NF protocols) to improve response inhibition. In fact, hypotheses that the SMR would be associated with increased Nogo-P3 amplitude (Hypothesis 1.1) and a decreased false alarm rate (Hypothesis 3.3) were not supported by the results. Furthermore, the false alarm rate was significantly increased in control NF and, according to hypothesis, marginally increased in beta1 NF (Hypothesis 3.3).

On the other hand, the results did not show the hypothesized involvement of beta1 NF with the facilitation of target detection and processing as would be evinced by increased in Go-P3 amplitude (Hypothesis 1.2), decreased mean RT (Hypothesis 2.1), increased *d'* (Hypothesis 3.1), increased hit rate (Hypothesis 3.2). However, the Go-P3 amplitude was significantly increased in SMR NF and marginally increased in control NF. The results were also suggestive of decreased mean RT and RT-SD in SMR NF, but not in the other NF protocols (Hypothesis 2.2).

5.4.1. Differential effects of low-beta NF on response inhibition

The present study was a first attempt to independently examine the specific contribution of SMR and beta1 NF protocols for improvements in response inhibition in a cued Go/Nogo paradigm. The present experiment failed to provide supportive evidence of the involvement of SMR NF in increasing the Nogo-P3 amplitude in a task that presented a high Go stimulus probability and required a previous response preparation. These results are in contradiction with previous research findings indicating an increased Nogo-P3 amplitude in ADHD children that were successful in regulating their brain activity in a mixed low-beta/theta NF protocol (Kropotov et al., 2005). Several methodological differences between the studies may account for the divergent conclusions. Firstly, there are obvious differences between the samples in terms of age and inhibitory deficits. While the deficits in inhibitory control were evident in Kropotov and collaborators' study (around 20% false alarm rate), in the present experiment the sample of high functioning adults produced only negligible false alarm rate at pre-training (<3% of the trials). Thus, the initial differences in efficiency of inhibitory control and the "floor effect" might explain the progress in ADHD children and the lack of it in high functioning adults. The higher Go stimulus probability relative to the

equiprobable cued-Go/Nogo task used by Kropotov and collaborators (2005), might have not been enough to impose the same level of inhibitory demands in high functioning adults as in ADHD children. An alternative explanation is that the differences in neuroplasticity between the two samples may have resulted in a functional reorganization that compensates initial deficits in response inhibition in children but not in young adults.

Another fundamental difference between the two studies relies on the NF protocols. While Kropotov and collaborators (2005) emphasized both SMR and beta1amplitude enhancement relative to adjacent low and high frequencies in the EEG spectrum (1-30Hz), in the present study the effects of enhancing SMR and beta1 frequencies were independently trained without concomitant inhibition of other EEG frequencies. Finally, the higher number of NF session in that previous study (varying between 15 and 22) provided more training opportunities that might have been related to the higher cognitive improvements. The limitations regarding the differences between NF protocols and number of sessions and will be further discussed in section 12.3.1.

Despite the absence of response inhibition improvements, the examination of pre- to post-training changes in false alarm rates partially supported the claims of a differential impact of SMR NF and beta1 NF protocols in response inhibition. In line with previous studies (Egner & Gruzelier, 2004; Gruzelier, Foks, Steffert, Chen & Ros, 2014), the current one failed to provide supportive evidence of superior improvements in response inhibition following SMR NF but suggested a deterioration of inhibitory functions following beta1 and control NF. Importantly, while in previous studies it was unclear whether the failure to demonstrate improvements following the SMR/theta NF was related to the variable demands in response inhibition imposed by the TOVA task, in the present study the Go response probability of the Go/Nogo task was clarified. However, because the higher Go stimulus probability *per se* might have not been enough to impose a high demand on response inhibition in healthy adults, it remains unclear whether SMR NF may be involved with improved response inhibition in contexts of high response inhibition demands. These results are line with the proposal that the enhanced beta1 activity might be associated with an increase in the percentage of commission errors possibly because of an excessive arousal state (Egner & Gruzelier, 2001). The present study adds to these previous findings by demonstrating that the increase in false alarm rate were observed in conditions of high Go

stimulus probability. Possible reasons for the increased false alarm rate in control NF will be discussed in section 6.2.2.

Importantly, the global increase in the percentage of commission errors from pre- to post-training may be explained by an increase in difficulty of response inhibition determined by the decreased mean RT across the three NF protocols (see below). This explanation is in line with the observation that faster RTs increase the probability of unsuccessful behavioral inhibition (Lappin & Eriksen, 1966) and correlate with the increased amplitude of the Nogo-P3 component (Dimoska, Johnstone, & Barry, 2006; Smith, Johnstone, & Barry, 2006). However, this explanation is unlikely to account for the increased false alarm rate in beta1 NF because, contrary to observed in the control and SMR NF, no changes in response speed were evident in that group.

In sum, the results of the current experiment provided supportive evidence for a differential impact of the NF protocols in response inhibition. Although the results did not confirm the hypothesized improvements in response inhibition following SMR NF, there is evidence to believe that beta1 NF (but more clearly control NF) was associated with the hypothesized deterioration of response inhibition.

5.4.2. Differential effects of low-beta NF on selective atention

The current experiment contributed to clarify the differential contribution of SMR and beta1 NF to the increased efficiency of task-relevant stimulus processing in the specific conditions of the cued Go/Nogo task. The results failed to support the hypothesis of a specific Go-P3 amplitude increment following beta1 NF. However, contrary to hypothesis, the Go-P3 amplitude was significantly increased in SMR NF and marginally increased in control NF (Hypothesis 1.2). These results are in line with previous evidence of increased Go-P3 amplitude in ADHD children under similar cued-Go/Nogo task conditions (Kropotov et al., 2005). However, that previous study was inconclusive regarding the differential contribution of SMR and beta1 NF to the effect because the NF protocol included both SMR and beta1 enhancement instructions. Furthermore, these results constitute a first evidence of specific increments in Go-P3 amplitude following SMR NF in healthy adults. A possible interpretation of this finding is that SMR NF may facilitate the top-down attentional control in situations of a previously prepared response to a frequent target stimulus, consistent with the role of sensorimotor beta frequencies in response preparation (Zhang, Chen, Bressler, & Ding, 2008). Although the significant decreases in mean RT may be better explained by unspecific NF factors or practice effects, this effect was apparently driven by pre- to post-training decreases in control and SMR NF. Moreover, only SMR NF was associated with significant pre- to post-training decreases in RT-SD despite the absence of significant differences relative to other NF protocols. These results are in line with previous increments in response speed and moment-to moment consistency observed in variable contexts of Go stimulus probability (Egner & Gruzelier, 2004). Taken together these results are consistent with the interpretation that SMR may contribute to increased allocation of attentional resources (Go-P3 amplitude) and processing efficiency of task-relevant stimulus (decreased mean RT and RT-SD) in the context of previously prepared and highly probable responses.

However, the present results did not support the hypothesis of a superior performance accuracy following SMR NF in disagreement with previous observations of increased d' and increased hit rate in the TOVA and in a DAT following SMR/theta NF (Egner & Gruzelier, 2004). The failure to replicate these findings in a cued and prepotent Go response context suggests that the specific response priming and Go response probability context of the task may modulate the effects of SMR NF on performance accuracy.

Moreover, the present results did not support the hypothesis that beta1 NF would be associated with decreased RT in a frequent Go response context (Hypothesis 2.1). This finding is in disagreement with previous studies in which the beta1/theta NF was associated with decreased RT in variable conditions of Go response probability in the TOVA (Egner & Gruzelier, 2004) and in a cued-CPT task (Bakhshayesh, Hänsch, Wyschkon, Rezai, & Esser, 2011). The varying conditions response priming and Go stimulus probability of the tasks may explain this contradictory evidence. Speculatively, a decrease in mean RT to targets following the beta1 enhancement may only become evident in conditions of lower Go stimulus probability preceded or not by warning cues or in conditions of high Go stimulus probability without response priming.

5.4.3. Summary

In sum, the present findings were consistent with the contribution of SMR but not of beta1 NF to an increased allocation of attentional resources and processing efficiency of taskrelevant stimulus. Despite the involvement of the sensorimotor beta frequencies in motor and cognitive aspects of response inhibition, none of the NF protocols was associated with improvements in response inhibition in the context of a prepotent Go response. However, the current results are compatible with the interpretation that SMR NF may contribute to prevent an increase in commission errors despite increments in response speed and, therefore, of increasing inhibitory demands. In disagreement with previous proposals, there was no evidence that beta1 NF was associated with increased response speed leading to an increase in commission errors. Thus, the current experiment did not support the theoretical proposal that SMR NF in the right hemisphere would be associated with improved response inhibition to nontarget stimulus and that increased beta1 activity in the left hemisphere would be associated with increasing (Othmer et al., 1999).

Chapter 6

General discussion and conclusions of study 1

6.1. Overview of the main findings

SMR NF was associated with increased attention to targets but not with inhibition

The results of the current study suggest that NF had different implications for the neural processing of target/Go stimulus in the two Go/Nogo tasks. Contrary to hypothesis, beta1 NF did not affect the Go-P3 amplitude in any of the tasks. However, SMR and, to some extent, control NF were associated with increased Go-P3 amplitude in the cued-Go/Nogo task but not in the three-stimuli oddball task. This difference in Go-P3 amplitude modulation in the two task was surprising given that this component indexes similar processes of allocation of attentional resources regardless of the Go stimulus probability (Pfefferbaum, Ford, Weller, & Kopell, 1985).

This difference may be explained by the higher response preparation imposed by the Go stimulus probability and the predictive warning cues (Bruin & Wijers, 2002; Thomas, Gonsalvez, & Johnstone, 2009) in the cued-Go/Nogo relative to the three-stimuli oddball task. Speculatively, the increased response preparation in the former task may have facilitated the process of allocation of attentional resources indexed by the Go-P3 amplitude increments associated with SMR and control NF. Unfortunately, the current study design does not allow for a direct test of whether a differential increase in response preparation in the foreperiod (i.e., between cue and stimulus presentation) might have contributed to differences in allocation of attentional resources in the two tasks. In fact, while in the cued-Go/Nogo task it would have been possible to extract measures of response preparation (such as the lateralized readiness potentials) in the three-stimuli oddball task the absence of warning cue conditions did not allow for an investigation of electrophysiological responses in the foreperiod. Along the same lines, the context of increased response preparation might have favored the increase response speed and consistency following SMR NF. In fact, despite

the non-significant differences relative to other NF protocols, the mean RT and RT-SD was decreased following SMR NF in the cued-Go/Nogo but not in the oddball task.

It could be argued that an increased response preparation state following control and SMR NF would have imposed higher demands on inhibition manifested by an increase in the Nogo-P3 amplitude (Groom & Cragg, 2015; Kok, Ramautar, De Ruiter, Band, & Ridderinkhof, 2004; Randall & Smith, 2011; Smith, Jamadar, Provost, & Michie, 2013). Although this seems to have been the case (see fig. 5.4.), the Nogo-P3 amplitude increments were not significantly different from those in beta1 NF. However, it must be noticed that because response preparation and Nogo-P3 amplitude are only moderately correlated (Smith, Johnstone, & Barry, 2006), the absence of an increased Nogo-P3 amplitude does not invalidate the interpretation that the control and SMR NF might have contributed to an increase in response preparation.

The failure to observe improvements in unsuccessful inhibition (i.e., in false alarm rates) following SMR NF in both Experiment 2 and 3 was surprising in light of previous studies (Egner & Gruzelier, 2001; Fuchs, Birbaumer, Lutzenberger, Gruzelier, & Kaiser, 2003; Kropotov et al., 2005). Interestingly, the different response preparation and inhibitory demands imposed by the tasks was reflected by increased false alarm rates in the control and beta1 NF during the cued-Go/Nogo but not in the three-stimuli oddball task (see figures 4.6. and 5.6.). Taken together, these results suggest that in conditions of higher response preparation and/or inhibitory demands SMR NF might have had protective effect against the increases in false alarm rate. Thus, SMR might facilitate the allocation of attentional resources and processing efficiency of targets in situations of increased response preparation without compromising inhibitory control, while other NF protocols (notably the control condition, but also the beta1) might be associated with more error-prone behavior in contexts of the increased response preparation.

Speculatively, SMR NF may promote the regulation of the cortico-basal-gangliathalamic circuits involved in the integration of competing speed and accuracy demands (Bogacz, Wagenmakers, Forstmann, & Nieuwenhuis, 2010). This hypothesis is consistent with the proposed effects of SMR NF in regulating the excitability of thalamocortical circuits (Egner & Sterman, 2006). However, the current experimental conditions do not allow for a clarification of whether SMR NF may promote a more efficient speed-accuracy trade-off. Specific suggestions to investigate this question in future studies will be proposed in section 12.3.3.

Further, because control NF was also associated with a marginal increase in Go-P3 amplitude one cannot rule out the possibility that similar increments observed in SMR NF may reflect unspecific effects of NF. Thus, a replication of these findings in a larger sample size is needed to clarify the potential impact of SMR NF on the allocation of attentional resources to target stimulus and subsequent consequences for response execution and inhibition in different response preparation task contexts (see also section 12.3.3.).

The cognitive effects of SMR NF do not necessarily reflect increased SMR activity

Another surprising aspect of the present study was that SMR NF was associated with significant cognitive outcomes even in the absence of training- and frequency-specific SMR amplitude increments (Experiment 1). A possible explanation to these findings is that the learning measures might have failed to capture crucial changes in SMR oscillatory activity that may be linked to the enhanced modulation of the Go-P3 component. As previously discussed in section 1.1.2., several oscillatory mechanisms may be responsible for the increased amplitude of late ERP components. In fact, the effects of SMR NF on the Go-P3 amplitude may be explained by changes in evoked power, phase resetting and/or instantaneous phase alignment of task-relevant ongoing oscillations which are not necessarily captured by amplitude learning measures of Experiment 1. Although there is no current support to the contribution of SMR NF to changes in phase resetting during stimulus processing, this possibility gains particular strength in light of new evidence that it may influence SMR phase coherence between central and posterior electrodes in the resting state (Kober et al., 2015). Speculatively, it is possible that the attempt to influence the local SMR amplitude will bring about changes in interregional and local phase alignment in the same and different frequencies. Future studies may attempt to investigate whether the Go-P3 amplitude increments may have resulted from changes in evoked and/or phase oscillatory activity in one or more frequency bands.

Beta1 and RF NF were associated with increased inhibitory difficulties

The current study also provided support to the hypothesis that increments in beta1 activity (Experiment 1) might be associated with a (marginal) increase in false alarm rate in

the cued-Go/Nogo task (Experiment 3) but not in the three-stimuli oddball task (Experiment 2). The higher inhibitory demands of the cued-Go/Nogo highlighted the possible contribution of beta1 NF to increasing difficulties in withholding a prepotent response. Importantly, unlike control NF, there was no evidence that this increasing inhibitory difficulty might be explained by an increase in response speed.

The increased difficulty of inhibition is surprising in light of the evidence that beta activity promotes antikinetic behavior (Gilbertson et al., 2005; Joundi, Jenkinson, Brittain, Aziz, & Brown, 2012). On the other hand, these pattern of results appears to be in line with recent proposals linking beta activity with the facilitation of cue anticipation and processing (Kilavik, Zaepffel, Brovelli, MacKay, & Riehle, 2013). Speculatively, the increased commission errors in performance enhancement contexts may be explained by the an excessive facilitation of sensorimotor transmission following cue onset. Interestingly, the presence of a warning cue did not have a detrimental effect in false alarm rate in ADHD children (Arns, de Ridder, Strehl, Breteler, & Coenen, 2009; Arns, Heinrich, & Strehl, 2014; Bakhshayesh et al., 2011). In that case in which the beta amplitude at pre-training might be relatively lower (Barry, Clarke, & Johnstone, 2003; Clarke, Barry, McCarthy, & Selikowitz, 1998), the increase in beta activity might have had a compensatory effect. Future work is needed to investigate whether beta1 NF might be related to detrimental performance effects in specific contexts of increased response preparation and inhibition demands. This could be achieved by orthogonally manipulating response priming (cue vs. no cue) and Go stimulus probability (high vs. low) as proposed in section 12.3.3.

6.2. Limitations and future directions

Several limitations of the current study warrant a caution interpretation of the findings and deserve further attention in future studies for their possible implications in learning and cognitive outcomes.

6.2.1. Failure to demonstrate independence of SMR and beta1 amplitude increments

The previously discussed patterns of cognitive change could not be related to frequency-specific increments in SMR and beta1 activity. Given that selective amplitude

increments between adjacent frequencies may only be obtained with the increasing number of sessions (Sterman & Friar, 1972), a possible explanation for this failure may be the reduced number of sessions (for a further discussion see section 12.3.1.). The reduced number of sessions may have also limited the possibility of replicating previous acrosssession SMR amplitude increments (Doppelmayr & Weber, 2011; Hoedlmoser et al., 2008; Schabus et al., 2014).

Alternatively, the possibility of differentiating SMR and beta1 amplitude increments may be critically dependent on individual differences in the definition of the low-beta frequency bands. Given that the beta band frequency exhibits considerable inter-individual variability (Pfurtscheller, Stancák, & Edlinger, 1997) the SMR and beta1 frequencies may overlap in the traditionally defined frequency bands, such that increments in one sub-band may lead to increments in the other. In order to prevent this overlap and increase the trainability in a given frequency band some authors have recommended the individual adjustment of the frequency bands (Enriquez-Geppert et al., 2014a; Escolano, Aguilar, and Minguez, 2011; Hanslmayr et al., 2005; Zoefel, Huster, & Herrmann, 2011).

A further limitation of the present study concerns the different placement of the feedback electrodes in the three NF protocols. Although the different electrode placement follows the influential proposal that the regulation cortical activation and deactivation would be better achieved by increasing beta1 activity in the left hemisphere and SMR activity in right hemisphere respectively (Othmer, Othmer, & Kaiser, 1999), this might have resulted in confounding the effects of different types of NF training and hemispheric differences in sensorimotor beta reactivity. In order to address this limitation, future studies may consider monitoring SMR and beta1 amplitude changes from multiple scalp locations in both hemispheres or use a consistent electrode location.

6.2.2. Failure to replicate selective attention and response inhibition findings

Several methodological differences of the current study may explain that only partial support was provided to the differential effects of SMR and beta1 NF on selective attention and response inhibition. Firstly, the reduced number of sessions might have implicated fewer opportunities to develop and consolidate the kind of neural changes that correlate with improved modulation of those cognitive processes. However, the observation of significant changes in performance suggests that factors other than insufficient training may have prevented the replication of some previous results.

Alternatively, the differences relative to previous studies may be related to the different sensitivity of the cognitive measures to the neurobehavioral changes induced by NF. For example, the increased P3b amplitude effect was observed by Egner & Gruzelier (2004) in an oddball task that imposed higher cognitive demands in terms of difficulty of target-standard discrimination and divided attention relative to those observed in Experiment 2. Moreover, the improvements in the false alarm rate associated with SMR learning (Egner & Gruzelier, 2001) were observed in a situation of variable Go stimulus probability and absence of warning cues that imposed lower response inhibition demands than Experiment 3. A related aspect is the presence of "ceiling and floor effects" in the behavioral measures – namely in hit and false alarm rates – which severely limited the possibility of determining whether the failure to observe the hypothesized cognitive effects was related to the NF manipulations or to the low sensitivity of the measures (for a further discussion see section 12.3.1.).

Another aspect that may explain the failure to replicate previous findings relates to the changes introduced in the NF protocols. Differently from most previous studies, the SMR and beta1 NF protocols did not emphasize theta and high-beta suppression. Since the suppression of these frequencies has been proposed to play a coadjuvant role in maintaining the attentional focus and postural composure during successful NF learning (Othmer et al., 1999), the failure to do so may have resulted in less than optimal SMR learning. However, because others have demonstrated the feasibility of enhancing SMR amplitude without concomitant theta and high beta suppression (e.g., Hoedlmoser et al., 2008; Kober, Witte, and Ninaus, 2013) this remains an open question. Regardless of its putative role in facilitating NF learning, theta suppression might have contributed to the changes in cognitive performance observed in previous studies. In fact, the involvement of theta oscillations in response inhibition (Harper, Malone, & Bernat, 2014) and deviance detection (Demiralp et al., 2001) suggests that the positive impact of the low-beta/theta NF protocols on cognitive performance may explained by enhancements in low-beta amplitude as much as by the suppression of theta amplitude. In agreement with this hypothesis, Study 2 examined the
effects of theta suppression NF on the same aspects of selective attention and response inhibition as the current study.

Finally, the absence of NF protocol specific cognitive improvements might be related to unanticipated cognitive effects in the control condition. In fact, RF NF was associated with unexpected increments in Go-P3 amplitude, response speed and false alarm rate. These cognitive changes might have been related to the higher cognitive and motivational demands of having to change the self-regulatory strategy from down- to up-regulation of random frequencies halfway through the training session. Although previous studies failed to report significant cognitive effects with similarly training strategies in RF NF (e.g., Doppelmayr & Weber, 2011), it is unclear whether these higher demands might have induced systematic changes in oscillatory activity with impact on cognitive performance rather than merely controlling for unspecific training effects (e.g., practice and placebo effects). Further studies are needed to ensure that experimental and control conditions are comparable in terms of motivational and/or cognitive aspects with possible implications for cognitive outcomes.

Chapter 7

Study 2: The effects of theta NF and beta1 NF on selective attention and response inhibition

7.1. Introduction

The two major goals of study 2 were to further clarify whether the theta and beta1 NF would be associated (1) with training-specific and frequency-specific theta amplitude decrement and beta1 amplitude increments and (2) with specific improvements in selective attention and response inhibition.

Moreover, Study 2 addressed several methodological limitations that might have been related to the failure to replicate previous cognitive improvements (Doppelmayr & Weber, 2011; Egner & Gruzelier, 2001, 2004; Fuchs et al., 2003; Kropotov et al., 2005) following the independent manipulation of SMR and beta1 enhancement in Study 1.

7.1.1. Beta1 NF effects on selective attention and response inhibition

Beta 1 learning

Study 2 attempted to better clarify whether the beta1 amplitude could be seleectively enhanced relative to other frequency-bands. In Study 1, despite the training-specific beta1 amplitude enhancement effect relative to controls, there was no evidence that the witinsession amplitude changes in the beta1 frequency were larger than in other frequencies. Thus, the current experiment addressed several metholodolgical limitations that could have prevented the observation of this frequency-specific effect.

Firstly, because the smaller number NF sessions relative to previous studies (see tables 1.1. and 1.2.) could have contributed to the absence of training-specific and cognitive enhancement effects, its number was increased (see section 8.2.3.) to provide comparable training opportunities relative to previous studies demonstrating successful modulation of theta (Vernon et al., 2003; Ros et al., 2009; Enriquez-Geppert et al., 2014) and beta

frequencies (Gruzelier et al., 2014a; Schabus et al., 2014; Kober et al., 2015). Moreover, it was assumed that the higher number of NF sessions would contribute to frequency-specific effects in the target frequencies (Lubar & Shouse, 1976; Sterman & Friar, 1972).

Furthermore, in order to increase the responsiveness to NF interventions and reduce the probability of overlap with adjacent but functionally distinct frequencies (e.g., between SMR and beta1) the EEG frequencies were customized to the Individual Alpha Frequency (IAF) following previous recommendations (Escolano, Aguilar, & Minguez, 2011; Ghoshuni et al., 2012; Hanslmayr, Sauseng, Doppelmayr, Schabus, & Klimesch, 2005; Zoefel, Huster, & Herrmann, 2011). Admitting that the beta frequency can be decomposed into three lowbeta (SMR, beta1, beta2) and three high beta (beta3, beta4, beta5) sub-bands, NF was focused on the middle range of the low-beta frequency in order to attempt to prevent an overlap with the adjacent high alpha and high beta bands. A complete description of the frequency bands will be provided in section 8.2.4. In order to keep with the nomenclature of the frequency bands used in the previous study, the SMR was considered to be the first frequency in the beta range, followed by beta1 and so on.

Moreover, it was not clear whether the training-specific effect in the beta1 frequency could be attributed to the within-session amplitude decrements in RF NF which were possibly related to the difficulty in maintaining a consistent self-regultory strategy throughout the session. Thus, the current study introduced some modifications in RF NF condition. Similarly to the previous study, real-time visual NF was provided in selected frequencies of the spectrum other than the target frequencies. However, the participants were allowed to develop of a consistent self-regulatory strategy (i.e., either up- or down-regulation) throughout the session. Also differently from the previous study, the order of selected frequency bands and the up- and down-regulation instructions were randomized and counterbalanced between-subjects and between-sessions.

Cognitive effects of beta1 NF

Despite the involvement of beta activity in top-down control of attention and motor control and previous demonstrations of cognitive improvements following beta1/theta NF, Study 1 failed to provide evidence of the independent contribution of beta1 enhancement to those findings. However, in face of the available evidence and the methodological limitations

identified in the previous study (see section 6.2.) it would be premature to dismiss the possible contribution of beta1 enhancement NF to the improvements in selective attention and response inhibition.

Moreover, in Study 1 the false alarm rate was marginally increased in beta1 NF probably reflecting an increased difficulty in inhibiting a previously prepared and prepotent response. This warranted a reexamination of whether or not enhancements in the beta1 amplitude may have a detrimental impact on response inhibition (Egner & Gruzelier, 2001, 2004).

7.1.2. Theta NF effects on selective attention and response inhibition

The present study was a first attempt to independently investigate the implications of central midline theta suppression for selective attention and response inhibition improvements. In the previous study, the absence of concomitant suppression of theta amplitude in the SMR and betal NF protocols was advanced as a possible explanation for the failure to provide supportive evidence of these improvements. Although theta suppression has been assigned a coadjuntive role in maintaining the necessary attentional focus for successful self-regulation of sensorimotor beta activity (Sterman and Shouse, 1980; Othmer et al., 1999), previous evidence of significant cognitive improvements associated with theta suppression (see section 1.2.4.3.) and the involvement of theta activity in cognitive control (see section 1.1.1.2. and 1.1.2.2.) suggest that it may play a more active role in the cognitive performance enhancement.

Theta learning

As noted above in section 1.2.3.1.3., previous studies low-beta/theta NF studies found supportive evidence of within-session theta amplitude decrements in both central (de Zambotti, Bianchin, Magazzini, Gnesato, & Angrilli, 2012; Gruzelier, Hirst, Holmes, & Leach, 2014; Ros et al., 2009; Vernon et al., 2003) and occipital regions (Beatty, Greenberg, Deibler, & O'Hanlon, 1974; Williams, Beatty, & O'Hanlon, 1975). However, other studies failed to corroborate these findings (Kober et al., 2015). Moreover, as discussed in section 1.2.3.2.3., there is no current evidence of across-session theta amplitude decrements in active (Doppelmayr & Weber, 2011; de Zambotti et al., 2012; Kober et al., 2015) or passive periods

(Becerra et al., 2012). As discussed in section 1.2.3.3.3., previous sudies were also inconclusive regarding the frequency-specificity of the theta suppression effects (Becerra et al., 2012). Thus, the current study attempted to clairfy whether within-session and/or across-session methods may capture the dynamic changes in theta amplitude induced by NF and whether those amplitude decrements were frequency-specific.

Cognitive effects of theta NF

The correlation of low tonic and high phasic theta amplitude with good cognitive performance (Klimesch, 1999) and the contribution of evoked theta power to the generation of the P3 component and automatic deviance detection (Demiralp et al., 2001; Yordanova et al., 2003) suggest that the suppression of ongoing oscillatory theta may be associated with increased P3 amplitude. Moreover, the association of theta activity with cognitive control suggests that the suppression of theta activity would be specifically novelty detection indexed by the P3a (Luu & Tucker, 2003; Cavanagh & Frank, 2014). Also based on previous evidence of that low pre-stimulus theta was associated with significant Nogo-P3 amplitude modulations in the Go/Nogo task (De Blasio & Barry, 2013; Harper, Malone, & Bernat, 2014; Kirmizi-Alsan et al., 2006) it was proposed that theta suppression NF would be associated with improvements in electrophysiological and behavioral indices of response inhibition. Additionally, there is currently evidence of the involvement of theta oscillations in performance consistency as measured by RT-SD in ADHD children and typically developing individuals across the life-span (McLoughlin, Palmer, Rijsdijk, & Makeig, 2014; Papenberg, Hämmerer, Müller, Lindenberger, & Li, 2013) which may explain the decrease RT-SD following SMR/theta NF (Egner & Gruzelier, 2004),

7.2. Study 2 outline

Study 2 sought to investigate the differential electrophysiological and behavioural effects of theta and beta1 NF protocols on selective attention and response inhibition in three experiments. Despite acknowledging the limitations resulting from the failure to independently manipulate Go response probability and response priming effects in the two Go/Nogo tasks these were kept unchanged in Study 2. In fact, this would have implied

modifying the Go response probability in the two task (to keep a proportional ratio between Go and Nogo stimulus) and introducing two new tasks (i.e., an oddball task with response priming and a non-cued Go/Nogo task) with implications for task complexity and length. This would also have reduced the comparability with Study 1 and with the previous literature.

Experiment 4 investigated whether (1) the experimental manipulations of thetasupression NF (theta NF) and beta1-enhancement NF (beta1 NF) were successful in inducing training- and frequency-specific amplitude changes in the frequencies of interest relative to a control condition (RF NF) and (2) to understand whether these effects would be differentially apprehended by within- and across-session methods in active and passive periods. Additionally, the present experiment addressed some methodological limitations of the previous study in control NF condition, in the number of NF sessions and in the definition of the target frequency bands.

Experiments 5 and 6 examined the electrophysiological and behavioural effects of theta and beta1 NF in the same two Go/Nogo tasks used in Study 1. Experiment 5 sought to examine the relative contribution of theta-inhibition and beta-enhancement to improvements in voluntary and involuntary selective attention in a three-stimuli oddball task (Hillyard, Squires, Bauer, & Lindsay, 1971; Hillyard & Kutas, 1983; Katayama, & Polich, 1996; Polich, 2007, 2012; Squires, Donchin, Herning, & McCarthy, 1977) as suggested by previous low-beta/theta NF (Arns et al., 2012; Bakhshayesh et al., 2011; Doppelmayr & Weber, 2011; Egner & Gruzelier, 2001, 2004; Kober et al., 2015).

Experiment 6 sought to examine the relative contribution of theta-inhibition and betaenhancement to improvements in electrophysiological and behavioral indices of response inhibition in a cued Go/Nogo task as proposed by previous low-beta/theta NF studies (Bakhshayesh et al., 2011; Egner & Gruzelier, 2001, 2004). In this task, the lower probability of the Nogo stimulus relative to the Go target stimulus and the pre-stimulus preparation of a motor response imposed specific demands on response inhibition (Eimer, 1993; Kok, 1986; Pfefferbaum, Ford, Weller, & Kopell, 1985; Randall and Smith, 2011; Smith, Johnstone, & Barry, 2006, 2008).

Chapter 8

Experiment 4: Theta and beta1 learning

8.1. Hypotheses

Hypothesis 1: Training-specificity of within-session theta and beta1 amplitude changes in active periods.

It was hypothesized that:

- (1) Theta amplitude would be decreased within-sessions in active periods in theta NF relative to beta1 NF and control NF.
- (2) Beta1 amplitude would be increased within-sessions in active periods in beta1 NF relative to theta NF and control NF.

Hypothesis 2: Training-specificity of across-session theta and beta1 amplitude changes in active periods.

It was hypothesized that:

- (1) Theta amplitude would be decreased across-sessions in active periods in theta NF relative to beta1 NF and control NF.
- (2) Beta1 amplitude would be increased across-sessions in active periods in beta1 NF relative to theta NF and control NF.

Hypothesis 3: Training-specificity of within-session theta and beta1 amplitude changes in passive periods.

It was hypothesized that:

- (1) Theta amplitude would be decreased within-sessions in passive periods in theta NF relative to beta1 NF and control NF.
- (2) Beta1 amplitude would be increased within-sessions in passive periods in beta1 NF relative to theta NF and control NF.

Hypothesis 4: Training-specificity of across-session theta and beta1 amplitude changes in passive periods.

It was hypothesized that:

- Theta amplitude would be decreased across-sessions in passive periods in theta NF relative to beta1 NF and control NF.
- (2) Beta1 amplitude would be increased across-sessions in passive periods in beta1 NF relative to theta NF and control NF.

Hypothesis 5: Frequency-specificity of theta and beta1 amplitude changes in active periods.

It was hypothesized that:

- (1) Theta NF would be associated with decreased mean theta amplitude but not with significant amplitude changes in other frequencies relative to beta1 NF and control NF on the average of active periods within-sessions.
- (2) Beta1 NF would be associated with increased mean beta1 amplitude but not with significant amplitude changes in other frequencies relative to theta NF and control NF on the average of active periods within-sessions.

Hypothesis 6: Frequency-specificity of theta and beta1 amplitude changes in passive periods.

It was hypothesized that:

- (1) Theta NF would be associated with decreased mean theta amplitude but not with significant amplitude changes in other frequencies relative to beta1 NF and control NF on the average of passive periods within-sessions.
- (2) Beta1 NF would be associated with increased mean beta1 amplitude but not with significant amplitude changes in other frequencies relative to theta NF and control NF on the average of passive periods within-sessions.

8.2. Methods and materials

8.2.1. Participants

Twenty-nine healthy adult students from Goldsmiths, University of London volunteered to take part in the experiment. Twenty-three participants were first-year psychology undergraduate students recruited through a Research Participation Scheme in return of course credit. Six participants were postgraduate students recruited through advertisement in the Campus and received monetary compensation of £50 at completion. All participants were right-handed and had normal or corrected-to-normal vision. None of the participants had previous experience with NF. Prior to the pre-assessment general information regarding the goals of the study and about EEG and NF procedures was given. Participants remained blind to the details of the experimental design and to the frequencybands of interest. Participants gave written informed consent for participation in the study, which had previously been approved by the Department of Psychology Ethics Commission. Participants were screened for neuropsychiatric disorders and psychoactive substances use (see Appendix A). Participants were randomly allocated to one of three experimental conditions via rotation (starting with control NF condition) resulting in a control NF group consisting of 9 participants (5 females; mean age= 21.44 years; SD=4.03 years), a thetainhibition NF group consisting of 10 participants (6 females; mean age= 22.60 years; SD=6.31 years) and beta1 NF group consisting of 10 participants (8 females; mean age= 21.7 years; SD=5.29 years). The mean age did not differ between groups as demonstrated by oneway ANOVA (F(2,26)=0.13, p=0.88, $\eta_p^2=0.01$).

8.2.2. General procedure

The general procedure was identical to Experiment 1 (see section 3.2.2) except that sessions in the current experiment participants received eight (instead of six) NF sessions. Moreover, 3 minutes of eyes closed EEG was recorded in the pre-training and post-training assessment sessions for determination of the IAF. Thus, the participants attended a total of ten sessions in which the first and the last session were dedicated to EEG/ERP and behavioural outcome tasks (for details see sections 4.2.2. and 5.2.2.).

8.2.3. Neurofeedback Procedure

The NF procedure was identical to Experiment 1 (see section 3.2.3.) except that the participants received either theta-suppression, beta1-enhancement or control NF (see below). The NF sessions were distributed over three to four weeks according to the participants availability.

8.2.4. Neurofeedback training

EEG recording. EEG recording was identical to Experiment 1 (see section 3.2.4.) except that the peak-to-peak amplitudes were calculated to continuously extract EEG components customized to the IAF ranging from delta to gamma (see details of the frequency bands definition below). Furthermore, the scalp electrode was placed at Cz against a left earlobe reference electrode and ground electrode placed at the right earlobe according to the 10-20 electrode placement system irrespective of the NF protocol.

Frequency Bands. For purposes of feedback and statistical analyses, frequency band limits from delta to gamma bands were calculated individually for each participant according to the IAF. Frequency band limits determination followed the theoretical subdivision of three alpha bands outlined by Klimesch (1999). The subdivision of the low-beta band into SMR and beta1 frequency followed the 3 Hz bins bandwidth commonly encountered in the NF literature (e.g., Egner & Gruzelier, 2004). The resultant twelve frequency bands were defined as follows:

- delta (IAF 8 Hz to IAF 6 Hz);
- theta (IAF 6 Hz to IAF 4 Hz);
- alpha1 (IAF Hz 4 Hz to IAF 2 Hz);
- alpha2 (IAF Hz 2Hz to IAF);
- alpha3 (IAF to IAF + 2 Hz);
- SMR (IAF + 2 Hz to IAF + 5 Hz);
- beta1 (IAF + 5 Hz to IAF + 8 Hz);
- beta2 (IAF + 8 Hz to IAF + 11 Hz);
- beta3 (IAF Hz + 11Hz to IAF + 14 Hz);
- beta4 (IAF Hz + 14Hz to IAF + 17 Hz);

- beta5 (IAF + 17 Hz to IAF + 20 Hz); and
- gamma (IAF + 20 Hz to IAF + 30 Hz).

Determination of the IAF. For the determination of IAF, the 3-minute long eyes closed EEG was resampled at 512 Hz and segmented in 2 second intervals (power resolution = 0.256 Hz). Fast Fourier Transform (FFT) was calculated on the full spectrum with a 10% Hanning window in BVA. The EEG signal was off-line band-pass filtered (0.1-100Hz) with a Butterworth Zero Phase 12 dB/oct filter. A notch filter (50Hz) was also applied. Ocular artefact correction was performed using Gratton and Coles method (Gratton, Coles, & Donchin, 1983). A sliding window of 500ms was applied and segments exceeding 100µV in any of the 64 scalp channels were rejected. IAF was determined by visually inspecting the pre-training EEG power spectrum for the peak of alpha power at Cz in order to maintain consistency with the NF training site. IAF ranged from 8 to 12 Hz (Mean=9.56, SD=0.95). A preliminary one-way ANOVA failed to reveal significant differences between the NF protocols in IAF (F(2,26)=0.89, p=0.42, $\eta_p^2=0.06$).

Neurofeedback task. The procedures of the neurofeedback task were identical to Experiment 1 (see section 3.2.4.) except that the video was shown prior to recording for one minute to reduce novelty effects of the feedback screens and allow for an adaption to the training settings. Also, to ensure the comparability of resting-state and training periods the same display was used in both conditions except that on the former the thresholds on the target-frequency band were disabled and the video kept on constant playback.

Feedback screens. In order to keep the participants motivated during the sessions the video displays were randomly selected from a set of eight. In this manner each participant was exposed to a different video every session. The feedback consisted of visual representations of peak-to-peak amplitude: two bar graphs, and a playing video. The bar graphs represented the frequency-band activity in the target-frequency band and the EMG band. The bars were green when the activity was on the correct side of a manually set threshold and turned red at overstepping it. A score point at the top-left of the screen was allotted for every two seconds of continuous above-threshold activity. The screens were randomly chosen from a set of eight so that participants were exposed to a different video every session. The videos

displayed animations that played in a continuous manner whenever the training requirements were met and stopped when they were not (see Reward contingencies section below). Appendix C shows an example of the feedback screens.

Feedback instructions and groups. The instructions for theta NF protocol emphasized keeping the bar graph representing the theta frequency's amplitude as low and for as long as possible, while the reverse was emphasized for the beta NF protocol. The participants in control NF protocol received suppression or enhancement instructions according to the pseudo-random counterbalanced order described above. In control NF group the participants received real-time feedback on the peak-to-peak amplitude of four different frequencies (alpha2, alpha3, beta3 and beta4). Note that to prevent possible leakage effects to the theta and beta1 frequencies the adjacent frequencies (e.g., delta and alpha1 and SMR and beta2) were purposively left out from the target frequencies used in control NF protocol.

Also, in order to cancel out possible effects of frequency order (e.g., alpha2 followed by alpha 3 and so on), of the type of frequency (low- vs. high-frequencies) and of training instruction (inhibition vs. enhancement) these factors were counterbalanced in a pseudo-random manner across sessions. Thus, from one session to the next the participants received real-time feedback in a different frequency, of a different type and with a different instruction. The participants in the control condition remained by blind regarding the changing training contingencies.

Reward contingencies. Reward contingencies were identical to the exposed in section 3.2.4.

8.2.5. EEG-Neurofeedback Data Processing

The processes of data preparation, artifact detection and peak-to-peak calculations were identical to those of Experiment 1 (see section 3.2.5.). The epoch-rejection rate per participant ranged from 3.81% to 7.10% (*Mean* = 5.51%, *standard deviation* = 0.77%). The percentage of rejected trials did not differ between groups as demonstrated by one-way ANOVA (F(2,26)=0.33, p=0.54, $\eta_p^2=0.04$).

8.2.6. Statistical Analyses

The statistical analysis procedures were identical to those of Experiment 1 (see section 3.2.6.).

8.3. Results

The within-session and across-session theta and beta1 amplitude changes in the three NF groups were separately analyzed for active and passive trials. The normalized within-session amplitude change in active periods was calculated for 8 periods. For passive periods there was only a single measure of normalized within-session change. Thus, four different analyses were performed (1) two-way 8 PERIOD x 3 PROTOCOL for within-session active periods, (2) one-way 3 PROTOCOL for within-session passive periods, (3) two-way 8 SESSION x 3 PROTOCOL for across-session active periods, and (4) two-way 8 SESSION x 3 PROTOCOL for across-session passive periods.

8.3.1. Training-specific learning

Hypothesis 1: Training-specificity of within-session theta and beta1 amplitude changes in active periods.

Within-session theta and beta1 amplitude changes in each of the NF protocols were investigated in separate two-way 8 PERIOD x 3 PROTOCOL ANOVAs with normalized amplitude change for each period relative to the first training period collapsed across sessions as the dependent measure.

Figures 8.1A and 8.1B show the within-session amplitude in the theta and beta1 frequencies for the three NF protocols. These were investigated in separate two-way 8 PERIOD x 3 PROTOCOL ANOVAs for theta and beta1 frequencies. The analyses indicated a marginally significant effect of PROTOCOL for theta (F(2,26)=2.97, p=0.07, $\eta_p^2=0.19$) but not for beta1 (F(2,26)=0.73, p=0.49, $\eta_p^2=0.05$). Contrary to hypothesis (Hypothesis 1), pairwise comparisons by independent *t*-tests failed to reveal training-specific decrements of mean theta amplitude in theta NF when compared to control NF (t(17)=0.75, p=1.00) and

beta1 NF (t(18)=1.66, p=0.33). However, the theta amplitude was marginally decreased in the control relative to beta1 NF (t(17)=2.36, p=0.08).

As shown in figures 8.1A and 8.1B, the analyses revealed significant PERIOD x PROTOCOL interaction effects in the theta (F(8.18,106.28)=2.26, p=0.03, $\eta_p^2=0.15$) but not in the beta1 frequency (F(6.10,79.35)=0.50, p=0.81, $\eta_p^2=0.04$). Descriptive statistics of theta and beta1 amplitude change and statistical results of between-subjects multiple comparisons for each period are presented in table 8.1. As shown in the table, theta amplitude change was marginally decreased in control relative to beta1 NF in periods 4, 6 and 7 (all other ps>0.13) and in theta relative to beta1 NF in period 7 (all other ps>0.15). None of the pairwise comparisons between control and theta NF were reliable (all ps>0.26).

Within-subject pairwise comparisons between periods relative to the first period performed for each NF protocol, revealed that in control NF the theta amplitude change was significantly decreased in all the periods (all *ps*<0.05). In theta NF, the theta amplitude was significantly decreased in periods 6 through 8 (all *ps*<0.05), marginally decreased in period 5 (*p*<0.10) and non-significantly decreased in periods 2 through 4 (all *ps*>0.36). Finally, in beta1 NF there was no evidence of theta amplitude decrements between the first and the remaining periods (all *ps*>0.33). The main effect of PERIOD was significant in the theta but non-significant in the beta1 frequency (theta: *F*(4.09,106.28)=15.63, *p*<0.001, η_p^2 =0.38; beta1: *F*(3.05,79.35)=0.67, *p*=0.58, η_p^2 =0.03).



Figure 8.1. Within-session theta and beta1 amplitude change in active periods for control NF, theta NF and beta1 NF. The graphs represent the mean amplitude change in eight active periods (from P1 to P8) collapsed across eight training sessions relative to the first active period of each session for theta (A) and beta1 frequencies (B). Error bars depict standard error of the mean. Statistical significance level is represented by asterisks (\dagger =p<0.10, *=p<0.05, **=p<0.01).

				Prot	Independent <i>t</i> -test								
		Cor	ntrol	Th	eta	Be	ta1	Control	Control	Theta			
Frequency	Period	М	SD	м	SD	М	SD	vs. Theta	vs. Beta1	vs. Beta1			
Theta	1	1.00	0.00	1.00	0.00	1.00	0.00	_	_	_			
	2	0.98	0.03	0.99	0.01	1.00	0.02	1.78	2.07	0.30			
	3	0.97	0.03	0.98	0.02	0.98	0.03	1.41	1.38	0.03			
	4	0.96	0.04	0.97	0.02	1.00	0.04	0.86	2.48 †	1.67			
	5	0.96	0.04	0.96	0.02	0.99	0.04	0.25	1.67	1.46			
	6	0.95	0.03	0.96	0.02	0.99	0.05	0.45	2.45 †	2.06			
	7	0.95	0.03	0.95	0.01	0.99	0.04	0.09	2.33 †	2.48 †			
	8	0.95	0.04	0.96	0.02	0.98	0.04	0.59	2.14	1.60			
Beta1	1	1.00	0.00	1.00	0.00	1.00	0.00	_	_	_			
	2	0.99	0.02	1.00	0.02	1.01	0.02	0.28	1.68	1.44			
	3	0.99	0.02	1.00	0.03	1.01	0.02	1.09	1.65	0.57			
	4	0.99	0.03	1.01	0.03	1.02	0.03	1.21	1.47	0.26			
	5	0.99	0.03	1.01	0.03	1.01	0.04	0.85	0.70	0.15			
	6	1.00	0.03	1.01	0.03	1.00	0.03	0.82	0.40	0.43			
	7	0.99	0.05	1.00	0.03	1.00	0.04	0.62	0.56	0.07			
	8	0.99	0.05	1.00	0.03	1.01	0.05	0.42	1.04	0.64			

Table 8.1. Mean (and SD) of theta and beta1 amplitude change (μ V) for each period (normalized to period 1 of each session) for control NF, theta NF and beta1 NF; independent *t*-tests (Bonferroni corrected *p*-values) betw een control NF, theta NF and beta1 NF.

†= p < 0.10, *= p < 0.05, **= p < 0.01

Hypothesis 2: Training-specificity of across-session theta and beta1 amplitude changes in active periods.

Figures 8.2A and 8.2B show the within-session theta and beta1 amplitude changes in active periods across-sessions in the three NF protocols. Across-session theta and beta1 amplitude changes as a function the type of NF protocol were investigated in separate 8 SESSION x 3 PROTOCOL mixed ANOVAs. Contrary to hypotheses, the SESSION x PROTOCOL interactions were non-significant for both frequencies (theta: F(14,182)=0.86, p=0.60, $\eta_p^2=0.06$; beta1: F(14,182)=0.96, p=0.50, $\eta_p^2=0.07$).

Means (and SD) of theta and beta1 amplitude change in active periods for each session are presented in table 8.2. As shown in the table, between-subject pairwise comparisons (Bonferroni corrected p-values) for each session did not provide evidence of significant within-session theta amplitude differences between control and theta NF (all ps>0.44), between control and beta1 NF (all ps>0.15), or between theta and beta1 NF (all ps>0.20). Similarly, for the beta1 frequency the post-hoc comparisons failed to show significant within-session beta1 amplitude differences between control and theta NF (all

ps>0.43), between control and beta1 NF (all *ps*>0.14), or between theta and beta1 NF (all *ps*>0.51).



Figure 8.2. Across-session theta and beta1 amplitude change for control NF, theta NF and beta1 NF in active periods. The graphs represent the amplitude change in eight training sessions (from S1 to S8) for theta (A) and beta1 frequencies (B). Amplitude change was normalized relative to the first active period of the first session. Error bars depict standard error of the mean. Statistical significance level is represented by a cross and an asterisk ($^+=p<0.10$; $^{*}=p<0.05$, $^{**}=p<0.01$).

The analyses also revealed a marginally significant effect of SESSION for the theta $(F(7,182)=1.86, p=0.08, \eta_p^2=0.07)$ but not for the beta1 frequency $(F(7,182)=1.25, p=0.28, \eta_p^2=0.05)$. However, the Bonferroni corrected post-hoc comparisons between sessions failed to provide evidence of significant changes in within-session theta amplitude (all *ps*>0.20). Post-hoc contract within-subject contrasts also failed to provide evidence of linear effects of SESSION in both frequencies (theta: $F(1,26)=2.48, p=0.13, \eta_p^2=0.09$; beta1: $F(1,26)=1.40, p=0.25, \eta_p^2=0.05$). The main effect of PROTOCOL revealed marginally significant average within-session amplitude changes in the theta but not in the beta1 frequency (same as Hypothesis 1).

				Prot	Independent <i>t</i> -test									
	-	Cor	ntrol	Th	heta Beta1		Control	Control	Theta					
Frequency	Session	М	SD	м	SD	м	SD	vs. Theta	vs. Beta1	vs. Beta1				
Theta	1	0.95	0.06	0.97	0.05	0.99	0.08	0.57	1.45	0.90				
(active)	2	0.94	0.07	0.97	0.05	1.01	0.09	0.93	2.05	1.16				
	3	0.98	0.07	0.94	0.04	1.00	0.09	1.08	0.78	1.91				
	4	0.92	0.06	0.96	0.07	0.99	0.09	0.95	2.10	1.18				
	5	0.94	0.05	0.93	0.08	0.97	0.08	0.33	0.85	1.22				
	6	0.97	0.12	0.96	0.06	1.01	0.11	0.25	0.85	1.13				
	7	0.98	0.07	0.99	0.05	0.96	0.04	0.45	0.67	1.15				
	8	0.89	0.05	0.95	0.10	0.93	0.10	1.49	0.94	0.57				
Beta1	1	0.97	0.08	1.02	0.05	1.04	0.09	1.51	2.08	0.58				
(active)	2	0.98	0.05	1.01	0.07	0.99	0.08	0.78	0.38	0.41				
	3	1.01	0.07	1.00	0.07	1.00	0.09	0.26	0.30	0.03				
	4	1.00	0.06	1.02	0.04	0.99	0.08	0.74	0.19	0.95				
	5	1.00	0.08	0.98	0.11	0.98	0.10	0.43	0.35	0.08				
	6	1.01	0.09	0.98	0.06	1.03	0.09	0.82	0.56	1.41				
	7	0.99	0.07	1.01	0.03	1.03	0.07	0.77	1.48	0.73				
	8	0.94	0.08	0.96	0.09	1.01	0.09	0.52	1.75	1.26				
Theta	1	0.95	0.06	0.97	0.05	0.99	0.08	0.57	1.45	0.90				
(passive)	2	0.94	0.07	0.97	0.05	1.01	0.09	0.93	2.05	1.16				
	3	0.98	0.07	0.94	0.04	1.00	0.09	1.08	0.78	1.91				
	4	0.92	0.06	0.96	0.07	0.99	0.09	0.95	2.10	1.18				
	5	0.94	0.05	0.93	0.08	0.97	0.08	0.33	0.85	1.22				
	6	0.97	0.12	0.96	0.06	1.01	0.11	0.25	0.85	1.13				
	7	0.98	0.07	0.99	0.05	0.96	0.04	0.45	0.67	1.15				
	8	0.89	0.05	0.95	0.10	0.93	0.10	1.49	0.94	0.57				
Beta1	1	0.97	0.08	1.02	0.05	1.04	0.09	1.51	2.08	0.58				
(passive)	2	0.98	0.05	1.01	0.07	0.99	0.08	0.78	0.38	0.41				
	3	1.01	0.07	1.00	0.07	1.00	0.09	0.26	0.30	0.03				
	4	1.00	0.06	1.02	0.04	0.99	0.08	0.74	0.19	0.95				
	5	1.00	0.08	0.98	0.11	0.98	0.10	0.43	0.35	0.08				
	6	1.01	0.09	0.98	0.06	1.03	0.09	0.82	0.56	1.41				
	7	0.99	0.07	1.01	0.03	1.03	0.07	0.77	1.48	0.73				
	8	0.94	0.08	0.96	0.09	1.01	0.09	0.52	1.75	1.26				

Table 8.2. Mean (and SD) of normalized theta and beta1 amplitude change (μ V) in active and passive periods for each session and forcontrol NF, theta NF and beta1 NF; independent *t*-tests (Bonferroni corrected *p*-values) betw een control NF, theta NF and beta1 NF.

†= p < 0.10, *= p < 0.05, **= p < 0.01

Hypothesis 3: Training-specificity within-session theta and beta1 amplitude changes in passive periods.

Within-session theta and beta1 amplitude changes in passive periods were investigated in one-way between-subjects ANOVA. Data and statistical results are presented in table 8.3. For the theta frequency the analysis revealed a marginally significant effect of PROTOCOL (F(2,26)=3.36, p=0.05, $\eta_p^2<0.21$). As shown in figure 8.4B, post-hoc comparisons indicated that the within-session theta amplitude was significantly decreased in theta relative to beta1 NF (t(18)=2.43, p=0.02), marginally decreased in control relative to beta1 NF (t(17)=2.01, p=0.06) but not significantly different between control and theta NF (t(18)=0.34, p=0.74). Contrary to hypothesis, the main effect of PROTOCOL was not significant in the beta1 frequency (F(2,26)=0.16, p=0.86, $\eta_p^2=0.01$).

Hypothesis 4: Training-specificity of across-session theta and beta1 amplitude changes in passive periods.

Figures 8.3A and 8.3B show the theta and beta1 amplitude changes in passive periods across-sessions in the three protocols. Across-session theta and beta1 amplitude changes in passive periods was investigated in separate two-way 8 SESSION x 3 PROTOCOL mixed-ANOVAs. Table 8.2. presents means (and SDs) of average theta and beta1 amplitude change in passive periods relative to the first resting state period of each session.

In the theta frequency the analysis failed to reveal significant changes across sessions as a function of the type of NF (SESSION x PROTOCOL:F(14,182)=0.86, p=0.60, $\eta_p^2=0.06$). However, there was a marginal effect of SESSION (F(7,182)=1.86, p=0.08, $\eta_p^2=0.07$). Polynomial contrasts failed to provide evidence of a significant linear increment in theta amplitude change across sessions irrespective of the NF protocol (F(1,26)=2.48, p=0.13, $\eta_p^2=0.09$). Bonferroni corrected pairwise comparisons also failed to provide evidence of significant theta amplitude changes between sessions (all ps>0.20).

In the beta1 frequency the main effect of SESSION (F(7,182)=1.25, p=0.28, $\eta_p^2=0.05$) and the interaction effect of SESSION x PROTOCOL (F(14,182)=0.96, p=0.50, $\eta_p^2=0.07$) were not reliable.

				Prot	ocol	ANOVA	Independent t-test						
	Frequency	Control		The	eta	Be	ta1		Control	Control	Theta		
Source Active		М	SD	М	SD	М	SD	· ·	vs. Theta	vs. Beta1	vs. Beta1		
Active	Delta	0.957	0.03	0.968	0.03	0.997	0.02	5.38 *	0.87	3.17 *	2.34 †		
	Theta	0.959	0.03	0.968	0.01	0.988	0.03	2.97 †	0.75	2.36 †	1.66		
	Alpha1	0.973	0.02	0.989	0.02	0.992	0.02	2.25	1.67	1.99	0.33		
	Alpha2	1.017	0.04	1.008	0.02	1.007	0.02	0.28 0.61 0		0.69	0.08		
	Alpha3	1.032	0.05	1.014	0.02	1.016	0.04	0.55 0.97 0		0.85	0.12		
	SMR	0.998	0.03	1.003	0.02	1.018	0.04	1.01 0.37		1.36	1.02		
	Beta1	0.993	0.03	1.004	0.02	1.008	0.03	0.73	0.88 1.17		0.29		
	Beta2	1.000	0.05	1.005	0.03	1.007	0.03	0.06	0.25	0.34	0.09		
	Beta3	0.974	0.06	0.996	0.02	1.007	0.04	1.69	1.69 1.24		0.59		
	Beta4	0.979	0.04	0.996	0.02	1.009	0.04	1.74	1.07	1.86	0.81		
	Beta5	0.992	0.02	0.999	0.02	0.997	0.03	0.20	0.62	0.40	0.23		
	Gamma	0.983	0.03	1.006	0.03	1.002	0.02	2.25	2.01	1.63	0.40		
Passive	Delta	0.996	0.04	0.977	0.06	1.004	0.04	0.80	0.86	0.34	1.23		
	Theta	0.967	0.05	0.960	0.04	1.011	0.05	3.36 †	0.34	2.01 †	2.43 *		
	Alpha1	0.983	0.05	0.961	0.04	1.001	0.04	2.14	1.10	0.91	2.06 *		
	Alpha2	1.036	0.09	0.988	0.04	1.009	0.08	1.00	1.42	0.80	0.63		
	Alpha3	1.036	0.14	1.020	0.07	1.028	0.12	0.05	0.05 0.32 0.1		0.18		
	SMR	0.997	0.07	0.995	0.04	1.041	0.09	1.33	0.06	1.35	1.45		
	Beta1	1.001	0.05	1.001	0.04	1.011	0.06	0.16 0.03		0.49	0.48		
	Beta2	0.993	0.08	0.998	0.06	1.013	0.06	0.24 0.17		0.66	0.50		
	Beta3	0.960	0.10	0.975	0.06	1.019	0.06	1.62	0.41	1.71	1.33		
	Beta4	0.961	0.08	0.976	0.04	1.012	0.08	1.32 0.46		1.57	1.14		
	Beta5	0.989	0.03	0.980	0.04	0.992	0.06	0.20	0.47	0.11	0.60		
	Gamma	0.996	0.04	0.977	0.06	1.004	0.04	0.80	0.86	0.34	1.23		

Table 8.3. Means (and SD) of the mean within-session amplitude change (μ V) in active and passive periods for each frequency; statistical values of the between-subjects one-way ANOVAs and multiple comparisons by independent *t*-tests (Bonferroni corrected p-values) between control NF, theta NF and beta1 NF.

† = p < 0.10; * = p < 0.05; ** = p < 0.01



Figure 8.3. Across-session theta and beta1 amplitude change for control NF, theta NF and beta1 NF in passive periods. The graphs represent the amplitude change in eight training sessions (from S1 to S8) for theta (A) and beta1 frequencies (B). Amplitude change was normalized relative to the first passive period of the first session. Error bars depict standard error of the mean.

8.3.1. Frequency-specific learning

In order to investigate frequency-specific effects, the mean within-session amplitude change for eight frequencies ranging from delta to gamma was calculated for each NF protocol. Means (and SD) of within-session amplitude changes in the twelve frequencies in active feedback and passive resting state periods are presented in table 8.3. Frequency-specific effects were analyzed with two-way 12 FREQUENCY x 3 PROTOCOL mixed-ANOVAs for active and passive periods.

Hypothesis 5: Frequency-specificity of theta and beta1 amplitude changes in active periods.

Figure 8.4A presents the within-session amplitude changes in the twelve frequencies in active periods for each of the three NF protocols. The two-way mixed ANOVA revealed a significant main effect of FREQUENCY (F(11,286)=10.87, p<0.001, $\eta_p^2=0.30$) and a nearly significant interaction effect of FREQUENCY x PROTOCOL (F(22,286)=1.57, p=0.05, $\eta_p^2=0.11$). The effects of the three protocols on each of the twelve frequencies were explored by one-way ANOVAs and *t*-tests. As can be seen in table 8.3., apart from the previously described effects in the theta frequency (section 8.3.1.), also significant differences in withinsession amplitude were found between the NF protocols in the delta frequency (F(2,26)=5.38, p=0.01, $\eta_p^2=0.29$) but not in other frequencies (all ps>0.13). Post-hoc comparisons revealed that the within-session delta amplitude was significantly decreased in the control relative to beta1 NF (t(17)=3.17, p=0.01) and marginally decreased in the theta relative to theta NF (t(18)=2.34, p=0.08) but was not significantly different in the control relative to theta NF (t(17)=0.87, p=1.00). The analysis also indicated that the main effect of PROTOCOL was not reliable (F(2,26)=1.17, p=0.33, $\eta_p^2=0.08$).



Figure 8.4. Within-session amplitude change in different frequency bands for control NF, theta NF and beta1 NF protocols. The graphs indicate the mean within-session amplitude change from delta to gamma frequencies for active periods (normalized to the first active period of each sessions) (A) and passive periods (normalized to the first passive period of each session) (B). Error bars depict standard error of the mean. Statistical significance level is represented by asterisks (* p<0.05 and **p<0.01).

Hypothesis 6: Frequency-specificity of theta and beta1 amplitude changes in passive periods.

The mean amplitude changes in passive periods for the frequencies ranging from delta to gamma for each of NF protocols are shown in figure 8.4B. The two-way mixed ANOVA indicated significant main effect of FREQUENCY (F(11,286)=2.82, p<0.01, $\eta_p^2 = 0.10$). However, the post-hoc comparisons failed to provide evidence of reliable differences between the twelve frequencies (all ps>0.32). The analysis also revealed a non-FREQUENCY x reliable interaction effect of PROTOCOL (*F*(22,286)=0.83, $p=0.69 \eta_p^2=0.06$). As shown in table 8.3., despite this non-significant interaction, the post-hoc comparisons additionally revealed a within-session alpha1 amplitude decrement in the theta relative to beta1 NF (t(18)=2.06, p<0.05) while other pairwise comparisons were not reliable (all ps>0.28). The analysis further indicated a non-significant main effect of PROTOCOL $(F(2,26)=0.95, p=0.40, \eta_p^2=0.07).$

8.4. Discussion

A major goal of the current study was to investigate the training- and frequencyspecificity of theta suppression and beta1 enhancement using within- and across-session methods. The results partially supported the hypothesis that the three NF protocol would have a differential impact on within-session amplitude changes in both active and passive periods, but only in the theta frequency (Hypothesis 1 and 3). However, they failed to provide supportive evidence of across-session amplitude changes in theta and beta1 frequencies irrespective of whether active or passive periods were considered (Hypothesis 2 and 4). Moreover, the type of NF also modulated the within-session delta amplitude in active periods (Hypothesis 5) but no significant within-session amplitude changes in other frequencies were observed in the passive periods (Hypothesis 6).

8.4.1. Within-session training- and frequency-specificity

8.4.1.1. Training-specific within-session amplitude changes

The three NF protocols had a differential impact on within-session amplitude changes in theta but not in beta1 frequency. Additionally, in the theta frequency the results evinced a distinct pattern of within-session amplitude changes in active and passive periods.

Theta

Contrary to hypothesis, the analysis indicated of a trend towards within-session theta amplitude decrements in RF NF control condition, but not in theta NF, when compared to beta1 NF (Hypothesis 1). Firstly, it should be noticed that this difference did not reach statistical significance and, therefore, warrants replication in a larger sample before further conclusions can be drawn. Nevertheless, this result is surprising in light of the particular training regime and of the previous literature using similar RF NF as a control condition. In fact, because RF NF training did not emphasize the suppression of theta activity (or any of the adjacent delta and alpha1 frequencies) this result cannot be explained by the direct effect of the NF intervention. Moreover, this theta suppression effect following RF NF is in disagreement with one previous study (Doppelmayr & Weber, 2011). However, the two studies are difficult to compare because of differences in the learning measures (within vs. across-session) and in the training instructions of the RF condition (counterbalanced up- and down-regulation vs. up-regulation).

Alternatively, theta suppression following RF NF may be considered epiphenomenal to unspecific brain processes involved in NF such as time on task and cognitive effort. However, time on task is unlikely to explain this effect because theta amplitude tends to increase rather than decrease over time in both resting and vigilance task conditions (Brismar, 2007; Paus et al., 1997; Wascher et al., 2013). Assuming the involvement of more active processes, theta suppression might be explained by a decrement in cognitive effort relative to the first active period of each session (i.e., the normalization period) possibly related to the difficulty in maintaining a consistent self-regulatory strategy throughout the session. This explanation would be consistent with the modulation of theta amplitude by cognitive effort (Gevins, Smith, McEvoy, & Yu, 1997; Klimesch et al., 2001; Wascher et al., 2013) and with previous evidence of fm-theta amplitude increments when NF training promoted the active use of specific self-regulatory strategies and required higher cognitive effort relative to Sham NF (Enriquez-Geppert et al., 2014a, 2014b). In turn, the absence of variations in within-session theta amplitude in Sham NF (Enriquez-Geppert et al., 2014a, 2014b; Kober et al., 2015) might be explained by the lower cognitive effort involved in these conditions. Given the speculative nature of this explanation, more studies are needed to investigate whether the variations in within-session theta amplitude might reflect changes in cognitive effort. Future studies should consider ensuring the comparability of control (e.g., sham or RF) and other experimental NF conditions in terms of cognitive effort by monitoring subjective measures of effort and self-regulatory strategies.

The failure to provide evidence of training-specific within-session theta amplitude decreases in active periods (Hypothesis 1) in theta NF is in disagreement with previous results obtained during SMR/theta (de Zambotti et al., 2012; Gruzelier et al., 2014a; Ros et al., 2009; Vernon et al., 2003) and theta suppression NF (Beatty, Greenberg, Deibler, and O'Hanlon, 1974; O'Halon, Royal, & Beatty, 1977; Williams, Beatty, & O'Hanlon, 1975). Several aspects may have contributed to these different outcomes. Firstly, the control for training-specificity was different between studies. Contrary to previous studies (de Zambotti et al., 2012; Gruzelier et al., 2014a; Ros et al., 2012; Gruzelier et al., 2014a; Ros et al., 2009; Vernon et al., 2003), the current one also monitored within-session theta amplitude changes in other NF protocols. Thus,

differently from the current study, training-specificity could not be inferred from those previous ones. Moreover, in those studies it was unclear whether the within-session theta amplitude decrements were an epiphenomenon of increasing the synchronization in the higher frequencies (i.e., SMR and other beta frequencies) or a direct manifestation of an attempt to desynchronize the lower ones.

Also differently from the present study, in Beatty and colleagues investigations the evidence of training-specific theta suppression was obtained relative to a theta enhancement NF condition. In the latter, the theta amplitude was enhanced relative to the baseline and therefore this condition could not provide an unbiased measure of within-session theta amplitude changes (O'Halon, Royal, & Beatty, 1977; Williams, Beatty, & O'Hanlon, 1975). Moreover, the training-specific theta suppression effect was obtained during a single-session in which the prolonged time on task may have provided more opportunities to counteract the increasing theta activity over time. In the current experiment, it may have been more difficult to further decrease theta amplitudes because the NF sessions had a shorter duration. Moreover, the theta amplitude might have been already low because of its scarcity in healthy adults that have not been sleep deprived or subject to tiring and monotonous tasks.

According to hypothesis, in passive periods the theta amplitude was significantly decreased within sessions in theta NF relative to beta1 (Hypothesis 3). However, the absence of such decrements in active periods questions whether this result can be interpreted as the culmination of training. Moreover, the nearly significant decreases in passive periods in RF relative to beta1 NF and the absence of differences between theta and RF NF suggest that unspecific effects of NF may equally explain these effects. These results could be interpreted as an increment in theta amplitude, possibly reflecting increased mental effort in beta1 NF.

Beta1

In the beta1 frequency there was no evidence of training-specific effects in active periods. These results are in disagreement with previous observations of within-session beta1 amplitude increments following beta1 NF relative to control NF condition (Experiment 1, Hypothesis 1). Three methodological differences relative to that previous study might explain this disagreement. Firstly, in the current experiment the beta1 frequency was defined relative to IAF (IAF + 5 Hz to IAF + 8 Hz), while in the previous one it was defined

according to an arbitrary sub-band categorization (15-18 Hz). However, because the mean IAF was 9.5 Hz (see section 8.2.4) this is unlikely to have resulted in actual differences in beta1 frequency band definition between studies. Another methodological difference concerns the training regime in the control condition. In the previous study, RF NF showed a more pronounced within-session beta1 amplitude decrement than that observed in the current experiment. This might have contributed to the significant differences between RF and beta1 NF protocols in the former but not in the latter study. Finally, the different position of the active electrode (Cz) relative to the previous study (C3) might equally explain the difficulties in enhancing the beta1 amplitude in the present experiment. Further work is needed to investigate whether the success of the beta1 amplitude enhancement depends on such factors as the electrode location and the type of control condition.

There was also no evidence supporting the hypothesis of training-specific beta1 amplitude increments in passive periods (Hypothesis 3). These results are in line with those of Study 1 (Experiment 1, Hypothesis 3) and similarly suggest that the resting state beta activity rapidly returns to baseline levels after beta1 NF.

In sum, the absence of training-specific theta amplitude decrements and betal amplitude increments relative to control NF suggests that these changes were undistinguishable from unspecific NF effects such as time on task and cognitive effort. The lack of further within-session amplitude changes may be related to "floor and ceiling effects" in theta and beta frequencies respectively justified by the relative scarcity of theta and abundance of beta in healthy adult subjects during vigilance states (Amzica & Lopes da Silva, 2011). Future studies are needed to investigate whether the pre-training absolute and/or relative theta and beta1 amplitude may play a critical role in NF learning as suggested for the SMR frequency (Rutterford & Pacheco, 2011).

8.4.1.2. Frequency-specific within-session amplitude changes

In the present experiment the frequency-specificity of theta and beta1 NF was examined in the spectrum of EEG frequencies ranging from delta to gamma. In active periods, there was evidence of specific within-session delta amplitude decrements in the RF and theta NF relative to beta1 NF. This result is in disagreement with the hypothesis that the theta and beta1 NF protocols would be associated with frequency-specific effects in the theta and beta1 frequencies respectively (Hypothesis 5). Again, this is a surprising outcome given that none of these NF protocols attempted to manipulate delta amplitude. Although this might reflect a partial overlap between delta and theta frequencies, unspecific effects of NF may also contribute to delta amplitude changes. In this regard, delta oscillations have been related to motivational processes involved in the detection of salient internal and external signals (Lakatos et al., 2008; Knyazev, 2012) that might be crucially important to feedback learning. Interestingly, the modulation of delta amplitude as a function of the type of NF was only observed in active periods suggesting that it might reflect the involvement of an active learning process. Future studies are needed to replicate these results and investigate whether the within-session modulation of delta amplitude might reflect distinct motivation processes between NF protocols.

Contrary to observed in the active periods, a non-significant FREQUENCY x PROTOCOL interaction suggested that the within-session amplitude change from pre- to post-training in the different frequencies was not significantly modulated by the type of NF protocol. Nevertheless, the pairwise comparisons revealed within-session theta and alpha 1 amplitude decrements in the theta relative to beta1 NF. These effects are unlikely to represent the culmination of the training process because no such effects were observed in active periods. Alternatively, this result suggests a differential effect of the two NF protocols in counteracting the increased amplitude of slow frequencies usually associated with increased mental fatigue and lower arousal (Wascher et al., 2013).

8.4.1.3. Across-session training-specific effects

Another major goal of the present experiment was to investigate across-session theta and beta amplitude changes in both active and passive periods. Contrary to hypothesis (Hypothesis 2), there was no evidence of across-session beta1 amplitude increments despite the higher number of NF sessions and the individualization of the beta1 frequency relative to Study 1 (Experiment 1, Hypothesis 2). These results are in line with a previous failure to increase beta1 amplitude across sessions (Doppelmayr & Weber, 2011). Also contrary to hypothesis (Hypothesis 2), in the theta frequency the within-session amplitude was not significantly decreased across-sessions irrespective of the NF protocol. These results are in agreement with previous studies that failed to observe across-session theta amplitude decrements after SMR/theta NF and beta1/theta NF (de Zambotti et al., 2012; Doppelmayr & Weber, 2011; Kober et al., 2015). In passive periods, there was no evidence of training-specific across-sessions amplitude changes in theta and beta1 frequencies (Hypothesis 4).

Taken together, the current evidence suggests that amplitude changes in the theta and beta1 frequencies are perhaps better captured by within-session than by across-session learning indices. Speculatively, the modulation of these frequencies may critically depend on the purposeful engagement of particular self-regulatory strategies which, once developed, are unlikely to have a facilitative effect from one session to next. Another aspect that may have contributed to the failure to observe linear amplitude changes across-session in both frequencies was the inconsistency of the learning strategies over time. Thus, future studies would do well in monitoring the consistency of these strategies across training sessions.

8.4.4. Summary

In sum, the present study confirmed that within-session amplitude changes in active and passive periods provide potentially complementary information regarding the NF learning process. In fact, theta NF was associated with significant within-session theta amplitude decrements in the passive but not in the active periods. However, these decrements are difficult to attribute to the effects of feedback (which was only provided in active periods) and to time on task (which has been associated with increments in theta amplitude). Moreover, the observation of theta amplitude decrements in active and passive periods associated with RF NF suggests that these may reflect unspecific effects of NF. Interestingly, both RF and theta NF were associated with within-session delta amplitude decrements in active but not in passive periods suggesting that delta frequency may play a role during active feedback learning. In the beta frequency, there was no evidence of training-specific withinsession learning in active and passive periods. The present study also failed to evince acrosssession changes in both theta and beta1frequencies, suggesting that within- and acrosssession measures may provide non-overlapping information regarding the learning process.

Chapter 9

Experiment 5: The effects of theta and beta1 on selective attention

9.1. Hypotheses

Hypothesis 1: Increased P3a and P3b amplitude as a function of theta and beta1 NF

It was hypothesized that:

- (3) P3b amplitude would be increased in beta1 NF relative to theta and control NF conditions and in theta NF relative to control NF (beta1 > theta > control).
- (4) P3a amplitude would be increased in theta NF relative to beta1 NF and control NF conditions and in beta1 NF relative to control NF (theta > beta1 > control).

Hypothesis 2: Decreased RT and RT-SD as a function of theta and beta1 NF

It was hypothesized that:

- (3) RT would be decreased in beta1 NF relative to the theta and control NF and in theta NF relative to control NF (beta1 < theta < control).
- (4) RT-SD would be decreased in theta NF relative to control and beta1 NF and in beta1 NF relative to control NF (theta < beta1< control).</p>

Hypothesis 3: Increased performance rates and perceptual sensitivity as a function of theta and beta1 NF

It was hypothesized that:

- (4) *d*' would be increased in beta1NF relative to the theta and control NF and in theta NF relative to control NF (beta1 > theta > control).
- (5) Hit rate would be increased in beta1 NF relative to theta and control NF and in theta NF relative to control NF (beta1> theta > control).
- (6) False alarm rate would be decreased in theta NF relative to control and beta1 NF and in control NF relative to beta1 NF (theta < control < beta1).</p>

9.2. Methods

9.2.1. Participants

The same twenty-nine participants of Experiment 4 took part in the present study (see section 8.2.1. for details). After excluding trials with artifacts and averaging the ERP waveforms one participant from beta1 NF was excluded from the analysis for having less than thirty correct trials in at least one of the stimulus conditions at either pre- or posttraining. Table 9.1. presents demographic data for the remaining twenty-eight participants. The mean age and the mean IAF did not differ between NF protocols as confirmed by oneway ANOVAs (Age: F(2,25)=0.11, p=0.89, $\eta_p^2 < 0.01$; IAF: F(2,25)=0.95, p=0.40, $\eta_p^2=0.07$).

Table 9.1. Descriptive statistics of age (mean and SD), gender (frequency) and IAF (mean, SD and range) by NF protocol after excluding one participant with insufficient number of correct trials per stimulus condition.

	Ag	ge		Gender			IAF							
NF	М	SD	Male	Female	Total	Μ	SD	Range						
Control	21.44	4.03	4	5	9	9.22	0.67	8 -10						
Theta	22.60	6.31	4	6	10	9.80	0.92	9 -12						
Beta1	22.00	5.52	2	7	9	9.44	1.13	9 - 11						
Total	22.04	5.23	10	18	28	9.50	0.92	8 - 12						

9.2.2. Stimulus and procedure

The three-stimuli oddball task was presented to the subjects before and after the NF sessions in counterbalanced order with the cued Go/Nogo task (Experiment 6). The order of presentation had no effect in either the P3 amplitude for target, nontarget and standard stimuli or the behavioral measures of RT and accuracy.

Stimuli and procedures were identical to those of Experiment 2 of Study 1 (see section 4.2.2.).

9.2.3. EEG recording and analysis

EEG recording.

EEG recording was identical to Experiment 2 (see section 4.2.3.).

EEG data analysis.

EEG pre-processing was performed in EEGLAB (Delorme and Makeig, 2004). EEG data was re-referenced to the average of the Left and Right earlobes and high-pass filtered (0.05 Hz). EEG continuous recordings were visually inspected to remove extreme EEG artifacts prior to an infomax Independent Component Analysis was performed for correction of ocular artifacts. EEG data was then segmented relative to stimulus onset (1000ms prestimulus to 1000 ms post-stimulus) and trials exceeding 100µV were automatically identified and rejected on visual inspection.

EEG analysis was performed in Fieldtrip (Oostenveld, Fries, Maris, and Schoffelen, 2011). Only correct trials (correct response to target and correct non-response to standard and nontarget on the 0 to 1000ms interval post-stimulus onset) were analyzed and submitted to baseline correction (-200 to 0ms relative to stimulus onset). A minimum of 30 trials per condition were used to calculate individual ERP components. One-hundred random permutations were performed on thirty correct trials of each condition in order to ensure that the same number of trials per condition were used and that the individual waveforms resulted from a random selection of the number of available trials per condition. Hence, the individual mean P3 amplitudes represent the average amplitude of a hundred randomly generated waveforms each of which was based on an average of 30 artifact-free segments.

The target-standard and nontarget-standard difference waves and respective mean P3 amplitude were calculated for frontal, central and parietal regions according to the procedures described in section 4.2.3.

Percent of correct ERP trials.

After excluding segments with artifacts, the percentage of correct trials for each type of stimulus was calculated for each participant and averaged across groups. Descriptive statistics of the mean percentage of correct responses to target, nontarget and standard stimuli are presented in table 9.2. Pre-training differences between the three NF protocols were

investigated with separate one-way ANOVA for each type of stimulus. As shown in the table, the NF protocols differed significantly in the percentage of correct standard and nontarget stimuli (*ps*<0.05), but not in the percentage of correct target stimulus (*p*>0.24). Post-hoc comparisons revealed a marginal lower percentage of correct non-responses to standard stimuli in the beta1 relative to RF NF (t(27)=2.46; p=0.06) but not between beta1 and theta NF or between theta and RF NF (all *ps*>0.16). The percentage of correct non-responses to nontargets was significantly lower in the beta1 relative to RF NF (t(27)=2.64; *p*=0.04) while other pairwise comparison were non-significant (all *ps*>0.16).

Additionally, the differences in the percentage of correct trials from pre- to posttraining as a function of the NF were analyzed in separate two-way 2 TIME x 3 PROTOCOL mixed-ANOVAs for each type of stimulus. As can be seen in table, the non-significant TIME x PROTOCOL interactions showed that the NF protocols did not differ significantly in the percentage of correct responses from pre- to post-training (all ps>0.28). However, the significant main effects of TIME indicate that the percentage of correct standard and nontarget stimuli increased from pre- (Standards: M=84.67%, SD=11.36%; Nontargets: M=89.50%, SD=8.46%) to post-training (Standards: M=90.18%, SD=8.70%; Nontargets: M=94.79%, SD=7.59%). On the other hand, the percentage of correct target stimulus was not significantly changed as a function of the moment of assessment. Moreover, a significant main effect of PROTOCOL indicated that in beta1 NF the overall percentage of correct standard stimulus was lower relative to control NF (t(16)=2.71; p=0.04) and marginally lower relative to theta NF (t(17)=2.53; p=0.06). Despite these pre- to post-training and NF protocol differences in the percentage of correct trials, the method of calculating the individual ERP waveforms ensured that the same number of correct trials was used at preand post-training in the three NF protocols (see previous section).

9.2.4. Behavioral data analysis

Reaction time data preparation

RT and RT-SD analysis were performed on the same correct target trials used for ERP analysis after the exclusion of artifacts.

Performance rates and d-prime data preparation

Performance rates (hit rate to targets, false alarm to standards and false alarm to nontargets) were calculated for all the available trials before artifact rejection. The procedure for the calculation of d-prime scores was as described in section 4.2.4.

9.2.5. Statistical analysis

Specifications of statistical packages and criteria were provided in section 3.2.6.

9.3. Results

9.3.1. Hypothesis 1: Increased P3a and P3b amplitude as a function of theta and beta1 NF

9.3.1.1. Pre-training differences between NF protocols

Figure 9.1. shows the grand average of the stimulus-locked ERP elicited by the targetstandard and nontarget-standard difference waves in frontal, central and parietal regions at pre-training irrespective of NF protocol. The figure shows a large positive deflection with maximal peak amplitude around 300ms in parietal electrodes for target-standard (P3b) and a smaller positive deflection around 300ms with maximal peak in central electrodes after nontarget-standard (P3a).



Figure 9.1. *Left.* Grand average of the stimulus-locked ERP of target-standard and nontarget-standard difference waves in midline frontal, central and parietal regions at pre-training irrespective of the NF protocol. The shaded area represents the 250-400ms measurement window. Positive amplitude is displayed upwards. *Right.* Distribution of the target-standard and nontarget-standard difference waves.

A three-way 2 STIMULUS (target-standard vs. nontarget-standard) x 3 REGION (frontal vs. central vs. parietal) x 3 PROTOCOL (Control vs. Theta vs. Beta1) mixed-design ANOVA for the mean P3 amplitude at pre-training yielded significant main effects of STIMULUS (F(1,25)=36.56, p<0.01, $\eta_p^2=0.59$) and REGION (F(1.39,34.89)=3.99, p=0.04, $\eta_p^2=0.14$) as well as a significant interaction of STIMULUS x REGION (F(1.29,32.11)=10.61, p<0.01, $\eta_p^2=0.29$). The main effect of PROTOCOL (F(2,25)=0.63, p=0.54, $\eta_p^2=0.05$) and the interactions involving the between-subject variable were non-significant (STIMULUS x PROTOCOL: F(2,25)=0.05, p=0.95, $\eta_p^2<0.01$; REGION x PROTOCOL: F(2.79,34.89)=0.59, p=0.61, $\eta_p^2=0.05$; STIMULUS x REGION x PROTOCOL: F(2.57,32.11)=1.73, p=0.18, $\eta_p^2=0.12$).

As shown in fig. 9.2., the post-hoc comparisons (Bonferroni corrected *p*-values) for the target-standard revealed that the P3 amplitude was significantly higher in the parietal (M=12.38, SD=4.42) relative to the frontal region (M=10.05, SD=4.97; t(27)=3.22, p=0.01) and marginally higher in the parietal relative to central region (M=11.26, SD=5.09; t(27)=2.54, p=0.05). The target-standard P3 amplitude was marginally higher in the central region relative to the frontal region (t(27)=2.52, p=0.06). However, for the nontargetstandard the P3 amplitude was not significantly different between parietal (M=6.42, SD=3.27) and central regions (M=6.66, SD=3.77; t(27)=0.35, p=1.00), between parietal and frontal regions (M=6.22, SD=3.56; t(27)=0.32, p=1.00) and between central and frontal regions (t(27)=1.27, p=0.64). Finally, the post-hoc comparisons revealed that the P3 amplitude was significantly higher for the target-standard difference waves relative to the nontarget-standard difference waves in the three regions (Frontal: t(27)=4.75, p<0.001; Central: t(27)=5.43, p<0.001; Parietal: t(27)=6.85, p<0.001).



Figure 9.2. Two-way interaction of Stimulus and Region in mean P3 amplitude at pre-training. The graph shows the mean P3 amplitude (μ V) for Target-Standard and Nontarget-Standard stimulus in three Regions (Frontal, Central and Parietal) irrespective of NF protocol. For Target-Standard the P3 amplitude was larger at Parietal electrodes, while for Nontarget-Standard the P3 amplitude did not differ significantly between regions. Error bars depict standard error of the mean.

Table 9.2.

Means (standard deviations) and statistical results of the pre-training one-way ANOVAs and the pre- to post-training two-way ANOVAs on Target-Standard (T-S) and Nontarget-Standard (N-S) P3 amplitude, Hit Rate (H), False Alarm Rate (FA) to Standards (S) and Nontargets (N), d-prime (d') to Target-Standard (T-S) and Target-Nontarget (T-N), Reaction Time (RT) Mean and Reaction Time Standard Deviation (RT SD), percentage of Correct Trials (CT) to Targets (T), Standards (S) and Nontargets (N) included in ERP analyses after artifact rejection for each of the three NF protocols. Significant differences (p<0.05) are highlighed in bold.

							One-	way AN	OVA				Two-	way Al	NOVA			
-		Pre-training			Post-training			Protocol			Time			Time x Protocol			Protocol	
Source	Control	Theta	Beta1	Control	Theta	Beta1	F	Р	η_p^2	F	Р	η_p^2	F	Р	η_p^2	F	Р	η_p^2
Mean P3																		
T-S	13.92 (4.65)	11.32 (5.43)	11.91 (2.48)	12.74 (3.42)	12.25 (9.06)	12.50 (6.09)	_	_	_	0.02	0.89	0.00	0.59	0.56	0.04	0.22	0.81	0.02
N-S	7.45 (3.89)	5.42 (4.11)	7.12 (3.19)	6.32 (2.14)	6.04 (4.16)	7.56 (4.45)	-	-	-	0.00	0.98	0.00	0.62	0.54	0.05	0.62	0.55	0.05
H (%)	97.09 (3.54)	100.00 (0.00)	99.09 (1.50)	98.61 (1.57)	99.06 (2.03)	98.75 (2.06)	4.40	0.02	0.26	0.04	0.85	0.00	2.87	0.08	0.19	2.39	0.11	0.16
FA (%)																		
S	0.14 (0.18)	0.11 (0.17)	0.18 (0.21)	0.14 (0.13)	0.80 (0.16)	0.07 (0.11)	0.32	0.73	0.02	2.24	0.15	0.08	0.94	0.40	0.07	0.27	0.77	0.02
N	1.78 (2.11)	0.90 (0.99)	1.44 (1.33)	0.67 (0.16)	0.80 (1.03)	0.89 (1.45)	0.80	0.46	0.06	2.88	0.10	0.10	0.72	0.50	0.05	0.38	0.69	0.03
d'																		
T-S	4.87 (0.49)	5.26 (0.16)	4.96 (0.40)	5.05 (0.27)	5.20 (0.28)	5.15 (0.33)	2.92	0.07	0.19	1.94	0.18	0.07	1.35	0.28	0.10	2.39	0.11	0.16
T-N	4.04 (0.48)	4.54 (0.14)	4.22 (0.33)	4.38 (0.32)	4.44 (0.25)	4.38 (0.36)	5.33	0.01	0.30	3.11	0.09	0.11	3.22	0.06	0.20	2.85	0.08	0.19
RT (ms)																		
Mean	566.30 (166.28)	514.22 (99.57)	545.11 (133.54)	591.92 (158.24)	491.65 (114.26)	606.28 (124.15)	0.36	0.70	0.03	2.20	0.15	0.08	2.90	0.07	0.19	1.09	0.35	0.08
RT SD	105.86 (26.78)	105.76 (20.78)	101.71 (13.54)	105.30 (27.62)	91.27 (29.23)	102.85 (30.95)	0.12	0.89	0.01	1.03	0.32	0.04	1.22	0.31	0.09	0.24	0.79	0.02
CT (%)																		
т	88.00 (8.94)	91.60 (4.20)	84.00 (13.64)	88.00 (9.74)	88.44 (8.99)	94.00 (9.71)	1.50	0.24	0.11	0.43	0.52	0.02	0.08	0.92	0.01	2.37	0.11	0.16
S	89.42 (5.76)	87.05 (9.64)	77.31 (14.29)	92.58 (5.46)	93.08 (8.11)	84.56 (9.96)	3.42	0.05	0.21	8.30	0.01	0.25	0.39	0.68	0.03	4.56	0.02	0.27
N	93.33 (3.00)	91.20 (6.27)	83.78 (11.46)	96.22 (2.73)	95.60 (8.32)	92.44 (9.99)	3.86	0.03	0.24	12.87	0.00	0.34	1.32	0.28	0.10	2.56	0.10	0.17

The pre- to post-training effects of the three NF protocols were analyzed in the region where the P3a and P3b components reached the maximal amplitude. This corresponded to the average of frontal, central and parietal regions for nontarget-standard (P3a) and to the parietal region for target-standard (P3b). The means (and standard deviations) of the P3 amplitude of target-standard and nontarget-standard differences waves in the three NF protocols at preand post-training are shown in table 9.2. These effects were investigated in separate two-way mixed-ANOVA with 2 TIME (pre- vs. post-training) x 3 PROTOCOL (Control vs. Theta vs. Beta1) for target-standard and nontarget-standard.

Figures 9.3. and 9.4. present the target-standard and nontarget-standard difference waveforms and the mean P3 amplitude for each of the three NF protocols at pre- and post-training respectively. As suggested by the figures, the statistical analysis did not support the hypothesis of a differential impact of the three NF protocols on the neural processing of targets (TIME x PROTOCOL: F(2,25)=0.59, p=0.56, $\eta_p^2=0.04$). As shown in the table 9.2., the main effects of TIME and PROTOCOL for target-standard difference waves were also not reliable (*ps*>0.81). Moreover, the post-hoc comparisons failed to provide evidence of pre-to post-training target-standard amplitude changes in any of the three NF protocols (all *ps*>0.44).

Similarly, the type of NF did not significantly modulate the nontarget-standard P3 amplitude from pre- to post-training (TIME x PROTOCOL: F(2,25)=0.62, p=0.54, $\eta_p^2=0.05$). Again the main effects of TIME and PROTOCOL were not reliable for the nontarget-standard difference waves (*ps*>0.55). The post-hoc comparisons failed to provide evidence of pre- to post-training nontarget-standard amplitude changes in any of the three NF protocols (all *ps*>0.37).


Fig. 9.3. Grand-average of the parietal target-standard difference and average nontarget-standard difference ERP waveforms for control, theta and beta1 NF at pre- and post-training. The shaded area represents the 250-400ms measurement window. Positive amplitude is displayed upwards.



Figure 9.4. Mean P3 amplitude of the target-standard and nontarget-standard difference waveforms for control, theta and beta1 NF at pre- and post-training. Both target-standard **(A)** and nontarget-standard P3 amplitudes **(B)** were not significantly modulated by the type NF protocol. Error bars depict standard error of the mean.

9.3.2. Hypothesis 2: Decreased RT and RT-SD as a function of theta and beta1 NF

Table 9.2. summarizes the data and statistical results of RT and RT-SD at pre- and post-training for the three NF protocols. As can be observed in table, two one-way ANOVAs revealed that there were no initial differences between NF protocols in mean RT and RT-SD (all *ps*>0.70). The differential impact of the three NF protocols in mean RT and RT-SD were analyzed in separate two-way 2 TIME (Pre-training vs. Post-training) x 3 PROTOCOL (Control vs. Theta vs. Beta1) mixed-ANOVAs.

9.3.2.1. Mean reaction time

As shown in fig. 9.5A, the mean RT was decreased in theta NF and increased in control and beta1 NF from pre- to post-training. As shown in table 9.2., there was a marginally significant TIME x PROTOCOL interaction (F(2,25)=2.90, p=0.07, $\eta_p^2=0.19$). Bonferroni corrected post-hoc comparisons indicated that the mean RT was significantly increased in beta1 NF (t(8)=2.41, p=0.02), while the mean RT changes in control and theta NF were not reliable (ps>0.32). As also shown in the table, the main effects of TIME and PROTOCOL were not reliable (ps>0.15).



Figure 9.5. Two-way interaction of 2 Time x 3 Protocol for reaction time. The graphs show the effects at preand post-training of the three NF protocols on mean reaction time (RT) and reaction time standard deviation (RT-SD). (**A**) The significant Time x Protocol interaction revealed an increase in mean RT in beta1 NF. (**B**) RT-SD was significantly decreased in theta NF. However, the Time x Protocol interaction was non-significant. Error bars depict standard error of the mean

9.3.2.2. Reaction time standard deviation

Figure 9.5.B shows that the RT-SD was decreased from pre- to post-training in theta NF but not in the other NF protocols. However, as shown in table 9.2., the two-way mixed-ANOVA did not reveal significant main effects or interactions (*ps*>0.31). Despite the absence of a significant TIME x PROTOCOL interaction, the post-hoc comparisons revealed a marginally decreased RT-SD in theta NF (t(9)=1.89, p=0.07) but no significant changes in the other NF protocols (*ps*>0.89).

9.3.3. Hypothesis 3: Increased performance rates and perceptual sensitivity as a function of theta and beta1 NF

Table 9.2. summarizes the data and statistical results for target-standard d', targetnontarget d', target hit rate, standard false alarm and nontarget false alarm at pre- and posttraining for the three NF protocols. As can be observed in the table, at pre-training separate one-way ANOVAs for each variable revealed significant differences between NF protocols in hit rate and d' between nontargets and standards and a marginal significant difference in d'between targets and standards. As shown in figures 9.6A, 9.6B and 9.7A, the d' between targets and standards (T-S), the d' between targets and nontargets (T-N) and the hit rate (H) were marginally or significantly lower in the control relative to theta NF (d' T-S: t(17)=2.28, p=0.09; d' T-N: t(17)=3.21, p=0.01; H: t(17)=2.90, p=0.02) but not significantly different between control and beta1 NF or between theta and beta1 NF (d' T-S: ps>0.26; d' T-N: ps>0.15; H: ps>0.18). No initial differences in standard and nontarget false alarm rates were observed between NF protocols (ps>0.46). The differential impact of the three NF protocols in these variables were analyzed in separate two-way 2 TIME x 3 PROTOCOL mixed-ANOVAs.

9.3.3.1. Target-standard d'

Figure 9.6A shows the effects of the three NF protocols on target-standard d' from pre-to post-training. As shown in table 9.2., the two-way TIME x PROTOCOL mixed-ANOVA did not yield any significant main effects or interactions (all *ps*>0.11). Post-hoc comparisons only indicated non-significant pre- to post-training target-standard d' differences in the three NF protocols (all *ps*>0.14).

9.3.3.2. Target-nontarget d'

As shown in figure 9.6B, from pre-to post-training the target-nontarget d' increased in control and beta1 NF and was slightly decreased in theta NF. The two-way ANOVA confirmed a marginal significant TIME x PROTOCOL interaction (F(2,25)=3.22, p=0.06, $\eta_p^2=0.20$). Post-hoc comparisons confirmed a significantly increased target-nontarget d' in control NF protocol (t(8)=2.63, p=0.01) and non-significant pre- to post-training differences in the other two NF protocols (ps>0.24). As shown in table 9.2., the main effects of TIME and PROTOCOL were also marginally significant.



Figure 9.6. Two-way interactions of 2 Time x 3 Protocol for d-prime (d'). The graphs show the d' targetstandard (**A**) and d' target-nontarget (**B**) at pre- and post-training for the three NF protocols. At pre-training, in control NF the d' target-standard was marginally lower and the d' target-nontarget was significantly lower relative to theta NF. The d' target-nontarget was significantly increased from pre- to post-training In control NF, but was not significantly changed in the other NF protocols. Error bars depict standard error of the mean.

9.3.3.3. Target hit rate

As shown in figure 9.7A, the mean hit rate increased in control NF and decreased in theta and beta1 NF from pre- to post-training. As shown in table 9.2., the statistical analysis revealed a marginal significant TIME x PROTOCOL interaction (F(2,25)=2.87, p=0.08, $\eta_p^2=0.19$). Post-hoc comparisons confirmed a marginally increased hit rate in control NF protocol (t(8)=1.88, p=0.06) and non-significant pre- to post-training differences in the other

two NF protocols (ps>0.21). As shown in table 9.2., the main effects of TIME and PROTOCOL were not reliable (ps>0.11).



Figure 9.7. Two-way interaction of 2 Time x 3 Protocol for performance rates. The graphs show the percentage of hit rate for targets (**A**), the percentage of false alarm rate for standards (**B**) and the percentage of false alarm rate for nontargets (**C**) at pre- and post-training for the three NF protocols. In control NF the hit rate was significantly decreased at pre-training and marginally increased from pre- to-post-training (**A**). The false alarm to standards decreased from pre- to post-training in beta1 NF (**B**) and the false alarm to nontargets decreased from pre- to post-training in control NF (**C**). Error bars depict standard error of the mean.

9.3.3.4. False alarm rate for standards

As shown in figure 9.7B, the false alarm rate for standards was decreased from pre-to post-training in theta and beta1 NF. As shown in table 9.2., the two-way TIME x

PROTOCOL mixed-ANOVA revealed no reliable main effects or interactions (all *ps*>0.10). Despite the absence of a significant TIME x PROTOCOL interaction, post-hoc comparisons revealed a marginally decreased false alarm rate in beta1 NF (t(8)=1.91, p=0.07) and non-significant changes from pre- to post-training in the control and theta NF (ps>0.50).

9.3.3.5. False alarm rate for nontargets

As shown in figure 9.7C, the false alarm rate for nontargets was decreased from preto post-training in theta and beta1 NF. As shown in table 9.2., the two-way TIME x PROTOCOL mixed-ANOVA revealed no reliable main effects or interactions (all *ps*>0.15). Despite the absence of a significant TIME x PROTOCOL interaction, post-hoc comparisons revealed a marginally decreased false alarm rate in control NF (t(8)=1.83, p=0.08) and nonsignificant changes from pre- to post-training in the theta and beta1 NF (ps>0.37).

9.4. Discussion

The ERP analysis did not support the hypotheses that the three NF protocols would be associated with differential neural response to infrequent nontarget (P3a) and target (P3b) stimuli (Hypothesis 1.1 and 1.2). However, the behavioral analyses suggested that the type of NF had differential impact on measures of selective attention and response inhibition. Contrary to hypothesis, the mean RT was increased in beta1 NF rather than decreased (Hypothesis 2.1). Moreover, the increments in target-nontarget *d*' and hit rate were observed in control NF condition which are likely to represent a rectification of lower performance at pre-training rather than a real performance advantage of RF NF (Hypothesis 3.1 and 3.2). Moreover, the current experiment indicated a differential impact of the type of NF protocol on false alarm rate. Also contrary to hypothesis (Hypothesis 3.3), beta1 and control NF but not theta NF were associated with marginal decreases in false alarm rate for standards and nontargets respectively.

The next sections will examine several methodological differences relative to previous studies that might have been related to the failure to replicate differential electrophysiological and behavioral effects of theta and beta1 NF on selective attention and response inhibition in the context of a three-stimuli auditory oddball.

9.4.1. Electrophysiological and behavioral effects of beta1 NF

ERP components

The present experiment investigated the hypothesis that beta1 NF would be associated with a superior increment in the P3 amplitude to infrequent target (P3b) stimuli relative to the other NF protocols (Hypothesis 1.1). In agreement with the results of Experiment 2, the current ones failed to replicate previous findings that suggested an advantage of beta1 NF in the modulation of the P3b amplitude (Egner & Gruzelier, 2004). In Experiment 2, it was proposed that the failure to replicate this effect could have been related to the lower number of NF sessions relative to that previous study (six vs. ten). However, the increment in the number of NF sessions, numbering eight in the present experiment, might have not been enough to significantly impact the neural generators of the P3b. Alternatively, the failure to observe specific NF effects in the P3b amplitude might be related to the absence of significant increments in beta1 activity in the present study (for a further discussion see section 11.2.1.). Additionally, the present experiment failed to support the hypothesis that the P3b amplitude increment effect would be better explained by theta suppression rather than by the enhancement of beta1 activity.

Behavioral responses

Furthermore, it was hypothesized that the higher efficiency of task-relevant stimulus processing following betal NF would be reflected by increments in response speed (Hypothesis 2.1), *d'* (Hypothesis 3.1) and hit rate (Hypothesis 3.2) from pre- to post-training. Contrary to prediction, betal NF was associated with increased mean RT (Hypothesis 2.1) relative to the control and theta NF protocols. This deterioration of processing efficiency in betal NF is in disagreement with previous studies indicating a decrease in mean RT after beta1/theta NF (Bakhshayesh et al., 2011; Egner & Gruzelier, 2004) and with the absence of significant changes in RT following both SMR and beta1 NF in Experiment 2. However, a nearly significant interaction effect warrants caution in the interpretation of these findings and replication in a large sample. Future studies may attempt to clarify whether the increased RT may be explained by increased motor inhibition in agreement with the proposed role of

the beta oscillation in maintaining the current motor/cognitive set in a low Go response probability context (Engel & Fries, 2010) and in promoting the inhibition of the motor system (Ritter, Moosmann, & Villringer, 2009; Picazio et al., 2014). Also it would be interesting to investigate whether this increased RT effect would be specific to infrequent responding contexts, characterized by prolonged inactivity which may contribute to decrease the motor cortex excitability.

Although the specific mechanisms by which beta1 NF may impact response speed are still unclear it is worth noting that a delay in the P3b peak latency could be detected at posttraining in the beta NF (see fig. 9.3.). The P3 latency has been taken a measure of stimulus evaluation and/or response selection time with some studies suggesting that it constitutes an independent processes (e.g., Kutas, McCarthy, & Donchin, 1977; Magliero, Bashore, Coles, & Donchin, 1984) and other showing a positive correlation with RT (e.g., Makeig et al., 2004). In order to clarify whether the increased mean RT may be explained by a delay in stimulus evaluation time or in response selection further post-hoc analyses are needed to investigate the effects of the three NF on P3b latency. An increase in the P3b latency would suggest that the participants in beta1 NF group required additional stimulus evaluation and/or response selection time. Alternatively, if beta1 NF affects mainly the post-decisional processes an increase in mean RT without changing the P3 latency would be expected. Future studies should also take into consideration that the P3b latency may be described in terms of one or more underlying components: a first component may reflect stimulus evaluation, while a second one may be related to response selection time (Verleger, 1997; Dien et al., 2004; Verleger et al., 2005, 2015).

Further, the hypothesis that theta suppression would explain the decreased mean RT in low-beta/theta NF was also not confirmed by the results. Thus, the NF conditions that may have been related to the decreases in mean RT in previous beta1/theta NF studies remain unclear. A possible explanation is that the decreased mean RT observed in previous studies may have resulted from the concertation of theta and high beta inhibition with beta1 enhancement emphasized in those NF protocols (e.g., Egner & Gruzelier, 2004). While the specific involvement of high beta inhibition in the cognitive enhancement effects remains unexplored, the present study suggests that, contrary to theta suppression, beta1 NF might be counterproductive in terms of increasing the processing speed of task-relevant information.

Also contrary to hypothesis, the present experiment failed to provide supportive evidence of increased target-standard d' and target-nontarget d' (Hypothesis 3.1) and hit rate (Hypothesis 3.2) in the beta1 relative to the other NF protocols. This is in agreement with previous studies that showed that beta1/theta NF was not associated with higher performance accuracy (Bakhshayesh et al., 2011; Egner & Gruzelier, 2004). The "ceiling effects" in performance accuracy observed in theta and beta1 NF may have prevented further increases in target detection and perceptual sensitivity.

Interestingly, the target-nontarget *d'* was significantly increased from pre- to posttraining in control NF relative to other NF protocols and nearly significant increases in hit rate and decreases in false alarm for nontargets. These differences may also be explained by an initial lack of compliance with the instructions which might have determined the significantly lower pre-training scores in these measures observed in control NF. Alternatively, these improvements may be related to the more efficient detection of taskrelevant stimulus and inhibition of distractors in control NF. Taking into consideration that RF NF was focused on the regulation of alpha and high beta, the increased performance accuracy in that condition would be consistent with the role of alpha (Dockree et al., 2007; Klimesch, 2012; Gruber et al., 2014) and beta (Wróbel, 2000; Bekisz and Wróbel, 2003; Lee et al., 2013; Sacchet et al., 2015) frequencies on attention processes. However, as shown in Experiment 4, there was no evidence of an advantage of control NF in the modulation of alpha and beta frequencies. Instead, the within-session delta and theta amplitude decrements may explain the general improvements in cognitive performance (Klimesch, 1999; Lakatos et al., 2008; De Blasio & Barry, 2013) observed in control NF condition.

Finally, in disagreement with a previous proposal that the beta1/theta NF would be associated with excessive arousal state determining a faster yet error prone behavior (Egner & Gruzelier, 2004), beta1 NF was not associated with an increase in the false alarm rate. On the contrary, despite the absence of a significant TIME x PROTOCOL interaction there was evidence that beta1 NF was associated with a decrease in false alarm rate from pre- to post-training (Hypothesis 3.3.). Future studies are needed to clarify the specific NF conditions that might have been related to improvements in response inhibition in the three-stimuli oddball task.

9.4.2. Electrophysiological and behavioral effects of theta NF

ERP components

The present experiment investigated the possibility that theta NF would be associated with a superior P3 amplitude increment to infrequent nontarget (P3a) stimuli relative to the other two NF protocols (Hypothesis 1.2). The failure to observe this effect was surprising in light of previous demonstrations of the involvement of evoked theta oscillatory activity in the generation of the P3 component (Demiralp, Ademoglu, Comerchero, & Polich, 2001; Yordanova et al., 2002, 2003). Thus, whether the suppression of theta activity may have been related to previous improvements in P3 amplitude (Egner & Gruzelier, 2004) remains unclear.

Behavioral responses

In agreement with the proposed association of low tonic theta amplitude with improved cognitive control, the present experiment suggested a decreased RT-SD after theta NF (Hypothesis 2.2). The present results suggest that theta NF may have promoted a reduction in the moment-to-moment fluctuations in behavioral performance. However, the absence of a significant TIME x PROTOCOL interaction warrants a cautious interpretation and replication of these findings.

In disagreement with the proposed relation between theta suppression and increased performance monitoring, there was no evidence of decreased false alarm rate in theta NF (Hypothesis 3.3.). The analyses revealed that neither the false alarm for standards nor the false alarm for nontargets were significantly decreased from pre- to post-training as a function of the type of NF. Again, the nearly flawless performance in theta NF at pre-training may have determined a "floor effect" that prevented further decrements in false alarm rates at post-training.

In sum, despite the involvement of theta and beta oscillations in goal-directed selective attention (Voytek et al., 2015; Womelsdorf & Everling, 2015) there is currently no evidence that the modulation of their amplitude by NF may impact fundamental neural processes associated with selective processing of task-relevant information. Future studies are needed to explore the hypothesis that an increment of the top-down control of stimulus selection may be related to the facilitation of the long-range coherent activity of theta and

beta frequencies (Womelsdorf & Everling, 2015) rather than with power modulations of these frequencies in local neural populations.

9.4.3. Summary

The current experiment provided inconclusive results regarding the differential impact of theta and beta1 NF on selective attention and response inhibition. The increased RT and decreased false alarm rate suggest that beta1 NF might be related to improvements in response inhibition. Additionally, the decreased RT-SD in theta NF suggests a diminished performance variability which may be explained by an increase in cognitive control. However, the absence of statistically significant TIME x PROTOCOL interactions recommends the replication of these results in a larger sample before any conclusions can be reached regarding the advantage of a specific NF for performance improvements.

Chapter 10

Experiment 6: The effects of theta and beta1 on response inhibition

10.1. Hypotheses

Hypothesis 1: Increased Go and Nogo-P3 amplitude as a function of theta and beta1 NF

It was hypothesized that:

- (3) Nogo-P3 amplitude would be increased in theta NF relative to beta1 and control NF and in beta1 NF relative to control NF (theta > beta1 > control).
- (4) Go-P3 amplitude would be increased in beta1 NF relative to theta and control NF conditions and in theta NF relative to control NF condition (beta1 > theta > control).

Hypothesis 2: Decreased RT and RT-SD as a function of theta and beta1 NF

It was hypothesized that:

- (5) RT would be decreased in beta1 NF relative to the theta and control NF and in theta NF relative to control NF (beta1 < theta < control).
- (6) RT-SD would be decreased in theta NF relative to control and beta1 NF and in beta1 NF relative to control NF (theta < beta1< control).</p>

Hypothesis 3: Increased performance rates and perceptual sensitivity as a function of theta and beta1 NF

It was hypothesized that:

- (7) d' would be increased in beta1NF relative to the theta and control NF and in theta NF relative to control NF (beta1 > theta > control).
- (8) Hit rate would be increased in beta1 NF relative to theta and control NF and in theta NF relative to control NF (beta1> theta > control).

(9) False alarm rate would be decreased in theta NF relative to control and beta1 NF and in control NF relative to beta1 NF (theta < control < beta1).</p>

10.2. Methods

10.2.1. Participants

The same twenty-nine participants of Experiment 4 took part in the present study (see Methods Experiment 4 for details). One participant was excluded from the analysis for having less than twenty correct trials in at least one of the four stimulus conditions (Left-cue Go, Left-cue Nogo, Right-cue Go, Right-cue Nogo) at either pre- or post-training. The same participants were included in experiments 5 and 6 (demographic data is presented in Table 9.1).

10.2.2. Stimulus and procedure

The stimuli and procedure were identical to those of Experiment 3 (see section 5.2.2.).

10.2.3. EEG recording and analysis

EEG recording.

EEG recording specifications were identical to Experiment 2 (see section 4.2.3).

EEG data analysis.

EEG data analysis and pre-processing were identical to Experiment 3 (see section 5.2.3.).

Percent of correct ERP trials.

Preliminary analyses with one-way ANOVA at pre-training failed to reveal any significant influence of the type of cue (right vs. left) in the percent of correct trials for both Go and Nogo stimuli (Go: F(1,27)=0.63, p=0.43, $\eta_p^2=0.02$; Nogo: F(1,27)=0.15, p=0.70, $\eta_p^2<0.01$). Subsequent analysis were performed on collapsed left- and right-hand primed Go and Nogo trials. Descriptive statistics of percent of rejected trials in the Go/Nogo task are

presented in table 10.1. As shown in the table, separate one-way ANOVA at pre-training failed to reveal any significant differences between NF protocols for the percent of correct Go and Nogo trials (*ps*>0.96). However, there was a significant increase in the percentage of correct Go trials from pre- (M=80.69%, SD=14.26%) to post-training (M=88.25%, SD=12.53%; *F*(1,25)=7.94, *p*<0.01, η_p^2 = 0.24) and a marginal increase in the percentage of correct Nogo trials from pre- (M=81.38%, SD=13.06%) to post-training (M=86.43%, SD=12.53%; *F*(1,25)=3.05, *p*=0.09, η_p^2 = 0.11). All other main and interaction effects were non-reliable (all *ps*>0.27).

10.2.4. Behavioral data recording and analysis

Reaction time data preparation

Reaction time data preparation was identical to Experiment 3 (see section 5.2.4). The rejection rate of Go trials with response latencies below 100 ms or above 1200ms was 0.05%.

d-prime data preparation

d-prime data preparation was identical to Experiment 3 (see section 5.2.4).

10.2.5. Statistical analysis

Statistical analyses was identical to Experiment 3 (see section 5.2.5).

10.3. Results

10.3.1. Hypothesis 1: Increased Go and Nogo-P3 amplitude as a function of theta and beta1 NF

In the present experiment the hypotheses relative to the differential impact of the three NF protocols in the Nogo-P3 and Go-P3 amplitude (Hypotheses 1.1 and 1.2) were investigated for original waveforms at pre-training (section 10.3.1.1) and from pre- to post-training (section 10.3.1.2).

10.3.1.1. Pre-training differences between NF protocols

Figure 10.1 shows the grand average of the stimulus-locked ERP elicited by the frequent Go and infrequent Nogo stimuli in frontal, central and parietal regions at pre-training irrespective of NF protocol.

The pre-training differences between the NF protocols in the mean P3 amplitude of three scalp regions were investigated in a three-way 2 STIMULUS (Go vs. Nogo) x 3 REGION (Frontal vs. Central vs. Parietal) x 3 PROTOCOL (Control vs. Theta vs. Beta1) mixed-design ANOVA. This analysis yielded significant main effects of STIMULUS (F(1,25)=47.76, p<0.01, $\eta_p^2=0.65$) and REGION (F(2,50)=36.73, p<0.01, $\eta_p^2=0.59$) as well as a significant interaction of STIMULUS x REGION (F(1.45,35.37)=38.29, p<0.01, $\eta_p^2=0.61$). The main effect of PROTOCOL (F(2,25)=0.66, p=0.53, $\eta_p^2=0.05$) and the interactions involving the between-subject variable were non-significant (STIMULUS x PROTOCOL: F(2,25)=1.38, p=0.27, $\eta_p^2=0.10$; REGION x PROTOCOL: F(4,50)=1.10, p=0.37, $\eta_p^2=0.08$; STIMULUS x REGION x PROTOCOL: F(2.83,35.37)=1.60, p=0.21, $\eta_p^2=0.11$).

As shown in figure 10.2, for the Go stimulus the post-hoc comparisons (Bonferroni corrected *p*-values) revealed significantly higher P3 amplitude in the parietal region (M=4.58, SD=2.76) relative to central (M=3.63, SD=3.52; t(27)=3.00, p=0.02) and frontal regions (M=1.06, SD=2.24; t(27)=8.31, p<0.001) and in the central region relative to the frontal region (t(27)=6.88, p<0.001). However, the Nogo-P3 amplitude was higher in the central region (M=7.09, SD=4.06) relative to parietal (M=5.77, SD=3.75; t(27)=5.22, p<0.001) and frontal regions (M=5.10, SD=3.01; t(27)=6.93, p<0.001) but not in the parietal region relative to the frontal region (t(27)=1.81, p=0.25). Moreover, the post-hoc comparisons revealed that the P3 amplitude was significantly higher for Nogo stimulus relative to Go stimulus in the three regions (Frontal: t(27)=9.25, p<0.001; Central: t(27)=7.46, p<0.001; Parietal: t(27)=2.43, p=0.02).



Figure 10.1. *Left.* Grand average of the stimulus-locked ERP original waveforms elicited by the Go and Nogo stimuli in midline frontal, central and parietal regions at pre-training irrespective of the NF protocol. *Right.* Distribution of the P3 amplitude for the Go and Nogo stimuli. Positive amplitude is displayed upwards.



Figure 10.2. Two-way interactions of Stimulus and Region in mean P3 amplitude at pre-training. The graph shows that the mean P3 amplitude (μ V) for Nogo stimulus was significantly higher in the Central when compared to Frontal and Parietal locations irrespective of NF protocol at pre-training. For the Go stimulus P3 amplitude was significantly higher in Parietal when compared to Frontal and Central Regions. Error bars depict standard error of the mean.

10.3.1.2. Pre- to post-training differences between NF protocols

Based on the previous analysis of the scalp distribution of the Nogo- and Go-P3 amplitude, the impact of the three NF protocols was analyzed in the regions where these ERP components were maximally observed. Figure 10.3. presents the original waveforms for Nogo stimulus in the central region and for Go stimulus in the parietal region before and after control, theta and beta1 NF.

The differences in mean parietal Go and central Nogo-P3 amplitude were investigated in separate two-way mixed-design ANOVAs with the factors 2 TIME (Pre- vs. Post-training) x 3 PROTOCOL (control vs. theta vs. beta1). Figure 10.4. shows the mean P3 amplitude in the three NF protocols at pre- and post-training.

As shown in table 10.1, the two-way ANOVA did not support the hypothesis of a differential impact of the NF on the mean Nogo-P3 amplitude (TIME x PROTOCOL: F(2,25)=0.81, p=0.46, $\eta_p^2=0.06$). The main effects of TIME and PROTOCOL were also non-

significant (ps>0.44). The post-hoc comparisons failed to provide evidence of pre- to post-training Nogo-P3 amplitude changes in any of the three NF protocols (all ps>0.12).



Figure 10.3. Grand-average of the parietal Go and central Nogo ERP waveforms for control, theta and beta1 NF at pre- and post-training. The shaded area represents the 250-450ms measurement window. Positive amplitude is displayed upwards.

Similarly, the two-way ANOVA performed for the Go stimulus indicated a nonsignificant effect of the type of NF on the mean P3 amplitude (TIME x PROTOCOL: F(2,25)=0.43, p=0.66, $\eta_p^2=0.03$). For the Go-P3 amplitude the main effects of TIME and PROTOCOL were also non-significant (*ps*>0.66). The post-hoc comparisons failed to provide evidence of pre- to post-training Go-P3 amplitude changes in any of the three NF protocols (all *ps*>0.23).



Figure 10.4. Mean parietal Go- and central Nogo-P3 amplitudes for control, theta and beta1 NF at pre- and post-training. **(A)** The parietal Go-P3 and the central Nogo-P3 amplitude **(B)** were not significantly different at pre- and post-training in any of the NF protocols. Error bars depict standard error of the mean.

10.3.2. Hypothesis 2: Decreased RT and RT-SD as a function of theta and beta1 NF

As can be observed in table 10.1., two one-way ANOVAs revealed that there were no initial differences between NF protocols in mean RT and RT-SD (all ps>0.32). The differential impact of the three NF protocols in mean RT and RT-SD were analyzed in two-way 2 TIME x 3 PROTOCOL mixed-ANOVAs.

Table 10.1.

Means (standard deviations) and statistical results of the pre-training one-way ANOVAs and the pre- to post-training two-way ANOVAs on Mean Nogo- and Go-P3 amplitude, Hit Rate (H), False Alarm Rate (FA), d-prime (d'), Reaction Time (RT) Mean and Reaction Time Standard Deviantion (RT SD), percentage of Correct Trials (CT) to Go and Nogo stimuli included in ERP analyses after artifact rejection for each of the three NF protocols. Significant differences (p<0.05) are highlighed in bold.

Source							One-	NOVA				Two-	way Al	NOVA				
	Pre-training			Post-training			Protocol			Time			Time x Protocol			Protocol		
	Control	Theta	Beta1	Control	Theta	Beta1	F	Р	η_p^2	F	Р	η_p^2	F	Р	η_p^2	F	Р	η_p^2
Mean P3																		
Nogo	7.59 (2.84)	5.50 (4.50)	8.17 (4.52)	7.93 (1.95)	6.81 (3.98)	8.01 (3.77)	_	_	_	1.04	0.32	0.04	0.81	0.45	0.06	0.84	0.44	0.06
Go	5.14 (2.38)	4.15 (2.28)	4.46 (3.50)	4.31 (2.25)	4.15 (2.52)	3.88 (2.93)	-	-	-	1.51	0.23	0.06	0.43	0.66	0.03	0.16	0.85	0.01
H (%)	96.22 (3.42)	95.96 (6.19)	98.53 (1.39)	97.27 (3.41)	94.27 (15.46)	98.66 (1.49)	1.02	0.38	0.08	0.01	0.92	0.00	0.25	0.78	0.02	0.80	0.46	0.06
FA (%)	3.60 (3.60)	2.91 (2.53)	2.43 (2.62)	3.37 (2.75)	1.84 (2.14)	1.56 (1.78)	0.36	0.70	0.03	2.33	0.14	0.09	0.28	0.76	0.02	1.02	0.38	0.08
d'	3.99 (0.84)	4.09 (0.79)	4.42 (0.64)	4.10 (0.87)	4.45 (0.75)	4.69 (0.63)	0.80	0.46	0.06	4.30	0.05	0.15	0.36	0.70	0.03	1.24	0.31	0.09
RT (ms)																		
Mean	395.86 (64.32)	402.66 (108.30)	410.17 (132.26)	386.04 (104.41)	348.13 (91.70)	389.70 (100.80)	0.04	0.96	0.00	5.52	0.03	0.18	1.30	0.29	0.09	0.16	0.86	0.01
RT SD	140.83 (35.73)	126.66 (29.34)	116.97 (34.59)	127.76 (44.16)	113.09 (40.27)	121.29 (36.24)	1.18	0.32	0.09	2.19	0.15	0.08	1.35	0.28	0.10	0.56	0.58	0.04
CT (%)																		
Go	81.67 (11.25)	79.69 (17.98)	80.83 (13.92)	86.60 (14.12)	93.00 (6.63)	84.65 (13.22)	0.04	0.96	0.00	7.94	0.01	0.24	1.36	0.27	0.10	0.24	0.79	0.02
Nogo	82.22 (10.47)	80.63 (15.65)	81.39 (13.76)	85.00 (13.30)	90.88 (6.80)	82.92 (16.12)	0.03	0.97	0.00	3.05	0.09	0.11	0.99	0.38	0.07	0.27	0.77	0.02

10.3.2.1. Mean reaction time

As shown in fig. 10.5A, the mean RT decreased from pre-to post-training in the three NF protocols. As shown in table 10.1., the ANOVA revealed a significant main effect of TIME (F(1,25)=5.52, p=0.03, $\eta_p^2=0.18$) indicating that the RT was significantly lower at post-training (M=373.68; SD=97.06) relative to pre-training (M=402.89; SD=101.75) irrespective of the NF protocol. Furthermore, the TIME x PROTOCOL interaction was non-significant (F(2,25)=1.30, p=0.29, $\eta_p^2=0.09$) thus failing to provide evidence of a superior reduction of mean RT in the theta relative to control and beta1 NF. However, the post-hoc comparisons indicated a pre- to post-training decrease in mean RT in theta NF (t(9)=2.71, p=0.01) but not in other NF protocols (all ps>0.34). Finally, the main effect of PROTOCOL (F(2,25)=0.16, p=0.86, $\eta_p^2=0.01$) was not significant.



Figure 10.5. Two-way interaction of 2 Time x 3 Protocol for reaction time. The graphs show the variation in mean reaction time (RT) and reaction time standard deviation (RT-SD) from pre- to post-training in the three NF protocols. **(A)**. The mean RT was significantly decreased from pre-to post-training irrespective of the NF. **(B)** The RT-SD decrease in the theta and control NF were non-significant. Error bars depict standard error of the mean.

Reaction time standard deviation

Figure 10.5B shows the differential impact of the three NF protocols on RT-SD. As shown in table 10.1., the two-way TIME x PROTOCOL mixed-ANOVA did not reveal significant main effects or interactions (all ps>0.15) suggesting that the RT-SD was not significantly different at pre- and post-training and was not influenced by the type of NF.

The post-hoc comparisons failed to provide evidence of pre- to post-training RT-SD changes in any of the three NF protocols (all ps>0.12).

10.3.3. Hypothesis 3: Increased performance rates and perceptual sensitivity as a function of theta and beta1 NF

Table 10.1. summarizes the data and statistical results for mean d', hit rate and false alarm rate at pre- and post-training for the three NF protocols. As can be observed in table, there were no differences between NF protocols at pre-training for any of the dependent variables (all *ps*>0.38). The differential impact of the three NF protocols in mean d', hit rate and false alarm rate were analyzed in two-way 2 TIME x 3 PROTOCOL mixed-ANOVAs.

10.3.3.1.d'

As can be observed in figure 10.6A, d' was increased from pre-to post-training in in the three NF protocol. As shown in table 10.1, the two-way ANOVA revealed a nearly significant effect of TIME (F(1,25)=4.30, p=0.05, $\eta_p^2=0.15$) indicating a higher d' at posttraining (M=4.41, SD=0.77) relative to pre-training (M=4.16, SD=0.76). The same analysis did not reveal other significant main effects or interactions (all ps>0.31). The post-hoc comparisons failed to provide evidence of pre- to post-training d' changes in any of the three NF protocols (all ps>0.10).

10.3.3.2. Hit rate

The mean percentage of hit rate alarms showed a ceiling effect at pre- (M=96.87%; SD=4.27%) and post-training (M=96.65%; SD=9.34%). Figure 10.6B shows the percent hit rate changes from pre- to post-training in the three NF protocols. As shown in table 10.1, the TIME x PROTOCOL mixed ANOVA failed to reveal significant main effects or interactions (all *ps*>0.46). The post-hoc comparisons failed to provide evidence of pre- to post-training hit rate changes in any of the three NF protocols (all *ps*>0.54).

10.3.3.3. False alarm rate

The mean percentage of false alarms showed a floor effect at pre- (M=2.43%; SD=2.53%) and post-training (M=3.77%; SD=2.99%). As shown in figure 10.6C, the false

alarm rate was decreased in the three NF protocols. However, as shown in table 10.1, the statistical analysis with a two-way TIME x PROTOCOL mixed ANOVA did not reveal significant main or interaction effects (all ps>0.14). The post-hoc comparisons failed to provide evidence of pre- to post-training false alarm rate changes in any of the three NF protocols (all ps>0.19).



Figure 10.6. Two-way interaction of 2 Time x 3 Protocol for d' and performance rates. The graphs show the effects of the three NF protocols on d' (A), hit rate (B) and false alarm rate (C) at preand post-training. (A) d' was significantly increased from pre- to post-training irrespective of the type of NF. There were no significant changes in hit rate (B) or in false alarm rate (C) from pre- to post-training in the three NF protocols. Error bars depict standard error of the mean.

10.4. Discussion

The results of the current experiment failed to provide evidence of the hypothesized differential impact of theta NF and beta1 NF protocols in response inhibition and target stimulus processing in the cued Go/Nogo task.

10.4.1. Differential effects of theta and beta1 NF on response inhibition

The efficiency of response inhibition was measured by the Nogo-P3 amplitude and by the false alarm rate. The present experiment failed to provide supportive evidence that theta NF (or other NF protocols) would be associated with improved successful response inhibition as indexed by an increase in the Nogo-P3 amplitude (Hypothesis 1.1) and unsuccessful response inhibition as manifested by a decreased false alarm rate (Hypothesis 3.3).

In Study 1 (see section 6.2.2.) it was hypothesized that the failure of to replicate Nogo-P3 amplitude increases (Kropotov et al., 2005) and false alarm rate decreases (Egner & Gruzelier, 2001; Gruzelier, Foks, Steffert, Chen, & Ros, 2014) following SMR and beta1 NF could have been related to the absence of theta suppression instructions. Moreover, the involvement of the theta frequency in response inhibition (Barry, De Blasio, De Pascalis, & Karamacoska, 2014; De Blasio & Barry, 2013; Kirmizi-Alsan et al., 2006) would suggest that theta suppression rather than the enhancement of low-beta frequencies might have been implicated in those improvements. However, despite noticeable increments in Nogo-P3 amplitude and decrements in false alarm rate in theta NF, these were not statistically significant relative to the other NF conditions. Thus, the results of the present experiment suggest that theta suppression alone was not a sufficient condition to observe significant improvements in response inhibition.

Several aspects may have contributed to these results. Firstly, the previous evidence of increments in the Nogo-P3 amplitude following low-beta/theta NF (Kropotov et al., 2005) was obtained in ADHD children and not in healthy adults. Contrary to ADHD children that typically present lower Nogo-P3 amplitude (Liotti, Pliszka, Higgins, Perez, & Semrud-Clikeman, 2010), the absence of such initial deficits among high functioning subjects may have contributed to the failure to observe further cognitive improvements.

Another explanation for the failure to replicate previous reductions in commission errors in healthy adults might be related to differences in the type of NF protocol and in the task conditions. In previous studies, SMR enhancement with concomitant theta and highbeta suppression was associated with reductions in commission errors in the TOVA (Egner & Gruzelier, 2001; Gruzelier et al., 2014b). In Study 1 (see Experiment 3), differently from other NF protocols, SMR NF was not associated with increased false alarm rate despite increased response speed. Thus, it is possible that the SMR/theta NF protocol might be associated with superior response inhibition improvements relative to theta suppression and SMR enhancement NF in isolation. For instance, theta suppression might have a facilitative role in SMR learning that was found to correlate with improvements in response inhibition (Egner & Gruzelier, 2001).

Moreover, the lack of response inhibition improvements in the current experiment can be explained by a floor effect in the false alarm rate which is a likely consequence of low response inhibition demands of the task to high functioning adults. However, previous studies have found response inhibition improvements with even less demanding Go/Nogo tasks and equally few commission errors (Egner & Gruzelier , 2001) suggesting that the task conditions may not be a significant impediment to observe progress in situations of optimal performance context. Further, it is worth noting that the correlative nature of those findings and the lack of control for unspecific NF effects (see section 1.2.5.) warrants a cautious interpretation of those previous findings. Thus, future research is needed to investigate whether theta and beta1 NF may have a significant impact on response inhibition in optimal performance contexts by using more sensitivity measures.

Additionally, the results failed to show a contribution of theta NF to improve other behavioral indices of inhibition such as RT-SD (Hypothesis 2.2). Given that Experiment 3 failed to provide evidence of a specific effect of SMR NF RT-SD, it was reasoned that theta suppression may have been, at least in part, associated with RT-SD decrements in SMR/theta NF (Egner & Gruzelier, 2004). However, in the present experiment conditions there was no evidence that theta NF was associated with such effect. Chapter 11 will further discuss the implications of theta NF for RT-SD as a function of the task context.

Finally, the present experiment contributed to clarify the role of beta1 NF on response inhibition. Following Experiment 3, the current experiment confirms that beta1

NF was not associated with significant changes in Nogo-P3 amplitude. Importantly, contrary to hypothesis (Hypothesis 3.3) the current experiment did not replicate the trend towards increased false alarm rate observed in previous ones (Egner & Gruzelier, 2001; Experiment 3). Further work is necessary to clarify whether these differences might be related to active electrode location effects. Interestingly, , unlike present and past studies that used Cz- left earlobe, CPz-FCz or C3-C4 montages (Bakhshayesh, Hänsch, Wyschkon, Rezai, & Esser, 2011; Doppelmayr & Weber, 2011; Egner & Gruzelier, 2004) those previous increments were associated with left-hemispheric beta1 amplitude increments (C3-contralateral earlobe).

In sum, the current experiment did not allow for a full clarification of the specific effects of betal enhancement and theta suppression on response inhibition. Despite addressing potential methodological limitations of the previous study (by increasing the number of NF sessions and individualizing the frequency bands), the present experiment still failed to replicate those effects. The small sample size of the current experiment might have also limited the statistical power to detect differences between the NF protocols. Thus, more studies are needed to attempt to replicate these findings with a larger sample size.

10.4.2. Differential effects of theta an beta1 NF on Go stimulus processing

The results did not show the hypothesized involvement of beta1 NF with the facilitation of target detection and processing efficiency in the cued Go/Nogo task. In line with the previous study (Experiment 3), the analysis failed to provide supportive evidence that beta1 NF was associated with increased Go-P3 amplitude (Hypothesis 1.2.). These results are in disagreement with previous increments in Go-P3 amplitude effect on an equiprobable Go/Nogo task observed in ADHD (Kropotov et al., 2005). However, as noticed for the Nogo-P3 amplitude, the absence of initial neural and behavioral deficits in the present sample may explain the failure to observe further increments in Go-P3 amplitude.

Contrary to hypothesis beta1 NF was not associated with decreased mean RT (Hypothesis 2.1) relative to the other NF conditions. Thus, the present results failed to corroborate previous findings implicating the beta1/theta NF with decreased RT in ADHD children but also in healthy adults (Bakhshayesh, Hänsch, Wyschkon, Rezai, & Esser,

2011; Egner & Gruzelier, 2004). Importantly, the results suggest theta suppression NF rather beta1 NF might be associated with a pre- to post-training decrease in mean RT. However, in the absence of a significant TIME x PROTOCOL interaction it cannot be claimed that theta NF had a superior effect to other NF protocols in improving the efficiency of Go stimulus processing.

The inconsistent findings relative to previous studies may be related to some important differences in the task conditions. While in the present experiment the Go stimulus probability was higher than that of the Nogo stimulus, in previous studies the Go stimuli had a low probability (Bakhshayesh, Hänsch, Wyschkon, Rezai, & Esser, 2011) or high and low Go stimulus probability conditions were analyzed together (Egner & Gruzelier, 2004). Chapter 11 will further discuss the differential impact of theta NF and beta1 NF on Go stimulus processing efficiency as a function of Go stimulus probability (see section 11.1.2.). Additionally, the difference relative to previous studies may be explained by the absence of concomitant theta and high-beta inhibition which might have affected beta1 learning and the coupling of lower and higher frequencies as will be discussed in section 11.1.1.

Finally, following Experiment 3 and other previous studies (Bakhshayesh et al., 2011; Egner & Gruzelier, 2001, 2004), the results failed to demonstrate an association of beta1 NF with increases in correct target detection or hit rate (Hypothesis 3.2). Taken together the present experiment confirmed that beta1 NF is unlikely to contribute to a more efficient processing of target detection and processing in the particular conditions of the cued-Go/Nogo task. As a corollary of the absence of significant changes in performance rates (hit and false alarm), the present experiment failed to provide evidence of increased perceptual sensitivity as a function of the type of NF protocol (Hypothesis 3.1).

10.4.3. Summary

The current experiment suggests that the theta suppression and beta1 enhancement alone were not associated with improvements in response inhibition and Go stimulus processing in heathy adults. Despite evidence of improvements in ADHD children (Bakhshayesh et al., 2011; Kropotov et al., 2005) and in non-controlled studies with healthy adults (Egner & Gruzelier, 2001) the NF conditions that might have been associated with these effects remain unclear. The low response inhibition demands of the task and reduced number of sessions and statistical power might have prevented the observation of significant effects NF. Further work is needed to replicate these results with a larger sample and to take into consideration the response inhibition demands of the task for healthy adults.

Chapter 11

General discussion and conclusions of study 2

11.1. Overview of the main findings

Study 2 aimed at investigating the role of theta-inhibition NF and beta1 NF in selective attention and response inhibition improvements in three experiments. Contrary to hypothesis, Experiment 4 showed that during NF (i.e., in active periods) delta and theta activity were increased in beta1 NF relative to RF NF and delta activity was marginally increased in beta1 NF relative to theta NF. The results of Experiments 5 and 6 confirmed that the three types of NF had a differential effect on performance as a function of the different Go/target probability and response priming conditions of the tasks.

Beta1 NF was associated with task specific increased RT

The type of NF had a significant impact on mean RT in the three-stimuli oddball but not in the cued-Go/Nogo task. In the latter task, despite pre-to post-training decreases in mean RT it could not be claimed that theta NF had a superior improvement effect relative to other NF protocols. On the other hand, in the oddball task beta1 NF was associated with a significant increase in mean RT to an unpredictable target, but such effect was not observed when the target had a higher probability and was preceded by a warning cue. This differential increase in mean RT suggests that beta1 NF might have interfered with the processes of stimulus evaluation, response selection and/or movement execution upon detection of a deviant stimulus but not when the target stimulus was frequent.

In the oddball task, beta1 NF was also associated with an apparent delay of the P3b latency which possibly reflects increased stimulus evaluation time (Kutas et al., 1977) that might, in turn, explain the increased mean RT. Following the proposal that the latency of late ERP components may be explained by changes in phase and evoked oscillatory activity, a possible line of enquiry for future work would be to investigate whether this delayed P3 latency may be related to changes in phase resetting and enhanced post-stimulus

evoked delta and theta activity. This explanation would be consistent with the increased delta and theta amplitude in beta1 NF observed in Experiment 4. Interestingly, the visual inspection of the waveforms does not suggest a delay in the P3b latency in cued-Go/Nogo task (Experiment 6) in which the target stimulus was frequent. This difference may be explained by the role of theta dynamics in situations that imply a need for control such as the detection of a deviant task-relevant stimulus in the oddball task but not in the detection of a frequent target (Cavanagh & Frank, 2014). Moreover, in face of the previous literature it seems legitimate to conclude that the within-session increment in slow wave activity may reflect a decrease in the arousal state (Arns, Conners, & Kraemer, 2012; Barry, 2004; Howells, Stein, & Russell, 2010; Monastra et al., 2005), that might have contributed to the increased latency of the P3b and mean RT.

Control NF was associated with task specific increased hit rate and d'

Control NF condition was associated with increased hit rate and d' in the threestimuli oddball (Experiment 5) but not in the cued-Go/Nogo task (Experiment 6). Moreover, control NF was associated with decreased within-session delta and theta (Experiment 4). Taken together these findings suggest that the decreased pre-stimulus theta activity might be associated with a reduction of performance fluctuations and increased vigilance to infrequent targets (Makeig and Jung, 1996). Contrary to the oddball task, the higher Go stimulus probability and the presence of warning cues in the cued-Go/Nogo task might have prevented initial performance declines. However, as noted in section 9.4., it is possible that these performance increments in the oddball task reflect only the correction of an initial failure to comply with the instructions rather than a real performance advantage.

Nevertheless, the hypothesis that these performance enhancements may be causally related to the decreased pre-stimulus theta activity merits further investigation. Future studies are needed to clarify whether differences in pre-stimulus theta power induced by NF may predict changes in hit and error rates in averaged trials time-frequency analysis (Makeig & Jung, 1996). Unfortunately, in the current experiment the low number of missed targets at pre- and post-training does not allow for a meaningful statistical comparison between different conditions.

Delta and theta up- and down-regulation may be associated with distinct effects on selective attention and response inhibition

In contrast with previous studies (Vernon et al., 2003; Ros et al., 2009; de Zambotti et al., 2012; Gruzelier et al., 2014a), it was demonstrated that the theta amplitude was significantly decreased (in control and theta NF) relative to other NF comparison conditions (beta1 NF). Moreover, the current study suggests that the up- and down-regulation of slow frequencies (delta and theta) might have distinct cognitive effects on selective attention and response inhibition. In the selective attention domain, the down-regulation of theta amplitude might have been related to improved performance accuracy in control NF, while the up-regulation of theta amplitude in beta1 NF might have been related to increased stimulus evaluation and/or response selection time.

However, the modulation of delta and theta activity did not influence the response inhibition processes in both Experiment 5 and 6. Interestingly, these results are in line with previous demonstrations that fm-theta up-regulation had a significant impact on executive functions that implicated proactive (e.g., memory updating) but not reactive cognitive control (e.g., response inhibition) (Wang & Hsieh, 2013; Enriquez-Geppert et al., 2014a). Further work is needed to determine whether the down- and up-regulation of delta and theta amplitudes might have distinct cognitive effects and whether they reflect specific or unspecific effects of NF. As discussed in section 8.4.1.1., theta modulation might reflect distinct cognitive effort requirements of the NF (Enriquez-Geppert et al., 2014b).

Alternatively, previous improvements in response inhibition might have been related to changes in EEG parameters other than decreased theta power. In fact, although theta power was often decreased following SMR/theta NF (Gruzelier, Hirst, Holmes, & Leach, 2014a; Ros et al., 2009; Vernon et al., 2003), no previous study demonstrated a causal relationship between those decrements and improved response inhibition. In fact, increased post-stimulus power and phase coherence in the theta (Cavanagh & Frank, 2014) and beta frequencies (Alegre et al., 2004; Picazio et al., 2014; Swann et al., 2009) could have contributed to the cognitive control and response inhibition improvements observed in previous studies.

Finally, the current experiment failed to provide supportive evidence of the involvement of theta suppression in promoting a decrease in RT-SD (Egner & Gruzelier,

2001, 2004; Fuchs, Birbaumer, Lutzenberger, Gruzelier, & Kaiser, 2003; Monastra, Monastra, & George, 2002). However, it is worth noting that RT-SD was marginally decreased in theta NF in the three-stimuli oddball but not in the cued-Go/Nogo task. Importantly, this improvement was obtained in a low target probability context that imposes a higher demand to performance monitoring abilities. However, the absence of a significant decrease in RT-SD relative to other NF conditions advises a cautious interpretation of these findings.

11.2. Limitations and future directions

The absence of training- and frequency-specific amplitude changes clearly limits the interpretation of the differential cognitive effects of the three NF protocols. The withinsession delta and theta amplitude changes suggest that beta1 NF might differ from the control and theta NF in terms of motivation and cognitive effort. However, further work would be needed to investigate whether this interpretation is supported by differences in subjective ratings of effort and motivation. Moreover, it is unclear whether differences in motivation and cognitive effort might have interfered with the main task in the theta and beta1 NF (i.e., to enhance beta1 activity and decrease theta activity relative to controls).

On the other hand, the within-session amplitude increments in slow frequencies may reflect increased mental fatigue and lower arousal (Wascher et al., 2013) in beta1 NF. However, it is unclear whether these amplitude increments underpinned the failure to replicate previous improvements in selective attention following beta1/theta NF (Bakhshayesh et al., 2011; Egner & Gruzelier, 2004). A task for future studies will be to clarify whether the beta1/theta NF might better control for within-session amplitude increments in the slow frequencies relative to beta1 NF and whether the two protocols may have distinct effects on cognitive performance.

Thus, the present findings were inconclusive regarding the independent role of theta suppression and beta enhancement on selective attention and response inhibition improvements in heathy adults. This might have been related to previously discussed "ceiling and floor effects" in theta and beta1 amplitude as well as in the cognitive performance measures. This limitation and possible ways to tackle it in future studies will be further discussed in section 12.3.1.

Chapter 12

Conclusions

12.1. General discussion of studies 1 and 2

The experimental work constituting this thesis can now be evaluated in relation to the two main purposes of the thesis: (1) to investigate the feasibility of successfully modulating the EEG amplitude of the theta and low-beta frequencies (section 12.1.1.) and (2) to clarify whether theta suppression and/or low-beta enhancement would be associated with improvements in selective attention and response inhibition (section 12.1.2.). After summarizing the main findings, several methodological limitations of the present experimental work will be discussed and directions for future studies will be proposed (section 12.1.3.).

12.1.1. Learned modulation of theta and beta frequencies

The feasibility of inducing training- and frequency-specific modulations of theta and low-beta activity was investigated with within- and across-session methods during active feedback and passive resting state periods. Two major conclusions could be derived from the current experimental work. Firstly, there was limited evidence of trainingspecificity. In fact, even when there was evidence of superior amplitude changes in one NF condition relative to another this was not true relative to a third one. As an example, training-specific beta1 amplitude increments were observed between beta1 and control NF but not between beta1 and SMR NF. Moreover, it was not possible to demonstrate that the amplitude changes were frequency-specific. This probably reflects the fact that (1) the amplitude changes in one oscillatory frequency may induced changes in other parameters of brain oscillatory activity in the same or in different frequencies and that (2) common or unspecific aspects to every type of NF may constrain the oscillatory activity in certain frequencies. Besides searching for comparable control conditions in terms of motivational and cognitive implications, it would be important to redefine the criteria for determination of NF learning based on newer assumptions of brain oscillatory activity. This would include (1) the investigation of cross-frequency amplitude-amplitude and phase-amplitude coupling phenomena and (2) disentangling the effects of neurostimulation on the physiological parameters of interest (i.e., power or coherence) from those of self-regulatory processes by comparing NF with other forms of neurostimulation that do not require an active engagement of the participants (e.g., TACS).

A second aspect that emerges from the present research is that active feedback and passive resting state describe different phenomena involved in NF. Even in the same frequency, the amplitude changes across active feedback do not necessarily match those in the passive resting state periods. For instance, in Study 1 there was evidence of training-specific beta1 amplitude increments relative to controls in active feedback periods but not in passive resting state. Conversely, Study 2 revealed that between theta and beta1 NF there was only evidence of training-specific theta amplitude decrements in passive periods but not in active ones. Since previous studies also noticed conflicting evidence between feedback and resting state learning indices (HoedImoser et al., 2008; de Zambotti et al., 2012; Witte et al., 2013), a task for future research may be to systematically investigate for each frequency which learning index may be causally linked to the desired cognitive changes. For example, theta amplitude increments in active but not in passive periods may be causally related to increased post-stimulus theta power.

12.1.2 Central midline theta and sensorimotor beta effects on selective attention and response inhibition

A major goal of the thesis was to investigate the differential effects of NF on selective attention and response inhibition in two Go/Nogo tasks that imposed specific demands on these processes. In this respect, the two experimental studies made important advancements to our current understanding of the independent contribution of theta suppression and sensorimotor beta enhancement for cognitive enhancement.

Selective attention

SMR NF was associated with increased target stimulus processing efficiency evinced by increased Go-P3 amplitude, decreased mean RT and decreased RT-SD. However, except for the increased Go-P3 amplitude there was no evidence that SMR NF was associated with improved cognitive performance relative to other NF protocols, namely

RF NF. These results corroborate previous findings that SMR but not betal NF may be related to improved target stimulus processing efficiency (Doppelmayr & Weber, 2011) and provide first evidence of the proposed functional differentiation between left-hemispheric betal NF and right-hemispheric SMR NF (Othmer et al., 1999). However, the cognitive implications of SMR NF could not be attributed to SMR amplitude enhancement. As discussed in section 6.1., SMR NF might have contributed to influence oscillatory mechanisms other than SMR amplitude. The increased amplitude of the Go-P3 component suggests that SMR NF might have influenced the evoked activity in low frequencies that contribute to the P3 component and/or the mechanism of phase resetting and phase alignment of task-relevant oscillations. Moreover, the fact that control NF was also associated with nearly significant and significant changes in Go-P3 amplitude and RT suggests that the cognitive improvements in SMR NF were not different from a placebo intervention.

Taking together the results of both studies, there was no evidence that beta1 NF was associated in improvements in electrophysiological or behavioral indices of selective attention. Unlike other NF protocols, beta1 NF was not associated with significant increases in Go-P3 amplitude or decreases in mean RT and RT-SD in Study 1. These results suggest that beta1 NF had poorer selective attention outcome compared to placebo RF and ineffective SMR NF interventions. Moreover, in Study 2 beta1 NF was associated with increase mean RT and with an apparent increase in the P3b latency in the oddball task, suggesting that the increased response latency to infrequent targets might be related to increased stimulus evaluation time as well as with a slower movement execution. Further work is needed to investigate whether the increased P3b latency effect holds true and to determine whether it might explain the increased mean RT. However, the absence of frequency-specific beta1 amplitude increments in Study 1 and increased delta and theta activity relative to the other two NF protocols in Study 2 warrants caution in attributing these outcomes to increased betal activity. Although these results are in disagreement with the previous suggestion that beta1 NF might contribute to increased cortical arousal responsible as indicated by increases P3b amplitude and decreased mean RT (Egner & Gruzelier, 2004), it must be noticed that this previous study also failed to provide supportive evidence of frequency-specific beta1 amplitude increments (Egner et al., 2004). Thus, the potential cognitive effects of increased beta activity remain unclear.

The results of Study 2 were inconclusive regarding the cognitive effects of central midline theta suppression. Firstly, the presence of significant delta amplitude decrements in control and theta NF relative to beta1 NF did not allow a full clarification of the specific effects of theta suppression. Then, it is unclear whether the theta amplitude decrements in passive periods obtained in theta NF can be explained by the NF intervention because no significant theta decrements were observed in active periods. Moreover, the increased perceptual sensitivity (in control NF) and decreased RT mean and SD (in theta NF) suggest that reductions in amplitude of low frequencies was associated with selective attention improvements. Conversely, when pre-stimulus theta amplitude increases as a function of NF training (e.g., as in beta1 NF) the detection of infrequent target stimulus may become slower.

Response inhibition

In line with previous studies (Gruzelier et al., 2014b; Egner & Gruzelier, 2004; Bakhshayesh et al., 2011), the results of the two present ones were inconclusive regarding the independent effects of theta suppression and SMR and beta1 enhancement on response inhibition. Study 1 investigated the proposed functional role of right-hemispheric SMR NF in improving response inhibition (Othmer et al., 1999; Egner & Gruzelier, 2004). In contrast with the other NF protocols, the false alarm rate did not increase after SMR NF suggesting that it might contribute to increased response preparation without affecting response inhibition. However, Study 2 did not replicate the increases in false alarm rate in other NF protocols and even provided evidence of marginal decreases in false alarm rates in the oddball task for both control and beta1 NF. Thus, the previous interpretation that SMR NF may prevent increments in false alarm rate may not hold true when compared to other NF protocols.

Study 2 investigated whether theta suppression may account for previous effects of SMR/theta and beta1/theta NF on response inhibition (e.g., Kropotov et al., 2005). As previously noted, both control and theta NF were associated with decreased delta and theta activity relative to beta1 NF. However, contrary to theta NF, control NF was related to
decreased false alarm rate to nontargets. Moreover, betal NF was also associated with decreased false alarm rate but to standard stimulus. Thus, the differential effects of NF modulation of theta amplitude on false alarm rate remain elusive. Study 2 also failed to replicate previous Nogo-P3 amplitude enhancement effects (Kropotov et al., 2005) in high functioning adults.

Following the proposal that enhanced beta1 activity could be associated with fasteryet-more-error-prone behavioral performance (Egner & Gruzelier, 2001, 2004), the two experiments also attempted to investigate the effects of beta1 NF on response inhibition. However, Study 1 and 2 provided conflicting evidence of the impact of beta1 NF on false alarm rate. Study 1 indicated that beta1 NF was associated with a marginal increase in false alarm rate in the cued-Go/Nogo task which was not replicated in Study 2. In the latter beta1 NF was even associated with decrease false alarm rate to standards. Several differences between studies might explain these different outcomes. Besides negligible differences resulting from the individualization of the frequency bands, in Study 1 the beta1 activity was enhanced in the left hemisphere as opposed to the central midline location in Study 2. Moreover, while the results of Study 1 could be related to training-specific beta1 amplitude increments, those of Study 2 could be better explained by increased delta and theta activity than by increased betal activity relative to other NF protocols. As previously noted, because betal amplitude was not selectively increased relative to other frequencies, it remains unclear whether the marginal increase in false alarm rate can be attributed to the beta1 activity. Finally, the difference in the number of beta1 NF sessions between studies might explain the different response inhibition effects. In the future, differences in performance related to the hemispheric specificity and with the number of sessions should also be object of systematic study.

In sum, the two studies were inconclusive regarding the specific effects of NF that might contribute to increased processing efficiency of target stimulus as evinced by increased Go-P3 amplitude, decreased mean RT and RT-SD, increased d' and increased hit rate. Although both SMR (in Study 1) and RF NF (in both Study 1 and 2) were associated with specific cognitive improvements these NF protocols could not be related to a specific pattern of EEG amplitude changes. The results were also inconclusive regarding the specific conditions associated with improved response inhibition. Increased error rates were

evinced following both beta1 and RF NF which had distinct effects on beta1 activity. This suggests that factors other than enhanced beta activity might have accounted for the increased false alarm rate proposed in previous studies (Egner & Gruzelier, 2001, 2004).

Importantly, the analysis here reported did not allow for the establishment of a causal relationship between the increased behavioral and electrophysiological responses and NF-induced changes in the oscillatory activity at pre- and/or post-stimulus. Further work is needed to investigate this temporal and causal relationship in interactive ERP paradigms in which the self-induced modulations of theta and beta amplitude may be used as an independent variable as suggested in section 1.1.1.

12.2. Theoretical and practical implications

The experimental work of this thesis attempted to contribute to the ongoing debate concerning the differential impact of SMR and beta1 NF frequencies enhancement (Doppelmayr & Weber, 2011; Egner & Gruzelier, 2001, 2004; Othmer et al., 1999) and theta suppression (Becerra et al., 2012; Gruzelier et al., 2014; Gruzelier, 2014c; Ros et al., 2009) on cognitive performance. However, the failure to provide evidence of frequency-specific SMR and beta1 enhancements and of theta suppression hindered the possibility of establishing a relationship between the different types of NF modulation and the specific cognitive effects.

Is there a causal relationship between NF and cognitive outcomes?

So far, the relationship between NF modulation and cognitive outcomes, when demonstrated, has been correlational in nature (e.g., Kober et al., 2015). This correlational evidence is insufficient to establish a cause-and-effect relationship between NF modulation and cognitive effects. For example, training-specific amplitude changes may be observed in frequencies other than those targeted by the NF interventions. In the experimental work presented here (Study 2), beta1 NF was associated with increased delta and theta amplitude relative to controls. Thus, a positive correlation between increased beta activity and improvements in cognitive performance would be spurious. A more direct way to establish a causal link would be to compare the effects of NF designed to modulate power and/or connectivity in certain frequency bands with those of other forms of neurostimulation (e.g., TACS). If TACS reverses or has similar effects to those of NF a causal link between a specific frequency modulation and behavior can start to be established. Moreover, in order for a cause-and-effect relationship to be established a temporal link between NF modulation and the resultant cognitive effect must be demonstrated. This would imply not only training participants to acquire voluntary control over certain parameters of brain activity (e.g., local power and coherence) but also to make stimulus presentation contingent on voluntarily controlled signals. Finally, for a causal link to be established the cognitive effects cannot be better explained by the stimulation in other candidate task-relevant frequencies.

Another important theoretical question concerns the mechanisms by which NFinduced changes in oscillatory activity, namely amplitude, might be causally linked to changes in the amplitude and latency of ERP components. It was proposed that the increased Go-P3 amplitude following SMR NF may reflect increased evoked power or phase-locking of task-relevant low frequency oscillations (Başar-Eroglu, Başar, Demiralp, & Schürmann, 1992; Demiralp et al., 2001; Doppelmayr, Klimesch, Pachinger, & Ripper, 1998; Polich, 1997; Yordanova, Kolev, & Başar, 1998; Yordanova et al., 2002; Yordanova, Rosso, & Kolev, 2003). A recent study suggests that local increases in central SMR amplitude may be related to decreased phase coherence of SMR activity between central and posterior sites (Kober et al., 2015). Thus, the possible influence of SMR NF on mechanisms of stimulus evoked amplitude and instantaneous phase alignment might explain its implications for ERP amplitude. Further analyses are needed to investigate whether changes in post-stimulus evoked power or in phase locking may have been related to increased Go-P3 amplitude in SMR NF.

It was also proposed that the increased RT latency following betal NF could be explained by an increase in stimulus evaluation as evinced by an increased latency of the P3b. Although NF-induced changes in ERP latency were not object of study in the current experimental work, an increased ERP latency may be explained by a decreased in oscillatory frequency of task relevant oscillations and/or by an increase in phase concentration around a later time point. Further work is needed to investigate whether the increased oscillatory activity in lower frequencies associated with beta1 NF might explain the increased P3b latency. In sum, there is still limited evidence of how NF-induced amplitude changes may be causally related to functional changes in ERP amplitude and latency. In order to explore whether such relationship exists future research would do well to (1) explore changes in parameters of oscillatory activity other than pre-stimulus power such as inter-trial phase coherence and post-stimulus evoked power, (2) compare the frequency-specific NF effects with those of TACS and (3) to investigate the temporal relationship between NF-induced amplitude changes and stimulus evoked activity in trial-by-trial analysis.

Can the concertation of multiple frequencies explain the cognitive effects of NF?

Another theoretical challenge for future research is to explain the how NF may promote changes at the level of functional networks and influence higher order cognitive activity. Although NF protocols that combine the enhancement (e.g., low-beta) and suppression (e.g., theta and high-beta) of different frequencies may present increasing interpretational difficulties relative to simpler ones, they may better reflect the complexity of brain operations requiring the coordination of multiple networks. It is possible that a significant impact on cognitive activity may only be achieved by coordinating distributed neuronal processes involving the oscillatory activity in multiple frequency bands (Canolty & Knight, 2010; Varela, Lachaux, Rodriguez, & Martinerie, 2001). For instance, the crossfrequency phase-amplitude coupling of theta and beta frequencies was found to provide a mechanism for sequential encoding of items in working memory (Engel & Fries, 2010), behavioral inhibition (Knyazev, 2012) and cortico-subcortical communication (Schutter, Leitner, Kenemans, & van Honk, 2006). Thus, it is possible that in order to influence complex cognitive functions, such as response inhibition, NF protocols may need to promote the concertation of different brain oscillations. Future research should compare the learning and cognitive outcomes of simpler (e.g., SMR NF) and more complex (e.g., SMR/theta) NF protocols and investigate their impact at the level of local and distributed networks.

What are the potential practical applications of theta and low-beta NF?

Since the ultimate goal of NF research is to validate protocols for enhancing cognitive functions in clinical and typically developing populations, we now turn to the

potential practical applications of the experimental work of this thesis. A first potential application of SMR NF (and to some extent theta-suppression NF) is in clinical disorders that have been characterized by reduced P3 amplitude, increased RT and/or RT-SD such as cognitive decline associated with initial phases of dementia and normal aging (Phillips et al., 2013; Smart et al., 2014), development disorders such as ADHD (Barry et al., 2003b; Liotti et al., 2010; Senderecka et al., 2012) and traumatic brain injury (Stuss et al., 2003). However, the brain mechanisms behind these cognitive improvements must be clarified before clinical protocols can be developed and tested in clinical trials. Moreover, the potential benefits to clinical improvements must be rigorously assessed.

Another area of potential interest is in optimal performance and cognitive enhancement contexts, such as in sports, in which the evidence of efficacy of NF in response preparation and selection has remained limited to date (Park, Fairweather, and Donaldson, 2015; Ring, Cooke, Kavussanu, McIntyre, & Masters, 2015; Thompson, Steffert, Ros, Leach, & Gruzelier, 2008). In this area the P3b amplitude and latency as well as in RT and RT-SD have been proposed to differentiate the performance of experts from that of novices (Nakamoto & Mori, 2008, 2012). Thus, theta suppression and SMR NF interventions may offer the promise of being translated into significant cognitive performance improvements.

12.3. Methodological limitations and future directions

12.3.1. Factors affecting learning-specificity and cognitive performance effects

Although methodological limitations have already been separately discussed for each study in chapters 6 and 11, they will be reconsidered here focusing on the commonalities between the studies. These can be classified as limitations regarding (1) the method of individualizing the frequency bands, (2) the modifications introduced in the NF protocols, (3) the number of sessions, (4) the presence of ceiling and floor effects (5), and the reduced statistical power.

Method of frequency band individualization

It could be argued that the failure to show the expected training-specific theta amplitude decrements and beta1 amplitude increments in Study 2 may be related to the particular method of individualizing the frequency bands. Although it has been suggested that the alpha frequency may be adopted as a common reference for individually adjusting the frequency bands (Klimesch, 1999), especially for the determination of the transition frequency between theta and alpha (Doppelmayr, Klimesch, Pachinger, & Ripper, 1998), other methods could have been used. In particular, when frequencies other than alpha are considered, the maximal synchronization peak during activation tasks may be an alternative feasible method to determine the individual frequency peak. This method has been successfully adopted in previous fm-theta enhancement NF studies which used a range of executive functions tasks to determine the individual fm-theta (Enriquez-Geppert et al., 2014a). Post-movement event-related synchronization following activation tasks (e.g., motor preparation or imagery) has also been adopted to determine the dominant sensorimotor beta frequency (Pfurtscheller et al., 1997; Pfurtscheller & Lopes da Silva, 1999; Neuper & Pfurtscheller, 2001). Thus, future studies would do well in following previous suggestions of functionally differentiating beta frequencies based on individually defined oscillatory patterns of beta reactivity (e.g., to motor inhibition) rather than using the classically defined frequency bands or beta band adjustments according to the IAF (Neuper & Pfurtscheller, 2001; Pfurtscheller & Lopes da Silva, 1999).

Modifications of the low-beta NF protocol

The attempt to clarify the independent contribution of theta suppression and of SMR and beta1 enhancement for the cognitive improvements determined the exclusion of theta suppression from the SMR and beta1 NF protocols. This modification might have been related to the failure in replicating previous electrophysiological and behavioral effects and to provide evidence of SMR (Study 1) and beta1 learning (Study 2). For instance, the increased slow wave activity might have been related to the decreased behavioral perfomance in beta1 NF in Study 2.

As previously mentioned, complex cognitive operations typically rely on the concertation between low and higher frequencies rather than on a single frequency.

Moreover, theta suppression might be of importance in maintaining the attentional focus and in signaling early drowsiness stages (Makeig & Jung, 1996) necessary for successful low-beta conditioning (Sterman & Shouse, 1980; Othmer et al., 1999). Thus, it is an empirical question for future studies to investigate whether SMR/theta and beta1/theta NF may be related to different cognitive outcomes and involve different changes in oscillatory activity when compared to SMR and beta1 NF.

Reduced number of NF sessions

The reduced number of NF sessions might explain the failure to provide supportive evidence of training- and frequency-specific effects as well as evidence of specific cognitive outcomes. Based on clinical non-randomized studies, it has been assumed that the magnitude of the amplitude change in the target frequencies and the differentiation between amplitude changes in the target and nontarget adjacent frequencies increases with the increasing number of sessions (Lubar & Shouse, 1976; Sterman & Friar, 1972).

However, it remains unclear whether a linear relationship between the number of NF sessions and an higher learning specificity and cognitive performance can be claimed. As an example, significant Go-P3 and training-specific beta1 amplitude increments were observed in Study 1 where the number of NF sessions was lower, but not in Study 2 where it was higher. These results suggest that the number of NF sessions required to yield significant learning and cognitive outcomes may vary from one EEG frequency to the other and probably depend on the complexity and amenability of the cognitive functions for plastic change. Future research would do well in designing controlled studies attempting to clarify the relation between training- and outcome-specificity and the number of NF sessions.

Ceiling and floor effects

Another important methodological limitation was the possible presence of ceiling and floor effects in the learning and cognitive outcome measures. Although in principle the electrophysiological measures (peak-to-peak amplitude and mean amplitude of the ERP components) may not be constrained by the same inherent measurement and physiological limits as the behavioral measures, in practice these measures may also be subject to ceiling effects (Picton et al., 2000). For instance, a ceiling effect was observed in the oddball task

with the progressive increment of the target-to-target interval, a known determinant of the P3 amplitude (Gonsalvez & Polich, 2002). In the current experimental work, the pretraining high levels of beta activity and low levels of theta activity in healthy young adults could have determined "ceiling and floor effects" that can explain the failure to demonstrate significant training-specific amplitude changes (e.g., in SMR NF or in theta NF) relative to controls.

The presence of "ceiling and floor effects" was most evident in the behavioral measures of the oddball and cued Go/Nogo tasks. For instance, the hit rate scores concentrated around maximal values (ceiling effect), while the false alarm rates concentrated around minimal values (floor effect). This has probably inflated the probability of Type II errors, that is, the possibility that differences between the NF protocols at post-training might have gone undetected because of the reduced variance in the independent variables. These effects might have resulted from the combination of a high functioning sample with the relatively low cognitive demands of the tasks.

In order to avoid these effects, future studies may consider increasing the task difficulty by manipulating the inter-stimulus interval, the difficulty of stimulus discrimination (in the three-stimuli oddball task) and the Go/Nogo stimulus probability (in the Go/Nogo task) the in order to prevent these effects. Alternatively, other paradigms (SST and Go/Nogo task with RT deadline) may be more adequate to investigate response inhibition improvements in healthy adults. Future studies may also consider testing the differential effects of theta and beta1 NF in conditions of pre-training increases in theta power or decreases in beta power. Such conditions of increased theta/beta ratio are frequently encountered in ADHD (Arns et al., 2012; Barry, Clarke, & Johnstone, 2003; Clarke, Barry, McCarthy, & Selikowitz, 1998) and in drowsiness (Makeig & Jung, 1996; Takahashi, Shinomiya, Mori, & Tachibana, 1997).

Statistical significance and power

An important limitation in interpreting some of the present findings was that the post-hoc comparisons had an exploratory nature when not protected by significant TIME x PROTOCOL interactions (e.g., reduced RT-SD in theta NF). This represents an increased risk of incurring in type I error (rejecting the null hypothesis of no statistical differences

between NF protocols, when the null hypothesis is true). The failure to observe significant interaction effects might have been related to the reduced statistical power, which represents an increased risk of incurring in type II errors (accepting the null hypothesis, when there is in fact a true statistical difference).

The reduced power is a severe limitation shared by most NF and cognitive neuroscience studies (Button et al., 2013). Based on the reported effect size, the statistical power for each of the statistical tests could be calculated². In face of the small observed power, one cannot be certain that the rejection of the alternative hypotheses corresponds to the absence of real differences in performance after NF. Relying on the statistical conventions, in order to observe a desired power of 0.80 in the three-stimulus oddball task (minimizing the risk of accepting the false null-hypothesis to 20%), the calculations recommended sample sizes as high as 120 subjects. Thus, future studies investigating the effects of NF may consider significantly increasing the samples sizes.

12.3.2. Difficulty in determining the effects of NF on voluntary and involuntary aspects of selective attention

A major methodological limitation of the present study was the impossibility of determining whether the NF protocols would differentially affect the voluntary and involuntary mechanism of selective attention. The posterior distribution of the P3 amplitude to nontargets (P3a) was in agreement with the proposal that in an easy target-standard discrimination context this component may reflect the updating of the memory representation rather than the involuntary capture of attention (Hagen et al., 2006).

In a previous study, the P3 amplitude enhancement effect was observed in anterior and posterior sites in conditions of high target-standard discrimination difficulty (Egner & Gruzelier, 2004) suggesting that both P3a and P3b components might have been enhanced. Future studies would do well to replicate these findings in a difficult target-standard discrimination three-stimuli oddball paradigm in which the distinct topography of the frontal-P3a and parietal-P3b components may provide more specific indications of the nature of the selective attention mechanisms involved (e.g., Sawaki & Katayama, 2009).

² The calculation of the statistical power using G*Power software (Faul, Erdfelder, Lang, and Buchner, 2007) was based on η_p^2 for F-tests. The software also provides an estimation of the required sample size for the desired level of power (effect sizes were calculated as in SPSS).

Alternatively, future studies could introduce a novel stimulus which is recognized to elicit the same fronto-central effect.

12.3.3. Failure to independently manipulate Go response probability and response priming

The P3 amplitude has a similar functional significance in the context of the oddball and in that of the cued-Go/Nogo task (Pfefferbaum et al., 1985). Therefore, it is surprising that SMR NF was associated with P3 amplitude increments to a frequent Go target but not to an infrequent one.

The three stimuli oddball and the cued Go/Nogo were designed to maximize the possibility of detecting selective attention and response inhibition effects in accordance with the main goals of the thesis. However, the Go stimulus probability and the precedence by a warning cue were not orthogonally manipulated in the two tasks. While in the three-stimuli oddball task the low probability Go response was never preceded by a warning cue, in the cued Go/Nogo task the high probability Go response as always preceded by a facilitating response priming. This resulted in different levels of response preparation and top down control in the pre-stimulus period (Boulinguez, Ballanger, Granjon, & Benraiss, 2009) in the two tasks (lower in the three-stimuli oddball and higher in the cued Go/Nogo). that are difficult to attribute to the independent effects of stimulus probability or to response priming. In order to tackle this limitation future studies should consider orthogonally manipulate the Go response probability and the degree of previous response preparation as within-subject factors in the context of the same task.

Moreover, the cued-Go/Nogo task to be used in future studies should allow for the testing of specific predictions regarding improvements in speed-accuracy trade-off as suggested by increased response speed without affecting the false alarm rate following SMR (Experiment 3) and theta NF (Experiment 6). This could be achieved by comparing conditions in which the cue prompts response speed or accuracy in a counterbalanced design. A manipulation of response speed would also have the advantage of better elucidating the possible effects of NF on successful and unsuccessful response inhibition. In fact, despite increases in response speed in both tasks (see figures 4.4., 5.5. and 10.5.) this might have not been enough to substantially increase the difficulty of response

inhibition (Dimoska, Johnstone, & Barry, 2006). Future studies should also consider manipulating the difficulty of inhibition by imposing restrictions on RT deadline (Benikos, Johnstone, & Roodenrys, 2013). Finally, the comparison of cue and no cue conditions allows for the testing of specific hypothesis regarding differential effects of beta activity in cue processing (Kilavik, Zaepffel, Brovelli, MacKay, & Riehle, 2013) which might explain the increased false alarm rate in beta1 NF (Experiment 3).

12.3.4. Statistical comparison of different ERP components

Another factor that may present difficulties in the interpretation of the current findings results from the impossibility of comparing effects from different brain regions within the same experiment. For instance, while in the three-stimuli oddball task both P3a and P3b presented a maximal parietal amplitude allowing for the statistical analysis of effects associated with the stimulus condition (i.e., targets vs. nontargets) in the cued-Go/Nogo task this was not possible given the regional differences of the effects (central for the Nogo stimulus and parietal for the Go stimulus). It could be argued that the option to run different analysis for each type of stimulus could risk increasing the type II error. However, because parietal and central effects could not be included in the same analysis, this allowed the possibility of running the analysis in the two tasks with the same level of type II error protection, therefore increasing their comparability. An alternative procedure would have been to follow-up an omnibus analysis including the factors region and stimulus with planned comparisons in the regions and stimulus of interest.

12.4. Final conclusions

The experimental data constituting this thesis has provided supportive evidence of the differential effects of different types of NF on psychophysiological indices of target information processing efficiency such as the Go-P3 amplitude, the mean RT and the RT-SD. Although these findings replicate previous indications of differential effects of SMR and beta1 NF (Egner & Gruzelier, 2004; Doppelmayr & Weber, 2011), there was no evidence of superior cognitive performance following SMR NF when compared to a control NF condition or of training-specific SMR amplitude increments. Moreover, these effects were only observed in the cued-Go/Nogo task.

Despite inconclusive evidence regarding training-specific theta amplitude decrements, the results were also suggestive of the contribution of theta suppression NF to increased processing efficiency of target stimulus information. However, the RT-SD in the oddball task and mean RT in the cued-Go/Nogo task were not significantly decreased relative to other NF conditions. Importantly, the failure to provide evidence of training-specific beta1 amplitude increments in one of the studies limited the possibility of drawing more definite conclusion regarding the role of beta1 NF in selective attention and response inhibition.

In sum, the current experimental work was not conclusive regarding the conditions that might have contributed to the improvements in target processing efficiency. Despite the fundamental importance of demonstrating the training- and frequency-specificity of the NF interventions, this may not be enough to establish a causality link between NF and changes in cognitive performance. In fact, the present research showed that changes in cognitive performance may occur independently from NF-induced amplitude modulations and that training-specific effects are not necessarily predictive of cognitive improvements. In this respect, it would be important to investigate which parameters of brain oscillatory activity elicited by NF may be causally related to the modulation of ERP amplitude and latency and with the changes in behavioral performance.

13. References

- Alegre, M., Gurtubay, I. G., Labarga, A., Iriarte, J., Valencia, M., & Artieda, J. (2004). Frontal and central oscillatory changes related to different aspects of the motor process: a study in go/no-go paradigms. *Experimental Brain Research*, 159(1), 14–22.
- Amzica, F., & Lopes da Silva, F. H. (2011). Cellular substrates of brain rhythms. In D. L. Schomer & F. H. Lopes da Silva (Eds.), *Niedermeyer's Electroencephalography: Basic Principles, Clinical Applications and Related Fields.* (6th ed., pp. 33–63). Philadelphia: Lippincott Williams & Wilkins.
- Angelakis, E., Stathopoulou, S., Frymiare, J. L., Green, D. L., Lubar, J. F., & Kounios, J. (2007). EEG neurofeedback: a brief overview and an example of peak alpha frequency training for cognitive enhancement in the elderly. *The Clinical Neuropsychologist*, 21(1), 110–129.
- Arns, M., Conners, C. K., & Kraemer, H. C. (2013). A Decade of EEG Theta/Beta Ratio Research in ADHD: A Meta-Analysis. *Journal of Attention Disorders*, 17(5), 374– 383.
- Arns, M., de Ridder, S., Strehl, U., Breteler, M., & Coenen, A. (2009). Efficacy of neurofeedback treatment in ADHD: the effects on inattention, impulsivity and hyperactivity: a meta-analysis. *Clinical EEG and Neuroscience*, 40(3), 180–189.
- Arns, M., Drinkenburg, W., & Kenemans, J. (2012). The effects of QEEG-informed neurofeedback in ADHD: an open-label pilot study. *Applied Psychophysiology and Biofeedback*, 37(3), 171–180.
- Arns, M., Heinrich, H., & Strehl, U. (2014). Evaluation of neurofeedback in ADHD: the long and winding road. *Biological Psychology*, 95, 108–115.
- Aston-Jones, G., & Cohen, J. D. (2005). Adaptive gain and the role of the locus coeruleusnorepinephrine system in optimal performance. *The Journal of Comparative Neurology*, 493(1), 99–110.
- Azizian, A., Freitas, A. L., Watson, T. D., & Squires, N. K. (2006). Electrophysiological correlates of categorization: P300 amplitude as index of target similarity. *Biological Psychology*, 71(3), 278–288.
- Bagdasaryan, J., & Quyen, M. L. Van. (2013). Experiencing your brain: neurofeedback as a new bridge between neuroscience and phenomenology. *Frontiers in Human Neuroscience*, 7, 680.
- Baker, S. N. (2007). Oscillatory interactions between sensorimotor cortex and the periphery. *Current Opinion in Neurobiology*, 17(6), 649–55.
- Bakhshayesh, A. R., Hänsch, S., Wyschkon, A., Rezai, M. J., & Esser, G. (2011). Neurofeedback in ADHD: a single-blind randomized controlled trial. *European Child* & Adolescent Psychiatry, 20(9), 481–491.
- Bari, A., & Robbins, T. W. (2013). Inhibition and impulsivity: Behavioral and neural basis of response control. *Progress in Neurobiology*, *108*, 44–79.
- Barry, R. J., Clarke, A. R., & Johnstone, S. J. (2003). A review of electrophysiology in attention-deficit/hyperactivity disorder: I. Qualitative and quantitative electroencephalography, *114*, 171–183.

- Barry, R. J., & De Blasio, F. M. (2012). EEG-ERP phase dynamics of children in the auditory Go/NoGo task. *International Journal of Psychophysiology*, 86(3), 251–261.
- Barry, R. J., De Blasio, F. M., De Pascalis, V., & Karamacoska, D. (2014). Preferred EEG brain states at stimulus onset in a fixed interstimulus interval equiprobable auditory Go/NoGo task: A definitive study. *International Journal of Psychophysiology*, 94(1), 42–58.
- Barry, R. J., Johnstone, S. J., & Clarke, A. R. (2003). A review of electrophysiology in attention-deficit/hyperactivity disorder: II. Event-related potentials. *Clinical Neurophysiology*, 114(2), 184–198.
- Barry, R. J., Rushby, J. a, Smith, J. L., Clarke, A. R., Croft, R. J., & Wallace, M. J. (2007). Brain dynamics in the active vs. passive auditory oddball task: exploration of narrowband EEG phase effects. *Clinical Neurophysiology*, 118(10), 2234–2247.
- Başar, E., Başar-Eroglu, C., Karakaş, S., & Schürmann, M. (2000). Gamma, alpha, delta, and theta oscillations govern cognitive processes. *International Journal of Psychophysiology*, *39*(2-3), 241–248.
- Başar-Eroglu, C., Başar, E., Demiralp, T., & Schürmann, M. (1992). P300-response: possible psychophysiological correlates in delta and theta frequency channels. A review. *International Journal of Psychophysiology*, *13*(2), 161–179.
- Bekisz, M., & Wróbel, A. (1999). Coupling of beta and gamma activity in corticothalamic system of cats attending to visual stimuli. *Neuroreport*, 10(17), 3589–3594.
- Becerra, J., Fernández, T., Roca-Stappung, M., Díaz-Comas, L., Galán, L., Bosch, J., Harmony, T. (2012). Neurofeedback in healthy elderly human subjects with electroencephalographic risk for cognitive disorder. *Journal of Alzheimer's Disease*, 28(2), 357–367.
- Bellgrove, M. a., Hester, R., & Garavan, H. (2004). The functional neuroanatomical correlates of response variability: Evidence from a response inhibition task. *Neuropsychologia*, 42(14), 1910–1916.
- Besserve, M., Philippe, M., Florence, G., Laurent, F., Garnero, L., & Martinerie, J. (2008). Prediction of performance level during a cognitive task from ongoing EEG oscillatory activities. *Clinical Neurophysiology*, 119(4), 897–908.
- Bekisz, M., & Wróbel, A. (2003). Attention-dependent coupling between beta activities recorded in the cat's thalamic and cortical representations of the central visual field. *European Journal of Neuroscience*, 17(2), 421–426.
- Birbaumer, N. (1977). Operant enhancement of EEG-theta activity: Aspiration and reality. In J. Beatty & H. Legewie (Eds.), *Biofeedback and Behavior* (pp. 135–146). Plenum Press, New York.
- Birbaumer, N., Ruiz, S., & Sitaram, R. (2013). Learned regulation of brain metabolism. *Trends in Cognitive Sciences*, 17(6), 295–302.
- Bogacz, R., Wagenmakers, E. J., Forstmann, B. U., & Nieuwenhuis, S. (2010). The neural basis of the speed-accuracy tradeoff. *Trends in Neurosciences*, *33*(1), 10–16.
- Botvinick, M. M. (2007). Conflict monitoring and decision making: reconciling two perspectives on anterior cingulate function. *Cognitive, Affective & Behavioral Neuroscience*, 7(4), 356–366.
- Botvinick, M. M., Cohen, J. D., & Carter, C. S. (2004). Conflict monitoring and anterior cingulate cortex: an update. *Trends in Cognitive Sciences*, 8(12), 539–546.
- Boulinguez, P., Ballanger, B., Granjon, L., & Benraiss, A. (2009). The paradoxical effect of warning on reaction time: Demonstrating proactive response inhibition with event-

related potentials. Clinical Neurophysiology, 120(4), 730-737.

- Braboszcz, C., & Delorme, A. (2011). Lost in thoughts: neural markers of low alertness during mind wandering. *NeuroImage*, 54(4), 3040–3047.
- Brickenkamp, R., & Zillmer, E. (1988). d2 Test of Attention. Göttingen, Germany: Hogrefe & Hube.
- Brismar, T. (2007). The human EEG--physiological and clinical studies. *Physiology & Behavior*, 92(1-2), 141–7.
- Bruin, K. J., Kenemans, J. L., Verbaten, M. N., & Van der Heijden, A. H. (2000). Habituation: an event-related potential and dipole source analysis study. *International Journal of Psychophysiology*, 36(3), 199–209.
- Bruin, K. J., & Wijers, A. A. (2002). Inhibition, response mode, and stimulus probability: A comparative event-related potential study. *Clinical Neurophysiology*, *113*(7), 1172–1182.
- Bruin, K. J., Wijers, A. A., & Van Staveren, A. S. J. (2001). Response priming in a go/nogo task: Do we have to explain the go/nogo N2 effect in terms of response activation instead of inhibition? *Clinical Neurophysiology*, *112*(9), 1660–1671.
- Buschman, T. J., & Miller, E. K. (2007). Top-down versus bottom-up control of attention in the prefrontal and posterior parietal cortices. *Science*, *315*(5820), 1860–1862.
- Buzsáki, G. C. (2006). Rhythms of the brain. New York: Oxford University Press.
- Cacace, A. T., & McFarland, D. J. (2003). Spectral dynamics of electroencephalographic activity during auditory information processing. *Hearing Research*, 176(1-2), 25–41.
- Canolty, R. T., Edwards, E., Dalal, S. S., Soltani, M., Nagarajan, S. S., Kirsch, H. E., Knight, R. T. (2006). High gamma power is phase-locked to theta oscillations in human neocortex. *Science*, 313(5793), 1626–1628.
- Canolty, R. T., & Knight, R. T. (2010). The functional role of cross-frequency coupling. *Trends in Cognitive Sciences*, 14(11), 506–515.
- Caria, A. (2016). Self-Regulation of Blood Oxygenation Level Dependent Response: Primary Effect or Epiphenomenon? *Frontiers in Neuroscience*, 10(March), 1–5.
- Cavanagh, J. F., & Frank, M. J. (2014). Frontal theta as a mechanism for cognitive control. *Trends in Cognitive Sciences*, 18(8), 414–421.
- Chen, R., Corwell, B., & Hallett, M. (1999). Modulation of motor cortex excitability by median nerve and digit stimulation. *Experimental Brain Research*, 129(1), 77–86.
- Chen, R., Yaseen, Z., Cohen, L. G., & Hallett, M. (1998). Time course of corticospinal excitability in reaction time and self-paced movements. *Annals of Neurology*, 44(3), 317–325.
- Cheyne, D. O. (2013). MEG studies of sensorimotor rhythms: A review. *Experimental Neurology*, 245, 27–39.
- Clarke, A. R., Barry, R. J., Dupuy, F. E., McCarthy, R., Selikowitz, M., & Johnstone, S. J. (2013). Excess beta activity in the EEG of children with attention-deficit/hyperactivity disorder: A disorder of arousal? *International Journal of Psychophysiology*, 89(3), 314–319.
- Clarke, A. R., Barry, R. J., McCarthy, R., & Selikowitz, M. (1998). EEG analysis in Attention-Deficit/Hyperactivity Disorder: a comparative study of two subtypes. *Psychiatry Research*, 81(1), 19–29.
- Clemente, C. D., Sterman, M. B., & Wyrwicka, W. (1964). Post-reinforcement EEG synchronization during alimentary behavior. *Electroencephalography and Clinical Neurophysiology*, 16(4), 355–365.

- Cohen, M. X., & Ranganath, C. (2007). Reinforcement learning signals predict future decisions. *The Journal of Neuroscience*, 27(2), 371–378.
- Colgin, L. L. (2013). Mechanisms and Functions of Theta Rhythms. Annual Review of Neuroscience, 295–314.
- Comerchero, M. D., & Polich, J. (1999). P3a and P3b from typical auditory and visual stimuli. *Clinical Neurophysiology*, *110*(1), 24–30.
- Corbetta, M., & Shulman, G. L. (2002). Control of goal-directed and stimulus-driven attention in the brain. *Nature Reviews. Neuroscience*, *3*(3), 201–215.
- Courchesne, E., Hillyard, S. A., & Galambos, R. (1975). Stimulus novelty, task relevance and the visual evoked potential in man. *Electroencephalography and Clinical Neurophysiology*, *39*(2), 131–143.
- Courchesne, E., Kilman, B. A., Galambos, R., & Lincoln, A. J. (1984). Autism: Processing of novel auditory information assessed by event-related brain potentials. *Electroencephalography and Clinical Neurophysiology*, *59*(3), 238–248.
- De Blasio, F. M., & Barry, R. J. (2013a). Prestimulus alpha and beta determinants of ERP responses in the Go/NoGo task. *International Journal of Psychophysiology*, 89(1), 9–17.
- De Blasio, F. M., & Barry, R. J. (2013b). Prestimulus delta and theta determinants of ERP responses in the Go/NoGo task. *International Journal of Psychophysiology*, 87(3), 279–288.
- De Gennaro, L., Ferrara, M., & Bertini, M. (2001). The boundary between wakefulness and sleep: quantitative electroencephalographic changes during the sleep onset period. *Neuroscience*, *107*(1), 1–11.
- de Zambotti, M., Bianchin, M., Magazzini, L., Gnesato, G., & Angrilli, A. (2012). The efficacy of EEG neurofeedback aimed at enhancing sensory-motor rhythm theta ratio in healthy subjects. *Experimental Brain Research*. 221(1), 69–74.
- Debener, S., Kranczioch, C., Herrmann, C. S., & Engel, A. K. (2002). Auditory novelty oddball allows reliable distinction of top-down and bottom-up processes of attention. *International Journal of Psychophysiology*, *46*(1), 77–84.
- Debener, S., Makeig, S., Delorme, A., & Engel, A. K. (2005). What is novel in the novelty oddball paradigm? Functional significance of the novelty P3 event-related potential as revealed by independent component analysis. *Brain Research. Cognitive Brain Research*, 22(3), 309–321.
- deCharms, R. C. (2008). Applications of real-time fMRI. *Nature Reviews. Neuroscience*, 9(9), 720–729.
- Delorme, A., & Makeig, S. (2004). EEGLAB: an open source toolbox for analysis of single-trial EEG dynamics including independent component analysis. *Journal of Neuroscience Methods*, 134(1), 9–21.
- Demiralp, T., Ademoglu, A., Comerchero, M., & Polich, J. (2001). Wavelet Analysis of P3a and P3b. *Brain Topography*, *13*(4), 251–267.
- Dempster, T., & Vernon, D. (2009). Identifying indices of learning for alpha neurofeedback training. *Applied Psychophysiology and Biofeedback*, 34(4), 309–328.
- Desimone, R., & Duncan, J. (1995). Neural mechanisms of selective visual attention. Annual Review of Neuroscience, 18, 193–222.
- Dien, J., Spencer, K. M., & Donchin, E. (2003). Localization of the event-related potential novelty response as defined by principal components analysis. *Psychophysiology*, 17, 637–650.

- Dien, J., Spencer, K. M., & Donchin, E. (2004). Parsing the late positive complex: mental chronometry and the ERP components that inhabit the neighborhood of the P300. *Psychophysiology*, *41*(5), 665–678.
- Dimoska, A., Johnstone, S. J., & Barry, R. J. (2006). The auditory-evoked N2 and P3 components in the stop-signal task: Indices of inhibition, response-conflict or error-detection? *Brain and Cognition*, 62(2), 98–112.
- Doebler, P., & Scheffler, B. (in press). The relationship of choice reaction time variability and intelligence: A meta-analysis. *Learning and Individual Differences*.
- Donchin, E., & Coles, M. G. H. (1998). Context updating and the p300. *Behavioral and Brain Sciences*, 21(1), 152–154.
- Donkers, F. C. L., & van Boxtel, G. J. M. (2004). The N2 in go/no-go tasks reflects conflict monitoring not response inhibition. *Brain and Cognition*, *56*(2), 165–176.
- Doppelmayr, M. M., Klimesch, W., Pachinger, T., & Ripper, B. (1998). The functional significance of absolute power with respect to event-related desynchronization. *Brain Topography*, *11*(2), 133–140.
- Doppelmayr, M., & Weber, E. (2011). Effects of SMR and Theta/Beta Neurofeedback on Reaction Times, Spatial Abilities, and Creativity. *Journal of Neurotherapy*, 15(2), 115–129.
- Duric, N. S., Assmus, J., Gundersen, D., & Elgen, I. B. (2012). Neurofeedback for the treatment of children and adolescents with ADHD: a randomized and controlled clinical trial using parental reports. *BMC Psychiatry*, *12*(1), 107.
- Egner, T., & Gruzelier, J. H. (2001). Learned self-regulation of EEG frequency components affects attention and event-related brain potentials in humans. *Neuroreport*, *12*(18), 4155–4169.
- Egner, T., & Gruzelier, J. H. (2004). EEG Biofeedback of low beta band components: frequency-specific effects on variables of attention and event-related brain potentials. *Clinical Neurophysiology*, *115*(1), 131–139.
- Egner, T., Zech, T. F., & Gruzelier, J. H. (2004). The effects of neurofeedback training on the spectral topography of the electroencephalogram. *Clinical Neurophysiology*, *115*(11), 2452–2460.
- Eimer, M. (1993). Effects of attention and stimulus probability on ERPs in a Go/Nogo task. *Biological Psychology*, 35(2), 123–138.
- Engel, A. K., & Fries, P. (2010). Beta-band oscillations--signalling the status quo? *Current Opinion in Neurobiology*, 20(2), 156–165.
- Engel, A. K., Fries, P., & Singer, W. (2001). Dynamic predictions: oscillations and synchrony in top-down processing. *Nature Reviews. Neuroscience*, 2(10), 704–716.
- Enriquez-Geppert, S., Huster, R. J., Figge, C., & Herrmann, C. S. (2014). Self-regulation of frontal-midline theta facilitates memory updating and mental set shifting. *Frontiers in Behavioral Neuroscience*, *8*, 1–13.
- Enriquez-Geppert, S., Huster, R. J., Scharfenort, R., Mokom, Z. N., Zimmermann, J., & Herrmann, C. S. (2014). Modulation of frontal-midline theta by neurofeedback. *Biological Psychology*, 95, 59–69.
- Escera, C., Alho, K., Winkler, I., & Naatanen, R. (1998). Neural Mechanisms of Involuntary Attention. *Journal of Cognitive Neuroscience*, 10(5), 590–604.
- Escera, C., Alho, K., Schröger, E., & Winkler, I. (2000). Involuntary attention and distractibility as evaluated with event-related brain potentials. *Audiology & Neuro-Otology*, *5*(3-4), 151–166.

- Escolano, C., Aguilar, M., & Minguez, J. (2011). EEG-based upper alpha neurofeedback training improves working memory performance. *Conference Proceedings: Annual International Conference of the IEEE Engineering in Medicine and Biology Society. IEEE Engineering in Medicine and Biology Society. Conference*, 2011(iii), 2327–2330.
- Falkenstein, M., Hohnsbein, J., Hoormann, J., & Blanke, L. (1991). Effects of crossmodal divided attention on late ERP components. II. Error processing in choice reaction tasks. *Electroencephalography and Clinical Neurophysiology*, 78(6), 447–455.
- Falkenstein, M., Hoormann, J., & Hohnsbein, J. (1999). ERP components in Go/Nogo tasks and their relation to inhibition. *Acta Psychologica*, *101*(2-3), 267–291.
- Fell, J., Dietl, T., Grunwald, T., Kurthen, M., Klaver, P., Trautner, P., Fernández, G. (2004). Neural bases of cognitive ERPs: more than phase reset. *Journal of Cognitive Neuroscience*, 16(9), 1595–1604.
- Fell, J., Fernández, G., Klaver, P., Elger, C. E., & Fries, P. (2003). Is synchronized neuronal gamma activity relevant for selective attention? *Brain Research. Brain Research Reviews*, 42(3), 265–272.
- Fetz, E. E. (1969). Operant Conditioning of Cortical Unit Activity. *Science*, 163(3870), 955–958.
- Friedman, D., Cycowicz, Y. M., & Gaeta, H. (2001). The novelty P3: an event-related brain potential (ERP) sign of the brain's evaluation of novelty. *Neuroscience & Biobehavioral Reviews*, 25(4), 355–373.
- Fries, P. (2005). A mechanism for cognitive dynamics: neuronal communication through neuronal coherence. *Trends in Cognitive Sciences*, 9(10), 474–480.
- Fries, P. (2015). Rhythms for Cognition: Communication through Coherence. *Neuron*, 88(1), 220–235.
- Fuchs, T., Birbaumer, N., Lutzenberger, W., Gruzelier, J. H., & Kaiser, J. (2003). Neurofeedback treatment for attention-deficit/hyperactivity disorder in children: a comparison with methylphenidate. *Applied Psychophysiology and Biofeedback*, 28(1), 1–12.
- Fuentemilla, L., Marco-Pallarés, J., & Grau, C. (2006). Modulation of spectral power and of phase resetting of EEG contributes differentially to the generation of auditory event-related potentials. *NeuroImage*, *30*(3), 909–916.
- Fuentemilla, L., Marco-Pallarés, J., Münte, T. F., & Grau, C. (2008). Theta EEG oscillatory activity and auditory change detection. *Brain Research*, *1220*, 93–101.
- Gaetz, W., Edgar, J. C., Wang, D. J., & Roberts, T. P. L. (2011). Relating MEG measured motor cortical oscillations to resting γ-aminobutyric acid (GABA) concentration. *NeuroImage*, 55(2), 616–621.
- García-Larrea, L., & Cézanne-Bert, G. (1998). P3, Positive slow wave and working memory load: a study on the functional correlates of slow wave activity. *Electroencephalography and Clinical Neurophysiology*, *108*, 260–273.
- Gehring, W. J., Goss, B., Coles, M. . G. H., Meyer, D. E., & Donchin, E. (1993). A neural system for error detection and compensation. *Psychological Science*, 4(6), 385–390.
- Gevensleben, H., Holl, B., Albrecht, B., Schlamp, D., Kratz, O., Studer, P., Heinrich, H. (2009). Distinct EEG effects related to neurofeedback training in children with ADHD: a randomized controlled trial. *International Journal of Psychophysiology*, 74(2), 149–157.

- Gevensleben, H., Holl, B., Albrecht, B., Vogel, C., Schlamp, D., Kratz, O., Heinrich, H. (2009). Is neurofeedback an efficacious treatment for ADHD? A randomised controlled clinical trial. *Journal of Child Psychology and Psychiatry*, 50(7), 780–789.
- Gevins, A., Smith, M. E., McEvoy, L., & Yu, D. (1997). High-resolution EEG mapping of cortical activation related to working memory: effects of task difficulty, type of processing, and practice. *Cerebral Cortex*, 7(4), 374–385.
- Ghoshuni, M., Firoozabadi, M., Khalilzadeh, M. A., Reza, M., & Golpayegani, H. (2012). The Effect of Sensorimotor Rhythm Enhancing Neurofeedback on Power of Adjacent Frequency Bands. *Biomedical Engineering: Applications, Basis and Communications*, 24(4), 307–312.
- Gilbertson, T., Lalo, E., Doyle, L., Di Lazzaro, V., Cioni, B., & Brown, P. (2005). Existing motor state is favored at the expense of new movement during 13-35 Hz oscillatory synchrony in the human corticospinal system. *The Journal of Neuroscience*, 25(34), 7771–7779.
- Gola, M., Kamiński, J., Brzezicka, A., & Wróbel, A. (2012). β band oscillations as a correlate of alertness--changes in aging. *International Journal of Psychophysiology*, 85(1), 62–7.
- Goldstein, A., Spencer, K. M., & Donchin, E. (2002). The influence of stimulus deviance and novelty on the P300 and Novelty P3. *Psychophysiology*, *39*, 781–790.
- Grabska-Barwińska, A., & Zygierewicz, J. (2006). A model of event-related EEG synchronization changes in beta and gamma frequency bands. *Journal of Theoretical Biology*, 238(4), 901–913.
- Gratton, G., Coles, M. G., & Donchin, E. (1983). A new method for off-line removal of ocular artifact. *Electroencephalography and Clinical Neurophysiology*, 55(4), 468–484.
- Greenberg, L. M., Kindschi, C. L., & Corman, C. L. (1996). TOVA test of variables of attention: clinical guide.
- Groom, M. J., & Cragg, L. (2015). Differential modulation of the N2 and P3 event-related potentials by response conflict and inhibition. *Brain and Cognition*, *97*, 1–9.
- Gross, J., Schmitz, F., Schnitzler, I., Kessler, K., Shapiro, K., Hommel, B., & Schnitzler, A. (2004). Modulation of long-range neural synchrony reflects temporal limitations of visual attention in humans. *Proceedings of the National Academy of Sciences of the United States of America*, 101(35), 13050–13055.
- Gruber, W. R., Zauner, A., Lechinger, J., Schabus, M., Kutil, R., & Klimesch, W. (2014). Alpha phase, temporal attention, and the generation of early event related potentials. *NeuroImage*, 103, 119–129.
- Gruzelier, J. H. (2014a). EEG-neurofeedback for optimising performance III: A review of methodological and theoretical considerations. *Neuroscience & Biobehavioral Reviews*, 44, 159–182.
- Gruzelier, J. H. (2014b). EEG-neurofeedback for optimising performance. I: A review of cognitive and affective outcome in healthy participants. *Neuroscience and Biobehavioral Reviews*, 44, 124–141.
- Gruzelier, J. H., Foks, M., Steffert, T., Chen, M. J.-L., & Ros, T. (2014). Beneficial outcome from EEG-neurofeedback on creative music performance, attention and wellbeing in school children. *Biological Psychology*, 95, 86–95.

- Gruzelier, J. H., Hirst, L., Holmes, P., & Leach, J. (2014). Immediate effects of Alpha/theta and Sensory-Motor Rhythm feedback on music performance. *International Journal of Psychophysiology*, 93(1), 96–104.
- Güntekin, B., & Başar, E. (2010). A new interpretation of P300 responses upon analysis of coherences. *Cognitive Neurodynamics*, 4(2), 107–118.
- Gusnard, D. A., Raichle, M. E., & Raichle, M. E. (2001). Searching for a baseline: functional imaging and the resting human brain. *Nature Reviews. Neuroscience*, 2(October), 685–694.
- Hagen, G. F., Gatherwright, J. R., Lopez, B. a, & Polich, J. (2006). P3a from visual stimuli: task difficulty effects. *International Journal of Psychophysiology*, 59(1), 8–14.
- Hanslmayr, S., Aslan, A., Staudigl, T., Klimesch, W., Herrmann, C. S., & Bäuml, K.-H. (2007). Prestimulus oscillations predict visual perception performance between and within subjects. *NeuroImage*, 37(4), 1465–1473.
- Hanslmayr, S., Klimesch, W., Sauseng, P., Gruber, W., Doppelmayr, M., Freunberger, R., Birbaumer, N. (2007). Alpha Phase Reset Contributes to the Generation of ERPs. *Cerebral Cortex*, 17, 1–8.
- Hanslmayr, S., Sauseng, P., Doppelmayr, M., Schabus, M., & Klimesch, W. (2005). Increasing Individual Upper Alpha Power by Neurofeedback Improves Cognitive Performance in Human Subjects. *Applied Psychophysiology and Biofeedback*, 30(1), 1–10.
- Harper, J., Malone, S. M., & Bernat, E. M. (2014). Theta and delta band activity explain N2 and P3 ERP component activity in a go/no-go task. *Clinical Neurophysiology*, *125*(1), 124–132.
- Harper, R. M., & Sterman, M. B. (1972). Subcortical unit activity during a conditioned 12– 14 Hz sensorimotor EEG rhythm in the cat. *Federation Proceedings*, *31*, 404.
- Herrmann, C. S., Rach, S., Neuling, T., & Strüber, D. (2013). Transcranial alternating current stimulation: a review of the underlying mechanisms and modulation of cognitive processes. *Frontiers in Computational Neuroscience*, 7, 1–13.
- Hillyard, S. A., & Kutas, M. (1983). Electrophysiology of cognitive processing. *Annual Review of Psychology*, 34(Haber 1974), 33-61.
- Hillyard, S. A., Squires, K. C., Bauer, J. W., & Lindsay, P. H. (1971). Evoked potential correlates of auditory signal detection. *Science*, *172*(3990), 1357–1360.
- Hoedlmoser, K., Pecherstorfer, T., Gruber, G., Anderer, P., Doppelmayr, M., & Klimesch,
 W. (2008). Instrumental Conditioning of Human Sensorimotor Rhythm (12-15 Hz) and Its Impact on Sleep as Well as Declarative Learning. *Sleep*, *31*, 1401–1408.
- Holz, E. M., Glennon, M., Prendergast, K., & Sauseng, P. (2010). Theta-gamma phase synchronization during memory matching in visual working memory. *NeuroImage*, 52(1), 326–335.
- Howe, R. C., & Sterman, M. B. (1972). Cortical-subcortical EEG correlates of suppressed motor behavior during sleep and waking in the cat. *Electroencephalography and Clinical Neurophysiology*, *32*, 681–695.
- Howells, F. M., Stein, D. J., & Russell, V. a. (2010). Perceived mental effort correlates with changes in tonic arousal during attentional tasks. *Behavioral and Brain Functions*, *6*, 39.
- Huang, R.-S., Jung, T.-P., Delorme, A., & Makeig, S. (2008). Tonic and phasic electroencephalographic dynamics during continuous compensatory tracking. *NeuroImage*, *39*(4), 1896–1909.

- Hultsch, D. F., MacDonald, S. W. S., & Dixon, R. a. (2002). Variability in reaction time performance of younger and older adults. *The Journals of Gerontology. Series B, Psychological Sciences and Social Sciences*, 57(2), 101–115.
- Ingvar, D. H., Sjölund, B., & Ardö, A. (1976). Correlation between dominant EEG frequency, cerebral oxygen uptake and blood flow. *Electroencephalography and Clinical Neurophysiology*, 41(3), 268–276.
- Intriligator, J., & Polich, J. (1994). On the relationship between background EEG and the P300 event-related potential. *Biological Psychology*, *37*(3), 207–218.
- Isreal, J. B., Chesney, G. L., Wickens, C. D., & Donchin, E. (1980). P300 and tracking difficulty: evidence for multiple resources in dual-task performance. *Psychophysiology*, *17*(3), 259–273.
- Iversen, J. R., Repp, B. H., & Patel, A. D. (2009). Top-down control of rhythm perception modulates early auditory responses. Annals of the New York Academy of Sciences, 1169, 58–73.
- Jasiukaitis, P., & Hakerem, G. (1988). The Effect of Prestimulus Alpha Activity on the P300. *Psychophysiology*, 25(2), 157–165.
- Jenkinson, N., & Brown, P. (2011). New insights into the relationship between dopamine, beta oscillations and motor function. *Trends in Neurosciences*, *34*(12), 611–618.
- Jensen, O., Goel, P., Kopell, N., Pohja, M., Hari, R., & Ermentrout, B. (2005). On the human sensorimotor-cortex beta rhythm: Sources and modeling. *NeuroImage*, 26, 347–355.
- Joundi, R. A., Jenkinson, N., Brittain, J.-S., Aziz, T. Z., & Brown, P. (2012). Driving Oscillatory Activity in the Human Cortex Enhances Motor Performance. *Current Biology*, 22(5), 403–407.
- Jung, T.-P., Makeig, S., Westerfield, M., Townsend, J., Courchesne, E., & Sejnowski, T. J. (2001). Analysis and visualization of single-trial event-related potentials. *Human Brain Mapping*, 14(3), 166–185.
- Kahana, M. J., Seelig, D., & Madsen, J. R. (2001). Theta returns. Current Opinion in Neurobiology, 11(6), 739–744.
- Kamp, S.-M., Brumback, T., & Donchin, E. (2013). The component structure of ERP subsequent memory effects in the Von Restorff paradigm and the word frequency effect in recall. *Psychophysiology*, *50*(11), 1079–1093.
- Karalunas, S. L., Geurts, H. M., Konrad, K., Bender, S., & Nigg, J. T. (2014). Annual Research Review: Reaction time variability in ADHD and autism spectrum disorders: measurement and mechanisms of a proposed trans-diagnostic phenotype. *Journal of Child Psychology and Psychiatry*, 55(6), 685–710.
- Katayama, J., & Polich, J. (1999). Auditory and visual P300 topography from a 3 stimulus paradigm. *Clinical Neurophysiology : Official Journal of the International Federation of Clinical Neurophysiology*, 110(3), 463–468.
- Keizer, A. W., Verment, R. S., & Hommel, B. (2010). Enhancing cognitive control through neurofeedback: a role of gamma-band activity in managing episodic retrieval. *NeuroImage*, 49(4), 3404–3413.
- Keizer, A. W., Verschoor, M., Verment, R. S., & Hommel, B. (2010). The effect of gamma enhancing neurofeedback on the control of feature bindings and intelligence measures. *International Journal of Psychophysiology*, 75(1), 25–32.

- Kelly, S. P., & O'Connell, R. G. (2014). The neural processes underlying perceptual decision making in humans: recent progress and future directions. *Journal of Physiology, Paris*, 109(1-3), 27-37.
- Kilavik, B. E., Zaepffel, M., Brovelli, A., MacKay, W. A., & Riehle, A. (2013). The ups and downs of beta oscillations in sensorimotor cortex. *Experimental Neurology*, 245, 15–26.
- Kirmizi-Alsan, E., Bayraktaroglu, Z., Gurvit, H., Keskin, Y. H., Emre, M., & Demiralp, T. (2006). Comparative analysis of event-related potentials during Go/NoGo and CPT: Decomposition of electrophysiological markers of response inhibition and sustained attention. *Brain Research*, 1104(1), 114–128.
- Klimesch, W. (1999). EEG alpha and theta oscillations reflect cognitive and memory performance: a review and analysis. *Brain Research. Brain Research Reviews*, 29(2-3), 169–195.
- Klimesch, W., Doppelmayr, M., Pachinger, T., & Russegger, H. (1997). Event-related desynchronization in the alpha band and the processing of semantic information. *Cognitive Brain Research*, 6(2), 83–94.
- Klimesch, W., Sauseng, P., Hanslmayr, S., Gruber, W., & Freunberger, R. (2007). Eventrelated phase reorganization may explain evoked neural dynamics. *Neuroscience and Biobehavioral Reviews*, *31*(7), 1003–1016.
- Klimesch, W., Schack, B., Schabus, M., Doppelmayr, M., Gruber, W., & Sauseng, P. (2004). Phase-locked alpha and theta oscillations generate the P1-N1 complex and are related to memory performance. *Brain Research. Cognitive Brain Research*, *19*(3), 302–316.
- Knyazev, G. G. (2012). EEG delta oscillations as a correlate of basic homeostatic and motivational processes. *Neuroscience and Biobehavioral Reviews*, *36*(1), 677–695.
- Knye, M., Roth, N., Westhus, W., & Heine, A. (2004). Continuous Performance Task (CPT)-manual. *Hogrefe, Gottingen*.
- Kober, S. E., Witte, M., Stangl, M., Väljamäe, A., Neuper, C., & Wood, G. (2015). Shutting down sensorimotor interference unblocks the networks for stimulus processing: An SMR neurofeedback training study. *Clinical Neurophysiology*, 126(1), 82–95.
- Koch, G., Franca, M., Fernandez Del Olmo, M., Cheeran, B., Milton, R., Alvarez Sauco, M., & Rothwell, J. C. (2006). Time Course of Functional Connectivity between Dorsal Premotor and Contralateral Motor Cortex during Movement Selection. *Journal of Neuroscience*, 26(28), 7452–7459.
- Kok, A. (2001). On the utility of P3 amplitude as a measure of processing capacity. *Psychophysiology*, 38(3), 557–77.
- Kok, A., Ramautar, J. R., De Ruiter, M. B., Band, G. P. H., & Ridderinkhof, K. R. (2004). ERP components associated with successful and unsuccessful stopping in a stop-signal task. *Psychophysiology*, *41*(1), 9–20.
- Kopp, B., Mattler, U., Goertz, R., & Tist, F. (1996). N2, P3 and the lateralized readiness potential in a nogo task involving selective response inhibition. *Electroencephalography and Clinical Neurophysiology*, *99*, 19–27.
- Kramer, A. F., Wickens, C. D., & Donchin, E. (1985). Processing of stimulus properties: Evidence for dual-task integrality. *Journal of Experimental Psychology: Human Perception and Performance.*, 11, 393–408.

- Kropotov, J. D., Grin-Yatsenko, V. A., Ponomarev, V.A., Chutko, L. S., Yakovenko, E. A. A., & Nikishena, I. S. (2005). ERPs correlates of EEG relative beta training in ADHD children. *International Journal of Psychophysiology*, 55(1), 23–34.
- Kühn, A. a, Williams, D., Kupsch, A., Limousin, P., Hariz, M., Schneider, G.-H., ... Brown, P. (2004). Event-related beta desynchronization in human subthalamic nucleus correlates with motor performance. *Brain : A Journal of Neurology*, 127(Pt 4), 735– 46.
- Kutas, M., Mccarthy, G., & Donchin, E. (1977). Augmenting Mental Chronometry: The P300 as a Measure of Stimulus Evaluation Time. *Science*, *197*(4305), 792–795.
- Lee, J. H., Whittington, M. A., & Kopell, N. J. (2013). Top-Down Beta Rhythms Support Selective Attention via Interlaminar Interaction: A Model. *PLoS Computational Biology*, 9(8), e1003164.
- Lévesque, J., Beauregard, M., & Mensour, B. (2006). Effect of neurofeedback training on the neural substrates of selective attention in children with attentiondeficit/hyperactivity disorder: a functional magnetic resonance imaging study. *Neuroscience Letters*, 394(3), 216–221.
- Liotti, M., Pliszka, S. R., Higgins, K., Perez, R., & Semrud-Clikeman, M. (2010). Evidence for specificity of ERP abnormalities during response inhibition in ADHD children: A comparison with reading disorder children without ADHD. *Brain and Cognition*, 72(2), 228–237.
- Lopes da Silva, F. (2013). EEG and MEG: Relevance to neuroscience. *Neuron*, 80(5), 1112–1128.
- Lopes da Silva, F. H. (2011). Biophysical Aspects of EEG an Magnetoencephalogram Generation. In D. L. Schomer & F. H. Lopes da Silva (Eds.), *Niedermeyer's Electroencephalography: Basic Principles, Clinical Applications and Related Fields.* (6th ed., pp. 91–110). Philadelphia: Lippincott Williams & Wilkins.
- Lubar, J. F. (1997). Neocortical dynamics: implications for understanding the role of neurofeedback and related techniques for the enhancement of attention. *Applied Psychophysiology and Biofeedback*, 22(2), 111–126.
- Lubar, J. F., & Bahler, W. W. (1976). Behavioral management of epileptic seizures following EEG biofeedback training of the sensorimotor rhythm. *Biofeedback and Self-Regulation*, 1(1), 77–104.
- Lubar, J. F., & Shouse, M. N. (1976). EEG and behavioral changes in a hyperkinetic child concurrent with training of the sensorimotor rhythm (SMR): a preliminary report. *Biofeedback and Self-Regulation*, 1(3), 293–306.
- Lubar, J. F., Swartwood, M. O., Swartwood, J. N., & O'Donnell, P. H. (1995). Evaluation of the effectiveness of EEG neurofeedback training for ADHD in a clinical setting as measured by changes in T.O.V.A. scores, behavioral ratings, and WISC-R performance. *Biofeedback and Self-Regulation*, 20(1), 83–99.
- Lubar, J. O., & Lubar, J. F. (1984). Electroencephalographic biofeedback of SMR and beta for treatment of attention deficit disorders in a clinical setting. *Biofeedback and Self-Regulation*, 9(1), 1–23.
- Luck, S. J. (2014). An introduction to event-related potential technique. (2nd ed.). MIT Press.
- Luft, C. D. B., Nolte, G., & Bhattacharya, J. (2013). High-learners present larger midfrontal theta power and connectivity in response to incorrect performance feedback. *The Journal of Neuroscience*, 33(5), 2029–2038.

- Luu, P., & Tucker, D. M. (2003). Executive Functions: Electrophysiological Clues. In A. Zani & A. . Preverbio (Eds.), *The Cognitive Electrophysiology of Mind and Brain* (pp. 199–223). San Diego: Academic Press.
- Luu, P., Tucker, D. M., & Makeig, S. (2004). Frontal midline theta and the error-related negativity: neurophysiological mechanisms of action regulation. *Clinical Neurophysiology*, 115(8), 1821–1835.
- Macmillan, N. A., & Creelman, C. D. (2005). *Detection Theory: A User's Guide*. Lawrence Erlbaum Associates.
- Magliero, A., Bashore, T., Coles, M. G. H., & Donchin, E. (1984). On the dependence of P300 latency on stimulus evaluation processes. *Psychophysiology*, 21, 171–186.
- Makeig, S., Debener, S., Onton, J., & Delorme, A. (2004). Mining event-related brain dynamics. *Trends in Cognitive Sciences*, 8(5), 204–210.
- Makeig, S., & Jung, T.-P. (1996). Tonic, phasic and transient EEG correlates of auditory awareness in drowsiness. *Cognitive Brain Research*, *4*, 15–25.
- Makeig, S., Westerfield, M., Jung, T. P., Enghoff, S., Townsend, J., Courchesne, E., & Sejnowski, T. J. (2002). Dynamic brain sources of visual evoked responses. *Science*, 295(5555), 690–694.
- Maltez, J., Hyllienmark, L., Nikulin, V. V, & Brismar, T. (2004). Time course and variability of power in different frequency bands of EEG during resting conditions. *Clinical Neurophysiology*, *34*(5), 195–202.
- Mann, C. A., Lubar, J. F., Zimmerman, A. W., Miller, C. A., & Muenchen, R. A. (1992). Quantitative analysis of EEG in boys with attention-deficit-hyperactivity disorder: controlled study with clinical implications. *Pediatric Neurology*, 8(1), 30–36.
- Mann, C. A., Sterman, M. B., & Kaiser, D. A. (1996). Suppression of EEG rhythmic frequencies during somato-motor and visuo-motor behavior. *International Journal of Psychophysiology*, 23(1), 1–7.
- Maoz, H., Aviram, S., Nitzan, U., Segev, A., & Bloch, Y. (2015). Association Between Continuous Performance and Response Inhibition Tests in Adults With ADHD. *Journal of Attention Disorders*, 28.
- Mazaheri, A., & Jensen, O. (2010). Rhythmic pulsing: linking ongoing brain activity with evoked responses. *Frontiers in Human Neuroscience*, *4*, 177.
- McLoughlin, G., Palmer, J. a, Rijsdijk, F., & Makeig, S. (2014). Genetic overlap between evoked frontocentral theta-band phase variability, reaction time variability, and attention-deficit/hyperactivity disorder symptoms in a twin study. *Biological Psychiatry*, 75(3), 238–47.
- Meisel, V., Servera, M., Garcia-Banda, G., Cardo, E., & Moreno, I. (2013). Neurofeedback and standard pharmacological intervention in ADHD: a randomized controlled trial with six-month follow-up. *Biological Psychology*, *94*(1), 12–21.
- Menon, V. (2011). Large-scale brain networks and psychopathology: a unifying triple network model. *Trends in Cognitive Sciences*, 15(10), 483–506.
- Min, B. K., Busch, N. A., Debener, S., Kranczioch, C., Hanslmayr, S., Engel, A. K., & Herrmann, C. S. (2007). The best of both worlds: Phase-reset of human EEG alpha activity and additive power contribute to ERP generation. *International Journal of Psychophysiology*, 65(1), 58–68.
- Miniussi, C., & Thut, G. (2010). Combining TMS and EEG offers new prospects in cognitive neuroscience. *Brain Topography*, 22(4), 249–256.

- Mitchell, D. J., McNaughton, N., Flanagan, D., & Kirk, I. J. (2008). Frontal-midline theta from the perspective of hippocampal "theta". *Progress in Neurobiology*, *86*(3), 156–185.
- Monastra, V. J., Lubar, J. F., & Linden, M. (2001). The development of a quantitative electroencephalographic scanning process for attention deficit-hyperactivity disorder: reliability and validity studies. *Neuropsychology*, *15*(1), 136–144.
- Monastra, V. J., Lynn, S., Linden, M., Lubar, J. F., Gruzelier, J., & LaVaque, T. J. (2005). Electroencephalographic Biofeedback in the Treatment of Attention-Deficit/Hyperactivity Disorder. *Applied Psychophysiology and Biofeedback*, 30(2), 95–114.
- Monastra, V. J., Monastra, D. M., & George, S. (2002). The Effects of Stimulant Therapy, EEG Biofeedback, and Parenting Style on the Primary Symptoms of Attention-Deficit / Hyperactivity Disorder 1. Applied Psychophysiology and Biofeedback, 27(4), 231– 249.
- Moruzzi, G., & Magoun, H. W. (1949). Brain stem reticular formation and activation of the EEG. *Electroencephalography and Clinical Neurophysiology*, *1*(4), 455–473.
- Müller, G. R., Neuper, C., Rupp, R., Keinrath, C., Gerner, H. J., & Pfurtscheller, G. (2003). Event-related beta EEG changes during wrist movements induced by functional electrical stimulation of forearm muscles in man. *Neuroscience Letters*, *340*(2), 143– 147.
- Näätänen, R. (1992). Attention and brain function. Hillsdale, New Jersey: Erlbaum.
- Näätänen, R., & Kreegipuu, K. (2012). The Mismatch Negativity (MMN). In E. S. Kappenman & S. J. Luck (Eds.), *The Oxford Handbook of Event-Related Potential Components* (pp. 143 157). New York: Oxford University Press.
- Näätänen, R., Paavilainen, P., Rinne, T., & Alho, K. (2007). The mismatch negativity (MMN) in basic research of central auditory processing: A review. *Clinical Neurophysiology*, *118*(12), 2544–2590.
- Nakamoto, H., & Mori, S. (2008). Effects of stimulus-response compatibility in mediating expert performance in baseball players. *Brain Research*, *1189*(1), 179–188.
- Nakamoto, H., & Mori, S. (2012). Experts in fast-ball sports reduce anticipation timing cost by developing inhibitory control. *Brain and Cognition*, 80(1), 23–32.
- Nan, W., Rodrigues, J. P., Ma, J., Qu, X., Wan, F., Mak, P.-I., Rosa, A. (2012). Individual alpha neurofeedback training effect on short term memory. *International Journal of Psychophysiology*, 86(1), 83–87.
- Neuper, C., & Pfurtscheller, G. (2001). Evidence for distinct beta resonance frequencies in human EEG related to specific sensorimotor cortical areas. *Clinical Neurophysiology*, *112*(11), 2084–2097.
- Nieuwenhuis, S., Aston-Jones, G., & Cohen, J. D. (2005). Decision making, the P3, and the locus coeruleus-norepinephrine system. *Psychological Bulletin*, *131*(4), 510–532.
- Nigbur, R., Ivanova, G., & Stürmer, B. (2011). Theta power as a marker for cognitive interference. *Clinical Neurophysiology*, *122*(11), 2185–2194.
- Nigg, J. T., Willcutt, E. G., Doyle, A. E., & Sonuga-Barke, E. J. S. (2005). Causal heterogeneity in attention-deficit/hyperactivity disorder: do we need neuropsychologically impaired subtypes? *Biological Psychiatry*, *57*(11), 1224–1230.
- Nikulin, V. V, & Brismar, T. (2005). Long-range temporal correlations in electroencephalographic oscillations: Relation to topography, frequency band, age and gender. *Neuroscience*, *130*(2), 549–558.

- Nikulin, V. V, & Brismar, T. (2006). Phase synchronization between alpha and beta oscillations in the human electroencephalogram. *Neuroscience*, *137*(2), 647–657.
- Nunez, P. L. (2000). Toward a quantitative description of large-scale neocortical dynamic function and EEG. *The Behavioral and Brain Sciences*, 23(3),399–437.
- O'Halon, J. F., Royal, J. W., & Beatty, J. (1977). Theta regulation and radar vigilance performance. In J. Beatty & H. Legewie (Eds.), *Biofeedback and Behavior* (pp. 147–165). New York: Plenum Press.
- O'Hanlon, J. F., & Beatty, J. (1977). Concurrence of electroencephalographic and performance changes during a simulated radar watch and some implications for the arousal theory of vigilance. In *Vigilance* (pp. 189–201). Plenum Press, New York.
- Oades, R. D., Zerbin, D., & Dittmann-Balcar, A. (1995). The topography of event-related potentials in passive and active conditions of a 3-tone auditory oddball test. *The International Journal of Neuroscience*, 81(3-4), 249–264.
- Oostenveld, R., Fries, P., Maris, E., & Schoffelen, J.-M. (2011). FieldTrip: Open source software for advanced analysis of MEG, EEG, and invasive electrophysiological datA. *Computational Intelligence and Neuroscience*,
- Othmer, S., Othmer, S. F., & Kaiser, D. A. (1999). EEG Biofeedback: An Emerging Model for Its Global Efficacy. In J. R. Evans & A. Abarbanel (Eds.), *In Introduction to Quantitative EEG and Neurofeedback* (pp. 243–310). San Diego: Academic Press.
- Papenberg, G., Hämmerer, D., Müller, V., Lindenberger, U., & Li, S.-C. (2013). Lower theta inter-trial phase coherence during performance monitoring is related to higher reaction time variability: a lifespan study. *NeuroImage*, 83, 912–920.
- Park, J. L., Fairweather, M. M., & Donaldson, D. I. (2015). Making the case for mobile cognition: EEG and sports performance. *Neuroscience & Biobehavioral Reviews*, 52, 117–130.
- Pascual-Leone, A., Amedi, A., Fregni, F., & Merabet, L. B. (2005). The Plastic Human Brain Cortex. *Annual Review of Neuroscience*, 28(1), 377–401.
- Paus, T., Zatorre, R. J., Hofle, N., Caramanos, Z., Gotman, J., Petrides, M., & Evans, A. C. (1997). Time-related changes in neural systems underlying attention and arousal during the performance of an auditory vigilance task. *Journal of Cognitive Neuroscience*, 9(3), 392–408.
- Pfefferbaum, A., Ford, J. M., Weller, B. J., & Kopell, B. S. (1985). ERPs to response production and inhibition. *Electroencephalography and Clinical Neurophysiology*, 60(5), 423–434.
- Pfurtscheller, G., & Lopes da Silva, F. H. (1999). Event-related EEG/MEG synchronization and desynchronization: basic principles. *Clinical Neurophysiology*, *110*(11), 1842–1857.
- Pfurtscheller, G., Stancák, A., & Edlinger, G. (1997). On the existence of different types of central beta rhythms below 30 Hz. *Electroencephalography and Clinical Neurophysiology*, *102*(4), 316–325.
- Pfurtscheller, G., Stancák, A., & Neuper, C. (1996). Post-movement beta synchronization. A correlate of an idling motor area? *Electroencephalography and Clinical Neurophysiology*, 98(4), 281–293.
- Phillips, M., Rogers, P., Haworth, J., Bayer, A., & Tales, A. (2013). Intra-individual reaction time variability in mild cognitive impairment and Alzheimer's disease: gender, processing load and speed factors. *PloS One*, 8(6), e65712.

- Picazio, S., Veniero, D., Ponzo, V., Caltagirone, C., Gross, J., Thut, G., & Koch, G. (2014). Prefrontal Control over Motor Cortex Cycles at Beta Frequency during Movement Inhibition. *Current Biology*, 24(24), 2940–2945.
- Picton, T. W., Bentin, S., Berg, P., Donchin, E., Hillyard, S. A., Johnson, R., Taylor, M. J. (2000). Guidelines for using human event-related potentials to study cognition: recording standards and publication criteria. *Psychophysiology*, 37(2), 127–152.
- Polich, J. (2007). Updating P300: An Integrative Theory of P3a and P3b. *Clinical Neurophysiology*, *118*(10), 2128–2148.
- Polich, J. (2012). Neuropsychology of P300. In S. J. Luck & E. S. Kappenman (Eds.), *Oxford handbook of event-related potential components* (pp. 159–188). Oxford University Press.
- Polich, J., & Criado, J. R. (2006). Neuropsychology and neuropharmacology of P3a and P3b. *International Journal of Psychophysiology*, *60*(2), 172–185.
- Polich, J., & McIsaac, H. K. (1994). Comparison of auditory P300 habituation from active and passive conditions. *International Journal of Psychophysiology*, 17(1), 25–34.
- Posner, M. I. (1980). Orienting of attention. *The Quarterly Journal of Experimental Psychology*, 32(1), 3–25.
- Price, G. W. (1997). The effect of pre-stimulus alpha activity on the auditory p300 paradigm: A prospective study. *Brain Topography*, 9(3), 169–176.
- Price, G. W. (2004). EEG-dependent ERP recording: Using TMS to increase the incidence of a selected pre-stimulus pattern. *Brain Research Protocols*, *12*(3), 144–151.
- Rahn, E., & Basar, E. (1993). Prestimulus EEG-Activity Strongly Influences the Auditory Evoked Vertex Response: A New Method for Selective Averaging. *International Journal of Neuroscience*, 69(1-4), 207–220.
- Ramchurn, A., de Fockert, J. W., Mason, L., Darling, S., & Bunce, D. (2014). Intraindividual reaction time variability affects P300 amplitude rather than latency. *Frontiers in Human Neuroscience*, 8, 557.
- Randall, W. M., & Smith, J. L. (2011). Conflict and inhibition in the cued-Go/NoGo task. *Clinical Neurophysiology*, *122*(12), 2400–2407.
- Ring, C., Cooke, A., Kavussanu, M., McIntyre, D., & Masters, R. (2015). Investigating the efficacy of neurofeedback training for expediting expertise and excellence in sport. *Psychology of Sport and Exercise*, 16, 118–127.
- Ritter, P., Moosmann, M., & Villringer, A. (2009). Rolandic alpha and beta EEG rhythms' strengths are inversely related to fMRI-BOLD signal in primary somatosensory and motor cortex. *Human Brain Mapping*, *30*(4), 1168–1187.
- Ros, T., Baars, B. J., Lanius, R. A., & Vuilleumier, P. (2014). Tuning pathological brain oscillations with neurofeedback: A systems neuroscience framework. *Frontiers in Human Neuroscience*, 8, 1008.
- Ros, T., Moseley, M. J., Bloom, P. A., Benjamin, L., Parkinson, L. A., & Gruzelier, J. H. (2009). Optimizing microsurgical skills with EEG neurofeedback. *BMC Neuroscience*, 10, 87.
- Ros, T., Munneke, M. A. M., Ruge, D., Gruzelier, J. H., & Rothwell, J. C. (2010). Endogenous control of waking brain rhythms induces neuroplasticity in humans. *The European Journal of Neuroscience*, 31(4), 770–778.
- Rossiter, T. R., & Lavaque, T. J. (1995). A Comparison of EEG Biofeedback and Psychostimulants in Treating Attention Deficit Hyperactivity Disorders. *Journal of Neurotherapy*, 1(1), 48–59.

- Roth, S. R., Sterman, M. B., & Clemente, C. D. (1967). Comparison of EEG correlates of reinforcement, internal inhibition and sleep. *Electroencephalography and Clinical Neurophysiology*, 23(6), 509–520.
- Ruiz, S., Buyukturkoglu, K., Rana, M., Birbaumer, N., & Sitaram, R. (2014). Real-time fMRI brain computer interfaces: Self-regulation of single brain regions to networks. *Biological Psychology*, 95(1), 4–20.
- Rushby, J. A., Barry, R. J., & Doherty, R. J. (2005). Separation of the components of the late positive complex in an ERP dishabituation paradigm. *Clinical Neurophysiology*, *116*(10), 2363–2380.
- Rushby, J. A., Barry, R. J., Clarke, A. R., & VaezMousavi, M. (2007). Arousal and activation effects on physiological and behavioral responding during a continuous performance task. *Acta Neurobiologiae Experimentalis*, 67(4), 461–470.
- Rutterford, N., & Pacheco, B. (2011). The influence of baseline and mean maximum amplitude in the ability to control SMR. *Neuroscience Letters*, 500, 46.
- Salari, N., Buchel, C., & Rose, M. (2014). Neurofeedback training of gamma band oscillations improves perceptual processing. *Experimental Brain Research*, 232(10), 3353–3361.
- Salisbury, D. F., Griggs, C. B., Shenton, M. E., & McCarley, R. W. (2004). The NoGo P300 "anteriorization" effect and response inhibition. *Clinical Neurophysiology*, 115(7), 1550–1558.
- Salisbury, D. F., Rutherford, B., Shenton, M. E., & McCarley, R. W. (2001). Buttonpressing affects P300 amplitude and scalp topography. *Clinical Neurophysiology*, *112*(9), 1676–84.
- SanMiguel, I., Morgan, H. M., Klein, C., Linden, D., & Escera, C. (2010). On the functional significance of Novelty-P3: facilitation by unexpected novel sounds. *Biological Psychology*, 83(2), 143–152.
- Sarter, M., Gehring, W. J., & Kozak, R. (2006). More attention must be paid: the neurobiology of attentional effort. *Brain Research Reviews*, 51(2), 145–160.
- Sarter, M., Givens, B., & Bruno, J. P. (2001). The cognitive neuroscience of sustained attention: where top-down meets bottom-up. *Brain Research Reviews*, 35(2), 146–160.
- Sasaki, K., & Gemba, H. (1986). Electrical activity in the prefrontal cortex specific to nogo reaction of conditioned hand movement with colour discrimination in the monkey. *Experimental Brain Research*, 64(3), 603–616.
- Sauseng, P., Hoppe, J., Klimesch, W., Gerloff, C., & Hummel, F. C. (2007). Dissociation of sustained attention from central executive functions: local activity and interregional connectivity in the theta range. *The European Journal of Neuroscience*, 25(2), 587–593.
- Sauseng, P., & Klimesch, W. (2008). What does phase information of oscillatory brain activity tell us about cognitive processes? *Neuroscience and Biobehavioral Reviews*, 32(5), 1001–1013.
- Sauseng, P., Klimesch, W., Gruber, W. R., & Birbaumer, N. (2008). Cross-frequency phase synchronization: A brain mechanism of memory matching and attention. *NeuroImage*, 40(1), 308–317.
- Sauseng, P., Klimesch, W., Gruber, W. R., Hanslmayr, S., Freunberger, R., & Doppelmayr, M. (2007). Are event-related potential components generated by phase resetting of brain oscillations? A critical discussion. *Neuroscience*, 146(4), 1435–1444.

- Sauseng, P., Klimesch, W., Schabus, M., & Doppelmayr, M. (2005). Fronto-parietal EEG coherence in theta and upper alpha reflect central executive functions of working memory. *International Journal of Psychophysiology*, 57(2), 97–103.
- Saville, C. W. N., Dean, R. O., Daley, D., Intriligator, J., Boehm, S., Feige, B., & Klein, C. (2011). Electrocortical correlates of intra-subject variability in reaction times: average and single-trial analyses. *Biological Psychology*, 87(1), 74–83.
- Sawaki, R., & Katayama, J. (2006). Stimulus context determines whether non-target stimuli are processed as task-relevant or distractor information. *Clinical Neurophysiology*, *117*(11), 2532–2539.
- Sawaki, R., & Katayama, J. (2008). Distractor P3 is associated with attentional capture by stimulus deviance. *Clinical Neurophysiology*, *119*(6), 1300–1309.
- Sawaki, R., & Katayama, J. (2009). Difficulty of discrimination modulates attentional capture by regulating attentional focus. *Journal of Cognitive Neuroscience*, 21(2), 359–371.
- Sayers, B. M., Beagley, H. A., & Henshall, W. R. (1974). The mechansim of auditory evoked EEG responses. *Nature*, 247(5441), 481–493.
- Schabus, M., Heib, D. P. J., Lechinger, J., Griessenberger, H., Klimesch, W., Pawlizki, A., Hoedlmoser, K. (2014). Enhancing sleep quality and memory in insomnia using instrumental sensorimotor rhythm conditioning. *Biological Psychology*, 95, 126–134.
- Schachar, R., Logan, G. D., Robaey, P., Chen, S., Ickowicz, A., & Barr, C. (2007). Restraint and cancellation: multiple inhibition deficits in attention deficit hyperactivity disorder. *Journal of Abnormal Child Psychology*, 35(2), 229–238.
- Schacter, D. L. (1977). EEG theta waves and psychological phenomena: A review and analysis. *Biological Psychology*, 5(1), 47–82.
- Scharnowski, F., Hutton, C., Josephs, O., Weiskopf, N., & Rees, G. (2012). Improving visual perception through neurofeedback. *Journal of Neuroscience*, 32(49), 17830– 17841.
- Scheeringa, R., Bastiaansen, M. C. M., Petersson, K. M., Oostenveld, R., Norris, D. G., & Hagoort, P. (2008). Frontal theta EEG activity correlates negatively with the default mode network in resting state. *International Journal of Psychophysiology*, 67(3), 242– 251.
- Schutter, D. J. L. G., Leitner, C., Kenemans, J. L., & van Honk, J. (2006). Electrophysiological correlates of cortico-subcortical interaction: a cross-frequency spectral EEG analysis. *Clinical Neurophysiology*, 117(2), 381–287.
- Seifert, A. R. R., & Lubar, J. F. F. (1975). Reduction of epileptic seizures through EEG biofeedback training. *Biological Psychology*, *3*(3), 157–184.
- Seitz, A. R. (2013). Cognitive neuroscience: Targeting neuroplasticity with neural decoding and biofeedback. *Current Biology*, 23(5), 210–212.
- Senderecka, M., Grabowska, A., Gerc, K., Szewczyk, J., & Chmylak, R. (2012). Eventrelated potentials in children with attention deficit hyperactivity disorder: an investigation using an auditory oddball task. *International Journal of Psychophysiology*, 85(1), 106–115.
- Shah, A. S., Bressler, S. L., Knuth, K. H., Ding, M., Mehta, A. D., Ulbert, I., & Schroeder, C. E. (2004). Neural Dynamics and the Fundamental Mechanisms of Event-related Brain Potentials. *Cerebral Cortex*, 14(5), 476–483.

- Sherlin, L. H., Arns, M., Lubar, J., Heinrich, H., Kerson, C., Strehl, U., & Sterman, M. B. (2011). Neurofeedback and Basic Learning Theory: Implications for Research and Practice. *Journal of Neurotherapy*, 15(4), 292–304.
- Shouse, M. N., & Lubar, J. F. (1979). Sensorimotor rhythm (SMR) operant conditioning and methylphenidate in the treatment of hyperkinesis. *Biofeedback and Self-Regulation*, 4, 299–311.
- Simons, R. F., Graham, F. K., Miles, M. A., & Chen, X. (2001). On the relationship of P3a and the Novelty-P3. *Biological Psychology*, *56*(3), 207–218.
- Simson, R., Vaughan, H. G., & Ratter, W. (1977). The scalp topography of potentials in auditory and visual Go/No-Go tasks. *Electroencephalography and Clinical Neurophysiology*, 43, 864–875.
- Smart, C. M., Segalowitz, S. J., Mulligan, B. P., & MacDonald, S. W. S. (2014). Attention capacity and self-report of subjective cognitive decline: A P300 ERP study. *Biological Psychology*, 103, 144–151.
- Smith, J. L. (2011). To go or not to go, that is the question: Do the N2 and P3 reflect stimulus- or response-related conflict? *International Journal of Psychophysiology*, 82(2), 143–152.
- Smith, J. L., Jamadar, S., Provost, A. L., & Michie, P. T. (2013). Motor and non-motor inhibition in the Go/NoGo task: An ERP and fMRI study. *International Journal of Psychophysiology*, 87(3), 244–253.
- Smith, J. L., Johnstone, S. J., & Barry, R. J. (2006). Effects of pre-stimulus processing on subsequent events in a warned Go/NoGo paradigm: Response preparation, execution and inhibition. *International Journal of Psychophysiology*, 61(2), 121–133.
- Smith, J. L., Johnstone, S. J., & Barry, R. J. (2007). Response priming in the Go/NoGo task: The N2 reflects neither inhibition nor conflict. *Clinical Neurophysiology*, 118(2), 343–355.
- Smith, J. L., Johnstone, S. J., & Barry, R. J. (2008). Movement-related potentials in the Go/NoGo task: The P3 reflects both cognitive and motor inhibition. *Clinical Neurophysiology*, 119(3), 704–714.
- Smith, J. L., Smith, E. A., Provost, A. L., & Heathcote, A. (2010). Sequence effects support the conflict theory of N2 and P3 in the Go/NoGo task. *International Journal of Psychophysiology*, 75(3), 217–226.
- Spencer, K. M., Dien, J., & Donchin, E. (1999). A componential analysis of the ERP elicited by novel events using a dense electrode array. *Psychophysiology*, *36*(3), 409–414.
- Spencer, K. M., Dien, J., & Donchin, E. (2001). Spatiotemporal Analysis of the Late ERP Responses to Deviant Stimuli. *Psychophysiology*, *38*, 343–358.
- Squires, K. C., Donchin, E., Herning, R. I., & McCarthy, G. (1977). On the influence of task relevance and stimulus probability on event-related potential components. *Electroencephalography and Clinical Neurophysiology*, *42*, 1–14.
- Squires, N. K., Squires, K. C., & Hillyard, S. A. (1975). Two varieties of long-latency positive waves evoked by unpredictable auditory stimuli in man. *Electroencephalography and Clinical Neurophysiology*, *38*(4), 387–401.
- Stancák, A., & Pfurtscheller, G. (1996). Event-related desynchronisation of central betarhythms during brisk and slow self-paced finger movements of dominant and nondominant hand. *Brain Research. Cognitive Brain Research*, 4(3), 171–183.

- Staufenbiel, S. M., Brouwer, A.-M., Keizer, A. W., & van Wouwe, N. C. (2014). Effect of beta and gamma neurofeedback on memory and intelligence in the elderly. *Biological Psychology*, 95, 74–85.
- Steriade, M. (2000). Corticothalamic resonance, states of vigilance and mentation. *Neuroscience*, 101(2), 243–276.
- Steriade, M. (2006). Grouping of brain rhythms in corticothalamic systems. *Neuroscience*, 137(4), 1087–1106.
- Steriade, M., Amzica, F., & Nunez, A. (1993). Cholinergic and noradrenergic modulation of the slow (approximately 0.3 Hz) oscillation in neocortical cells. *Journal of Neurophysiology*, 70(4), 1385–1400.
- Sterman, M. B. (1996). Physiological origins and functional correlates of EEG rhythmic activities: implications for self-regulation. *Biofeedback and Self-Regulation*, 21(1), 3–33.
- Sterman, M. B., & Friar, L. (1972). Suppression of seizures in an epileptic following sensorimotor EEG feedback training. *Electroencephalography and Clinical Neurophysiology*, 33(1), 89–95.
- Sterman, M. B., Goodman, S. J., & Kovalesky, R. A. (1978). Effects of sensorimotor EEG feedback training on seizure susceptibility in the rhesus monkey. *Experimental Neurology*, 62(3), 735–747.
- Sterman, M. B., Howe, R. C., & Macdonald, L. R. (1970). Facilitation of spindle-burst sleep by conditioning of electroencephalographic activity while awake. *Science*, *167*(3921), 1146–1148.
- Sterman, M. B., & Shouse, M. N. (1980). Quantitative analysis of training, sleep EEG and clinical response to EEG operant conditioning in epileptics. *Electroencephalography* and Clinical Neurophysiology, 49(5-6), 558–576.
- Studer, P., Kratz, O., Gevensleben, H., Rothenberger, A., Moll, G. H., Hautzinger, M., & Heinrich, H. (2014). Slow cortical potential and theta/beta neurofeedback training in adults: effects on attentional processes and motor system excitability. *Frontiers in Human Neuroscience*, 8, 555.
- Stuss, D. T., Murphy, K. J., Binns, M. A., & Alexander, M. P. (2003). Staying on the job: the frontal lobes control individual performance variability. *Brain : A Journal of Neurology*, 126(Pt 11), 2363–2380.
- Sutton, S., Braren, M., Zubin, J., & John, E. R. (1965). Evoked-potential correlates of stimulus uncertainty. *Science*, 150(3700), 1187–1198.
- Swann, N., Tandon, N., Canolty, R., Ellmore, T. M., McEvoy, L. K., Dreyer, S., ... Aron, A. R. (2009). Intracranial EEG reveals a time- and frequency-specific role for the right inferior frontal gyrus and primary motor cortex in stopping initiated responses. *The Journal of Neuroscience*, 29(40), 12675–12685.
- Takahashi, K., Saleh, M., Penn, R. D., & Hatsopoulos, N. G. (2011). Propagating waves in human motor cortex. *Frontiers in Human Neuroscience*, *5*, 40.
- Takahashi, N., Shinomiya, S., Mori, D., & Tachibana, S. (1997). Frontal midline theta rhythm in young healthy adults. *Clinical EEG (Electroencephalography)*, 28(1), 49–54.
- Tan, G., Thornby, J., Hammond, D. C., Strehl, U., Canady, B., Arnemann, K., & Kaiser, D. A. (2009). Meta-Analysis of EEG Biofeedback in Treating Epilepsy. *Clinical EEG* and Neuroscience, 40(3), 1–8.
- Thomas, S. J., Gonsalvez, C. J., & Johnstone, S. J. (2009). Sequence effects in the

Go/NoGo task: Inhibition and facilitation. *International Journal of Psychophysiology*, 74(3), 209–219.

- Thompson, T., Steffert, T., Ros, T., Leach, J., & Gruzelier, J. (2008). EEG applications for sport and performance. *Methods (San Diego, Calif.)*, 45(4), 279–288.
- Thut, G., Miniussi, C., & Gross, J. (2012). The functional importance of rhythmic activity in the brain. *Current Biology*, 22(16), 658–663.
- Tucker, D. M., & Williamson, P. A. (1984). Asymmetric neural control system in human self-regulation. *Psychological Review*, *91*, 185–215.
- Varela, F., Lachaux, J., Rodriguez, E., & Martinerie, J. (2001). The BRAINWEB: Phase synchronization and large-sale integration. *Nature Reviews Neuroscience*, *2*, 229–239.
- Vaurio, R. G., Simmonds, D. J., & Mostofsky, S. H. (2009). Increased intra-individual reaction time variability in attention-deficit/hyperactivity disorder across response inhibition tasks with different cognitive demands. *Neuropsychologia*, 47(12), 2389– 2396.
- Verleger, R. (1988). Event-related potentials and cognition: A critique of the context updating hypothesis and an alternative interpretation of P3. *Behavioral and Brain Sciences*, *11*(03), 343–427.
- Verleger, R. (1997). On the utility of P3 latency as an index of mental chronometry. *Psychophysiology*, *34*, 131-156.
- Verleger, R., Baur, N., Metzner, M. F., & Smigasiewicz, K. (2014). The hard oddball: Effects of difficult response selection on stimulus-related P3 and on response-related negative potentials. *Psychophysiology*, 51(11), 1089–1100.
- Verleger, R., Jaskowski, P., & Wascher, E. (2005). Evidence for an Integrative Role of P3b in Linking Reaction to Perception. *Journal of Psychophysiology*, 20(2), 165–181.
- Vernon, D., Egner, T., Cooper, N., Compton, T., Neilands, C., Sheri, A., & Gruzelier, J. (2003). The effect of training distinct neurofeedback protocols on aspects of cognitive performance. *International Journal of Psychophysiology*, 47(1), 75–85.
- Vogel, W., Broverman, D. M., & Klaiber, E. L. (1968). EEG and mental abilities. *Electroencephalography and Clinical Neurophysiology*, 24(2), 166–175.
- Von Stein, A., & Sarnthein, J. (2000). Different frequencies for different scales of cortical integration: From local gamma to long range alpha/theta synchronization. *International Journal of Psychophysiology*, 38(3), 301–313.
- Wang, J.-R., & Hsieh, S. (2013). Neurofeedback training improves attention and working memory performance. *Clinical Neurophysiology*, 124(12), 2406–2420.
- Wang, X., & Ding, M. (2011). Relation between P300 and event-related theta-band synchronization: A single-trial analysis. *Clinical Neurophysiology*, 122(5), 916–924.
- Wangler, S., Gevensleben, H., Albrecht, B., Studer, P., Rothenberger, A., Moll, G. H., & Heinrich, H. (2011). Neurofeedback in children with ADHD: specific event-related potential findings of a randomized controlled trial. *Clinical Neurophysiology*, 122(5), 942–950.
- Wascher, E., Rasch, B., Sänger, J., Hoffmann, S., Schneider, D., Rinkenauer, G., Gutberlet, I. (2013). Frontal theta activity reflects distinct aspects of mental fatigue. *Biological Psychology*, 96C, 57–65.
- Weber, E., Köberl, A., Frank, S., & Doppelmayr, M. (2011). Predicting successful learning of SMR neurofeedback in healthy participants: methodological considerations. *Applied Psychophysiology and Biofeedback*, 36(1), 37–45.
- Wilkinson, R. T., & Seales, D. M. (1978). EEG event-related potentials and signal

detection. Biological Psychology, 7(1-2), 13-28.

- Williams, H. L., Granda, A. M., Jones, R. C., Lubin, A., & Armington, J. C. (1962). EEG frequency and finger pulse volume as predictors of reaction time during sleep loss. *Electroencephalography and Clinical Neurophysiology*, 14(1), 64–70.
- Williams, P., Beatty, J., & O'Hanlon, J. F. (1975). Self-control of occipital theta activity and task performance. *Science*, *188*(4187), 478–489.
- Witte, M., Kober, S. E., Ninaus, M., Neuper, C., & Wood, G. (2013). Control beliefs can predict the ability to up-regulate sensorimotor rhythm during neurofeedback training. *Frontiers in Human Neuroscience*, *7*, 478–486.
- Womelsdorf, T., & Everling, S. (2015). Long-Range Attention Networks: Circuit Motifs Underlying Endogenously Controlled Stimulus Selection. *Trends in Neurosciences*, 38(11), 682–700.
- Womelsdorf, T., & Fries, P. (2007). The role of neuronal synchronization in selective attention. *Current Opinion in Neurobiology*, 17(2), 154–160.
- Womelsdorf, T., Schoffelen, J.-M., Oostenveld, R., Singer, W., Desimone, R., Engel, A. K., & Fries, P. (2007). Modulation of neuronal interactions through neuronal synchronization. *Science*, 316(5831), 1609–1612.
- Wróbel, A. (2000). Beta activity: a carrier for visual attention. Acta Neurobiologiae Experimentalis, 60(2), 247–260.
- Wronka, E., Kaiser, J., & Coenen, A. M. L. (2008). The auditory P3 from passive and active three-stimulus oddball paradigm. *Acta Neurobiologiae Experimentalis*, 68(3), 362–172.
- Wyrwicka, W., & Sterman, M. B. (1968). Instrumental conditioning of sensorimotor cortex EEG spindles in the waking cat. *Physiology & Behavior*, *3*(5), 703–707.
- Yamagishi, N., Callan, D. E., Goda, N., Anderson, S. J., Yoshida, Y., & Kawato, M. (2003). Attentional modulation of oscillatory activity in human visual cortex. *NeuroImage*, 20(1), 98–113.
- Yordanova, J., & Kolev, V. (1998). Single-sweep analysis of the theta frequency band during an auditory oddball task. *Psychophysiology*, 35(1), 116–126.
- Yordanova, J., Kolev, V., & Polich, J. (2001). P300 and alpha event-related desynchronization (ERD). *Psychophysiology*, *38*(1), 143–152.
- Yordanova, J., Kolev, V., Rosso, O. A., Schurmann, M., Sakowitz, O. W., Ozgoren, M., & Basar, E. (2002). Wavelet entropy analysis of event-related potentials indicates modality-independent theta dominance. *Journal of Neuroscience Methods*, 117(1), 99– 109.
- Yordanova, J., Rosso, O. A., & Kolev, V. (2003). A transient dominance of theta eventrelated brain potential component characterizes stimulus processing in an auditory oddball task. *Clinical Neurophysiology*, 114(3), 529–540.
- Yordanova, J. Y., Kolev, V. N., & Başar, E. (1998). EEG theta and frontal alpha oscillations during auditory processing change with aging. *Electroencephalography* and Clinical Neurophysiology, 108(5), 497–505.
- Zhang, Y., Chen, Y., Bressler, S. L., & Ding, M. (2008). Response preparation and inhibition: the role of the cortical sensorimotor beta rhythm. *Neuroscience*, *156*(1), 238–246.
- Zoefel, B., Huster, R. J., & Herrmann, C. S. (2011). Neurofeedback training of the upper alpha frequency band in EEG improves cognitive performance. *NeuroImage*, 54(2), 1427–1431.

Appendices

14.1. Appendix A: Screening questionnaire

Neurofeedback Frequency-Specific Effets on Late Event-Related Potentials

Screening Questionnaire

ID number:_____ Age:_____ Gender:_____ CODE:_____

Before taking part in this experiment it is important that we know some things about your medical history as this might influence EEG recording and make you ineligible to take part.

Please answer the questions below by ticking the boxes. After you've done it, please verify that you're eligible for the experiment.

Have any of the following ever happened to you?	Yes	No
Traumatic brain injury (with loss of counsciouness)		
Acute intoxication (leading to coma or severe loss of counsciouness)		
Drug or alcohol addiction		
Neurological disorder (brain tumor, epilepsy, stroke, multiple sclerosis, other)		
Psychiatric disorder (major depression, anxiety disorder, bipolar disorder, psychosis, other)		

Are you currently taking any medication with a psychoactive effect?	Yes	No
Antidapressants		
Antidepressants		
Stimulants		
Antipsychotics		
March et al. 11 - and (in the disc and in the et al. and in the state of the state		
Mood stabilizers (including antiephepues or anticonvulsants)		
Anxiolytics (benzodiazepines, other)		
Depressants (sedatives, hypnotics, other)		

Hand preference	Left	Right

Thank you for having completed the questionnaire. If you ticked NO in every box you may participate in the experiment. Please, bring the filled questionnaire with you for the initial session or send it to <u>m.pimenta@gold.ac.uk</u> (Miguel Pimenta).

14.2. Appendix B: Neurofeedback instructions

Instructions

In the coming weeks you will take part in a neurofeedback training study. As you now know from the EEG session, your brain works partly on electrical activity which we can measure using these electrodes. The electrical activity in your brain happens in certain frequency ranges, for example, there is slower activity when you're asleep, and faster activity when you are trying to solve a math problem.

This image shows you the electrical activity that's going on in your brain right now. The higher the peaks the more activity you are creating at certain frequencies. As you can see there is continuous movement, there is natural variation over time of the EEG (and in brain waves). So, it is believed that we can gain some degree of control over this activity. In this study we are trying to find out if that is indeed the case and to what extent people can influence their brain activity. In a moment you will be shown a screen which responds to a specific frequency range. Over the weeks you will be training to influence your brain's activity within this frequency range by means of the feedback the screen provides to you. Try to keep the EEG activity within the range for as long as possible as well as to be aware of what makes it work.

We can only have a certain degree of control over our brain activity. The way you can gain control over your brain activity is by learning from the feedback you're being given. So you have to pay attention to what happens in the screen, but you shouldn't try too hard. There will not always be a correspondence between what you are intending to do and what your brain does. Also, you shouldn't play tricks, just pay passive attention to what happens and see if you can keep the game from stopping. We cannot tell you exactly how to do this because it is different for everybody, and do not be disappointed if you don't get it or if you can't stay above this threshold the whole time. There will always be some variation.

Try as much as possible to relax your jaw, neck and shoulder muscles as it might create electromuscular activity that will interfere with the EEG reading. Information on electromuscular activity will also be displayed.

Each training session will have 10 trials. Each trial is 2.50 in duration. During the first and last trials try to keep the EMG within range and don't bother with the EEG activity. Each time we will compare whatever you do during your training to the previous period.

14.3. Appendix C: Examples of feedback displays

Figures A, B and C are examples of visual representations used for NF. In the first study, the participants received NF in the target frequency through two types of visual displays (A and B). In the first half of the session, the moving video in right side of display A represented successful SMR or betal enhancement or the successful suppression of the target frequency in the control NF. The bar graph in the left side of the screen represented the EMG activity. In the second half of the session, the enlarged blue circle represented the enhancement of the target frequency, while the green square represented EMG. In the second study, the visual representation of successful enhancement or inhibition was provided both by the video and the leftmost bar graph as shown in display C. The rightmost bar graph represented EMG activity. As shown in the figure, the bar graph turned green when the the peak-to-peak amplitude was enhanced above the threshold (invisible) or supressed below the threshold (horizontal line crossing the vertical axis) according to instructions. In both studies, the participants obtained points for fulfilling both target and EMG conditions.

Α






